FRANCHISE DISCLOSURE DOCUMENT

BrightStar Senior Living Franchising, LLC (an Illinois limited liability company) 2275 Half Day Road, Suite 210 Bannockburn, Illinois 60015 Telephone: 888.684.8250 www.brightstarfranchise.com email: franchise@brightstarseniorliving.com



The franchise offered in this disclosure document is to operate a 10-to-12-bed assisted-living and memorycare community under the "BRIGHTSTAR CARE HOMES[®]" trademark providing various services for seniors and other individuals in need, including residential assisted-living, memory-care, and ancillary services.

The total investment necessary to begin operation of a 10-to-12-bed BRIGHTSTAR CARE HOMES[®] Community franchise is \$1,225,916 to \$2,202,720 (if you intend to buy the Community's land and building) and \$201,308 to \$329,850 (if you intend to lease the Community's land and building). This includes \$50,000 to \$55,000 that must be paid to the franchisor or affiliate.

This disclosure document summarizes certain provisions of your franchise agreement and other information in plain English. Read this disclosure document and all accompanying agreements carefully. You must receive this disclosure document at least 14 calendar-days before you sign a binding agreement with, or make any payment to, the franchisor or an affiliate in connection with the proposed franchise sale. **Note, however, that no governmental agency has verified the information contained in this document.**

The terms of your contract will govern your franchise relationship. Don't rely on the disclosure document alone to understand your contract. Read all of your contract carefully. Show your contract and this disclosure document to an advisor, like a lawyer or an accountant.

Buying a franchise is a complex investment. The information in this disclosure document can help you make up your mind. More information on franchising, such as "<u>A Consumer's Guide to Buying a Franchise</u>," which can help you understand how to use this disclosure document, is available from the Federal Trade Commission. You can contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, D.C. 20580. You can also visit the FTC's home page at www.ftc.gov for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

Issuance date of this Franchise Disclosure Document: April 29, 2024

How to Use This Franchise Disclosure Document

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTION	WHERE TO FIND INFORMATION
How much can I earn?	Item 19 may give you information about outlet sales, costs, profits or losses. You should also try to obtain this information from others, like current and former franchisees. You can find their names and contact information in Item 20 or Exhibit G.
How much will I need to invest?	Items 5 and 6 list fees you will be paying to the franchisor or at the franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.
Does the franchisor have the financial ability to provide support to my business?	Item 21 or Exhibit F includes financial statements. Review these statements carefully.
Is the franchise system stable, growing, or shrinking?	Item 20 summarizes the recent history of the number of company-owned and franchised outlets.
Will my business be the only BRIGHTSTAR CARE HOMES business in my area?	Item 12 and the "territory" provisions in the franchise agreement describe whether the franchisor and other franchisees can compete with you.
Does the franchisor have a troubled legal history?	Items 3 and 4 tell you whether the franchisor or its management have been involved in material litigation or bankruptcy proceedings.
What's it like to be a BRIGHTSTAR CARE HOMES franchisee?	Item 20 or Exhibit G lists current and former franchisees. You can contact them to ask about their experiences.
What else should I know?	These questions are only a few things you should look for. Review all 23 Items and all Exhibits in this disclosure document to better understand this franchise opportunity. See the table of contents.

What You Need To Know About Franchising Generally

<u>Continuing responsibility to pay fees</u>. You may have to pay royalties and other fees even if you are losing money.

Business model can change. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

Supplier restrictions. You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

Operating restrictions. The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>**Competition from franchisor**</u>. Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

<u>Renewal</u>. Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

When your franchise ends. The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

Some States Require Registration

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Exhibit A.

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

Special Risks to Consider About This Franchise

Certain states require that the following risk(s) be highlighted:

- 1. <u>Out-of-State Dispute Resolution</u>. The franchise agreement requires you to resolve disputes with the franchisor by mediation, arbitration, and/or litigation only in the franchisor's home state (currently Illinois). Out-of-state mediation, arbitration, or litigation may force you to accept a less favorable settlement for disputes. It may also cost more to mediate, arbitrate, or litigate with the franchisor in the franchisor's home state (currently Illinois) than in your own state.
- 2. **Spousal Liability.** Your spouse must also sign a document that makes your spouse liable for all financial obligations under the Franchise Agreement even though your spouse has no ownership interest in the franchise. This guarantee will place both your and your spouse's marital and personal assets, perhaps including your house, at risk if your franchise fails.
- 3. **Financial Condition.** The franchisor's financial condition, as reflected in its financial statements (see Item 21), calls into question the franchisor's financial ability to provide services and support to you.
- 4. <u>Short Operating History</u>. The franchisor is at an early stage of development and has a limited operating history. This franchise is likely to be a riskier investment than a franchise in a system with a longer operating history.
- 5. **Supplier Control.** You must purchase all or nearly all of the inventory and supplies necessary to operate your business from Franchisor, its affiliates, or from suppliers that Franchisor designates at prices that the Franchisor or they set. These prices may be higher than prices you could obtain elsewhere for the same or similar goods. This may reduce the anticipated profit of your franchised business.

Certain states may require other risks to be highlighted. Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

<u>THE FOLLOWING PROVISIONS APPLY ONLY TO TRANSACTIONS GOVERNED</u> <u>BY THE MICHIGAN FRANCHISE INVESTMENT LAW</u>

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS. IF ANY OF THE FOLLOWING PROVISIONS ARE IN THESE FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU.

(a) A prohibition on the right of a franchisee to join an association of franchisees.

(b) A requirement that a franchisee assent to a release, assignment, novation, waiver, or estoppel which deprives a franchisee of rights and protections provided in this act. This shall not preclude a franchisee, after entering into a franchise agreement, from settling any and all claims.

(c) A provision that permits a franchisor to terminate a franchise prior to the expiration of its term except for good cause. Good cause shall include the failure of the franchisee to comply with any lawful provision of the franchise agreement and to cure such failure after being given written notice thereof and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.

(d) A provision that permits a franchisor to refuse to renew a franchise without fairly compensating the franchisee by repurchase or other means for the fair market value at the time of expiration of the franchisee's inventory, supplies, equipment, fixtures, and furnishings. Personalized materials which have no value to the franchisor and inventory, supplies, equipment, fixtures, and furnishings not reasonably required in the conduct of the franchise business are not subject to compensation. This subsection applies only if: (i) the term of the franchise is less than 5 years and (ii) the franchise is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise or the franchise does not receive at least 6 months advance notice of franchisor's intent not to renew the franchise.

(e) A provision that permits the franchisor to refuse to renew a franchise on terms generally available to other franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.

(f) A provision requiring that arbitration or litigation be conducted outside this state. This shall not preclude the franchisee from entering into an agreement, at the time of arbitration, to conduct arbitration at a location outside this state.

(g) A provision which permits a franchisor to refuse to permit a transfer of ownership of a franchise, except for good cause. This subdivision does not prevent a franchisor from exercising a right of first refusal to purchase the franchise. Good cause shall include, but is not limited to:

(i) The failure of the proposed transferee to meet the franchisor's then-current reasonable qualifications or standards.

(ii) The fact that the proposed transferee is a competitor of the franchisor or subfranchisor.

(iii) The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.

(iv) The failure of the franchisee or proposed transferee to pay any sums owing to the franchisor or to cure any default in the franchise agreement existing at the time of the proposed transfer.

(h) A provision that requires the franchisee to resell to the franchisor items that are not uniquely identified with the franchisor. This subdivision does not prohibit a provision that grants to a franchisor a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the franchisor the right to acquire the assets of a franchise for the market or appraised value of such assets if the franchisee has breached the lawful provisions of the franchise agreement and has failed to cure the breach in the manner provided in subdivision (c).

(i) A provision which permits the franchisor to directly or indirectly convey, assign, or otherwise transfer its obligations to fulfill contractual obligations to the franchisee unless provision has been made for providing the required contractual services.

If the franchisor's most recent financial statements are unaudited and show a net worth of less than \$100,000, the franchisor shall, at the request of a franchisee, arrange for the escrow of initial investment and other funds paid by the franchisee until the obligations to provide real estate, improvements, equipment, inventory, training, or other items included in the franchise offering are fulfilled. At the option of the franchisor, a surety bond may be provided in place of escrow.

THE FACT THAT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION, OR ENFORCEMENT BY THE ATTORNEY GENERAL.

Any questions regarding this notice should be directed to:

State of Michigan Consumer Protection Division Attn: Franchise 670 G. Mennen Williams Building 525 West Ottawa, Lansing, Michigan 48933 (517) 335-7567

Despite subparagraph (f) above, we intend to enforce fully the provisions of the arbitration section contained in our Franchise Agreement. We believe that subparagraph (f) is unconstitutional and cannot preclude us from enforcing our arbitration section. If you acquire a franchise, you acknowledge that we will seek to enforce that section as written, and that the terms of the Franchise Agreement will govern our relationship with you, including the specific requirements of the arbitration section.

TABLE OF CONTENTS

ITEM 1	THE FRANCHISOR, AND ANY PARENTS, PREDECESSORS, AND AFFILIATES
ITEM 2	BUSINESS EXPERIENCE
ITEM 3	LITIGATION
ITEM 4	BANKRUPTCY
ITEM 5	INITIAL FEES7
ITEM 6	OTHER FEES
ITEM 7	ESTIMATED INITIAL INVESTMENT
ITEM 8	RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES
ITEM 9	FRANCHISEE'S OBLIGATIONS
ITEM 10	FINANCING
ITEM 11	FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND TRAINING
ITEM 12	TERRITORY
ITEM 13	TRADEMARKS
ITEM 14	PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION46
ITEM 15	OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS
ITEM 16	RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL
ITEM 17	RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION51
ITEM 18	PUBLIC FIGURES
ITEM 19	FINANCIAL PERFORMANCE REPRESENTATIONS
ITEM 20	OUTLETS AND FRANCHISE INFORMATION
ITEM 21	FINANCIAL STATEMENTS
ITEM 22	CONTRACTS64
ITEM 23	RECEIPTS

EXHIBITS

- A. List of State Administrators and Agents for Service of Process
- B. BRIGHTSTAR CARE HOMES Community Franchise Agreement
- C. Operations Manual Table of Contents
- D. Confidentiality, Non-Disclosure and Non-Competition Agreement
- E. State Specific Addenda
- F. Financial Statements
- G. List of Franchisees
- H. Release of Claims

I. Microsoft Teams/One-Drive/Web Version Office Package Opt-in Addendum State Effective Dates

J. Receipts (2 copies)

ITEM 1

THE FRANCHISOR, AND ANY PARENTS, PREDECESSORS, AND AFFILIATES

To simplify the language in this Franchise Disclosure Document, "we," "us," and "our" refer to BrightStar Senior Living Franchising, LLC, the franchisor. "You" means the entity that acquires the franchise and, where noted, includes your owners. We refer to your prospective BRIGHTSTAR CARE HOMES Community as the "Community."

A. The Company, Our Predecessors and Affiliates.

We are an Illinois limited liability company with our principal offices at 2275 Half Day Road, Suite 210, Bannockburn, Illinois 60015. We were organized on March 28, 2013, and have offered franchises for BRIGHTSTAR CARE HOMES Communities since August 2021 and for BrightStar Senior Living Communities since January 2015. BrightStar Senior Living Communities provide the same services as BRIGHTSTAR CARE HOMES Communities but in a larger facility (anywhere from a 36 to 44-unit facility as opposed to the 10-to-12-bed BRIGHTSTAR CARE HOMES Community). We have never operated a BRIGHTSTAR CARE HOMES Community or a BrightStar Senior Living Community and do not operate, and have not offered franchises in, any other business. Our affiliate currently operates one BrightStar Senior Living Community in Mason, Ohio. (The BrightStar Senior Living Community franchise opportunity is not offered in this disclosure document.) We do not have any predecessors.

We began offering the BRIGHTSTAR CARE HOMES Community franchise opportunity in August 2021. We initially offered the franchise opportunity only to certain experienced BrightStar Care Agency program franchisees (described below) operating in certain states/venues. However, in 2022 we also began offering the BRIGHTSTAR CARE HOMES Community franchise opportunity to candidates who are not existing BrightStar Care Agency franchisees.

Our affiliate BrightStar Franchising, LLC ("BrightStar Franchising"), whose principal business address is the same as ours, has offered franchises for BrightStar Care agencies since August 2005. BrightStar Care agencies offer 4 principal services to clients: in-home Non-Medical (Companion) care services; in-home Medical Personal Care services; in-home Medical Skilled Care services; and Supplemental Healthcare Staffing. Some BrightStar Care franchisees offer all four services, and some do not. (The BrightStar Care agency program franchise opportunity is offered in a separate franchise disclosure document.) BrightStar Franchising had granted 347 BrightStar Care agency franchises as of December 31, 2023. BrightStar Franchising never has operated or offered franchises for BrightStar Senior Living Communities. However, as part of the pilot of the BRIGHTSTAR CARE HOMES Community concept, BrightStar Franchising granted (beginning in 2018) one existing BrightStar Care franchise the right to open and operate BRIGHTSTAR CARE HOMES Communities under an amendment to its existing BrightStar Care Franchise Agreement. BrightStar Franchising has not offered franchises in other lines of business.

Our parent is BrightStar Group Holdings, Inc., a Delaware corporation formed on September 29, 2010 ("BrightStar Holdings") whose principal offices are the same as ours. Other than serving as

our parent (and the parent of our affiliated entities), BrightStar Holdings does not operate any business and does not offer, and has not offered, franchises in any line of business.

We have 2 affiliates besides BrightStar Franchising that we must disclose in this Item 1: 24-7 Bright Star Healthcare, LLC ("Bright Star Nevada") and BrightStar Technology Group, LLC ("BrightStar Technology"), whose principal offices are the same as ours.

Bright Star Nevada owns the "BrightStar" trademarks (among other intellectual property) and, as described further in Item 13 below, licenses us the right to use and sublicense those trademarks. Bright Star Nevada has never operated a BRIGHTSTAR CARE HOMES Community or a BrightStar Senior Living Community or offered franchises in any line of business.

BrightStar Technology may provide you with ongoing support and assistance with respect to technology, branded email, and the website and intranet. If so, you will pay BrightStar Technology all fees associated with its technology support. BrightStar Technology does not operate any other business. BrightStar Technology has not previously offered franchises in any line of business.

Our agents for service of process are listed in Exhibit A to this disclosure document.

B. The BRIGHTSTAR CARE HOMES Community Franchise.

You will develop and operate up to 3 senior-living and memory-care or full memory-care Communities under the BRIGHTSTAR CARE HOMES trademark providing various services for seniors and other individuals in need, including residential assisted-living, memory-care, and ancillary services, according to the administrative and operational components noted in your Franchise Agreement (Exhibit B) (the "BrightStar Program" or "Program"). The BRIGHTSTAR CARE HOMES franchise opportunity involves the acquisition, construction, development, and operation of one or more homes in a residential setting, each of which will provide supervised care for up to 10 to 12 residents. Our Franchise Agreement grants you the right to construct, develop, and operate up to 3 BRIGHTSTAR CARE HOMES Communities under the System and Marks at locations to be identified. Unless this disclosure document must in certain contexts distinguish among the 3 BRIGHTSTAR CARE HOMES Communities that you are entitled to construct, develop, and operate under the Franchise Agreement, all references to "a Community" or "the Community" will be deemed to mean each of your Communities. The Franchise Agreement governs your construction, development, and operation of each and all of your Communities.

We identify the BrightStar Program by certain trade names, service marks, trademarks, logos, emblems, and indicia of origin, including the marks "BRIGHTSTAR CARE HOMES," distinctive trade dress, and such other trade names, trademarks, and service marks we now, or in the future may, designate in writing for use in the BrightStar Program (the "Marks"). We and our affiliates continue to develop, use, and control the use of the Marks in order to identify for the public the source of services and products marketed under the BrightStar Program and represent the Program's high standards of quality, appearance, and service.

If we deem it necessary for the operation of BRIGHTSTAR CARE HOMES Communities, the BrightStar Program in the future may include one or more aspects of our or our affiliate's technology (the "System Technology"). System Technology may encompass an integrated-management system providing various technology solutions, including client-relationship management, staff-relationship management, billing management, web management, franchise management, and all modifications.

It is not necessary that you have experience in the healthcare or senior-housing industry before acquiring your BRIGHTSTAR CARE HOMES Community franchise. You must obtain and maintain during the franchise term all licensure, licenses, and permits required to operate a BRIGHTSTAR CARE HOMES Community.

Your Communities will provide various services for seniors and other individuals in need, including residential assisted-living, memory-care, and ancillary services, or the Communities may only provide services for those with a memory-care need. You must have qualified healthcare professionals on staff and available to your clients at all times. You will offer your services throughout the year (24 hours a day, 7 days a week); your business is not seasonal.

C. Competition.

You will compete with other businesses providing similar services, including franchise programs competing with BRIGHTSTAR CARE HOMES Communities, non-franchised chains, and "individual" residences of this type. The principal market for the services of BRIGHTSTAR CARE HOMES Communities is the population of aging persons who may require some assistance with the activities of daily living or require care due to conditions such as Alzheimer's or dementia and therefore can benefit from a communal-living environment and the meals and maintenance services of BRIGHTSTAR CARE HOMES Communities. The market is a growing market driven by the aging of the general population.

D. Specialized Industry Laws.

You must comply with all wage and hour laws and regulations as well as all federal, state, and local laws and regulations that apply to your operations, including those pertaining to the health care industry, professional and community licensing, workers' compensation, corporate, tax, environmental, sanitation, insurance, no smoking, EEOC, HIPAA, OSHA, non-discrimination, employment, and sexual harassment. You must obtain and maintain any health care or employment-related permits, licenses, certifications, or other indications of authority necessary to operate your BRIGHTSTAR CARE HOMES Communities to the fullest extent of the model possible within the spectrum of residential assisted-living and memory-care services available in the state in which the BRIGHTSTAR CARE HOMES Community license. A Clinical Laboratory Improvement Amendment waiver ("CLIA" waiver) might be required if you perform services requiring a CLIA waiver. We may require you to obtain a particular permit, license, or accreditation. Some states have imposed a moratorium on issuing assisted-living licenses or may

require a Certificate of Need. You must investigate the availability and requirements for obtaining all necessary licenses in your state.

You may not participate in Medicare, Medicaid, or other governmental payor programs.

ITEM 2 BUSINESS EXPERIENCE

Andrew Ray - Chief Executive Officer

Andrew Ray has been our Chief Executive Officer since February 2024 and has held the same position with BrightStar Franchising, LLC and BrightStar Holdings, both of Bannockburn, IL, since that date. He was our Chief Operating Officer from October 2022 until January 2024. Mr. Ray was President of NextCare Holding in Mesa, Arizona (DBA BrightStar Care of Mesa) from September 2019 until October 2022. Mr. Ray was in between positions from November 2018 until September 2019.

Shelly Sun – Founder and Executive Chairwoman

Shelly Sun has been our Executive Chairwoman since February 2024. She was our Chief Executive Officer from our inception in March 2013 until January 2024. Ms. Sun also is: (i) Executive Chairwoman of BrightStar Franchising of Bannockburn, Illinois since February 2024, having served as its Chief Executive Officer from January 2005 until January 2024; (ii) the Chief Executive Officer of BrightStar Group Holdings, Inc. of Bannockburn, Illinois and has held that position since September 2010; and (iii) the Chief Executive Officer and Managing Member of BrightStar Technology Group, LLC of Bannockburn, Illinois and has held that position since August 2009.

Dean Ulizio – Chief Strategy Officer

Dean Ulizio has been our Chief Strategy Officer since January 2020 and has held the same position with BrightStar Franchising of Bannockburn, Illinois since that date. Mr. Ulizio was our Executive Vice President of Global Strategic Development, and held the same position with BrightStar Franchising, from September 2016 through December 2019.

<u>Pete First – Chief Development Officer</u>

Pete First has been our Chief Development Officer since March 2022 and has held the same position with BrightStar Franchising of Bannockburn, Illinois since that date. Mr. First was Senior Vice President of Franchise Development for both us and BrightStar Franchising from March 2020 through March 2022 and Vice President of Franchise Development from September 2018 through March 2020.

David Pallaschke – Chief Financial Officer

David Pallaschke has been our Chief Financial Officer since November 2019 and has held the same position with BrightStar Franchising of Bannockburn, Illinois since that date. From March 2015 through November 2019, Mr. Pallaschke was Senior Vice President Finance/Clinic Performance & Analytics for ATI Physical Therapy of Bolingbrook, IL.

<u>Teresa Celmer – Chief Marketing Officer</u>

Teresa Celmer has been our Chief Marketing Officer since March 2023 and was our Senior Vice President Marketing from March 2020 through February 2023. She has held (or held) the same positions with BrightStar Franchising of Bannockburn, Illinois during those timeframes. From July 2019 through April 2020, Ms. Celmer was Vice President of Marketing for both us and BrightStar Franchising. From June 2015 through June 2019, Ms. Celmer was Brand Director, Consumer Marketing for Ace Hardware Corporation in Oak Brook, Illinois.

Susan Ehrlich – Director of Clinical Operations

Susan Ehrlich, RN BSN CPD, has been our Director of Clinical Operations since October 2023. From July 2019 through September 2023, Ms. Ehrlich held positions of National Director of Quality and Care Services from August 2022 through September 2023, Regional Director of Care Services from March 2020 through August 2022, and North Division – Divisional Care Service Specialist from July 2019 through March 2020 for Enlivant in Chicago, Illinois, and Quality Improvement Manger of Williamsburg Healthcare from February 2017 to July 2019 in Crawfordville, Indiana.

ITEM 3 LITIGATION

The following litigation involves the operations of our affiliated BrightStar Care agency franchise system and BrightStar Franchising. No litigation about us must be disclosed in this Item.

Starcatcher Healthcare, LLC vs. BrightStar Franchising, LLC and Shelly Sun (No. CV 13-02051-PHX-MEA, filed on October 8, 2013, in the United States District Court for the District of Arizona). Starcatcher Healthcare ("Starcatcher"), a now-former franchisee that operated 2 BrightStar Care businesses, filed a lawsuit against BrightStar Franchising and Ms. Sun alleging breach of contract and the implied covenant of good faith and fair dealing, negligent misrepresentation, common law fraud, violation of the Arizona Consumer Fraud Act, and violation of the Illinois Franchise Disclosure Act in connection with its acquisition and operation of its 2 franchises. Starcatcher's claims were based on alleged deficiencies with software, failure to prevent encroachment by another franchisee, and failure to spend the general marketing fund in a manner benefiting Starcatcher. Starcatcher sought declaratory judgment, rescission of the parties' franchise agreements, unspecified damages, and costs and attorneys' fees. After various motions, BrightStar Franchising and Ms. Sun filed an answer denying the allegations. The parties settled the lawsuit on November 28, 2014, with full mutual releases of all claims. While BrightStar Franchising reimbursed approximately \$137,000 of fees and costs incurred by Starcatcher in the dispute, BrightStar Franchising did not make any lump-sum payment to Starcatcher on the substantive claims it asserted. As part of the settlement, Starcatcher exercised its right under a then-existing area development agreement to acquire a third franchise from BrightStar Franchising for an additional Arizona market. Starcatcher also agreed to contribute to the General Marketing Fund in substantial compliance with current system requirements. BrightStar Franchising and Starcatcher also modified Starcatcher's other 2 existing franchise agreements to reflect that, under a new (2-year minimum) "Pilot Program," Starcatcher's BrightStar Care businesses would be "beta sites" to collaborate with BrightStar Franchising to develop and customize new software and processes to support the managed skilled care portion of the BrightStar Care business. In consideration for Starcatcher's investment of time and money to participate in the Pilot Program, BrightStar Franchising agreed to waive or reduce various fees due under the parties' franchise agreements during and/or after the Pilot Program. Starcatcher dismissed the lawsuit, and the case is over.

Cosmo Fraser and Adam Fraser vs. BrightStar Franchising, LLC, Shelly Sun, Thomas Gilday, Scott Oaks, et al. (American Arbitration Association Case No. 02-16-0005-0209, filed March 15, 2017). The Frasers, former franchisees of BrightStar Franchising who operated one BrightStar Care franchise in Georgia, filed this arbitration against the defendants. They alleged violation of the Illinois Consumer Fraud and Deceptive Practices Act, violation of the Illinois Franchise Disclosure Act, statutory fraud under Georgia law, common law fraud, and negligent representation in connection with the Frasers' acquisition of their franchise. The Frasers alleged that the defendants failed to disclose before granting the franchise that the franchised territory was not a "new" territory, previously had been operated by 2 different franchisees who allegedly had failed, and allegedly was smaller than most franchised territories BrightStar Franchising granted and also allegedly misrepresented the financial performance for first-year franchisees. The Frasers sought rescission of the franchise agreement and rescissionary or compensatory damages in excess of \$500,000, punitive damages, attorneys' fees and costs, and other relief. The parties settled the case on March 1, 2018, in order to avoid further legal proceedings. The defendants paid the Frasers a total of \$215,000, the parties exchanged mutual releases, and the case was dismissed with prejudice.

<u>BrightStar Franchising, LLC v. Angelito D. Caballa and Silver Servants, LLC</u> (American Arbitration Association Case No. 01-23-0005-6659, filed December 6, 2023). BrightStar Franchising filed this arbitration against a franchisee whose franchise agreement was terminated and its guarantor, seeking payment of liquidated termination damages and other amounts due under the franchise agreement totaling \$126,386. BrightStar Franchising also seeks an award of the costs and attorneys' fees incurred in connection with the dispute.

BrightStar Franchising, LLC v. SFV, LLC, Ivan Wai-Wing Yeung, and Chun Fei ("Roger") Lui (American Arbitration Association Case No. 02-23-0004-3410, filed November 8, 2023). BrightStar Franchising filed this arbitration against a franchisee whose franchise agreement was terminated and its guarantors, seeking payment of post-termination liquidated damages ranging

from \$400,000 to \$500,000, return and disclosure of certain information, and costs and attorneys' fees. On or about December 3, 2023, the franchisee filed counterclaims for wrongful termination in violation of California Business & Professions Code Section 20020 and breach of contract, and the franchisee and its guarantors asserted a counterclaim for violation of California Business & Professions Code Section 17200. They seek a total monetary award of \$999,999 and injunctive relief. BrightStar Franchising intends to pursue its claims and defend all counterclaims vigorously.

<u>BrightStar Franchising, LLC v. Howard Wayne Bosh and Jerod W. Bosh</u> (American Arbitration Association, Case No. 02-23-0001-9769, filed May 1, 2023). BrightStar Franchising initiated action to enforce post-expiration obligations. The parties mediated and settled the matter.

Other than the actions listed above, no litigation is required to be disclosed in this Item.

ITEM 4 BANKRUPTCY

No bankruptcy is required to be disclosed in this Item.

ITEM 5 INITIAL FEES

The initial franchise fee for the three (3) BRIGHTSTAR CARE HOMES Communities that the Franchise Agreement entitles you to develop and operate currently is \$50,000. If you or your affiliate is not an existing BrightStar Care Agency franchisee, the full initial franchise fee is due when you sign the Franchise Agreement. If you choose not to develop the second and/or third BRIGHTSTAR CARE HOMES[®] Communities that the Franchise Agreement entitles you to develop and operate, no part of the \$50,000 initial franchise fee is refundable.

If you or your affiliate is an existing BrightStar Care Agency franchisee, you must pay us \$30,000 when you sign the Franchise Agreement. You must pay us an additional \$10,000 when we accept the site for your second Community to be developed under the Franchise Agreement and the final \$10,000 when we accept the site for your third Community to be developed under the Franchise Agreement. The initial franchise fee is not refundable under any circumstances. However, if you choose not to develop (by the applicable deadlines) the second and/or third BRIGHTSTAR CARE HOMES Communities that you are entitled to develop, the additional \$10,000 payments will not be due.

We may make 1 trip to your market at our own cost to visit sites you are considering for your Communities so that we can accept or not accept the sites. If we must make more than 1 on-site evaluation, you must reimburse our reasonable expenses, including the costs of travel, lodging, and meals. We estimate that such expenses will not exceed \$5,000 per visit. This payment is not refundable.

ITEM 6 OTHER FEES

Column 1	Column 2	Column 3	Column 4
Type of fee*	Amount**	Due Date	Remarks
Royalty/Continuing Fee	5% of monthly Net Billings during preceding calendar month (Note 1)	your bank account via EFT 10 days after date of our invoice	Note 1 defines "Net Billings." We may collect more frequently than monthly on 30 days' prior notice to you. See also Notes 3 and 6.
General Marketing Fee	Beginning on Community's Opening Date, the greater of 2.5% of the prior month's Net Billings or \$500 per month per Community for up to 3 Communities operating under the Franchise Agreement (Note 2)	Deducted from	We may collect more frequently than monthly on 30 days' prior
Technology & Email	If we decide to charge this fee, it	Deducted on 15 th	We do not charge this fee as of
Service Fee	will be due beginning on the Community's Opening Date. The fee amount is the greater of \$100 per month or .83% of the prior month's Net Billings for <u>each</u> Community operating under Franchise Agreement (Note 3) We have the right to increase this fee if our costs increase, in which case the \$100 fixed fee portion of the calculation may go up to no more than \$500 per month (per Community) during the course of the franchise term.	day of each month from your bank account via EFT	this disclosure document's issuance date.

Column 1	Column 2	Column 3	Column 4
Type of fee*	Amount**	Due Date	Remarks
Email Service Fee	email provided (Note 3). This fee	day of each month from your	This fee will apply if we do not require you to use certain System Technology in operating your Communities and therefore do not charge the Technology & Email Service Fee described above.
Electronic Client Management Software	\$150 per month per Community for up to 10 clients per Community. An additional \$10 per month for each client over 10 in each Community (Note 4)	day of each	
Vendor Evaluation	Up to \$5,000 if the vendor is approved.	As incurred	Evaluation of proposed new vendors to support the brand. Due only if we approve the vendor.
Supplemental Additional Training Programs	Not to exceed \$500 per day per attendee	Upon receipt of invoice	We provide initial training for up to 4 people at no additional charge; we may charge you for training replacements if your original trainees cannot complete training, for training additional or newly hired personnel, and for supplemental training programs. We may require your key personnel to attend supplemental training programs at your expense.
Additional On-site Training to be Provided on an as- needed Basis	Not to exceed \$500 per day per trainer, plus travel, meals, hotel, and related expenses for each trainer	Upon receipt of invoice	We may charge you for additional or special guidance, assistance, or training you need or request.

Column 1	Column 2	Column 3	Column 4
Type of fee*	Amount**	Due Date	Remarks
Annual Meetings	You must pay registration fees whether or not you attend, not to exceed \$2,000 per person per annual meeting, plus travel, room, board, and salary expenses of your attendees	Due upon invoice	We reserve the right to hold annual franchise meetings virtually. If we hold the annual franchise meeting (including branch leadership conference) virtually, you must attend and pay the registration fee regardless of your participation in the virtual annual meetings.
Product and Service Purchases	See Item 8	As incurred	You will buy products and services from us, our affiliates, designated and approved vendors whose items and services meet our standards and specifications, and/or other suppliers to the industry. Prices depend on supplier and item/service involved.
Insurance reimbursement	Out-of-pocket premium cost reimbursement plus 18% administrative fee	As incurred	You must reimburse us if we obtain required insurance for you.
Late Fee	\$250	May be auto- debited from your account	Due for each late or dishonored payment.
Interest	Lesser of 1.5% per month or highest commercial contract interest rate law allows.	May be auto- debited from your account	Due on all overdue amounts more than 7 days late.
Transfer Fee	\$15,000 plus any fees paid directly to a broker if buyer found through broker	Before or at transfer closing	Due if you transfer Franchise Agreement and Communities or your owners transfer controlling ownership interest in you or your owners.
Step-In Rights	Reimbursement of our costs and overhead while operating Community or Communities plus management fee of up to \$2,000 per day	As incurred	We have the right to operate one or more of your Communities in certain circumstances.

Column 1	Column 2	Column 3	Column 4
Type of fee*	Amount**	Due Date	Remarks
Management Fee	Manager's then-current daily salary plus our direct expenses	As incurred	Due if we manage Community temporarily after Control Person's death or disability.
Indemnification	Varies under circumstances and depends on nature of third-party claim	As incurred	You must reimburse us if we are held liable for claims from your Community's operation or incur costs to defend them (when we are not at fault).
Costs and Attorneys' Fees	Varies under circumstances and depends on nature of your non- compliance	As incurred	Due when we incur costs and expenses to enforce Franchise Agreement against you, whether or not we begin formal legal proceeding.
Microsoft Teams/One- drive/Web version office package Option	Beginning in the month you execute the Microsoft Team/One- drive/Web version office package Opt-in Addendum, you will pay \$7.80 per user per month (subject to change upon written notice)	Deducted monthly from your bank account by us via EFT	This is optional. The per-user, per-month fee is over and above any email fee.
Tax Reimbursement	Out-of-pocket cost reimbursement	As incurred	You must reimburse us for any taxes we must pay a state taxing authority on account of either your operation or your payments to us (except for our own income taxes).
Re-Inspection Fee	Up to \$2,500	As incurred	Due for each follow-up Community inspection to confirm you have corrected operating deficiencies we brought to your attention.
De-Identification Reimbursement	Out-of-pocket cost reimbursement	As incurred	You must reimburse our costs to de-identify your Communities after termination or expiration of Franchise Agreement if you fail to do so as required.

Column 1	Column 2	Column 3	Column 4
Type of fee*	Amount**	Due Date	Remarks
De-Identification Royalty Fee	\$1,000 per day		Due if you fail to comply with your Community de-branding and de-identification obligations after Franchise Agreement expires or is terminated.
Termination Damages	Note 5	15 days after effective date of	
		termination	

<u>NOTES</u>

* Except for product and service purchases described in Item 8, and except as otherwise noted in this Item 6, all fees are imposed and collected by and payable to us. Except as noted above, all fees currently are uniformly imposed. No fee is refundable.

** We have the right during the franchise term to increase certain fixed fees or payments described in the Franchise Agreement. An annual increase may occur only once during any calendar year and may not exceed the corresponding cumulative percentage increase in the Index since the effective date of the Franchise Agreement or, as the case may be, the date on which the last Annual Increase became effective for the particular fixed fee or payment being increased. Annual Increases will be made at the same time during the calendar year. "Index" means the Consumer Price Index for All Urban Consumers (CPI-U) for the U.S. City Average for all Items (1982 – 1984 = 100), not seasonally adjusted, as published by the United States Department of Labor, Bureau of Labor Statistics or in a successor index. The fixed fees/payments subject to increase during the franchise term are the General Marketing Fee, the late fee, the re-inspection fee, minimum required local marketing expenditures (described in Item 11), transfer fees, the "Step-In" per diem fee, and the de-identification royalty fee.

NOTE 1. Royalty fees are based on a percentage of Net Billings. "Net Billings" means the aggregate of all revenues and other income from whatever source (whether in the form of cash, credit, agreements to pay or other consideration, and barter transactions, and whether or not payment is received at the time of sale, or any amounts prove uncollectible) arising or derived by you or any other person from business conducted by or at, or originating from, the Community. Net Billings also includes all proceeds from any business interruption insurance.

Excluded from Net Billings are: (i) sales and other taxes separately stated that you collect from clients and actually pay to taxing authorities; (ii) the discount value of any coupon, voucher, or other allowance we authorize at the time you redeem the client's coupon, voucher or allowance, and (iii) no-mark-up items such as personal protective equipment, testing costs, or credit card fees

where the amount billed to client is at the franchisee's cost. All royalties will be collected via EFT 28 days after the end of the weekly billing period.

NOTE 2: In addition to the General Marketing Fee, you must comply with certain local advertising and marketing obligations described in Items 7 and 11 below.

NOTE 3: If we decide to begin charging the Monthly Technology & Email Service Fee (and we currently do not), that fee will be due and payable to BrightStar Technology or another affiliate via EFT beginning on the Community's Opening Date (unless we start charging this fee after your Opening Date, in which case it would start at that time). The Monthly Technology & Email Service Fee excludes any third-party approved solutions (e.g., Accounting/Payroll Software outside of System Technology), Electronic Health Records and Medication Management Software, Hireology, CRM, Electronic Client Management Software, HR/workforce or WorkBright), which may incur a separate fee, either billed directly by the vendor or billed collectively and passed through by us. If charged, the Monthly Technology & Email Service Fee will be due each month on the 15th day of the month for each Community operating under the Franchise Agreement during the franchise term. If we do not charge the Monthly Technology & Email Service Fee, BrightStar Technology will provide you with branded email. The branded email fee will be \$10 per email per user per month and will be due each month on the 15th day of the month sequence the franchise Agreement's effective date.

NOTE 4: Beginning on the Community's Opening Date, we will collect the fees from each Community and disburse to the vendor, or the fee will be charged directly to the Community.

NOTE 5: Termination damages are due if you cease operating a Community before the Franchise Agreement expires. The amount depends on the reason why the Community ceased operations and your actions afterward. Termination damages can range from \$100,000 to \$400,000 depending on the circumstances but typically will be calculated to equal the product of (a) either 36 or the number of months then remaining in the initial franchise term as of the effective date of termination (had the Franchise Agreement or your right to operate the particular Community not been terminated), whichever is shorter, multiplied by (b) the sum of the average monthly Royalty/Continuing Fees, plus the average monthly General Marketing Fees, plus the average Monthly Technology & Email Service Fee (if applicable) that were due and payable to us with respect to each affected Community during the 12 months before the month of termination.

NOTE 6: If you or your affiliate is a BrightStar Care Agency franchisee, you or that affiliate must ensure that your or its BrightStar Care Agency continues to receive the necessary attention during the development and operation of your Communities. If the Net Billings of one or more of your or your affiliate's BrightStar Care Agencies decline on a year-over-year basis by 20% or more during the development and operation of your Communities, you or your affiliate must send us a plan within 15 days following our notice to you of the decline with a description of the proposed cure for that decline, which then must be implemented within the following 90 days. The plan must include, at a minimum, additional infrastructure/roles or other revenue-producing activities proposed to support the operations in order to recover or recoup the year-over-year decline. Your or your affiliate's failure to recover or recoup the decline in the performance of the BrightStar Care Agency within 12 months from the date on which we first notify you of the year-over-year decline

in Net Billings of at least 20% will be good cause for our termination of your BrightStar Care Agency Franchise Agreement.

ITEM 7 ESTIMATED INITIAL INVESTMENT

YOUR ESTIMATED INITIAL INVESTMENT

(The first table covers your estimated initial investment for a 10-to-12-bed BRIGHTSTAR CARE HOMES Community if you buy the land and building)

Column 1	Column 2	Column 3	Column 4	Column 5
Type of expenditure	Amount	Method of payment	When due	To whom payment is to be made
Initial Franchise Fee (1)	\$50,000	Lump Sum	 \$50,000 upon signing the Franchise Agreement. If you or an affiliate is an existing BrightStar Care Agency franchisee, \$30,000 is due upon signing the Franchise Agreement for first Community and an additional \$10,000 for each of the second and third Communities is due when we accept sites for them 	Us
Architecture / Engineering / Development Fee & Expenses (2)	\$80,000 - \$225,000	Lump Sum	As invoiced	Third-Party Vendors
Site Demolition	\$0 - \$50,000	Lump Sum	As invoiced	Third-Party Vendors
Legal Expenses (3)	\$1,500 - \$5,000	Lump Sum	As invoiced	Third-Party Vendors
Fees for Applications, Permits Review, and Recording (4)	\$13,000 - \$20,000	Lump Sum	As invoiced	Government Agencies
Community's Construction (5a)	\$715,000 - \$1,121,000	Lump Sum	As invoiced	Third-Party Vendors

Column 1	Column 2	Column 3	Column 4	Column 5
Type of expenditure	Amount	Method of payment	When due	To whom payment is to be made
Land (5b)	\$200,000 - \$450,000	Lump Sum	As invoiced	Third-Party Vendors
Landscape Architect Fees (6)	\$35,000 - \$50,000	Lump Sum	As invoiced	Third-Party Vendors
Furniture, Fixtures & Equipment (7)	\$8,000 - \$32,000	Lump Sum	As invoiced	Third-Party Vendors
Technology/Computer Requirements (8)	\$4,000 - \$12,000	Lump Sum	As invoiced	Third-Party Vendors
Supplies & Materials Inventory (9)	\$15,000 - \$25,000	Lump Sum	As invoiced	Third-Party Vendors
Banking Fees, Interest, and Land Taxes During Construction (10)	\$49,000 - \$53,000	Lump Sum	As invoiced	Third-Party Vendors
Pre-Opening Marketing Costs (11)	\$16,000 - \$24,000	As incurred	Starting no later than 6 months before opening	Third-Party Vendors
Pre-Opening Labor Costs (12)	\$9,000 - \$13,000	As incurred	As incurred	Employees
Employee Travel and Living Expenses Associated with Training (13)	\$4,000 - \$7,500	As incurred	As incurred	Third-Party Vendors
Recruiting Spend (14)	\$900 - \$1,500	Lump Sum	As invoiced	Third-Party Vendors
General Marketing Fee (15)	\$1,500	As invoiced	As invoiced	Us
State Required Licensure (16)	\$1,400 - \$2,050	Lump sum	As incurred	Licensing Authority
Registered Nurse Hired Before Opening (17)	\$0 - \$9,375	As incurred	As incurred	Employee
Additional Funds – 3 months (18)	\$22,616 - \$50,795	As incurred	As incurred	Third-Party Vendors and Employees
Total Estimated Initial Investment (20)	\$1,225,916 - \$2,202,720			

YOUR ESTIMATED INITIAL INVESTMENT

(The second table covers your estimated initial investment for a 10-to-12-bed BRIGHTSTAR CARE HOMES Community if you lease the land and building)

Column 1	Column 2	Column 3	Column 4	Column 5
Type of expenditure	Amount	Method of payment	When due	To whom payment is to be made
Initial Franchise Fee (1)	\$50,000	Lump Sum	 \$50,000 upon signing the Franchise Agreement. If you or an affiliate is an existing BrightStar Care Agency franchisee, \$30,000 is due upon signing the Franchise Agreement for first Community and an additional \$10,000 for each of the second and third Communities is due when we accept sites for them 	Us
Legal Expenses (3)	\$1,500 - \$5,000	Lump Sum	As invoiced	Third-Party Vendors
Fees for Applications, Permits Review, and Recording (4)	\$0 - \$7,000	Lump Sum	As invoiced	Government Agencies
Furniture, Fixtures & Equipment (7)	\$8,000 - \$32,000	Lump Sum	As invoiced	Third-Party Vendors
Technology/Computer Requirements (8)	\$4,000 - \$12,000	Lump Sum	As invoiced	Third-Party Vendors
Supplies & Materials Inventory (9)	\$15,000 - \$25,000	Lump Sum	As invoiced	Third-Party Vendors
Pre-Opening Marketing Costs (11)	\$16,000 - \$24,000	As incurred	Starting no later than 6 months before opening	Third-Party Vendors
Pre-Opening Labor Costs (12)	\$9,000 - \$13,000	As incurred	As incurred	Employees
Employee Travel and Living Expenses Associated with Training (13)	\$4,000 - \$7,500	As incurred	As incurred	Third-Party Vendors
Rent & Lease Deposit	\$58,461 - \$73,077	Lump sum	As invoiced	Lessor

Column 1	Column 2	Column 3	Column 4	Column 5
Type of expenditure	Amount	Method of payment	When due	To whom payment is to be made
Recruiting Spend (14)	\$900 - \$1,500	As incurred	As incurred	Third-Party Vendors
General Marketing Fee (15)	\$1,500	As invoiced	As invoiced	Us
State Required Licensure (16)	\$1,400 - \$2,050	Lump sum	As incurred	Licensing Authority
Registered Nurse Hired Before Opening (17)	\$0 - \$9,375	As incurred	As incurred	Employee
Additional Funds – 3 months (19)	\$31,547 - \$66,848	As incurred	As incurred	Third-Party Vendors, Lessor, and Employees
Total Estimated Initial Investment (20)	\$201,308 - \$329,850			

<u>NOTES</u>

- No expenditure in these tables is refundable.
- The above estimates do not take into consideration any revenue derived during the first 3 months of operation.
- The above estimates are based upon the Communities developed and operated in Idaho under a pilot program by a BrightStar Care franchisee with an enclosed home size of approximately 4,000 to 5,500 square feet and a 0.75 to 1-acre parcel of land. If your planned Community's design deviates in any way from these Communities (and we consent to the deviations), certain costs described in Item 7 will be higher.
- (1) Initial Franchise Fee: We describe the initial franchise fee (and when it is due) in Item 5.

(2) Architect Fees and Expenses: You will need an architect to adapt the prototype to the local site and requirements for new construction or to design the renovations of existing buildings. The architect's work will include sketches and drawings providing a graphic delineation of the project with respect to height, bulk, scale, and architectural character of the prototype building and its relationship to surrounding properties. This estimate includes architectural, civil, mechanical, electrical, plumbing, structural, and landscaping construction drawings. Copies of the architect's work will be made available to us, you, and government agencies having jurisdiction over the proposed site. The cost shown includes the architect's fee, plus estimated reimbursable expenses

such as copying, renderings, travel, etc. The low end of this estimate takes into consideration that you utilize a developer, and it may be paying these fees.

Development Fees & Expenses: You may obtain development services from an approved vendor, who would manage the entire building delivery process as your representative. The cost shown includes the vendor's fee, plus estimated reimbursable expenses such as copying, renderings, travel, etc.

- After we approve localized Construction Documents, the vendor will help you obtain either competitive bids or negotiated proposals and determine the successful bid or proposal. During the construction phase, the vendor will:
 - Help you and the general contractor obtain permits and other approvals from agencies having jurisdiction over your project;
 - Review shop drawings, material samples, and other submittals for general conformance with the design concept;
 - Consider requests for substitutions;
 - Provide other construction phase related services such as processing applications for payment, acting upon requests for change orders by contractors, preparing final punch-lists and reviews, etc.; and
 - Work with the general contractor to obtain a Certificate of Occupancy and issue a Certificate of Substantial Completion at the conclusion of the project.

The cost shown includes the vendor's fee. The Architecture and Engineering Services Fees shown above are based on working from prototype building design documents. Modifications requiring additional Architecture and Engineering Services Fees and expenses will include changing configuration of the building, adding or deleting residential units, modifying residential units, adding or deleting amenity spaces, or re-organizing interior space layout. In addition, changes in codes (for example, building, life safety, and health department) and specific jurisdictional requirements might require adjustments to the fees shown above.

(3) Initial legal review of employment documents that should be reviewed annually to ensure they stay current with local and state employment laws.

(4) Fees for Applications, Review, and Recording: Various local government or regulating agencies may impose fees for reviewing your project during the entitlement phase. The low end of this estimate takes into consideration that you utilize a developer, and it may be paying these fees.

(5a) Community Construction: The Community's construction must comply with our thencurrent specifications. Construction costs vary widely depending on market conditions. The range of construction costs provided represents an estimate based on the actual costs of the Communities in Boise, Idaho. Your costs will depend on market conditions and local government requirements.

(5b) Land: We recommend a parcel of real estate approximately 0.75 to 1 acre in size. Land costs vary widely depending on market conditions. While we believe this to be generally representative of the land costs our franchisees may incur, your costs may differ significantly (higher or lower) from these estimates and will depend on local market conditions.

(6) Landscape Architect Fees: You will obtain landscape services to design your exterior landscape.

(7) Furniture, Fixtures & Equipment: Furnishings include furniture, case goods, lamps, artwork, window treatments, televisions, and accessories to furnish the Community's rooms, common areas, and utility areas. Cost of furnishings also includes equipment for nurse call system, building access control, shelving, and the central kitchen. The estimated costs for Furniture, Fixtures & Equipment differ between the Leasing and Purchase scenarios because, under the Leasing scenario, it is assumed that the Lessor (Developer) would include certain portions of the Furniture, Fixtures & Equipment in its total project cost.

(8) We describe the Technology/Computer Requirements in Item 11, which generally include computer hardware, software, peripheral items, property-management and other electronic-information systems, and technology solutions, including third-party accounting and payroll, electronic health records, medication management, and client relationship management software.

(9) Supplies & Material Inventory: You must purchase supplies, materials, and uniforms for dining services, maintenance, housekeeping, wellness/clinical, and life enrichment.

(10) Banking Fees, Interest, Land Taxes During Construction/Remodeling: You will pay land taxes during construction. You will also pay financing fees and interest to a lending institution if you are financing the project. This amount is approximately a 12-month range.

(11) Pre-Opening Marketing Costs: Pre-Opening Marketing costs include digital (website, PPC, display, etc.) and social media marketing, print, promotional wear, brochures, banners, sales office, promotions, groundbreaking ceremonies, signage, and other costs associated with the Community's opening.

(12) Pre-Opening Labor Costs: Pre-Opening Labor Costs include hiring the leadership team and direct care staff before the Community's opening. If you are an existing BrightStar Care Agency franchisee, some of this cost may be captured under your BrightStar Care Agency.

(13) Employee Travel and Living Expenses Associated with Training: The amount will vary based on the type of accommodations you select, dining preferences, travel preferences, and differences in compensation arrangements with your employees while the employees are being trained.

(14) Recruiting Spend: You will run a minimum of 3 ads per month for 3 months.

(15) General Marketing Fee: You will begin paying the General Marketing Fee on your Opening Date.

(16) State Licensure: Licensure requirements vary by state. Your state's requirements will determine your licensure costs. Some states may have Plan Review fees as a part of construction.

(17) Registered Nurse before Opening: Depending on your state's licensure requirement, you might need to hire a Registered Nurse (full or part-time) before Opening to include on your licensure application.

(18) Additional Operating Funds—3 months (Land and Building Purchase Option): This estimates the funds needed to cover your initial expenses for the first 3 months of operation (besides the items identified separately in the table). It does not include any scheduled loan payments for any items you have financed. It includes payroll costs but not any owner's draw or salary.

This is only an estimate, and you might need additional working capital during the Community's first 3 months of operation and for a longer time period afterward. We relied on our affiliate's franchisee's experience in developing and operating the BRIGHTSTAR CARE HOMES Communities in Boise, Idaho to compile this Additional Funds estimate.

(19) Additional Operating Funds—3 months (Leasing Option): This estimates the funds needed to cover your initial expenses for the first 3 months of operation (besides the items identified separately in the table). It includes payroll costs but not any owner's draw or salary. This includes all lines of required insurance except for property/renters' insurance.

This is only an estimate, and you might need additional working capital during the Community's first 3 months of operation and for a longer time period afterward. We relied on our affiliate's franchisee's experience in developing and operating the BRIGHTSTAR CARE HOMES Communities in Boise, Idaho to compile this Additional Funds estimate.

(20) Total: You should review these figures carefully with a business advisor before deciding to acquire the franchise. We do not offer financing directly or indirectly for any part of the initial investment. The availability and terms of financing depend on many factors, including the availability of financing generally, your creditworthiness and collateral, and lending policies of financial institutions from which you request a loan.

ITEM 8 RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

General Comments

You must operate your Communities in strict compliance with the methods, standards, and specifications we prescribe in our confidential operating manual and other confidential materials we prepare for a BRIGHTSTAR CARE HOMES Community (collectively, the "Operations Manual"), all of which we may change. The Operations Manual covers nearly all aspects of your Community's operation, including processes and procedures, resident service techniques, and administration. You may offer only approved Services and products at the Community, including housing, meals, life enrichment activities, personal care, and medication care.

All fixtures, furnishings, furniture, equipment, signs, other operating assets and supplies, and services you acquire for the Community must meet our standards and specifications. We have the right to require you to buy certain equipment, services, and products (e.g., food service, medical supplies, and office supplies) only from us, our affiliates, or other suppliers or distributors we approve or designate, which in certain instances may be a single-sourced supplier. We have the right to restrict these items' sources to protect trade secrets and other intellectual property rights, assure quality, assure a reliable supply of products meeting our standards, achieve better terms of purchase and delivery service, control usage of the Marks by third parties, and monitor the manufacture, packaging, processing, and sale of these items.

Currently, you must use our designated suppliers for medical supplies (where alternative sources of supply are unavailable), marketing materials, credit card processing services, payroll services (unless we approve your request to use an alternative supplier), drug screening, insurance brokerage for all types of insurance coverage except employee benefits, operations software, and email. You also must use our required vendor partner for any SEM/PPC advertising investments to ensure optimal campaign performance and brand continuity. If we do not require you to use our System Technology (and we currently do not) or other designated operations software, you must obtain your own accounting software, payroll, Electronic Health Records and Medication Management Software, and CRM for management of your client census as well as the supporting hardware and software necessary to operate your Community. If we require you to use our System Technology, you will have the option to request our permission to use an alternative vendor for payroll services. If we grant your written request, you will be allowed to use such vendor's payroll software. If we require, you must also obtain your accounting software from us or our affiliate (currently BrightStar Technology) or any third party we designate. You must also purchase at your sole expense any software necessary to meet state-specific requirements of a state program or segment of business unique to your location. Any request to use a non-designated supplier must be submitted to us in writing. We will review all written requests and respond within 15 business days. You may not use a non-designated supplier without our express written permission.

We recommend, but do not require you to use, specific suppliers for signage, equipment, and supplies. We may develop certain proprietary products you must buy from us or our affiliates and offer for sale at your Communities. We and our affiliates currently are the designated supplier of

branded email account, website, and intranet administration and content management but are not designated or approved suppliers of any other items.

We will issue and modify standards and specifications based upon marketplace and reimbursement changes, the collective experience of our affiliate's franchisee and our principals' experience in operating BRIGHTSTAR CARE HOMES Communities. Standards and specifications may impose minimum requirements for production, performance, reputation, prices, quality, design, and appearance. Our Operations Manual or other communications will identify our standards and specifications for you and (where appropriate) suppliers or only for suppliers (in the latter case where, for example, we give our standards and/or specifications to a supplier under a confidentiality agreement). There might be situations where you can obtain items from any supplier who can satisfy our requirements and therefore would be an approved supplier. We may change standards and specifications upon written notice to you or through changes in the Operations Manual. You may incur increased costs to comply with these changes; however, no change will materially alter your fundamental rights under the Franchise Agreement.

Except for those instances where we have designated a single-sourced supplier (which may be us or an affiliate), if you wish to buy any unapproved item or services, including inventory, or acquire approved items or services from an unapproved supplier, you must provide us the proposed supplier's name, address, and telephone number, a description of the item you wish to purchase, and the item's purchase price, if known. At our request, you must provide us, for testing purposes, a sample of the item you wish to purchase. We periodically will establish procedures for your requests and may limit the number of approved items, services and/or suppliers as we think best. We may charge a vendor evaluation fee only if we approve the supplier. We need not, and have no intent to, approve your request if we already have designated specific items, services and/or suppliers or otherwise have restricted the supply system. Supplier approval might depend on product quality, delivery frequency and reliability, service standards, financial capability, customer relations, concentration of purchases with limited suppliers to obtain better prices and service and/or a supplier's willingness to pay us, our affiliates and/or our system to do business with our system. We will notify you in writing (via email or otherwise) of our approval or disapproval of a proposed supplier, product, or service within 30 days after receiving all requested information. Supplier approval might be temporary until we evaluate the supplier in more detail. We may revoke our approval of particular products or suppliers when we determine that such products or suppliers no longer meet our standards or for other business reasons, we deem best. You must stop purchasing products from a supplier as soon as you receive our notice. You may not use any unapproved products or services. You must use products purchased from approved suppliers solely for operating your Community and not for any competitive business purpose. We have the absolute right to limit the suppliers with whom you may deal. One of our officers owns an interest in BrightStar Technology. No officer owns an interest in any other supplier.

You must pay the then-current price for the items you purchase from us or our affiliates. In some instances, the cost for the items you purchase from us or our affiliate may be higher than the cost of other similar supplies and products on the market.

You must use only business stationery, business cards, marketing materials, advertising materials, printed materials, and forms we approve in advance and meeting our standards. You must send us at least 15 days before planned publication or use samples of all Marketing you want to use that we have not previously approved. ("Marketing" means marketing, advertising, promotional, client relationship management (CRM), public relations, and brand building and protection activities and programs; recruiting sites; sales training programs and content; advocacy; and training courses based upon discount for system purchase and distribution.) We will notify you in writing within 10 days whether we approve or disapprove the materials. You may not use any Marketing materials we have not expressly approved or have disapproved. (Our failure to respond within 10 days will be deemed disapproval.) You must use our required vendor partner for any SEM/PPC advertising investments to ensure optimal campaign performance and brand continuity.

Insurance

You must carry insurance meeting our minimum coverage conditions. For each line of coverage below, you must (at your own cost) name "BrightStar Group Holdings, Inc., its subsidiaries, officers, directors, and employees, and any other person or entity we designate" as additional insureds. Your current insurance requirements are as follows:

(a) Professional Liability with limits <u>not</u> less than \$1,000,000 per occurrence / \$3,000,000 aggregate per policy year. If you purchase this coverage on a claims-made policy form, you must purchase a minimum 3-year extended reporting endorsement (aka "tail") upon sale or closure of your business.

(b) General Liability with limits not less than \$1,000,000 per occurrence / \$3,000,000 aggregate per policy year. The following minimum sub-limits must be met: \$1,000,000 Personal & Advertising Injury, \$3,000,000 Products/Completed Ops Aggregate, \$100,000 Damage to Rented Premises, and \$5,000 Medical Expense. If you purchase this coverage on a claims-made policy form, you must purchase a minimum 3-year extended reporting endorsement (i.e., "tail") upon sale or closure of your business.

(c) Abuse/Molestation Liability with limits not less than \$1,000,000 per occurrence and aggregate per policy year.

(d) Hired and Non-Owned Automobile coverage not less than \$1,000,000 combined single limit each accident.

(e) Special Form property insurance in an amount appropriate to cover the full replacement value of the Community (if owned) and its contents, including flood and any other insurance that is appropriate for the Community's location if the primary policy has exclusions for such events. Business Income and Extra Expense must be included at a minimum of 1.5 times annual revenue assuming full occupancy and covering an 18-month period following the loss.

(f) Workers' Compensation and Employer's Liability insurance with minimum limits not less than \$500,000 per accident; \$500,000 policy limit by disease; and \$500,000 per employee by disease or higher limit as required by law in your state.

(g) \$50,000 Employee Dishonesty and \$50,000 Theft of Client Property.

(h) Cyber Liability with 500,000 minimum limit to include both administrative costs of the privacy event and 3^{rd} party losses.

(i) Umbrella Liability with a \$2,000,000 minimum limit to extend over professional liability, general liability, and employer's liability. The umbrella may contain a sub-limit of \$1,000,000 for abuse/molestation.

(j) Employment Practices Liability (EPL) with \$1,000,000 minimum limit covering indemnification and defense costs for employee allegations of harassment, discrimination, and wrongful termination practices. Coverage must include a "3rd Party Endorsement" to respond to client allegations of similar wrongful acts, to include at minimum \$100,000 for Wage & Hour defense costs. You must purchase a minimum 12month extended reporting endorsement (i.e., "tail") upon sale or closure of your business.

(k) Any other insurance not listed here but required by applicable law, rule, regulation, ordinance, or licensing requirements and any updates made from time to time in the operations manuals.

(1) Policy and Coverage level deductibles may not exceed \$25,000 for any coverages required unless we grant a written waiver.

These insurance specifications represent minimum required coverage and limits. We do not warrant these requirements will cover any and all losses potentially sustained while operating the Community. Additional coverage and/or higher limits might be advisable based upon your Community's location or unique operations. You should explore all available options with your insurance agent to ensure your insurance program is appropriate for your individual risk tolerance and financial position. (You also must comply with your landlord's insurance requirements if different from those described above.)

In order to monitor claims activity on a national level and effectively assess program exposures, you must collect Loss History Statements ("Loss Runs") from the carriers and remit Loss Runs to us when requested. Certain states also require a surety bond for licensure purposes. These minimum amounts may be revised periodically in the Operations Manual to reflect inflation, general industry standards, or our future experience with claims.

The Community must be at a site that we accept. (Item 11 describes the site selection process.) If you will occupy the Community's premises under a lease, you must send us the lease for review and written acceptance before you sign it. We will have 14 calendar days after receiving the lease to accept or reject it. Our acceptance will be conditioned on your execution of a Collateral

Assignment of Lease in the form we prescribe and the lease's inclusion of certain terms and conditions.

You are responsible for constructing and developing the Community. We will give you template plans and drawings ("Plans") for a BRIGHTSTAR CARE HOMES Community's physical structure, exterior elements, and interior layout. Plans are not defined as full architectural drawings but will be a package providing enough detail to engage with an architect to create such plans. The Plans will include mandatory and suggested standards and specifications for, among other things, size, dimensions, design, image, decor, fixtures, equipment, signs, furnishings, and color scheme. However, the Plans might not reflect the requirements of any federal, state, or local law, code, or regulation, including those arising under the Americans with Disabilities Act ("ADA") or similar rules governing public accommodations for disabled persons. You must hire an architect and/or engineer whom we accept to prepare a site survey and all required construction plans and specifications to suit the site and make sure they comply with our requirements, the ADA and similar rules, other applicable ordinances, building codes, permit requirements, and occupancy requirements and restrictions. Depending on state regulations, the full architectural plans might be owned by the architect and can be separately purchased from the architect. You must send us, for review and acceptance, your site survey and construction plans and specifications before submitting them for permitting and beginning construction and all revised or "as built" plans and specifications during construction. Our review is only to ensure your compliance with our design and layout requirements.

You must identify, and we first must accept, your proposed general contractor before you begin constructing the Community. You alone are responsible for the performance of architects, engineers, contractors, and subcontractors you hire to construct, develop, and maintain the Community and for ensuring that sufficient insurance coverage is in place before, during, and after the construction process.

Except as provided above, there are no goods, services, supplies, fixtures, equipment, inventory, computer hardware and software, real estate, or comparable items related to establishing or operating the Community that you currently must buy or lease from us (or an affiliate) or designated suppliers or in compliance with our standards and specifications. We estimate that the costs of your purchases and leases from designated or approved sources, or according to our standards and specifications, are approximately 88% to 94% of the total cost to establish, and approximately 88% to 96% of the total cost to operate, your Community. Any purchases from us and our affiliates, whether required or voluntary, generally may be at prices exceeding our costs so that we can make a profit (plus applicable taxes and shipping charges) (specific pricing depends on the particular item/service involved).

We and our affiliates have the right to receive payments from suppliers on account of their actual or prospective dealings with us, our affiliates, you, and other franchisees and to use all amounts received without restriction for any purposes we and our affiliates deem appropriate. However, we (and our affiliate) did not receive any revenue during 2023 on account of purchases made by a BRIGHTSTAR CARE HOMES Community franchisee directly from us (or it) or from a third-party supplier.

There currently are no purchasing or distribution cooperatives. We and our affiliates currently negotiate purchase arrangements with suppliers (including price terms) for pre-construction evaluations, construction managers, market surveyors, architects, interior decorators, and operations software. In doing so, we and our affiliates seek to promote the overall interests of the franchise system and affiliate-owned operations and our interests as the franchisor. We do not provide material benefits to you (for example, renewal or granting additional franchises) for purchasing particular products or services or using particular suppliers.

ITEM 9 FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the franchise and other agreements. It will help you find more detailed information about your obligations in these agreements and in other items of this disclosure document.

	Obligation	SECTION IN AGREEMENT	Disclosure Document Item
a.	Site selection and acquisition/lease	1.2 of Franchise Agreement and Site Selection Addendum (Exhibit H)	7, 8, 11, and 12
b.	Pre-opening purchases/leases	1.4, 3.1.2, 6.4, 6.8.1, 6.8.2, 6.8.4, 6.9, and 6.11 of Franchise Agreement	5, 7, 8, and 11
c.	Site development and other pre-opening requirements	3.1.2, 6.1, 6.2, 6.3, 6.6, and 6.8.8 of Franchise Agreement	7, 8, and 11
d.	Initial and ongoing training	6.5 and 6.16 of Franchise Agreement	6, 7, and 11
e.	Opening	6.1 and 6.2 of Franchise Agreement	11 and 17
f.	Fees	1.4, 4, 6.5.4, 6.5.5, 6.5.6, 6.5.7, 6.5.8, 6.9.1, 6.9.2, 6.9.4, 6.14, 6.16, 8, 9.3, 10.1, 10.2, 11.2, 13.4.1.8, 13.5.2, 14.9, 15.2, 16.2, 17.4, 19.1, 20.1, and 20.2 of Franchise Agreement	5, 6, and 7
g.	Compliance with standards and policies/operating manual	6, 7, and 9 of Franchise Agreement	8 and 11
h.	Trademarks and proprietary information	5, 6.17, 8.4, 12.2, and 12.3 of Franchise Agreement	11, 13, and 14

	Obligation	SECTION IN AGREEMENT	Disclosure Document Item
i.	Restrictions on products/services offered	1.2, 6.3, and 6.8.3 of Franchise Agreement	8, 11, 12, and 16
j.	Warranty and customer service requirements	6.6 of Franchise Agreement	11
k.	Territorial development and sales quotas	Not Applicable	Not Applicable
1.	Ongoing product/service purchases	1.4, 6.4, 6.8.1, 6.8.2, 6.8.4, 6.9, and 6.11 of Franchise Agreement	6 and 8
m.	Maintenance, appearance, and remodeling requirements	2.2.1, 6.7, 6.14, 9.1, 9.2, 9.3, and 10.2 of Franchise Agreement	8 and 17
n.	Insurance	17 of Franchise Agreement	6, 7, and 8
0.	Advertising	1.2, 4.4, 6.3, and 8 of Franchise Agreement	6, 7, 8, and 11
p.	Indemnification	20.1 of Franchise Agreement	6
q.	Owner's participation/ management/staffing	6.5 and 6.6 of Franchise Agreement	11 and 15
r.	Records and reports	6.8.5, 6.8.6, 6.8.7, and 11 of Franchise Agreement	Not Applicable
s.	Inspections and audits	6.14 and 11.2 of Franchise Agreement	6 and 11
t.	Transfer	13 of Franchise Agreement	17
u.	Renewal	2.2 of Franchise Agreement	17
v.	Post-termination obligations	15 of Franchise Agreement	17
w.	Non-competition covenants	12.4, 13.6, and 15.3 of Franchise Agreement	15 and 17
x.	Dispute resolution	16 of Franchise Agreement	17

	OBLIGATION	SECTION IN AGREEMENT	DISCLOSURE DOCUMENT ITEM
у.	Other:		
	Compliance with Anti- Terrorism Laws	27 of Franchise Agreement	17
	Spousal Consent	26 of Franchise Agreement	15
	Personal Data	6.8.7 of Franchise Agreement	Not Applicable
	Obtain Licensure	6.8.8 and 6.10 of Franchise Agreement	Not Applicable
	Compliance with Pricing Standards and Policies	6.8.9 of Franchise Agreement	16
	Condemnation	10 of Franchise Agreement	Not Applicable
	Personal Guarantee and Agreement to be Bound	18.3 and Exhibit D of Franchise Agreement	15

ITEM 10 FINANCING

We do not offer direct or indirect financing. We do not guarantee your note, lease, or obligation.

ITEM 11 FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND TRAINING

Except as listed below, we are not required to provide you with any assistance.

- A. After you sign your Franchise Agreement:
 - a. We will give you our site selection criteria for the Community.
 - b. We may make 1 trip to your market at our own cost to visit sites you are considering so that we can accept or not accept the sites. If we must make more than 1 on-site evaluation, you must reimburse our reasonable expenses, including the costs of travel, lodging, and meals.

(Franchise Agreement—Exhibit H) There are certain "milestones" you must satisfy after signing the Franchise Agreement. The over-arching requirement is that you must find and

secure a suitable site for your first Community within a designated area within 1 year after signing the Franchise Agreement. The milestones you must satisfy during the 1 year are the following:

1. Acceptance of Community Site Prior to Purchase or Lease (Franchise Agreement— Exhibit H)

- a. You must find and secure at your own expense, and we must accept, the land or building for your first Community within 12 months after the Franchise Agreement's effective date. Your failure to do so is a default under the Franchise Agreement which, if not cured, allow us to terminate the Franchise Agreement.
- b. You must send us the following information when seeking our acceptance of a Community site:
 - i. Site/Building details, including floor plan and/or plot diagram based on survey or plat map.
 - ii. Feasibility analysis of zoning and entitlement for projected number of units including Homeowners Association or other covenants or restrictions.
 - iii. Documentation of required licensure for the site and whether any special use permit is required.
 - iv. Research into competition within 5 miles of site.
 - v. Initial Pricing Assumptions based on competitive shopping.

We will have 21 calendar days after receiving such information and materials to accept or reject the site as a location for the Community. No proposed site will be deemed accepted unless we have expressly accepted it in writing.

If you will occupy the Community's premises under a lease, you must send us the lease for review and written acceptance before you sign it. We will have 21 calendar days after receiving the lease to accept or reject it. Our acceptance will be conditioned on your execution of a Collateral Assignment of Lease in the form we prescribe and the lease's inclusion of certain terms and conditions. You must give us a copy of the executed lease and Collateral Assignment of Lease within 14 days after signing them.

2. Approval of Architectural Drawings Prior to Start of Construction (Franchise Agreement—Section 3.1.2)

a. Before starting construction/renovation, you must send us for approval or revision the Community's interior and exterior architectural plans and drawings. The plans should include details on exterior and interior materials, such as siding, windows, roofing, doors, and plumbing, all of which must comply with our then-current BRIGHTSTAR CARE HOMES Community standards. The Community's final asbuilt construction must be consistent with the approved plans. 3. Passing Pre-Opening Inspection and Addressing Any Identified Deficiencies (Franchise Agreement—Sections 3.1.2 and 6.1)

- a. Before your Opening Date and before you accept any residents, you must arrange for our on-site inspection of the finished property and operational readiness (the "Pre-Opening Inspection").
- b. Any deficiencies identified during the Pre-Opening Inspection must be corrected before the Opening Date and before you accept any residents. We may allow certain deficiencies to be addressed after the Opening Date.

If you do not complete to our satisfaction each of tasks in these 3 steps on or before its respective deadlines (and do not cure the default within any applicable timeframe), we have the right to terminate the Franchise Agreement.

B. After you sign your Franchise Agreement, but before you open your Community, we will:

1. Grant you access to the Operations Manual via our intranet. (Franchise Agreement – Sections 3.1.1 and 7). We may modify the Operations Manual by written or online supplements. Our Operations Manual currently contains a total of 180 pages; its Table of Content is Exhibit C.

2. Give you the plans and drawings for the BRIGHTSTAR CARE HOMES Community's prototype physical structure, exterior elements, and interior layout. These Plans will include mandatory and suggested standards and specifications for, among other things, size, dimensions, design, image, decor, fixtures, equipment, signs, furnishings, and color scheme. (Franchise Agreement – Section 3.1.2) (These plans and drawings are not your Construction Documents. You will hire a vendor we accept, who will create Construction Documents specific to your Community.) We do not deliver or install items such as equipment, signs, fixtures, opening inventory, and supplies.

Help you with goal-setting and business planning. (Franchise Agreement – Section 3.1.3)

4. Provide a training program, described later in this Item. (Franchise Agreement – Sections 3.1.4 and 6.5)

5. While we are not obligated to assist you in the regulatory process (i.e., licensing), we have resources available as a reference for you.

C. During your operation of the Community, we will:

1. Consult with you in response to your inquiries about specific administrative and operating issues. We may decide how best to communicate with you, whether by telephone, in writing, electronically or in person. (Franchise Agreement – Section 3.2.1)

2. Administer the General Marketing Fund described in more detail below and approve advertising you create for your local use. (Franchise Agreement – Sections 3.2.2 and 8.1)

3. Make goods and services available to you either directly or through designated and/or approved suppliers. (Franchise Agreement – Section 3.2.3)

4. Periodically revise the Operations Manual to incorporate new developments and changes in the System and give you electronic access to all updates. (Franchise Agreement – Sections 3.2.4 and 7.3)

5. Let you use confidential information. (Franchise Agreement – Sections 7, 12.2, and 12.3)

6. Let you use the Marks. (Franchise Agreement – Section 5)

7. We will recommend, or otherwise assist you in establishing, the prices that your Community should charge for its various services. We also have the right to regulate (to the extent that federal and state antitrust laws allow) price advertising policies and maximum, minimum, or other pricing requirements for the Community's services, including requirements for national, regional, and local promotions. (Franchise Agreement – Sections 3.1.3 and 6.8.9)

Advertising Services

A. <u>General Marketing Fund (the "Fund")</u>. (Franchise Agreement – Section 8.1)

You must contribute the General Marketing Fee to the Fund. The General Marketing Fee currently is the greater of 2.5% of monthly Net Billings during the preceding calendar month or \$500 per month per community for up to 3 Communities under the Franchise Agreement during the franchise term. BRIGHTSTAR CARE HOMES Communities that we and our affiliates own and operate will contribute to the Fund on the same basis as franchisees (in terms of how the contribution is calculated). We may deposit into the Fund any allowances paid by suppliers doing business with BRIGHTSTAR CARE HOMES Communities. While all franchisees currently will be obligated to contribute at the same rate, the Fund had no operations during 2023.

We will direct all Marketing that the Fund finances, with sole control over the creative concepts, graphics, materials, communication media, and endorsements used and their geographic, market, and media placement and allocation. (As noted above, "Marketing" means marketing, advertising, promotional, client relationship management (CRM), public relations, and brand building and protection activities and programs; recruiting sites; sales training programs and content; advocacy; and training courses based upon discount for system purchase and distribution.)

The Fund may pay for creating, preparing, and producing video, audio, and written materials, graphics, and electronic and social media; developing, implementing, operating, and maintaining the Franchise System Website, Intranet and/or related strategies; administering national, regional,

multi-regional, and local Marketing, including, without limitation, purchasing media advertising, conducting direct mail and other direct marketing campaigns, doing on-line Internet advertising and marketing (including search engine marketing / pay per click charges to search engines, banner advertising sources, and advertising host sites), conducting research and other Marketing tactics and using advertising, promotion, CRM, graphic design, marketing, and research agencies and other advisors to provide assistance; supporting public relations, market research, client satisfaction surveys, and other Marketing and research activities; paying dues for membership and participation in franchising and industry associations; paying third-party vendors to customize Marketing materials for local use by franchisees; and engaging in other brand enhancement activities. The Fund periodically may give you samples of Marketing formats and materials at no cost. We may sell you multiple copies of these materials at our direct production cost, plus any related shipping, handling, and storage charges.

We will account for the Fund separately from our other monies (although we need not keep Fund contributions in a separate bank account) and not use the Fund for our general operating expenses. However, we may use the Fund to pay the expenses we incur in activities reasonably related to directing the Fund and its programs, including market research; public relations; creating, preparing, and producing Marketing materials and collecting and accounting for Fund contributions; reasonable salaries and benefits of personnel who manage and administer the Fund; a management fee for us (or an affiliate); the Fund's other administrative costs, including taxes we must pay on Fund contributions; travel expenses of personnel while on Fund business; meeting costs; overhead relating to Fund business; and franchisee conferences.

We may assist franchisees to maintain high quality standards through client surveys, customer interviews, and similar initiatives ("Surveys"). The Fund may pay for these Surveys, although we charge you for these costs if your Survey results fall below our minimum standards. We will deposit these amounts into the Fund.

Although the Fund is not a trust, we will hold all Fund contributions for the contributors' benefit and use contributions only for their permitted purposes (described above). We have no fiduciary obligation to you for administering the Fund or for any other reason. The Fund may spend in any fiscal year more or less than the total Fund contributions in that year, borrow from us, our affiliates, or others (paying reasonable interest) to cover deficits, or invest any surplus for future use. Deficits may be repaid from future contributions. We may accumulate monies in the Fund for the time periods we deem best, with no obligation to spend any or all monies received in any fiscal year during that fiscal year. At the end of our fiscal year, any unused Fund contributions will roll over for use during the following fiscal year. If the Fund is in a deficit position from time to time, future contributions to the Fund may be used to cover that deficit.

The Fund will not be used principally to develop materials and programs to solicit franchisees. However, media, materials, and programs, including the Franchise System Website, prepared using Fund contributions may describe the franchise program, reference the availability of franchises and related information, and process franchise leads. We will prepare an annual, unaudited statement of Fund collections and expenses and either give you a copy upon written request or post the statement on the Franchise System Website or Intranet. We may have the Fund audited annually, at the Fund's expense, by an independent certified public accountant we select. We may incorporate the Fund or operate it through a separate entity when we think best. The successor entity will have all of the rights and duties described here.

The Fund is to maximize awareness, and enhance system protection, of the Marks and increase occupancy at BRIGHTSTAR CARE HOMES Communities. Although we will try to use the Fund to develop Marketing materials and execute Marketing activities and programs benefiting all BRIGHTSTAR CARE HOMES Communities, we need not ensure that Fund expenditures in or affecting any geographic area are proportionate or equivalent to Fund contributions by BRIGHTSTAR CARE HOMES Communities operating in that geographic area or that any BRIGHTSTAR CARE HOMES Communities operating in that geographic area or that any BRIGHTSTAR CARE HOMES Community benefits directly or in proportion to its Fund contributions from the development of Marketing materials or the execution of Marketing activities and programs. (We also have no obligation to advertise in your market area separate from the Fund.) We may use collection agents and institute legal proceedings at the Fund's expense to collect Fund contributions. We also may forgive, waive, settle, and compromise all claims by or against the Fund. We assume no other direct or indirect liability or obligation to you for collecting amounts due to, maintaining, directing, or administering the Fund.

We may at any time defer or reduce a franchisee's Fund contributions and, upon 30 days' prior written notice to you, reduce or suspend Fund contributions and operations for one or more periods of any length and terminate (and, if terminated, reinstate) the Fund. If we terminate the Fund, we will at our option spend all remaining Fund monies on permitted Marketing or distribute all unspent monies to Fund contributors in proportion to their respective contributions during the preceding 12-month period.

We may use the Fund for joint or collective advertising campaigns with our direct or indirect parent corporations, subsidiaries, or affiliated companies designed to promote the BrightStar Mark and other Marks (including, for example, campaigns and other advertising activities promoting both BRIGHTSTAR CARE HOMES[®] Communities and BRIGHTSTAR SENIOR LIVING Communities).

Anyone who buys a franchise after you do and signs a different form of Franchise Agreement might pay a different General Marketing Fee than you pay.

There currently are no local or regional advertising cooperatives and no franchisee advertising councils that advise us on advertising policies. However, we have the power at any time to form, change, dissolve, or merge a franchisee advertising council.

B. <u>Local Advertising</u>. (Franchise Agreement – Sections 6.3, 8.2, 8.3, and 8.5)

Beginning with your required Pre-Sale Activities, you must spend at least \$2,000 for each planned client room in each Community for pre-opening Marketing activities. Beginning on each Community's Opening Date and during each month afterward, you must spend at least \$100 per month for local Marketing for each room in the Community if your occupancy rate is below 85%. If your occupancy rate is 85% or greater, you must spend at least \$50 per month for local Marketing for each room in the community. (We may increase these dollar amounts based on inflation and local market conditions.) Costs related to charges from referral companies (e.g. A Place for Mom,

etc.) do not count toward your local Marketing spend requirements. Your local Marketing (including through digital, video, and social media platforms) must be created from our libraries of approved marketing and advertising materials and be conducted in a dignified manner and conform to our standards and requirements.

You must comply with all exterior local signage regulations. In most instances, a residential neighborhood will not permit business signage.

You must place or display at the Community's premises (interior and exterior) only the signs, emblems, lettering, logos, and displays and Marketing materials we approve in writing. You may not use any Marketing materials unless they are from our libraries of approved marketing and advertising materials. Any Marketing materials not from our approved libraries must be requested from us through a creative brief. If we approve your creation of local Marketing materials, those materials must be submitted for our approval and align with our brand standards. You may not use any Marketing materials we have not expressly approved or have disapproved. (Our failure to respond within 10 days is considered disapproval.) All Marketing must prominently display, and comply with our standards for using, the Marks. We may require you to stop using any Marketing materials within specific timeframes, at your sole cost and expense.

In addition to the Marketing obligations described above, at least 90 days before your first Community opens, you must review listings and invest in Google Business Profile, SEM/PPC, Meta/Facebook and Yellowpages.com. The advertisement must conform to our standards and specifications. We may use the Fund to cover online advertising for all franchisees. Until that time, you must pay for online advertising as part of your local Marketing requirement. You must use our required vendor partner for any SEM/PPC advertising investments to ensure optimal campaign performance and brand continuity.

You must participate in system-wide and applicable regional promotions and advertising campaigns we originate or approve. These may include promotions via the Internet, e-commerce, electronic media, or other technologically advanced media. You also must participate at your sole expense in, and comply with the rules and procedures of, all client loyalty, gift certificate, and similar programs we create.

C. <u>Website and Intranet Matters</u>. (Franchise Agreement – Section 8.4)

<u>Franchise System Website</u>. You may not without our prior written consent develop, maintain, link to, or authorize any website mentioning or describing you or the Community, or displaying any Mark, other than on a website we host. "Website" means any part of the Internet (including social media platforms) used as a commercial computer network by the public and any successor technology enabling the public to research and/or purchase services or goods by means of electronic commerce. We may establish a website providing information about the System and its Services and products (the "Franchise System Website").

If we establish a Franchise System Website, we may give you a separate interior webpage (accessible only through the Franchise System Website) referencing your Community and/or otherwise allow you to participate in the Franchise System Website. You will be trained on the

CMS (content management system) to update your webpage. If you elect to hire an external vendor to make changes on your behalf, we must pre-approve the vendor. We must pre-approve your webpage before it goes live on the Franchise System Website and have the continuing right to monitor and pre-approve your webpage's form, content, and quality. Your webpage must comply with our System Standards at all times.

We will control, and may use Fund contributions to develop, maintain, operate, update, and market, the Franchise System Website, including your webpage. We will update information or add anything on your webpage as frequently as we deem appropriate. You must notify us whenever any information on your webpage changes or is not accurate.

<u>Intranet</u>. We may establish and maintain an Intranet and System standards for the Intranet's use. If you do not comply with any Intranet System standard, we may temporarily suspend your access to any chat room, bulletin board, list-serve, or similar feature the Intranet includes until you fully cure the breach.

Typical Length of Time Before Opening

First Community

You must open your first Community for business within 550 days after signing the Franchise Agreement. The term "open" means that the Community is fully licensed to operate as an assisted-living/memory-care Community in your market area, the Community is prepared to receive its first residents, and we approve the Community to open. You may not open and begin operating the Community without our prior written consent. We may terminate the Franchise Agreement if you fail to commence operations within 550 days after the Franchise Agreement's effective date. (Franchise Agreement – Sections 6.1 and 14.4.2) The specific opening timeframe will depend on various factors, including build-out schedule, installing equipment and furniture, completing required training, obtaining licensure from applicable regulatory agencies, and printing/mailing lead time for pre-opening recruiting plan and material.

You must complete certain steps in the first Community's construction and development process on or before each of the following specific deadlines:

(1) Once we accept the site, you have 6 months to begin construction. If you or your affiliate purchased (or intends to purchase) the real property for the premises and will own the Community's physical structure, the term "Begin Construction" means that (a) you (or your affiliate) have secured all financing required to construct, develop, and operate the Community in compliance with System standards, (b) you (or your affiliate) have selected and formally hired (with a signed contract) a general contractor, (c) you (or your affiliate) have obtained all relevant building permits, and (d) the construction company has taken concrete steps consistent with beginning construction work in earnest, such as erecting a fence or clearing the site. [If you intend to lease (rather than purchase) the real property or building for the premises and the Community's physical structure, the term "Begin Construction" means that (a) the developer has taken concrete steps consistent with beginning construction work in earnest, such as erecting a fence or clearing the site. (b) the developer has approved the lessee, and (c) you have secured all required financing,

executed a 20-year lease for the premises, and paid any relevant deposits. (Franchise Agreement—Section 6.1)

(2) The premises and Community must have reached Complete Construction within 550 days after the Franchise Agreement's effective date. The term "Complete Construction" means that you have received a Certificate of Occupancy from local authorities (Franchise Agreement—Section 6.1); and

(3) The Community must Open for business within 30 days after you receive the Certificate of Occupancy (the "Opening Date"). The term "Open" means that (a) the Community is fully licensed to operate as an assisted-living/memory-care Community in your market area, (b) the Community is prepared to receive its first residents, and (c) we approve the Community to open. (Franchise Agreement—Section 6.1)

You may not open and begin operating any Community without our prior written consent. Your failure to satisfy any of the deadlines identified in clauses (1) through (3) above will be a default under the Franchise Agreement.

Second and Third Communities (Franchise Agreement-Section 6.2)

You must comply with our standards and specifications for the construction, development, and opening of your second and third Communities, including the applicable requirements described above for the first Community. In order to retain the exclusive right to establish and operate BRIGHTSTAR CARE HOMES[®] Communities whose premises are physically located within the protected territory described in your (or your affiliate's) BRIGHTSTAR CARE® Agency Franchise Agreement(s) or, if you or your affiliate is not an existing BrightStar Care Agency franchisee, within the Site Selection Area described in the Franchise Agreement, you must "Open" your second and third Communities for business by the following deadlines:

(i) the second Community must Open for business on or before the 2-year anniversary of the date on which the first Community actually opens for business; and

(ii) the third Community must Open for business on or before the 18-month anniversary of the date on which the second Community actually opens for business.

Despite the deadlines specified above for you to retain your location exclusivity in the protected territory described in your (or your affiliate's) BRIGHTSTAR CARE® Agency Franchise Agreement(s), your right to construct, develop, and commence operating your second and third Communities expires 5 years after the Franchise Agreement's effective date. You have no right to commence operating an additional Community after the 5-year anniversary of the Franchise Agreement's effective date.

<u>Technology/Computer Requirements</u>. (Franchise Agreement – Section 1.4)

You must purchase, license, and use the computer hardware, software, peripheral items, propertymanagement and other electronic-information systems, and technology solutions that we specify for the operation of BRIGHTSTAR CARE HOMES Communities (which may include, upon notice from us, the System Technology), including third-party accounting and payroll, electronic health records, medication management, Electronic Client Management, and client relationship management ("CRM") software. All such technology-related requirements are referred to collectively as the "Technology/Computer Requirements."

The Technology/Computer Requirements include at minimum 1 desktop computer or laptop running current Microsoft Windows operating system and associated peripherals, including monitors, office suite and other software, local network router, other network accessories, a printer, a scanner, and any installation services to install in your office. We also strongly recommend that all computers utilize disk-encryption software, which on some computer brands is available at no additional cost. The optional mobile tablets and/or laptops for your Director(s) of Nursing to use in conjunction with Accounting/Payroll software, Electronic Health Records and Medication Management Software, CRM, and the Technology/Computer Requirements are used to record and analyze all clinical, business, and accounting information for your Community's operation.

If we require you to use certain Microsoft software products, software vendors may require your acceptance of their license agreements. We reserve the right to designate changes or enhancements to the Technology/Computer Requirements used in your Community, the computer hardware, software, and other equipment. When we designate the change or enhancement to the Technology/Computer Requirements, you might need to make certain payments to us or our designated suppliers. You must procure your accounting software and cash application and collection software from us or any third party we approve and the System Technology, if applicable, from us or one of our affiliates. You alone are responsible for maintaining and supporting your local computer equipment and mobile devices.

The initial cost to purchase the Technology/Computer Requirements will range from \$4,000 to \$12,000. The monthly maintenance, repair, or upgrade of System Technology, if applicable, and any annual cost for optional or required maintenance, support, upgrades, and updates to System Technology generally would be covered by the Monthly Technology & Email Service Fee. By contrast, all hardware and computer network maintenance and upgrades of other software are your responsibility and must be done in a timely manner. There are no contractual limitations on the frequency or cost of upgrades or changes in the Technology/Computer Requirements we may impose. Upgrades could cost between \$1,000 and \$3,000 or more annually.

You must use the specified Technology/Computer Requirements and may not substitute other software without our express written permission. We reserve the right to specify different hardware and software systems in the future, including System Technology that we or our affiliates develop exclusively for the BRIGHTSTAR CARE HOMES Community franchise Program.

We reserve the right to access independently at all times (24 hours per day, 7 days per week) all information collected or compiled by your use of the Technology/Computer Requirements in your Community's operation. There are no contractual limitations on this right.

Training. (Franchise Agreement – Sections 6.5, 6.6, 6.16, and 13.4.1.4)

Your Control Person must attend at your expense, within 8 weeks after the Franchise Agreement's effective date, our BRIGHTSTAR CARE HOMES Getting Started Training (Phases 1 and 2) at our headquarters in Bannockburn, Illinois or another location we specify. The Getting Started Training (Phase 1) currently is scheduled for 2 days of training and online pre-work, and Getting Started Training (Phase 2) currently is scheduled for 1 day of training and online pre-work (depending on the trainee). We may supplement or shorten this training by e-learning through an Online Training System ("Online Training"). Attendance and learning at Getting Started Training (Phases 1 and 2) are the focus; it is not a training program for which we measure satisfactory completion.

Before the Opening Date and after securing the site for your first Community, your Control Person, Administrator (if the Control Person is not also the Administrator), and salesperson each must attend and complete to our satisfaction the portions of BRIGHTSTAR CARE HOMES Road to Opening Training program ("Road to Opening Training") designated for their positions. Road to Opening Training currently is scheduled for 2 to 4 days (depending on the trainee) at our headquarters in Bannockburn, Illinois or another location we specify. Road to Opening Training may be supplemented or reduced by Online Training.

Our training program may include a "train the trainer" module so that your senior-level personnel can learn how to train your other employees. Once we certify your trainer(s), we reserve the right to audit their live classes/presentations and other training methods used to ensure consistent delivery of materials and brand standards. Your Control Person and Administrator may request additional or repeat training at the end of Getting Started Training and Road to Opening Training if he or she does not feel sufficiently trained to operate a BRIGHTSTAR CARE HOMES Community. We and you will jointly determine the duration of any additional training. You must pay our then-current charges for additional or repeat training (not to exceed \$500 per day per attendee). If we determine that the Control Person and/or Administrator cannot satisfactorily complete the portions of Getting Started Training and/or Road to Opening Training designated for him or her (and he or she, or a replacement, cannot satisfactorily complete a repeat training program), we may terminate the Franchise Agreement.

Replacement Administrators (even if also a replacement Control Person), Directors of Nursing, Office Manager, and other employees we designate likewise must complete applicable training programs to our satisfaction after you hire them (typically within 60 days). You must pay our thenapplicable fee for all training (not to exceed \$500 per day per attendee) and are responsible for all expenses (transportation, hotel, meals, etc.) and related salary expenses during training programs.

We may require the Control Person, your Administrator (who may be you), Director of Nursing, Office Manager, and other employees we designate to attend and satisfactorily complete supplemental training each year during the franchise term at the times and locations we designate. We may charge reasonable registration or similar fees for these courses (not to exceed \$500 per day per attendee). You are responsible for all expenses (transportation, hotel, meals, etc.) and related salary expenses incurred to attend training. (See Item 6)

The following details our training programs as of this disclosure document's issuance date:

TRAINING PROGRAM

Getting Started Training (Phase 1)

Column 1	Column 2	Column 3	Column 4
	Hours of Classroom	Hours of On- The-Job	
Subject	Training	Training	Location
BRIGHTSTAR CARE HOMES Concept and Senior Living Industry Overview	2	0	Bannockburn, IL, another location we designate, or virtual
Role of the Franchisee	1	0	Bannockburn, IL, another location we designate, or virtual
Development Process	3	0	Bannockburn, IL, another location we designate, or virtual
Regulatory Framework-Zoning, Permitting, and Building Approvals	2	0	Bannockburn, IL, another location we designate, or virtual
Regulatory Framework-Licensing	2	0	Bannockburn, IL, another location we designate, or virtual
Care Homes Financials	1.5	0	Bannockburn, IL, another location we designate, or virtual

Getting Started Training (Phase 2)

Column 1	Column 2	Column 3	Column 4
	Hours of Classroom	Hours of On- The-Job	
Subject	Training	Training	Location
Managing Risks and Protecting	1.5	0	Bannockburn, IL, another location
the Brand			we designate, or virtual
Care Homes Process	1	0	Bannockburn, IL, another location
			we designate, or virtual
Creating the BRIGHTSTAR	2	0	Bannockburn, IL, another location
CARE HOMES Culture			we designate, or virtual

In addition to the training identified above, the Control Person, Administrator, and Salesperson each might spend up to 10 hours of online learning before attending Getting Started Training (Phases 1 and 2).

Road to Opening Training: Day 1 & 2

Column 1	Column 2	Column 3	Column 4
	Hours of Classroom	Hours of On- The-Job	
Subject	Training	Training	Location
Boot Camp Overview & Introductions	.5	0	Bannockburn, IL, another location we designate, or virtual
Review Pre-work	.5	0	Bannockburn, IL, another location we designate, or virtual
Human Resources & Staffing	1.5	0	Bannockburn, IL, another location we designate, or virtual
Marketing	1.5	0	Bannockburn, IL, another location we designate, or virtual
Sales	2	0	Bannockburn, IL, another location we designate, or virtual
Q&A Session from Day 1	1	0	Bannockburn, IL, another location we designate, or virtual
Resident Care Overview	1	0	Bannockburn, IL, another location we designate, or virtual
Kitchen and Meals	2	0	Bannockburn, IL, another location we designate, or virtual
Q&A Session	1	0	Bannockburn, IL, another location we designate, or virtual

Road to Opening Training: Day 3 & 4

Column 1	Column 2	Column 3	Column 4
	Hours of	Hours of On-	
	Classroom	The-Job	
Subject	Training	Training	Location
Welcome, Q&A Session	1	0	Bannockburn, IL, another location we designate, or virtual
Resident Care	3	0	Bannockburn, IL, another location we designate, or virtual
Q&A Session from Day 3	1	0	Bannockburn, IL, another location we designate, or virtual
Technology	6		Bannockburn, IL, another location we designate, or virtual

In addition to the training identified above, the Administrator might spend up to 20 hours of online learning before and/or after attending Road to Opening Training #2.

All training is conducted by our company officers, owners, directors, key employees, and consultants, each of whom has significant experience in the areas he or she will teach. The training leaders are:

- Dean Ulizio Chief Strategy Officer who has 17 years of experience in home care, senior living communities, and franchising.
- Pete First Chief Development Officer– who has over 20 years of experience in franchising, real estate, and business management.
- Susan Ehrlich RN BSN Director of Clinical Operations who has over 15 years of experience in clinical operations as a director and registered nurse.
- Shannon Stemen Regional Director, Sales and Marketing who has over 14 years of experience in skilled nursing and long-term care facilities as a Regional and Executive Director.

All training is offered on an as-needed basis during the year. We reserve the right to modify the training programs at any time. The Operations Manual and Classroom Training Manuals will be the principal training materials.

Your Control Person or another representative we authorize must attend all annual franchisee meetings (including leadership conferences) we conduct (at locations we designate) to address subjects relevant to the BrightStar System. We may use the annual meetings to offer continuing or advanced-level training instruction. We will determine the meeting's length and place and required attendees. We may charge a registration fee whether or not you attend (not to exceed \$2,000 per attendee), and you must pay all travel and salary expenses. (See Item 6) You or other authorized representatives need not attend more than 3 business days of any annual meeting. We may combine annual meetings for BRIGHTSTAR CARE HOMES Community franchisees with other brand conferences, which we may require you to attend.

Any training we provide is to provide you and your key representatives more information about brand standards and best practices. At all times, any individual you send to training is your employee, and you (not we) are solely responsible for all employment and personnel matters related to any employee you send to training. We do not reserve the right and expressly disavow any right or ability to control any employees you send to training. By sending an individual to training, you designate such employee as a key representative of your Agency to whom we can speak.

ITEM 12 TERRITORY

You will construct, develop, and operate your Communities at specific sites we first must accept. If you are or your affiliate is a BrightStar Care Agency franchisee, your Communities' sites must be within your or your affiliate's BrightStar Care "protected territory." If you are not, or your affiliate is not, a BrightStar Care Agency franchisee, your Communities' sites must be within the Site Selection Area described in the Franchise Agreement. The Site Selection Area will be defined

by a collection of contiguous zip codes with a total combined population of approximately 200,000 to 300,000 people. In all cases, you will look for your first Community's site after signing our Franchise Agreement. You have 1 year after the Franchise Agreement's effective date to locate and secure that site. (Item 11 describes the site selection process.)

You may operate the Community only at the accepted site and may not relocate without our approval, which we may grant or deny as we deem best. Whether or not we allow relocation depends on what is in the Community's and our system's best interests. However, given the nature of this business and required investment, approved relocations will be rare absent condemnation or material property damage.

The Franchise Agreement grants you the right to construct, develop, and operate up to 3 BRIGHTSTAR CARE HOMES[®] Communities. All 3 Communities that the Franchise Agreement entitles you to develop and operate will be located in either your or your affiliate's BrightStar Care "protected territory" (if you are or your affiliate is a BrightStar Care Agency franchisee) or the Site Selection Area (if you are not or your affiliate is not a BrightStar Care Agency franchisee).

You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control. If you construct and develop by their opening deadlines, and then operate in compliance with the Franchise Agreement, all 3 Communities that the Franchise Agreement entitles you to develop and operate, then we and our affiliates will not establish and operate, or grant to others the right to establish and operate, BRIGHTSTAR CARE HOMES[®] Communities whose premises are physically located within, as applicable, either the "protected territory" described in the Brightstar Care Agency Franchise Agreement(s) or the Site Selection Area described in your Franchise Agreement. However, if you do not satisfy this obligation, we have the right to take away this territorial exclusivity.

You must operate the Communities, and may provide Services and sell products, only at the premises of the Communities. You may not provide Services or sell products away from the premises of the Communities; sell Services or products through other distribution channels, including e-commerce; use the Marks in any other business activities we have not expressly authorized; or advise others who operate competitive businesses. You may engage in Marketing and related activities we authorize within the Communities' designated market area" (DMA). We have the unrestricted right to regulate these activities to protect our interests, other BRIGHTSTAR CARE HOMES Community franchisees, and the System. You may not engage in advertising, promotional, marketing, and related activities outside the DMA.

Except for your territorial exclusivity described above, we and our affiliates have the right during the franchise term to engage in any and all other activities we and they desire, at any time or place, whether or not those activities compete with your Communities. These unlimited rights include the right to:

(a) Establish and operate, and allow others to establish and operate, BRIGHTSTAR CARE HOMES Communities at any locations, regardless of their proximity to your Communities (including—if you do not open all 3 Communities that you have the right to develop under the

Franchise Agreement within the required development periods—within the "protected territory" described in your or your affiliate's BRIGHTSTAR CARE® Agency Franchise Agreement(s) or, if you or your affiliate does not operate a BRIGHTSTAR CARE® Agency, within the Site Selection Area described in your Franchise Agreement).

(b) Offer, sell, operate, or distribute and license others to offer, sell, operate, or distribute, directly or through franchised, licensed, or other businesses, at wholesale or retail, and at and from any locations, BrightStar® branded and non-BrightStar® branded Services, other services, and products. These Services, other services, and products may be offered and sold through similar or dissimilar channels or methods of distribution, including the Internet and businesses operating under the BrightStar® name or another name (*e.g.*, durable medical equipment centers and/or adult/child daycare facilities and hospices). For the avoidance of all doubt, franchised and non-franchised BrightStar® Care agencies may operate in any area, no matter how close to your Communities, without any restrictions.

(c) Merge with, acquire, or be acquired by ("Merger/Acquisition Activity") any businesses of any kind operating under other systems and trademarks, which businesses may (and, after the Merger/Acquisition Activity, may continue to) offer, sell, and distribute, and allow others to offer, sell, and distribute, goods and services – identical, similar, or dissimilar to Services and products provided by your Communities – through franchised or non-franchised businesses, at wholesale or retail, at and from any locations.

(d) Engage in all other activities the Franchise Agreement does not expressly prohibit.

We need not compensate you if we engage in these activities. You have no options, rights of first refusal, or similar rights to acquire additional franchises. Continuation of your franchise rights does not depend on your achieving a certain sales volume, market penetration, or other contingency.

We and our affiliates have the right (as described above) to operate and franchise businesses under identical, similar, or different trademarks selling services and products identical or similar to, or different from, those offered and sold by BRIGHTSTAR CARE HOMES Communities.

As described in Item 1, our affiliates currently operate, and we offer and sell franchises for, BRIGHTSTAR SENIOR LIVING facilities, which provide the same services as BRIGHTSTAR CARE HOMES Communities but in a larger facility (anywhere from a 36 to 44-unit facility as opposed to the 10-to-12-bed BRIGHTSTAR CARE HOMES Community). We do not anticipate any material conflicts between these franchise opportunities regarding territory, clients, and franchisor support because we expect that BRIGHTSTAR SENIOR LIVING facilities will operate in different market areas (capable of supporting a larger facility and justifying the larger investment). It is in our best interests that both franchise opportunities operate successfully. Nevertheless, we will take reasonable steps to resolve any conflicts that do arise. We do not have separate offices or training facilities for these two types of assisted-living and memory-care communities.

As also described in Item 1, our affiliate, BrightStar Franchising, offers franchises throughout the United States for BrightStar Care agencies, which offer 4 principal services to clients: in-home

Non-Medical (Companion) care services; in-home Medical Personal Care services; in-home Medical Skilled Care services; and Supplemental Healthcare Staffing. We and BrightStar Franchising do not anticipate any material conflicts between the brands, which both feature the BrightStar name, regarding territory, clients, and franchisor support because BRIGHTSTAR CARE HOMES Communities offer services to clients in need who generally cannot remain in their homes for limited or extended timeframes (and therefore would not be able to take advantage of the in-home services offered by BrightStar Care agencies). Because the brands are affiliated, it is in our and BrightStar Franchising's best interests that both brands operate successfully. Nevertheless, we will take reasonable steps to resolve any conflicts that do arise. BrightStar Franchising's principal business address is the same as ours. It does not have separate offices or training facilities.

ITEM 13 TRADEMARKS

You will have the limited right to use the Proprietary Marks we designate for the BRIGHTSTAR CARE HOMES Community franchise program. Our affiliate, Bright Star Nevada, owns the following registrations on the Principal Register of the United States Patent and Trademark Office (USPTO):

MARK	MARK TYPE	REGISTRATION NUMBER	REGISTRATION DATE
BrightStar Care Homes	Standard Character Mark	7133822	August 8, 2023
BrightStar Care	Standard Character Mark	4042547	October 18, 2011
BrightStar	Design Plus Words, Letters, and/or Numbers	7214797	November 7, 2023
BrightStar Care HOME CARE MEDICAL STAFFING A Higher Standard	Design Plus Words, Letters, and/or Numbers	4659696	December 23, 2014
BrightStar	Standard Character Mark	3608702	April 21, 2009
CareTogether	Standard Character Mark	4019022	August 30, 2011

MARK	MARK TYPE	REGISTRATION NUMBER	REGISTRATION DATE
BrightStar Senior Living	Standard Character Mark	4538086	May 27, 2014
BrightStar Concierge	Standard Character Mark	4444043	December 3, 2013
BrightStar Connections	Standard Character Mark	4588598	August 19, 2014
A HIGHER STANDARD	Standard Character Mark	7083304	June 20, 2023

Bright Star Nevada has filed or intends to file when due all required affidavits and renewals for the Proprietary Marks listed above.

We derive our right to use and sublicense the Proprietary Marks from a Trademark License Agreement (the "Trademark License Agreement") dated January 28, 2015, with Bright Star Nevada. Under the terms of the Trademark License Agreement, we have the exclusive right in the United States to use and sublicense the Marks in connection with the offer and sale of franchises to third parties.

Specifically, Bright Star Nevada has licensed us the right to use, and sublicense use of the Marks identified above with the operation of BRIGHTSTAR CARE HOMES Communities offering residential assisted-living, memory-care, and ancillary services for seniors and other individuals in need. The Trademark License Agreement is for the United States and is exclusive. The Trademark License Agreement may be terminated by mutual agreement of the parties. Additionally, either party may terminate the Trademark License Agreement in the event of: (i) any material breach of an obligation by the other party, which breach is capable of being, but is not, cured within 60 days after written notice of the breach to the breaching party; or (ii) any situation in which a party commits a material breach of the Trademark License Agreement that is not capable of being cured within 60 days and the non-breaching party provides written notice of the breach and notice of termination. Except for the Trademark License Agreement, there are no agreements currently in effect that significantly limit our rights to use or sublicense use of the Marks. The Trademark License Agreement may be modified periodically by Bright Star Nevada and us. If the Trademark License Agreement is terminated, you might no longer have the right to use the Marks.

There are no currently effective material determinations of the USPTO, the Trademark Trial and Appeal Board, the trademark administrator of any state, or any court, and no pending infringement, opposition, or cancellation proceedings or material litigation, involving the Marks. We do not actually know of either superior prior rights or infringing uses that could materially affect your use of the principal Marks in the state where we allow you to use them.

You must follow our rules when you use the Marks, including giving proper notices of trademark registration and obtaining necessary fictitious or assumed name registrations. You may not use

any Mark in your corporate or legal business name; with modifying words, terms, designs, or symbols (except for those we license to you); in offering or selling any unauthorized services or products; or as part of any unauthorized domain name, homepage, electronic address, or other electronic media. If we discover your unauthorized use of the Marks, we may require you to destroy all offending items (with no reimbursement from us).

You must notify us immediately of any apparent infringement or challenge to your use of any Mark or of any person's claim of any rights in any Mark or confusingly similar trademark. You may not communicate with any person other than us, our affiliates, and our respective attorneys, and your attorneys, regarding any infringement, challenge, or claim. We and our affiliates may take the action we and they deem appropriate (including no action) and control exclusively any litigation, USPTO proceeding, or other proceeding concerning any Mark. You must help us, and our affiliates protect and maintain our and their interests in any litigation or USPTO or other proceeding. We will reimburse your costs for taking any requested action.

If it becomes advisable at any time for us and/or you to modify, discontinue using and/or replace any Mark and/or to use one or more additional, substitute, or replacement trade or service marks together with or instead of any previously designated Mark, you must comply with our directions within a reasonable time after we notify you. We need not reimburse your direct expenses for changing the Community's signs, your lost revenue due to any modified or discontinued Mark, or your expenses to promote a modified or substitute trademark or service mark.

We will reimburse you for all damages, claims, and expenses you incur or for which you are liable in any proceeding challenging your right to use any Mark licensed under the Franchise Agreement if your use of the Marks has been consistent with the Franchise Agreement, the Operations Manual, and our System standards and you timely notify us and our affiliates of, and comply with, our and our affiliates' directions in responding to, the proceeding. At our and our affiliates' option, we may defend and control the defense of any proceeding arising from your use of any Mark.

ITEM 14 PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION

We do not own any patents or patent applications that are material to the franchise. We and our affiliates claim copyrights in the Operations Manual (which contains our trade secrets and confidential information), Community blueprints and other design features, advertising and marketing materials, our Franchise System Website, and similar items used in operating BRIGHTSTAR CARE HOMES Communities. We and our affiliates have not registered these copyrights with the United States Copyright Office but currently need not do so to protect them. You may use copyrighted items only as we specify while operating your Community (and must stop using them at our direction).

There currently are no effective adverse determinations of the USPTO, the United States Copyright Office, or any court regarding our copyrighted materials. No agreement limits our right to use or allow others to use copyrighted materials. We do not actually know of any infringing uses of our copyrights that could materially affect your use of our copyrighted materials in your state.

We and our affiliates need not protect or defend copyrights, although we intend to do so if in the BRIGHTSTAR CARE HOMES Community system's best interests. We need not defend you, or participate in your defense, against claims arising from your use of our copyrights or indemnify you for expenses or damages you incur in a copyright proceeding. We and our affiliates have the right to control the prosecution or defense of any copyright proceeding, whether we learn of the matter from you or on our own.

Our Operations Manual and other materials contain our confidential information (some of which constitutes trade secrets under applicable law). This information includes site selection criteria; specifications for Services; training and operations materials and manuals; methods, formats, specifications, standards, systems, procedures, sales and marketing techniques, knowledge, and experience used in developing and operating BRIGHTSTAR CARE HOMES Communities; marketing programs and materials for BRIGHTSTAR CARE HOMES Communities; knowledge of specifications for and suppliers of operating assets and other items; the operating results and financial performance of BRIGHTSTAR CARE HOMES Communities other than your Community; Technology/Computer Requirements; client communication and retention programs and data used or generated by those programs; and graphic designs and related intellectual property.

You may not use our confidential information in an unauthorized manner. You must take reasonable steps to prevent its improper disclosure to others and use non-disclosure and non-competition agreements with the appropriate officers, directors, and other personnel having access. We may regulate or review the forms of agreements you use solely to ensure you adequately protect our confidential information and the competitiveness of BRIGHTSTAR CARE HOMES Communities. We will be a third-party beneficiary of those agreements with independent enforcement rights. You must keep copies of these agreements and send them to us upon request. Under no circumstances, however, will we control or regulate the forms of employment agreements you use with your employees or otherwise be responsible for your labor relations or employment practices.

If you or your employees or principals develop any new concept, process, or improvement for the operation or promotion of BRIGHTSTAR CARE HOMES Communities, you must promptly notify us and send us all necessary related information, without compensation. Any such concept, process, or improvement will become our sole property, and we will be the sole owner of all related patents, patent applications, trademarks, copyrights, and other intellectual property rights. You and your principals assign to us any and all such rights, including the right to modify such concept, process, or improvement, and otherwise waive and/or release all related rights of restraint and moral rights. You and your principals must help us obtain and enforce the intellectual property rights in such concept, process, or improvement in any and all countries and execute all documents necessary for us to obtain and enforce such rights. If these requirements are invalid or unenforceable, we will have a worldwide, perpetual, non-exclusive, fully paid license to use and sublicense the use of the concept, process or improvement to the extent such use or sublicense otherwise would directly or indirectly infringe your rights.

ITEM 15 OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS

Your Communities must at all times be under the direct supervision of your designated Control Person, who must be one of your owners. You must obtain our approval if you want someone else to be your Control Person during the franchise term. Your Control Person must meet our standards and successfully complete our initial training program. Your "Control Person" is the individual who has authority to actively direct the Community's business affairs and sign all contracts and is responsible for overseeing the Community's general management. You must designate your Control Person on the Control Person Addendum attached to the Franchise Agreement (Exhibit K). You must request in writing permission to hire an operating manager to be your Control Person if you want your owner to step out of the day-to-day operations. You must give us advance written notice before you change your designated Control Person. We must confirm that your proposed replacement Control Person meets our then-current Control Person standards and requirements.

You must hire a sufficient number of qualified, competent personnel in order to offer prompt, courteous, high quality, and efficient service to the Community's residents and their family members and otherwise operate the Community in compliance with the BrightStar System and all applicable federal, state, and local requirements.

Federal, state, and local requirements might require additional staffing. The list below is our minimum staffing requirements for a Community with 12 residents or less.

Required initial personnel for a Community includes:

- An Administrator (for up to 3 Communities). Some states require the Administrator to be licensed. (It is your responsibility to determine your state's requirements.)
- A part-time salesperson. (If you or your affiliate is a BrightStar Care Agency franchisee, these duties may be assumed by an existing salesperson in your organization.)
- A part-time Director of Nursing (DON) (unless state license requires fulltime). Recommended that the DON hold a RN license or higher with a minimum of 2 years of experience working in an assisted-living or a memory-care residence. (If you or your affiliate is a BrightStar Care Agency franchisee, these duties may be assumed by an existing DON in your organization so long as capacity within that role exists.)
- A staff of full and part-time caregivers on site to cover the residents' needs 24/7. A staffing ratio of 1:8 is our minimum standard requirement during traditional 1st and 2nd shift hours (i.e., 6:00 a.m.-10:00 p.m.). A staffing ratio of 1:14 during overnight hours is acceptable if appropriate based on resident needs. As resident needs increase, adding part-time or full-time staff on any

shift may be needed to provide a higher standard of resident care. You must determine your state's requirements and staff accordingly.

Optional personnel to consider supporting business growth:

- House Manager overseeing the day-to-day operations and sales efforts of the Community
- Housekeeper (part-time)
- Life enrichment coordinator (part-time)
- Cooking support (part-time)

Unless state regulations have higher standards for nursing oversight, you also must employ at least one Registered Nurse to oversee the clinical components of each Community's operation, including resident care.

Our System standards do not include any personnel policies or procedures that we (at our option) may make available for your optional use. You alone will determine to what extent, if any, these policies and procedures might apply to your Community's operations. We do not dictate or control labor or employment matters for franchisees and their employees. You alone are responsible for all employment decisions regarding the Community, including hiring, firing, training, promotion, wage and hour requirements, recordkeeping, supervision, and discipline. You must communicate clearly with your employees in employment agreements, human resources manuals, written and electronic correspondence, paychecks, and other materials that you (and only you) are their employer and that we, as the franchisor of BRIGHTSTAR CARE HOMES Communities, are not their employer or joint employer and do not engage in any employer-type activities (including those described above) for which only franchisees are responsible. You also must obtain an acknowledgment (in the form we specify or approve) from all Community employees that we are not their employer.

Besides your Control Person, your executive and managerial staff and other employees need not have an ownership interest in you or the Community. (Item 11 describes our training requirements for all personnel.) However, appropriate officers, directors, and other personnel must agree in writing to preserve confidential information to which they have access and not to compete with you, us, and other franchisees. We may regulate or review the forms of agreements you use—solely to ensure the protection of confidential information and the competitiveness of BRIGHTSTAR CARE HOMES Communities—and be a third-party beneficiary of those agreements with independent enforcement rights. Under no circumstances, however, will we control or regulate the forms of employment agreements you use with your employees or otherwise be responsible for your labor relations or employment practices.

Each owner of yours (and their spouses) must personally guarantee all of your obligations under the Franchise Agreement and agree to be bound personally by every contractual obligation, both monetary and non-monetary, including the covenant not to compete. This "Personal Guarantee and Agreement to be Bound Personally" and our form of Spousal/Life Partner Consent are at the end of the Franchise Agreement. You, not we, are solely responsible for all employment and personnel matters related to the DON. We do not reserve the right and expressly disavow any right or ability to control your DON. By hiring an individual into the DON role, you designate your DON as a key representative of your Agency to whom we can speak.

ITEM 16

RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL

You may sell only the Services and products we authorize. You must offer all Services and products we prescribe for BRIGHTSTAR CARE HOMES Communities. We may change the Services and products you must or may offer upon notice to you. There is no limit on the number or type of changes we may make. We will notify you of all these changes in writing or electronically through our Intranet. You must operate the Community, and may provide Services and sell products, only at the Community's premises. You may not sell Services or products through other distribution channels, including e-commerce. You must comply with our pricing standards and policies, which may prohibit certain types of charges or billing practices we determine are misleading or otherwise detrimental to the System and require you to price consistently. We may regulate the prices you charge for Services and products to the extent the law allows (including, if applicable, restrictions on your use of coupons and other price discounting practices).

Your Community will provide housing and supportive services; supervision of residents' activities; personalized assistance; health-related services to meet the needs of residents who are unable to perform, or who need assistance with, Activities of Daily Living ("ADLs"); assistance with accessing and coordinating health care services; administering medications; and ongoing assistance in accessing social and recreational services.

Unless otherwise prohibited by licensure in your state, and once we authorize you to do so, you must begin offering and selling residence and other care opportunities to be provided by the Community before the Community's Opening Date ("Pre-Sale Activities"). This will commence no later than 90 days prior to your projected Opening Date, provided that we have accepted the location from which you will engage in the Pre-Sale Activities, and you and your required staff have completed to our satisfaction required pre-opening training. You also must obtain all necessary bonds and otherwise comply with all applicable laws and licensure restrictions relating to Pre-Sale Activities.

We may develop other business activities or modify existing business activities in response to changes in the business environment, including those resulting from technological advances, expansion into new markets, and other factors not presently anticipated. We reserve the right to change or modify the Marks, the BRIGHTSTAR CARE HOMES Community concept, the Operations Manual, and any Technology/Computer Requirements. We may adopt and use new or modified trade names, trademarks, service marks, logos, equipment, furniture, furnishings, products, techniques, concepts, or Technology/Computer Requirements. We may add new and different Services and products and withdraw Services or products or change their names or image;

redesign the trade dress, software programs, and equipment, furniture, or fixture standards; or discontinue them as we consider appropriate.

ITEM 17

RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION

THE FRANCHISE RELATIONSHIP

This table lists certain important provisions of the franchise and related agreements. You should read these provisions in the agreements attached to this disclosure document.

	PROVISION	SECTION IN FRANCHISE OR OTHER AGREEMENT	SUMMARY
a.	Length of the franchise term	2.1	Franchise term begins when we sign Franchise Agreement and expires 20 years from Opening Date of your first Community developed under the Agreement.
b.	Renewal or extension of the term	2.2	If you are in substantial compliance, you may acquire successor franchise for 10 years on our then-current terms.
с.	Requirements for franchisee to renew or extend		Give us timely notice; maintain possession of site of one or more of your Communities; remodel Communities according to our then-current standards (regardless of cost); correct operating and other deficiencies; and sign new franchise agreement and other documents we use to grant franchises, including release (if state franchise law allows). Terms of new franchise agreement you sign for successor franchise may differ materially from any and all of those contained in your Franchise Agreement, including Royalty/Continuing Fee, General Marketing Fee, and (if applicable) Monthly Technology & Email Service Fee. If you operate more than one Community under the Franchise Agreement during the initial term with respect to one or more of the Communities, or are not then in substantial compliance with the Agreement with respect to one or more of the Communities, we may choose not to grant you any successor franchise or, in our sole judgment, may choose to grant you a successor franchise for less than all of your Communities and require the non-

	PROVISION	SECTION IN FRANCHISE OR OTHER AGREEMENT	SUMMARY
			compliant Community or Communities to cease operations.
d.	Termination by franchisee		If we materially breach Franchise Agreement and do not cure default after notice from you (subject to state law).
e.	Termination by franchisor without cause	Not Applicable	We may not terminate your franchise rights without cause.
f.	Termination by franchisor with cause	14	We may terminate your franchise rights only if you or your owners commit one of several violations.
g.	"Cause" defined – curable defaults	and 14.6	We have the right to terminate the Franchise Agreement after providing you a 5-day cure period if you fail to: (i) maintain required insurance coverage or to repay us for insurance we acquire for you; (ii) have your Control Person supervise the Community's day-to-day operation personally or employ a sufficient number of qualified, competent personnel; (iii) maintain strict quality controls; (iv) fail to endorse and deliver to us or another franchisee any payments erroneously made to you; (v) maintain required hours of operation; or (vi) maintain authorization for our electronic transfer of funds from your account. We have right to terminate the Franchise Agreement after providing you a 15-day cure period if: (i) you fail to pay any monies you owe us or our affiliates; (ii) any audit reveals that you have understated your royalty or marketing payments by more than 2%; (iii) you offer any unauthorized and unapproved services or products at the Community; or (iv) you order or purchase unapproved marketing materials, non-health insurance coverage, supplies, or operating assets or buy these items from an unapproved supplier. We have right to terminate the Franchise Agreement after providing you a 30-day cure period if you fail to begin operating the Community by the Opening Date or otherwise to comply with the Franchise Agreement.

PROVISION	SECTION IN FRANCHISE OR OTHER AGREEMENT	SUMMARY
		Violations of law and failure to maintain licenses and permits generally must be cured within timeframe required by law.
		We have right to terminate the Franchise Agreement if the Net Billings of your or your affiliate's BrightStar Care Agency decline on a year-over-year basis by at least 20% while you are developing and operating your Communities and you or your affiliate fails to implement a plan to recoup that decline within 90 days and then successfully recoups the decline within 12 months following our first notice to you of the decline.
		Whenever we retain the right to terminate the Franchise Agreement due to your default occurring with respect to a specific Community (and not with respect to all of the Communities that you then are operating under the Franchise Agreement), we have the option (but not the obligation), instead of terminating the Franchise Agreement, to terminate your right to operate that particular Community, effective upon delivery of written notice of termination to you.
h. "Cause" defined – non-curable defaults	14.1	Non-curable defaults include failure to complete training; felonies; criminal and other misconduct; fraud or misrepresentation; unauthorized surrender or transfer; repeated breaches (even if cured); misuse of Marks or confidential information; violation of non-competition provision; assignment for benefit of creditors and other insolvency-type defaults; abandonment; engaging in terrorist activities; personal use of Community property; repeated insufficient funds or missing payroll at any time; unethical or immoral behavior; or loss of possession of Community site.

PROVISION	SECTION IN FRANCHISE OR Other Agreement	SUMMARY
i. Franchisee's obligations on termination / non-renewal	15	Obligations include paying outstanding amounts, including termination damages (if applicable); complete de-identification and de-branding, including returning or destroying certain branded items and modifying Community site; assigning telephone and other numbers to us (including under a Conditional Assignment of Telephone Number(s)); returning confidential information; sending us list of your employees, clients, and vendors and outstanding obligations to them; and vacating premises (also see (o) and (r) below); we may control de-identification process at your cost if you do not voluntarily take required action; we may assume operation while deciding whether to purchase Community.
j. Assignment of contract by franchisor	13.1	No restriction on our right to assign; we may assign without your approval.
k. "Transfer" by franchisee – defined	13.2	Includes transfer of Franchise Agreement, sale of Communities or substantially all of their assets, transfer of Communities' underlying real estate (including Communities' buildings/physical structure); transfer of ownership interest in you or controlling ownership interest in entity with ownership interest in you, or change in your actual management control.
 Franchisor approval of transfer by franchisee 	13.2 and 13.4	No transfer without our prior written consent. You may not transfer less than all of the Communities governed by the Franchise Agreement, and any proposed transfer of the Communities governed by the Franchise Agreement must be made to the same transferee.
m. Conditions for franchisor approval of transfer	13.4	New franchisee qualifies; the transferee and/or its owners are not a private equity firm or a Search Fund; you pay us, our affiliates, and third-party vendors all amounts due and submit all required reports; new franchisee (and its owners and affiliates) are not in a competitive business; training completed by transferee's Control Person; new franchisee has right to occupy Communities for expected franchise term; new franchisee obtains and maintains all licenses and licensure; you or transferee signs our then-current franchise agreement and other documents (any and all

PROVISION	SECTION IN FRANCHISE OR OTHER AGREEMENT	SUMMARY
		terms of which may differ materially from those in your Franchise Agreement, except that new franchise agreement's term will equal your Franchise Agreement's unexpired term or, at our option, a new 20-year term and the Royalty/Continuing Fee, General Marketing Fees, and certain technology fees (if applicable) payable under the new franchise agreement during its initial term will be the same as in your Franchise Agreement if that initial term is equal to your Franchise Agreement's unexpired initial term); you correct existing Community deficiencies of which we notify you on punch-list; transferee agrees to upgrade and remodel Community within specified timeframe after transfer; transfer fee paid (see Item 6); we approve transferee's capital structure; you subordinate amounts due to you; you de-identify; and you sign release (if state franchise law allows) (also see (r) below). Transfer of underlying real estate (if owned by unaffiliated third party) may be made only if proposed transferee first satisfies us that your possession and
		operation of Community after transfer, and our rights, will not be affected; transfer of underlying real estate (if owned by you or affiliate) may be made only with transfer of Community and Franchise Agreement.
n. Franchisor's right of first refusal to acquire franchisee's business		We may match any offer for the Franchise Agreement and the Communities (including their buildings/physical structures if you or affiliate owns them), an ownership interest in you, or a controlling ownership interest in entity with ownership interest in you.
o. Franchisor's option to purchase franchisee's business		We may buy the Communities' assets (including their buildings/physical structures if you or your affiliate owns them) at fair market value, or receive assignment or sublease of Communities' occupancy rights.
p. Death or disability of franchisee		Assignment of ownership interest in you to approved party within 9 months after owner's death or disability; we may manage Communities if qualified management not in place.

	PROVISION	SECTION IN FRANCHISE OR OTHER AGREEMENT	SUMMARY
q.	Non-competition covenants during the term of the franchise		No diverting business and no ownership interest in, or performing services for, competitive business anywhere ("competitive business" means (i) any business whose primary purpose is housing or caring for persons over the age of 60, or (ii) any business that provides supplemental healthcare staff to institutional clients, such as hospitals, Medicare agencies, hospice agencies, assisted-living centers, nursing homes, and clinics, or (iii) any business that provides comprehensive care, including medical and/or non-medical services, to private-duty clients within their home, or (iv) any business granting franchises or licenses to others to operate the type of business specified in clauses (i), (ii), or (iii)).
r.	Non-competition covenants after the franchise is terminated or expires		No direct or indirect ownership interest in, or performing services for, competing business for 24 months (i) at former Community's site, (ii) within 35-mile radius of former Community site, or (iii) within 25-mile radius of another BRIGHTSTAR CARE HOMES Community that was in operation or under construction on effective date of termination, expiration, or transfer; no soliciting business from clients or referral sources of your former Community or Communities, or with whom the BrightStar System did business during the franchise term, or contacting any of our suppliers or vendors in connection with your ownership, management, operation, or maintenance of, or engagement or consulting with, any Competitive Business.
s.	Modification of the agreement		No modifications generally, but we may change Operations Manual and System standards.
t.	Integration / merger clause		Only the Franchise Agreement's terms (including System standards in Operations Manual) are binding (subject to state law). Any representations or promises outside of the disclosure document and Franchise Agreement may not be enforceable.

PROVISION	SECTION IN FRANCHISE OR OTHER AGREEMENT	SUMMARY
u. Dispute resolution by arbitration or mediation		Except for certain claims, all disputes first must be submitted to our senior executives for internal dispute resolution and, if not resolved, for mediation. Mediation will be held at our office. Disputes that cannot be resolved through mediation or meeting are resolved through arbitration (subject to state law).
v. Choice of forum		All mediation, arbitration, and litigation occur in city and state where we have our headquarters when the action is filed (currently Illinois) (subject to state law).
w. Choice of law		Except for federal law, Illinois law applies (subject to state law).

ITEM 18 PUBLIC FIGURES

We do not use any public figure to promote our franchise.

ITEM 19 FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information, and if the information is included in the disclosure document. Financial performance information that differs from that included in Item 19 may be given only if: (1) a franchisor provides the actual records of an existing outlet you are considering buying; or (2) a franchisor supplements the information provided in this Item 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

This financial performance representation discloses the actual historical revenue and occupancy rates over certain operating timeframes for 3 BRIGHTSTAR CARE HOMES Communities. The first Community is in Eagle (Ada County), Idaho 83616 ("Prickly Pear Community"). The second Community is in Boise (Ada County), Idaho 83709 ("Taft Community"). The third Community is in Boise (Ada County), Idaho 83706 ("Stephen Community"). The Prickly Pear Community is a 9-unit BRIGHTSTAR CARE HOMES Community; the Taft Community is a 10-unit BRIGHTSTAR CARE HOMES Community. The Stephen Community is a 10-unit BRIGHTSTAR CARE HOMES Community. All 3 Communities were in operation for at least 1 full calendar year as of December 31, 2023.

An additional Community located in Boise (Southwest Ada County), Idaho 83709 ("Summerset Community"), whose operating results are <u>not</u> included in this Item 19, is an 8-unit BRIGHTSTAR

CARE HOMES Community that opened on August 18, 2018. It was the first BRIGHTSTAR CARE HOMES Community that opened. It has operated continuously since that date and serves up to 8 residents. Unit mix at the time of opening consisted of all assisted-living beds in 8 units. Summerset Community is licensed as a Residential Assisted Living Facility, which is the highest level of licensure available for residential assisted-living services in the State of Idaho. All residents pay out-of-pocket or use private insurance (no government payor sources are accepted or used). Summerset Community is located in the southwest portion of the city, approximately 10 miles southwest of downtown Boise. After Summerset Community opened for business, and as the BRIGHTSTAR CARE HOMES Community pilot program has evolved, it has been determined that 10 to 12 bed Communities with a focus on the growing memory-care population is the preferred model for expansion. Therefore, the focus of our franchise offering for new BRIGHTSTAR CARE HOMES Communities is consistent with the Prickly Pear, Taft, and Stephen Communities described below. For that reason, Summerset Community's 2023 operating results are not included in this financial performance representation.

While a fifth BRIGHTSTAR CARE HOMES Community opened for business in Boise, Idaho in August 2023, its results are not included in this financial performance representation because it did not operate for a full calendar year as of December 31, 2023.

The Prickly Pear Community opened on May 22, 2019, and has operated continuously since that date. It is a 9-unit memory-care residence licensed to serve up to 9 residents. Unit mix at the time of opening consisted of 8 memory-care beds in 8 units. The Community was newly constructed in 2018. In January 2021, this home was relicensed as a 9-unit Community. Prickly Pear Community is licensed as a Residential Assisted Living Facility, which is the highest level of licensure available for assisted-living services in the State of Idaho. All residents pay out-of-pocket or use private insurance (no government payor sources are accepted or used). Prickly Pear Community is located in the northwest portion of Boise, approximately 14 miles northwest of downtown Boise.

The Taft Care Community opened on August 13, 2020, and has operated continuously since that date. It is a 10-unit memory-care residence licensed to serve up to 10 residents. Unit mix at the time of opening consisted of 10 memory-care beds in 10 units. The Community was newly constructed in 2020. Taft Community is licensed as a Residential Assisted Living Facility, which is the highest level of licensure available for assisted-living services in the State of Idaho. All residents pay out-of-pocket or use private insurance (no government payor sources are accepted or used). Taft Community is located in the north portion of Boise, approximately 4 miles north of downtown Boise.

The Stephen Community opened on August 12, 2021, and has operated continuously since that date. It is a 10-unit memory-care residence licensed to serve up to 10 residents. Unit mix at the time of opening consisted of 10 memory-care beds in 10 units. The Community was newly constructed in 2021. The Stephen Community is licensed as a Residential Assisted Living Facility, which is the highest level of licensure available for assisted-living services in the State of Idaho. All residents pay out-of-pocket or use private insurance (no government payor sources are accepted or used). Stephen Community is located in the southeast portion of Boise, approximately 3 miles southeast of downtown Boise.

The immediate neighborhoods surrounding the Prickly Pear, Taft, and Stephen Communities are oriented toward single-residential development and are typical of the type of neighborhood for BRIGHTSTAR CARE HOMES Community development.

We believe that the Prickly Pear, Taft, and Stephen BRIGHTSTAR CARE HOMES Communities are prototypical of the types of Communities we recommend that our franchisees develop and operate in terms of services, size, and configuration; however, size may vary. These BRIGHTSTAR CARE HOMES Communities offer the same services and products to the public as the Communities that franchisees will operate under Franchise Agreements with us.

Our management prepared this financial performance representation based on the unaudited information provided by our affiliate's franchisee during the periods it has owned and operated the 3 BRIGHTSTAR CARE HOMES Communities. The franchisee has owned and operated Prickly Pear Community since May 22, 2019, Taft Community since August 13, 2020, and Stephen Community since August 12, 2021.

Revenue Progression:

Prickly Pear Community:

	2019*	2020	2021	2022	2023**
Revenue					
(\$)	309,793	599,561	540,051	570,166	411,514

*2019 was a partial year, as the Prickly Pear Community opened May 22, 2019. **2023 had occupancy only January through July due to water damage repairs.

Taft Community:

	2020*	2021	2022	2023
Revenue				
(\$)	173,950	837,609	796,350	744,293

*2020 was a partial year, as the Taft Care Community opened August 13, 2020.

Stephen Community:

	2021*	2022	2023
Revenue			
(\$)	126,079	706,768	653,938

*2021 was a partial year, as the Stephen Community opened August 12, 2021.

Occupancy Ramp-Up

The Prickly Pear, Taft, and Stephen Communities had the following occupancy ramp-up experiences, respectively:

Prickly Pear	Occupancy												
	Full Year	2018	2019	2020	2021	2022	2023						
	Average	N/A	79%	94%	87%	88%	67%						
ID - Boise	Monthly	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec
(Prickly Pear	2019	Jan	160	IVIGI	ЧЧ	iviay	44%	67%	67%	78%		-	100%
-	2019						4470	0770	0770	1070	100%	100%	100%
Care Home)	2020	100%	100%	100%	100%	100%	100%	100%	100%	78%	78%	78%	100%
	2021	78%	100%	100%	89%	89%	89%	100%	78%	89%	89%	78%	67%
	2022	78%	100%	78%	78%	89%	78%	78%	78%	100%	100%	100%	100%
	2023	78%	78%	78%	78%	56%	44%	56%					

*In 2023, Prickly Pear had occupancy only in January through July due to water damage repairs.

Taft	Occupancy												
	Full Year	2018	2019	2020	2021	2022	2023						
	Average	N/A	N/A	56%	100%	100%	84%						
ID - Boise (Taft	Monthly	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec
Care Home)	2020								50%	30%	50%	70%	80%
	2021	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
	2022	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%	100%
	2023	80%	90%	80%	80%	80%	90%	90%	80%	80%	90%	90%	80%

Stephen	Occupancy												
	Full Year	2018	2019	2020	2021	2022	2023						
	Average	N/A	N/A	N/A	42%	92%	73%						
ID - Boise													
(Stephen Care	Monthly	Jan	Feb	Mar	Apr	May	Jun	Jul	Aug	Sep	Oct	Nov	Dec
Home)	2021	0%	0%	0%	0%	0%	0%	0%	30%	30%	30%	50%	70%
	2022	80%	70%	90%	90%	100%	100%	90%	90%	90%	100%	100%	100%
	2023	80%	90%	80%	60%	60%	60%	70%	80%	80%	80%	70%	70%

Explanatory Notes to Previous 3 Tables

"Occupancy %" represents the number of Residents in each month divided by the number of capacity rooms in each Community (9 rooms in Prickly Pear Community, 10 rooms in Taft Community, and 10 rooms in Stephen Community).

Some outlets have sold this amount. Your individual results may differ. There is no assurance that you will sell as much.

Written substantiation of all financial information presented in this financial performance representation will be made available to you upon reasonable request. This financial performance representation was prepared based on unaudited information.

Other than the preceding financial performance representation, we do not make any financial performance representations. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing outlet, however, we may provide you with the actual records of that outlet. If you receive any other financial performance information or projections of your future income, you should report it to the franchisor's management by contacting Pete First, Chief Development Officer, BrightStar Senior Living Franchising, LLC, 2275 Half Day Road, Suite 210, Bannockburn, Illinois 60015, (877) 689-6898, the Federal Trade Commission, and the appropriate state regulatory agencies.

ITEM 20 OUTLETS AND FRANCHISE INFORMATION

All year-end numbers appearing in the tables below are as of December 31 in each year. As noted in Item 1, as part of the pilot of the BRIGHTSTAR CARE HOMES Community concept, BrightStar Franchising has granted one existing BrightStar Care franchisee the right to open and operate BRIGHTSTAR CARE HOMES Communities under an amendment to its existing BrightStar Care Franchise Agreement. The open and operating franchised BRIGHTSTAR CARE HOMES Communities disclosed below for 2021, 2022, and 2023 are those Communities authorized by BSF Franchising.

Table No. 1

Column 1	Column 2	Column 3	Column 4	Column 5
Outlet Type	Year	Outlets at the Start of the Year	Outlets at the	Net Change
		Start of the Tear	Year	
	2021	3	4	+1
Franchised	2022	4	4	0
	2023	4	5	+1
Commons	2021	0	0	0
Company- Owned	2022	0	0	0
Owned	2023	0	0	0
	2021	3	4	+1
Total Outlets	2022	4	4	0
	2023	4	5	+1

Systemwide Outlet Summary For years 2021 to 2023

Table No. 2

Transfers of Outlets from Franchisees to New Owners (other than the Franchisor) For years 2021 to 2023

Column 1	Column 2	Column 3
State	Year	Number of Transfers
	2021	0
All States	2022	0
	2023	0
	2021	0
Total	2022	0
	2023	0

Table No. 3

Status of Franchised Outlets For years 2021 to 2023

Col. 1	Col. 2	Col. 3	Col. 4	Col. 5	Col. 6	Col. 7	Col. 8	Col. 9
State			Outlets Opened	Termina- tions	Non- Renewals	Reacquired by Franchisor	Ceased Opera- tions – Other Reasons	Outlets at End of the Year
	2021	3	1	0	0	0	0	4
Idaho	2022	4	0	0	0	0	0	4
	2023	4	1	0	0	0	0	5
	2021	3	1	0	0	0	0	4
Totals	2022	4	0	0	0	0	0	4
	2023	4	1	0	0	0	0	5

Table No. 4

Col. 1	Col. 2	Col. 3	Col. 4	Col. 5	Col. 6	Col. 7	Col. 8
State	Year	Outlets at Start of the Year	Outlets Opened	Outlets Reacquired From Franchisee	Outlets Closed	Outlets Sold to Franchisee	Outlets at End of the Year
	2021	0	0	0	0	0	0
All States	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2021	0	0	0	0	0	0
Totals	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0

Status of Company-Owned Outlets For years 2021 to 2023

Table No. 5

Projected Openings As Of December 31, 2023

Column 1	Column 2	Column 3	Column 4		
	Franchise Agreements	Projected New	Projected New		
~	Signed But Outlets	Franchised Outlets In	Company-Owned Outlets		
State	Not Opened*	The Next Fiscal Year	In The Next Fiscal Year		
California	1	1	0		
Idaho**	2	2	0		
Louisiana	1	0	0		
Minnesota	1	0	0		
Texas	1	0	0		
Utah	1	1	0		
Washington	1	0	0		
Total	8	4	0		

*Each signed Franchise Agreement (other than in Idaho) gives the franchisee the right to develop and operate up to 3 BRIGHTSTAR CARE HOMES Communities to be governed by that same Franchise Agreement.

**While the Idaho franchisee's existing Franchise Agreement gives the franchisee the right to develop and operate under that same Franchise Agreement the number of BRIGHTSTAR CARE HOMES Communities that it desires to develop and operate, Column 2 references the two BRIGHTSTAR CARE HOMES Communities in Idaho that were actually under construction at the end of 2023 and expected to open in 2024.

A list of the current BRIGHTSTAR CARE HOMES Community franchisees is attached as Exhibit G to this Franchise Disclosure Document. There were no franchisees who had Communities terminated, cancelled, or not renewed, or who otherwise voluntarily or involuntarily ceased to do business under the Franchise Agreement, during the last fiscal year or who have not communicated with us within 10 weeks of this Franchise Disclosure Document's issuance date. If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

During the last 3 fiscal years, no current or former franchisees have signed confidentiality clauses restricting them from discussing with you their experiences as a franchisee in our franchise system. There are no trademark-specific franchisee organizations associated with the BRIGHTSTAR CARE HOMES Community franchise system.

ITEM 21 FINANCIAL STATEMENTS

Exhibit F contains our audited financial statements for the fiscal years ended December 31, 2023, January 1, 2023, and January 2, 2022.

ITEM 22 CONTRACTS

The following contracts/documents are exhibits:

EXHIBIT B BRIGHTSTAR CARE HOMES Community Franchise Agreement

- A. Premises
- B. Authorization to Initiate Debit Entries for Fees and Payments Due from Franchisee
- C. Franchisee Ownership and Management Information
- D. Personal Guarantee and Agreement to be Bound
- E. Franchisee Representations
- F. Spousal / Life Partner Consent
- G. Conditional Assignment of Telephone Numbers
- H. Site Selection Addendum

- I. Franchisor and Franchisee Acknowledgments
- J. Business Associate Agreement
- K. Control Person Addendum
- EXHIBIT D Confidentiality, Non-Disclosure and Non-Competition Agreement
- EXHIBIT E State Specific Addenda
- EXHIBIT H Release of Claims
- EXHIBIT I Microsoft Teams/One-Drive/Web Version Office Package Opt-in Addendum

ITEM 23 RECEIPTS

Exhibit J of this Franchise Disclosure Document contains a detachable document, in duplicate, acknowledging a prospective franchisee's receipt of this Disclosure Document. You should sign both copies of the Receipt. You should retain one signed copy for your records and return the other signed copy to: BrightStar Senior Living Franchising, LLC, 2275 Half Day Road, Suite 210, Bannockburn, Illinois 60015, or via facsimile to (866) 360-0393.

EXHIBIT A

TO BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC FRANCHISE DISCLOSURE DOCUMENT

LIST OF STATE ADMINISTRATORS AND LIST OF AGENTS FOR SERVICE OF PROCESS

STATE AGENCIES/AGENTS FOR SERVICE OF PROCESS

Listed here are the names, addresses and telephone numbers of the state agencies having responsibility for the franchising disclosure/registration laws. We may not yet be registered to sell franchises in any or all of these states.

If a state is not listed, we have not appointed an agent for service of process in that state in connection with the requirements of the franchise laws. There may be states in addition to those listed below in which we have appointed an agent for service of process.

There also may be additional agents appointed in some of the states listed.

CALIFORNIA

Website: <u>www.dfpi.ca.gov</u> Email: <u>ask.DFPI@dfpi.ca.gov</u>

Commissioner of Department of Financial Protection & Innovation Department of Financial Protection & Innovation Toll Free: 1 (866) 275-2677

Los Angeles

320 West 4th Street, Suite 750 Los Angeles, California 90013-2344 (213) 576-7500

Sacramento

2101 Arena Boulevard Sacramento, California 95834 (866) 275-2677

San Diego

1455 Frazee Road, Suite 315 San Diego, California 92108 (619) 525-4233

San Francisco

One Sansome Street, Suite 600 San Francisco, California 94104-4428 (415) 972-8559

HAWAII

(for service of process)

Commissioner of Securities Department of Commerce and Consumer Affairs Business Registration Division 335 Merchant Street, Room 203 Honolulu, Hawaii 96813 (808) 586-2722

(for other matters)

Commissioner of Securities Department of Commerce and Consumer Affairs Business Registration Division 335 Merchant Street, Room 205 Honolulu, Hawaii 96813 (808) 586-2722

ILLINOIS

Illinois Attorney General 500 South Second Street Springfield, Illinois 62706 (217) 782-4465

INDIANA

(for service of process)

Indiana Secretary of State 201 State House 200 West Washington Street Indianapolis, Indiana 46204 (317) 232-6531

(state agency)

Indiana Secretary of State Securities Division Room E-111 302 West Washington Street Indianapolis, Indiana 46204 (317) 232-6681

MARYLAND

(for service of process)

Maryland Securities Commissioner at the Office of Attorney General-Securities Division 200 St. Paul Place Baltimore, Maryland 21202-2021 (410) 576-6360

(state agency)

Office of the Attorney General-Securities Division 200 St. Paul Place Baltimore, Maryland 21202-2021 (410) 576-6360

MICHIGAN

Michigan Attorney General's Office Consumer Protection Division Attn: Franchise Section G. Mennen Williams Building, 1st Floor 525 West Ottawa Street Lansing, Michigan 48933 (517) 335-7567

MINNESOTA

Commissioner of Commerce Department of Commerce 85 7th Place East, Suite 280 St. Paul, Minnesota 55101 (651) 539-1500

NEW YORK

(for service of process)

Attention: New York Secretary of State New York Department of State One Commerce Plaza, 99 Washington Avenue, 6th Floor Albany, New York 12231-0001 (518) 473-2492

(Administrator)

NYS Department of Law Investor Protection Bureau 28 Liberty Street, 21st Floor New York, New York 10005 (212) 416-8236

NORTH DAKOTA

(for service of process)

Securities Commissioner North Dakota Securities Department 600 East Boulevard Avenue State Capitol, 14th Floor, Dept. 414 Bismarck, North Dakota 58505-0510 (701) 328-4712

(state agency)

North Dakota Securities Department 600 East Boulevard Avenue, Suite 414 Bismarck, North Dakota 58505 (701) 328-2910

OREGON

Oregon Division of Financial Regulation 350 Winter Street NE, Suite 410 Salem, Oregon 97301 (503) 378-4140

RHODE ISLAND

Securities Division Department of Business Regulations 1511 Pontiac Avenue John O. Pastore Complex-Building 69-1 Cranston, Rhode Island 02920 (401) 462-9500

SOUTH DAKOTA

Division of Insurance Securities Regulation 124 S. Euclid, Suite 104 Pierre, South Dakota 57501 (605) 773-3563

VIRGINIA

(for service of process)

Clerk, State Corporation Commission 1300 East Main Street First Floor Richmond, Virginia 23219 (804) 371-9733

(for other matters)

State Corporation Commission Division of Securities and Retail Franchising Tyler Building, 9th Floor 1300 East Main Street Richmond, Virginia 23219 (804) 371-9051

WASHINGTON

(for service of process)

Director Department of Financial Institutions Securities Division 150 Israel Road SW Tumwater, Washington 98501 (360) 902-8760

(for other matters)

Department of Financial Institutions Securities Division P. O. Box 41200 Olympia, Washington 98504-1200 (360) 902-8760

WISCONSIN

(for service of process)

Administrator, Division of Securities Department of Financial Institutions 4822 Madison Yards Way, North Tower Madison, Wisconsin 53705 (608) 266-2139

(state administrator)

Division of Securities Department of Financial Institutions 4822 Madison Yards Way, North Tower Madison, Wisconsin 53705 (608) 266-9555

EXHIBIT B

TO BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC FRANCHISE DISCLOSURE DOCUMENT

BRIGHTSTAR CARE HOMES COMMUNITY FRANCHISE AGREEMENT

BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC

BRIGHTSTAR CARE HOMES® COMMUNITY FRANCHISE AGREEMENT

BRIGHTSTAR CARE HOMES FA (FINAL 2024) ACTIVE\1606838711.2

TABLE OF CONTENTS

Page

1.	<u>GRA</u>	NT OF FRANCHISE	2
	1.1	Grant	2
	1.2	Communities Operated Only from the Premises; Scope of Advertising	2
	1.3	Our and our Affiliates' Reserved Rights.	. 3
	1.4	Technology/Computer Requirements.	. 4
2.	TERN	<u> 1 AND RENEWAL</u>	5
	2.1	Initial Term	5
	2.2	Successor Franchise Term.	5
3.	OPER	RATING ASSISTANCE	7
	3.1	Assistance Prior to Opening	7
	3.2	Ongoing Assistance.	.9
4.	FEES	AND OTHER PAYMENTS	9
	4.1	Initial Franchise Fee	9
	4.2	Royalty/Continuing Fee	10
	4.3	Net Billings1	10
	4.4	General Marketing Fee	10
	4.5	Monthly Technology and Email Service Fee	11
	4.6	Vendor Evaluation Fee	11
	4.7	Annual Increase in Certain Fixed Fees.	12
	4.8	Method of Payment and Electronic Funds Transfer.	12
	4.9	Payment and Interest on Late Payments	12
	4.10	Application of Payments	13
	4.11	Taxes on Payments and Currency	13
5.	LICE	NSED MARKS	13
	5.1	Ownership	13
	5.2	Authorized Use	13
	5.3	Infringement	14
	5.4	Operation Under Licensed Marks	15

TABLE OF CONTENTS

(continued)

	5.5	Modification/Replacement of Licensed Marks	. 15
	5.6	Non-Exclusive License.	. 15
6.	<u>STAN</u>	DARDS OF OPERATION	. 15
	6.1	Construction, Development, and Opening of First Community	. 15
	6.2	Construction, Development, and Opening of Second and Third Communities	16
	6.3	Pre-Sales.	17
	6.4	Quality Standards for Fixtures, Equipment and Supplies	17
	6.5	Training	17
	6.6	Hiring and Supervision.	18
	6.7	Maintenance of Premises.	21
	6.8	Operation of Community.	21
	6.9	Purchases	23
	6.10	Hours of Business.	24
	6.11	Printed Materials	24
	6.12	Identification of Community	24
	6.13	Third Party Actions	24
	6.14	Inspection of Community Premises.	24
	6.15	Possible Variation in Certain Standards.	25
	6.16	Attendance at Annual Meetings	25
	6.17	Intellectual Property Belongs to Us.	25
7.	<u>OPER</u>	ATIONS MANUAL	. 26
	7.1	Compliance with Operations Manual.	. 26
	7.2	Confidential Use.	. 26
	7.3	Revisions	. 27
8.	ADVE	ERTISING AND MARKETING	27
	8.1	General Marketing Fund	. 27
	8.2	Local Marketing	29

TABLE OF CONTENTS

(continued)

	8.3	Marketing Generally.	29
	8.4	Website and Intranet Matters.	30
	8.5	Online Advertising	31
9.	MOD	IFICATIONS	32
	9.1	Modifications to the BRIGHTSTAR CARE HOMES [®] Community Program	32
	9.2	Modifications not Requiring Significant Changes in Fixtures or Equipment.	32
	9.3	Modifications Requiring Significant Changes in Fixtures or Equipment; Upgrading the Community and CapEx Account.	32
	9.4	Test Marketing	. 33
	9.5	Dual Branding	33
10.	CONI	DEMNATION AND DAMAGE	33
	10.1	Condemnation	. 33
	10.2	Damage	. 34
	10.3	Extension of Initial Term	. 35
11.	<u>STAT</u>	EMENTS, RECORDS AND FEE PAYMENTS	. 35
	11.1	Maintenance of Records.	. 35
	11.2	Examination of Records	. 36
	11.3	Reports and Financial Statements	. 36
12.		ENANTS NOT TO COMPETE AND CONFIDENTIALITY	
	<u>OBLI</u>	<u>GATIONS</u>	
	12.1	You Defined	. 36
	12.2	Confidential Information.	. 37
	12.3	Non-Use and Non-Disclosure of Confidential Information	. 37
	12.4	Non-Compete Covenants.	. 38
13.	<u>TRAN</u>	ISFER AND ASSIGNMENT	. 39
	13.1	Transfer by Us	. 39
	13.2	Transfer by You.	40

TABLE OF CONTENTS (continued)

Page

	13.3	Representations as to Ownership	41
	13.4	Conditions to Our Consent to Transfer.	41
	13.5	Transfer and Operation in the Event of Death or Disability	44
	13.6	Our Right of First Refusal	44
14.	DEFA	<u>ULT AND TERMINATION</u>	46
	14.1	Termination With Notice and Without Opportunity to Cure	46
	14.2	Termination Upon Notice and Five (5) Days' Opportunity to Cure	48
	14.3	Termination Upon Notice and Fifteen (15) Days' Opportunity to Cure	49
	14.4	Termination With Notice and Thirty (30) Days' Opportunity to Cure	49
	14.5	Termination With Notice and One-Year Opportunity to Cure.	50
	14.6	Termination of Right to Operate a Community	50
	14.7	Nonwaiver	50
	14.8	Termination by You	50
	14.9	Step-In Rights.	51
15.	<u>POST</u>	-TERM OBLIGATIONS	52
	15.1	Obligations upon Termination or Expiration	52
	15.2	Termination Damages	54
	15.3	Our Right to Purchase Community's Assets.	55
16.	DISPU	JTE RESOLUTION AND CORRESPONDING PROCEDURES	57
	16.1	Meeting Between Senior Executives.	57
	16.2	Mediation.	58
	16.3	Arbitration	58
	16.4	Exceptions to Arbitration	60
	16.5	Third Party Beneficiaries.	60
	16.6	No Right to Offset	60
	16.7	Venue	60
	16.8	Limitation on Actions.	60
	16.9	Waiver of Punitive, Exemplary, and Certain Other Damages	61

TABLE OF CONTENTS (continued)

	16.10	Jury Trial Waiver.	. 61
17.	17. <u>INSURANCE</u>		. 61
	17.1	Lines of Insurance	. 61
	17.2	Insurance Certificates	. 63
	17.3	No Relief from Indemnity Requirement.	. 63
	17.4	Administrative Fee	. 63
18.	YOUE	R OWNERSHIP AND ORGANIZATION	. 63
	18.1	Disclosure of Ownership Interests.	. 63
	18.2	Organizational Documents	63
	18.3	Personal Guarantee Covenants and Assumption of Obligations.	. 63
19.	TAXE	ES, PERMITS AND INDEBTEDNESS	. 64
	19.1	Taxes	. 64
	19.2	Permits	. 64
	19.3	Full and Sole Responsibility for Debts and Obligations	. 64
20.	INDE	MNIFICATION AND INDEPENDENT CONTRACTOR	. 64
	20.1	Indemnification	. 64
	20.2	Cost of Enforcement.	. 65
	20.3	No Fiduciary Relationship; Independent Contractor Status.	. 65
21.	WRIT	TEN APPROVALS, WAIVERS, AND AMENDMENT	. 66
	21.1	Approval Process.	. 66
	21.2	No Waiver	. 66
	21.3	Amendments.	. 67
22.	NOTI	<u>CES</u>	. 67
23.	GOVE	ERNING LAW	. 68
24.	<u>SEVE</u>	RABILITY AND CONSTRUCTION	. 68
	24.1	Severability.	. 68
	24.2	Execution in Counterparts	68
	24.3	Headings and Captions.	. 69

TABLE OF CONTENTS (continued)

Page

	24.4 Interpretation of Rights and Obligations	69
25.	ENTIRE AGREEMENT	69
26.	SPOUSAL/LIFE PARTNER CONSENT	70
27.	ANTI-TERRORIST ACTIVITIES	70
28.	ACKNOWLEDGMENTS	71
29.	NO WAIVER OR DISCLAIMER OF RELIANCE IN CERTAIN STATES	71

EXHIBITS

- A. Premises
- B Authorization to Initiate Debit Entries for Fees and Payments Due from Franchisee
- C. Franchisee Ownership and Management Information
- D. Personal Guarantee and Agreement to be Bound
- E. Franchisee Representations
- F. Spousal / Life Partner Consent
- G. Conditional Assignment of Telephone Numbers
- H. Site Selection Addendum
- I. Franchisor and Franchisee Acknowledgments
- J. Business Associate Agreement
- K. Control Person Addendum

BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC

BRIGHTSTAR CARE HOMES[®] COMMUNITY FRANCHISE AGREEMENT

This BRIGHTSTAR CARE HOMES[®] Community Franchise Agreement ("Agreement") is made on ______, by and between BrightStar Senior Living Franchising, LLC, an Illinois limited liability company with its principal place of business at 2275 Half Day Road, Suite 210, Bannockburn, Illinois 60015 ("we", "us", or "our"), and ______

_____, with its principal place of business at ______ ("you" or "your").

RECITALS

WHEREAS, we and our affiliates have developed (and continue to develop and modify) a system for constructing, operating, identifying, and promoting assisted-living and memory-care communities under the BRIGHTSTAR CARE HOMES[®] trademark and related commercial symbols (collectively, the "Licensed Marks") providing various services for seniors and other individuals in need, including residential assisted-living, memory-care, and ancillary services (collectively, the "Services"). We have created and developed, and currently offer, a franchise opportunity for BRIGHTSTAR CARE HOMES[®] Communities that provide the Services, use the Licensed Marks, and use our and our affiliates' distinctive business formats, methods, procedures, signs, designs, layouts, standards, and specifications (collectively, the "BrightStar System" or "System"), all of which we and our affiliates periodically may improve, further develop, and otherwise modify. If we deem it necessary for the operation of BRIGHTSTAR CARE HOMES® Communities, the BrightStar System in the future may include, upon notice from us to you, one or more aspects of our or our affiliate's technology (the "System Technology"). Our System Technology may encompass an integrated-management system providing various technology solutions, including client-relationship management, staff-relationship management, billing management, web management, franchise management, and all modifications thereto that may be made from time to time; and

WHEREAS, you desire to obtain the right to construct, develop, and operate up to three BRIGHTSTAR CARE HOMES[®] Communities (each, a "Community" and, collectively, the "Communities") using the System, all of which will be governed by this Agreement; and

WHEREAS, the Licensed Marks have gained and will continue to gain public acceptance and goodwill, and we and our affiliates may create, use, and license new trademarks, service marks, and commercial symbols for BRIGHTSTAR CARE HOMES[®] Communities. The "BRIGHTSTAR[®]" and BRIGHTSTAR CARE HOMES[®] Licensed Marks currently are owned by 24-7 BrightStar Healthcare, LLC, our affiliate, which has licensed us to use and sublicense those Licensed Marks (and other Licensed Marks) in connection with BRIGHTSTAR CARE HOMES[®] Communities; and

WHEREAS, you have applied for a franchise to construct, develop, and operate up to three BRIGHTSTAR CARE HOMES[®] Communities using the BrightStar System, and we have approved that application in reliance upon all of your representations in the application; and

WHEREAS, you acknowledge that adhering to this Agreement's terms and our standards and specifications is essential to the Communities' operation and the BrightStar System.

NOW, THEREFORE, in consideration of the foregoing recitals and the mutual promises, commitments, and understandings contained in this Agreement, and for other good and valuable consideration, the receipt and adequacy of which we and you acknowledge, we and you hereby agree as follows:

1. <u>GRANT OF FRANCHISE</u>

1.1 <u>Grant.</u>

We hereby grant to you upon this Agreement's express terms and conditions, and you hereby accept, a franchise to construct, develop, and operate up to three BRIGHTSTAR CARE HOMES[®] Communities under the System and Licensed Marks at the locations identified or to be identified in Exhibit A to this Agreement (each referred to as "Premises"). We have the right to supplement, improve, or otherwise modify the BrightStar System from time to time, including, without limitation, by offering and selling new or different Services and products, and you agree to comply with all changes we specify. Unless this Agreement must in certain contexts distinguish among the three BRIGHTSTAR CARE HOMES[®] Communities that you are entitled to construct, develop, and operate under this Agreement, all references to "a Community" or "the Community" will be deemed to mean each of your Communities. This Agreement governs your construction, development, and operation of each and all of your Communities.

1.2 <u>Communities Operated Only from the Premises; Scope of Advertising.</u>

You acknowledge and agree that this franchise relates solely to the operation of your Communities at their designated and accepted Premises and gives you no right to construct, develop, or operate any additional, expanded, or modified BRIGHTSTAR CARE HOMES® Community at another location. If we have not accepted a location for your operation of your first Community as of the date you sign this Agreement, the parties will enter into the Site Selection Addendum attached as Exhibit H to this Agreement, the terms of which will govern the parties' site selection obligations. Exhibit H also will govern the parties' site selection obligations for your second and third Communities. You must operate your Communities, and may provide Services and sell products, only at the accepted Premises of such Communities. You must use each Premises solely for the Community's operation in the manner and under the standards prescribed in this Agreement, the Operations Manual, or otherwise. You may not (a) provide Services or sell products away from the Premises, (b) sell Services or products through other distribution channels, including e-commerce, (c) use the Licensed Marks in any other business activities we have not expressly authorized, or (d) advise others who operate Competitive Businesses (defined in Section 12.4.1 below). You may engage in advertising, promotional, marketing, and related activities we authorize within the Community's designated market area ("DMA"). We have the unrestricted right to regulate all such activities as we deem best to protect our interests, other BRIGHTSTAR

CARE HOMES[®] Community franchisees, and the System. You may not engage in advertising, promotional, marketing, and related activities outside the DMA.

1.3 Our and our Affiliates' Reserved Rights.

Your rights under this Agreement are non-exclusive. That means we and our affiliates retain the right during this Agreement's term to engage in any and all other activities we and they desire, at any time or place, whether or not those activities compete with your Communities. These unlimited rights include, but are not limited to, the right to:

(a) Establish and operate, and grant to others the right to establish and operate, BRIGHTSTAR CARE HOMES[®] Communities at any locations, regardless of their proximity to your Communities, provided, however, that if:

> (i) (x) you or your affiliate operates a BRIGHTSTAR CARE® Agency, (y) your Communities are located within the "Protected Territory" described in your or your affiliate's BRIGHTSTAR CARE® Agency Franchise Agreement(s), and (z) you construct and develop by their opening deadlines (see Section 6.1 below), and then operate in compliance with this Agreement, all three Communities that this Agreement entitles you to develop and operate, then we and our affiliates will not establish and operate, or grant to others the right to establish and operate, BRIGHTSTAR CARE HOMES[®] Communities whose premises are physically located within the "Protected Territory" described in the BRIGHTSTAR CARE® Agency Franchise Agreement(s); or

> (ii) (x) you or your affiliate does not operate a BRIGHTSTAR CARE® Agency, (y) your Communities are located within the Site Selection Area described in Exhibit H to this Agreement, and (z) you construct and develop by their opening deadlines (see Section 6.1 below), and then operate in compliance with this Agreement, all three Communities that this Agreement entitles you to develop and operate, then we and our affiliates will not establish and operate, or grant to others the right to establish and operate, BRIGHTSTAR CARE HOMES[®] Communities whose premises are physically located within the Site Selection Area.

(b) Offer, sell, operate, or distribute and license others to offer, sell, operate, or distribute, directly or through franchised, licensed, or other businesses, at wholesale or retail, and at and from any locations (but subject to the limited provisos in clause (a) above), BrightStar[®] branded and non-BrightStar[®] branded Services, other services, and products. These Services, other services, and products may be offered and sold through similar or dissimilar channels or methods of distribution, including the Internet and businesses operating under the BrightStar[®] name or another name (e.g., durable medical equipment centers and/or child daycare facilities and hospices). For the avoidance of all doubt, franchised and non-

franchised BrightStar[®] Care agencies may operate in any area, no matter how close to your Communities, without any restrictions.

- (c) Merge with, acquire, or be acquired by ("Merger/Acquisition Activity") any businesses of any kind operating under other systems and trademarks, which businesses may (and, after the Merger/Acquisition Activity, may continue to) offer, sell, and distribute, and allow others to offer, sell, and distribute, goods and services – identical, similar, or dissimilar to Services and products provided by the Communities – through franchised or non-franchised businesses, at wholesale or retail, at and from any locations.
- (d) Engage in all other activities this Agreement does not expressly prohibit.
- 1.4 <u>Technology/Computer Requirements.</u>

You agree to purchase, license, and use the computer hardware, software, peripheral items, property-management and other electronic-information systems, and technology solutions that we specify for the operation of BRIGHTSTAR CARE HOMES[®] Communities (which may include, upon notice from us, the System Technology), including, without limitation, third-party accounting and payroll, electronic health records, medication management, Electronic Client Management, and client relationship management ("CRM") software (together, the "Third-Party Materials"). All such technology-related requirements imposed by us from time to time are referred to collectively in this Agreement as the "Technology/Computer Requirements."

You acknowledge and agree that from time to time we may require additional Third-Party Materials or may substitute other Third-Party Materials for those originally or previously required or approved. To the extent any Third-Party Materials are packed with, incorporated into, or embedded in the System Technology (if the System Technology becomes part of the BrightStar System) or otherwise provided by us or BrightStar Technology under this Agreement, you agree to comply with the terms and conditions of the third-party licenses associated with such Third-Party Materials. You are also responsible for purchasing, at your sole expense, any third-party software we require as well as any software necessary to meet state-specific requirements of a state program or segment of business unique to your location and/or state.

We may modify the specifications and components of the Technology/Computer Requirements during the Initial Term. Such modifications and/or other technological developments or events might require you to purchase, lease, and/or license new or modified computer hardware, software, and/or peripheral items and to obtain service and support for the Technology/Computer Requirements. Although we cannot estimate the future costs of the Technology/Computer Requirements or required service or support, and although these costs might not be fully amortizable over the remaining portion of this Agreement's Initial Term (defined in Section 2.1 below), you agree to obtain all items comprising the Technology/Computer Requirements (and additions and modifications) and required service or support. Within sixty (60) days after we deliver notice to you, you must obtain the Technology/Computer Requirements that we designate and ensure that your Technology/Computer Requirements, as modified, are functioning properly. We have no obligation to reimburse you for the costs of any Technology/Computer Requirements. You may not use any unapproved computer software or security credentials. You must give us all security credentials and notify us of any changes to those credentials.

We and our affiliates may condition any license to you of required or recommended System Technology on your signing the software license agreement or similar document, or otherwise agreeing to the terms (for example, by acknowledging your consent to and accepting the terms of use of a click-through or other shrink-wrapped license agreement), we and our affiliates prescribe to regulate your use of, and the parties' respective rights and responsibilities with respect to, System Technology. We and our affiliates may charge you, if applicable, the Monthly Technology & Email Service Fee described in Section 4.5 below.

Despite the fact that you must buy, use, and maintain the Technology/Computer Requirements according to our standards and specifications, you have sole and complete responsibility for: (a) acquiring, operating, maintaining, and upgrading the Technology/Computer Requirements; (b) the manner in which our Technology/Computer Requirements interface with your and any third-party's computer systems; (c) any and all consequences if the Technology/Computer Requirements are not properly operated, maintained, and upgraded; and (d) ensuring your employees' compliance with our acceptable use policies. The Technology/Computer Requirements must permit twenty-four (24) hours per day, seven (7) days per week electronic communications between you and us, including access to the Internet and our then-current Franchise System Website and Intranet (if applicable). We always will have unlimited, independent access to the Technology/Computer Requirements to the extent the law allows.

2. <u>TERM AND RENEWAL</u>

2.1 <u>Initial Term.</u>

This Agreement will take effect on the date we sign it (the "Effective Date"), and its term will extend for twenty (20) years from the Opening Date (defined in Section 6.1 below) of your first Community developed under this Agreement (the "Initial Term"). The Initial Term is subject to earlier termination under Section 14. You must operate your Communities in compliance with this Agreement for the entire Initial Term unless this Agreement is properly terminated according to its express terms.

2.2 <u>Successor Franchise Term.</u>

2.2.1 Your Right to Acquire Successor Franchise. When this Agreement expires twenty (20) years from the Opening Date of your first Community: (a) if you (and each of your owners) have substantially complied with this Agreement during the Initial Term; and (b) if you (and each of your owners) are, both on the date you notify us in writing of your election to acquire a successor franchise (as provided below) and on the date on which the successor franchise term is scheduled to commence, in substantial compliance with this Agreement and all System standards; and (c) provided you maintain possession of the Premises of one or more of your

Communities and agree (regardless of cost) to remodel and/or expand the Communities, add or replace improvements and operating assets, and otherwise modify the Communities as we require to comply with System standards then applicable for new BRIGHTSTAR CARE HOMES[®] Communities, then you may acquire a successor franchise to operate the Communities as BRIGHTSTAR CARE HOMES[®] Communities for a new term commencing immediately upon the Initial Term's expiration and expiring either ten (10) years from that date or on the date your right to occupy the Premises of your Communities expires, whichever occurs first. (We need not grant you a successor franchise if you cannot maintain possession of the Premises of your Communities and therefore would have to secure substitute sites for the Communities.)

You must sign the franchise agreement we then use to grant franchises for BRIGHTSTAR CARE HOMES[®] Communities (modified as necessary to reflect that it is for a successor franchise), any and all of the provisions of which, including the Royalty/Continuing fee, the General Marketing Fees, and the Monthly Technology and Email Service Fee (if applicable) may differ materially from any and all of those contained in this Agreement.

If you (and each of your owners) did not substantially comply with this Agreement during the Initial Term with respect to one or more of your Communities or are not, both on the date you notify us in writing of your election to acquire a successor franchise and on the date on which the successor franchise term is scheduled to commence, in substantial compliance with this Agreement and all System standards with respect to one or more of your Communities, we need not grant you a successor franchise, whether or not we had, or chose to exercise, the right to terminate this Agreement during the Initial Term under Section 14. Alternatively, we may in our sole judgment choose to grant you a successor franchise for less than all of your Communities and require the non-compliant Community or Communities to cease operations.

2.2.2 <u>Grant of a Successor Franchise</u>. You agree to give us written notice of your election to acquire a successor franchise no more than five hundred forty (540) days and no less than three hundred sixty (360) days before this Agreement is scheduled to expire. We will give you written notice ("Our Notice"), not more than ninety (90) days after we receive your election, of our decision: (1) to grant you a successor franchise; (2) to grant you a successor franchise only on the condition you correct existing deficiencies of the Communities or in your operation of the Communities; or (3) not to grant you a successor franchise, or not to grant you a successor franchise for all of your Communities, based on our determination that you and your owners did not substantially comply with this Agreement during the Initial Term with respect to one or more of the Communities or were not in substantial compliance with this Agreement and all System standards with respect to one or more of the Communities or acquire a successor franchise.

We will not unreasonably withhold or delay our consent to a successor franchise if the conditions specified in this Section 2.2.2 have been satisfied. If applicable, Our Notice will: (a) describe the remodeling, expansion, improvements and/or modifications required to bring the Communities into compliance with then-applicable System standards for new BRIGHTSTAR CARE HOMES[®] Communities; and (b) state the actions you must take to correct operating deficiencies and the time period in which you must correct those deficiencies. If we elect not to

grant you a successor franchise, or not to grant you a successor franchise for all of your Communities, Our Notice will describe the reasons for our decision. If we elect to grant you a successor franchise for one or more of your Communities, your right to acquire the successor franchise is subject to your material compliance with this Agreement through the date of its expiration, in addition to your compliance with the obligations described in Our Notice.

If Our Notice states that you must cure certain operational or other deficiencies at one or more of the Communities as a condition to a successor franchise, we will give you written notice of our decision not to grant you a successor franchise, based upon your failure to cure those deficiencies, at least one hundred eighty (180) days before this Agreement expires. However, we need not give you this one-hundred-eighty (180) days' notice if we decide not to grant you a successor franchise due to your breach of this Agreement during the one-hundred-eighty (180) day period before it expires. If we fail to give you: (i) notice of operational or other deficiencies at the Communities within ninety (90) days after we receive your timely election to acquire a successor franchise (if we elect to grant a successor franchise under clauses (2) and (b) above); or (ii) notice of our decision not to grant you a successor franchise, or not to grant you a successor franchise for all of your Communities, at least one hundred eighty (180) days before this Agreement expires, if this notice is required, we may unilaterally extend the Initial Term for the time period necessary to give you, as applicable, either reasonable time to correct deficiencies or the one-hundred-eighty (180) days' notice of our refusal to grant a successor franchise. If you fail to notify us of your election to acquire a successor franchise within the prescribed time period, we need not grant you a successor franchise for any of your Communities.

2.2.3 <u>Agreements/Releases</u>. If you satisfy all of the other conditions for a successor franchise, you and your owners must sign the forms of franchise agreement and related documents we then customarily use to grant franchises for BRIGHTSTAR CARE HOMES[®] Communities (modified as necessary to reflect that it is for a successor franchise), any and all of the provisions of which, including the Royalty/Continuing fee, the General Marketing Fees, and the Monthly Technology and Email Service Fee (if applicable), may differ materially from any and all of those contained in this Agreement. You and your owners further agree to sign general releases, in a form satisfactory to us, of any and all claims against us and our owners, affiliates, officers, directors, employees, agents, successors, and assigns. We will consider your or your owners' failure to sign these agreements and releases and deliver them to us for acceptance and countersigning within the timeframe we specify to be your election not to acquire a successor franchise.

3. **OPERATING ASSISTANCE**

3.1 Assistance Prior to Opening.

Before the Opening Date, we will provide you with the following assistance:

3.1.1 Grant access to the Operations Manual via our Intranet.

3.1.2 Give you template plans and drawings ("Plans") for a BRIGHTSTAR CARE HOMES[®] Community's physical structure, exterior elements, and interior layout. You are responsible for constructing and developing each Community. The Plans will include mandatory and suggested standards and specifications for, among other things, size, dimensions, design, image, decor, fixtures, equipment, signs, furnishings, and color scheme. However, the Plans might not reflect the requirements of any federal, state, or local law, code, or regulation, including those arising under the Americans with Disabilities Act (the "ADA") or similar rules governing public accommodations for disabled persons. You must identify for our review and acceptance a proposed architect to prepare a site survey and all required construction plans and specifications to suit the site and make sure these plans and specifications comply with our requirements, the ADA and similar rules, other applicable ordinances, building codes, permit requirements, and occupancy requirements and restrictions. You may not engage an architect whom we do not accept. We own the Plans and all adaptations of the Plans for your site and Community. You must send us, for review and acceptance, your site survey and construction plans and specifications before you submit them for permitting and begin constructing the Community and all revised or "as built" plans and specifications during construction. Because our review is limited to ensuring your compliance with System design and layout requirements, we might not assess compliance with federal, state, or local laws and regulations, including the ADA.

Before beginning construction, you must identify, and we first must accept, the proposed general contractor. You may not engage a general contractor whom we do not accept. We periodically may inspect the site during the Community's construction and development process. You (and not we) are responsible for the performance of architects, engineers, contractors, and subcontractors hired to construct, develop, and maintain the Community. You also must ensure, and confirm to us, that sufficient insurance coverage is in place before, during, and after the construction process.

You must do the following, at your own expense, to construct, develop, and begin operating each Community:

- (a) secure all financing required to construct, develop, and operate the Community in compliance with System standards;
- (b) obtain all zoning, building, utility, sign, health, sanitation, business, and other permits, licenses, and certifications required for lawful construction of the Community;
- (c) build and decorate the Community according to approved plans and specifications and report construction progress to us at such times and by such means we specify;
- (d) obtain all customary contractors' sworn statements and partial and final lien waivers for construction, remodeling, decorating, and installation services;
- (e) purchase or lease, and install, all required flooring, fixtures, furniture, furnishings, signs, and equipment (including Technology/Computer Requirements); and

(f) purchase an opening inventory of required, authorized, and approved products, materials, and supplies from us, our affiliates, and designated or approved sources.

You must notify us of the scheduled construction completion date no later than forty-five (45) days before the completion date. You may not open and begin operating a Community without our prior written consent.

3.1.3 After the Effective Date, review your financial projections, including help with labor modeling and pricing your rent and care packages.

3.1.4 A full training program, as described in Section 6.5 below.

3.2 <u>Ongoing Assistance.</u>

After the Opening Date, we will provide you the following assistance:

3.2.1 Regular consultation and advice in response to your inquiries about specific administrative and operating issues. We may decide how best to communicate such consultation and advice to you, whether by telephone, in writing, electronically, or in person. The method we choose may be different from the methods we use for other franchisees.

3.2.2 Administer the General Marketing Fund (described in Section 8.1 below) and approve advertising you create for your local use.

3.2.3 Make goods and services available to you either directly or through designated and approved suppliers.

3.2.4 Periodically revise the Operations Manual to incorporate new developments and changes in the BrightStar System and give you electronic access to all updates.

4. <u>FEES AND OTHER PAYMENTS</u>

4.1 <u>Initial Franchise Fee.</u>

You agree to pay us a nonrecurring and nonrefundable initial franchise fee of Fifty-Thousand Dollars (\$50,000) when you sign this Agreement. This fee is fully earned by us when paid. If you choose not to develop the second and/or third BRIGHTSTAR CARE HOMES[®] Community that this Agreement entitles you to develop and operate, no part of the initial franchise fee is refundable.

If you or your affiliate is an existing BrightStar Care Agency franchisee, you or your affiliate agrees to pay us a nonrecurring and nonrefundable initial franchise fee of Fifty-Thousand Dollars (\$50,000). This fee must be paid, and is fully earned by us, on the following dates:

4.1.1 Thirty-Thousand Dollars (\$30,000) is due when you sign this Agreement.

4.1.2 Ten-Thousand Dollars (\$10,000) is due on the date we accept the site for your second Community to be developed under this Agreement.

4.2.3 Ten-Thousand Dollars (\$10,000) is due on the date we accept the site for your third Community to be developed under this Agreement.

If you or your affiliate is an existing BrightStar Care Agency franchisee and you choose not to develop the second and/or third BRIGHTSTAR CARE HOMES[®] Community that this Agreement entitles you to develop and operate, the additional Ten-Thousand Dollar (\$10,000) portions of the initial franchise fee will not be payable.

4.2 <u>Royalty/Continuing Fee.</u>

Royalty Fees will be based on a percentage of Net Billings beginning on the Opening Date. Beginning on the Opening Date and for the remainder of the Initial Term, you must pay us, without offset, credit, or deduction of any nature, a monthly royalty/continuing fee (the "Royalty/Continuing Fee") equal to five percent (5%) of Net Billings during the preceding calendar month. The Royalty/Continuing Fee is due and payable via electronic funds transfer ten (10) days after the date of our invoice.

4.3 <u>Net Billings.</u>

"Net Billings" means the aggregate of all revenues and other income from whatever source (whether in the form of cash, credit, agreements to pay or other consideration, and barter transactions, and whether or not payment is received at the time of sale or any amounts prove uncollectible) arising or derived by you or any other person from business conducted by or at, or originating from, the Community. Net Billings also include all proceeds from any business interruption insurance. Excluded from Net Billings are: (a) sales and other taxes separately stated that you collect from clients and actually pay to taxing authorities; (b) the discount value of any coupon, voucher, or other allowance we authorize at the time you redeem the client's coupon, voucher, or allowance; and (c) no mark-up items such as personal protective equipment, testing costs, or credit card fees where the amount billed to client is at your cost.

4.4 <u>General Marketing Fee.</u>

Beginning on the Opening Date and for the remainder of the Initial Term, you must pay us, without offset, credit, or deduction of any nature, a monthly General Marketing Fee equal to the greater of (a) two and one-half percent (2.5%) of your Net Billings during the preceding calendar month or (b) \$500 per month per Community for up to three (3) Communities operating under this Agreement (the "General Marketing Fee"). The General Marketing Fee is due and payable via electronic funds transfer ten (10) days after the date of our invoice. We reserve the right to collect General Marketing Fees more frequently than monthly upon thirty (30) days' prior written notice to you.

4.5 <u>Monthly Technology and Email Service Fee.</u>

If we implement the Monthly Technology & Email Service Fee, in whole or in part, in connection with your operation of your Communities, then beginning on each Community's Opening Date (unless we require you to begin using the System Technology at a later time, in which case such fee will begin upon the date we specify) and for the remainder of the Initial Term, you must pay BrightStar Technology, without offset, credit or deduction of any nature, a monthly Technology and Email service fee (the "Monthly Technology & Email Service Fee") on the 15th day of each month. If charged by us, the Monthly Technology & Email Service Fee does not include any fees associated with Third-Party Materials. The Monthly Technology & Email Service Fee currently will be the greater of (a) \$100 per month or (b) .83% of the prior month's Net Billings for each Community operating under this Agreement during the Initial Term. However, we have the right to increase the fixed fee portion of the fee (per Community) if our costs increase, in which case the \$100 fixed fee portion of the calculation may go up during the course of the franchise term.

4.5.1 If you acquire the Community operated under this Agreement as a result of a transfer, and we require use of System Technology, you must begin paying the Monthly Technology & Email Service Fee as of the effective date of the transfer.

4.5.2 If we have not started to charge you the Monthly Technology & Email Service Fee, BrightStar Technology will provide you and your staff with branded email. The fee for branded email will be \$10 per email per month ("Branded Email Fee") and starts as soon as you sign this Agreement.

4.5.3 Payment will be prorated for any partial month during which the Community was open for business and payable as per the terms of our or BrightStar Technology's invoice. We reserve the right to have BrightStar Technology, or any subsequent designee, collect the Monthly Technology & Email Service Fee or Branded Email Fee more frequently than monthly upon 30 days' prior written notice to you.

4.5.4 If we do not require you to use System Technology, we will notify you of the electronic client-management software you must use. Payments for the designated software will be deducted from your bank account on the 15th day of each month via EFT.

4.6 <u>Vendor Evaluation Fee.</u>

If you request our approval to allow you to use an unapproved vendor for products or services and that request requires us to invest time and expense to evaluate the unapproved vendor, then you must pay us up to Five Thousand Dollars (\$5,000) for our time and costs evaluating if we choose to add the unapproved vendor to our approved vendor list. However, no vendor evaluation fee is payable if we do not approve the unapproved vendor.

4.7 <u>Annual Increase in Certain Fixed Fees.</u>

Besides the right we and our affiliates reserve elsewhere in this Agreement to modify periodically certain fees and payments described in this Agreement (which modifications are not limited by this Section 4.7), we and our affiliates reserve the right to increase annually the fees and payments specified in Sections 4.4 (General Marketing Fee), 4.5 (Monthly Technology & Email Service Fee), 4.6 (Vendor Evaluation Fee), 4.9 (Late Fee), 6.14 (Inspection Fee), 8.2 (Local Marketing), 10.1 (Condemnation), 13.4.1.8 (Transfer Fee), 14.9 (Step-In Per Diem Fee), and 15.1.2.4 (De-Identification Non-Compliance Fee) based on changes in the Index (defined below) ("Annual Increase"). An Annual Increase to such fees and payments may occur only once during any calendar year and may not exceed the corresponding cumulative increase in the Index since the Effective Date or, as the case may be, the date on which the last Annual Increase became effective for the particular fixed fee or payment being increased. Any and all Annual Increases will be made at the same time during the calendar year. "Index" refers to the Consumer Price Index for All Urban Consumers (CPI-U) for the U.S. City Average for all Items (1982 – 1984 = 100), not seasonally adjusted, as published by the United States Department of Labor, Bureau of Labor Statistics, or in a successor index.

4.8 Method of Payment and Electronic Funds Transfer.

Unless otherwise agreed between us and you, all fees and other amounts due to us or any affiliate must be paid in the form of an electronic or similar funds transfer in the appropriate amount(s) from your bank account. We reserve the right to require you to pay any fees due under this Agreement at any intervals we designate and by such means we specify from time to time. You agree to execute and deliver to your bank and us all documents necessary to authorize such withdrawals and to make payments or deposits as directed by us or any of our affiliates. The authorization forms for electronic transfer of funds are attached as Exhibits B-1 and B-2. You further agree not to terminate such authorization while this Agreement is in effect. You agree not to close such bank account without prior notice to us and first establishing a substitute bank account permitting such withdrawals. If a direct electronic funds transfer or other withdrawal program is not available at the bank at which you currently do business, you will take all reasonable and necessary steps to establish an account at a different bank with such a program.

4.9 Payment and Interest on Late Payments.

You agree to pay us a Two Hundred Fifty Dollar (\$250) late fee for each required payment not made on or before its original due date and for each payment not honored by your financial institution. (You also must reimburse our bank charges for your dishonored payments.) This late fee is not interest or a penalty but compensates us for increased administrative and management costs due to your late payment. In addition, all amounts you owe us that are more than seven (7) days late will bear interest, accruing as of their original due date, at one and one-half percent (1.5%) per month or the highest commercial contract interest rate the law allows, whichever is less. We may debit your bank account automatically for late fees and interest. This Section is not our agreement to accept any late payments or our commitment to extend credit to, or otherwise finance your operation of, a Community.

4.10 <u>Application of Payments.</u>

We may, despite your designation, apply any of your payments to any of your past due indebtedness to us and our affiliates. We may set off any amounts you or your owners owe us or our affiliates against any amounts that we or our affiliates owe you or your owners. You may not withhold payment of any amounts you owe us or our affiliates due to our alleged nonperformance of any of our obligations under this Agreement.

4.11 <u>Taxes on Payments and Currency.</u>

You and your owners are solely responsible for all taxes, however denominated or levied upon you or the Community, in connection with the business you conduct under this Agreement. We will have no liability for any sales, use, service, occupation, excise, gross receipts, income, property, or other taxes due to the business you conduct (except for our own income taxes). You must pay these taxes and reimburse us for any taxes we must pay to any taxing authority on account of either your operation or payments you make to us (except for our own income taxes). You also must reimburse us for any taxes you must withhold in connection with the receipt or accrual of service fees, royalties, or any other amounts payable by you to us under this Agreement. Any additional required payment pursuant to the preceding sentences must be made in an amount necessary to provide us with after-tax receipts (taking into account any additional payments required under this Agreement) equal to the same amounts we would have received under the provisions of this Agreement had such additional tax liability or withholding not been imposed or required.

5. <u>LICENSED MARKS</u>

5.1 <u>Ownership.</u>

You expressly acknowledge our and our affiliates' rights in and to the Licensed Marks and agree not to represent in any manner that you have acquired any ownership rights in the Licensed Marks. You agree not to use any Licensed Mark, or any confusingly similar marks, names, or indicia, in your own corporate or business name, e-mail address, or domain name, except as authorized in this Agreement. Any and all goodwill associated with the BrightStar System and the Licensed Marks inures directly and exclusively to the benefit of us and our affiliates. Upon termination or expiration of this Agreement, no monetary amount will be attributable to any goodwill associated with your use of the Licensed Marks.

5.2 <u>Authorized Use.</u>

Your use of the Licensed Marks other than as expressly authorized by this Agreement, without our prior written consent, constitutes an infringement of our and our affiliates' rights in the Licensed Marks. Your right to use the Licensed Marks does not extend beyond the expiration or termination of this Agreement. During and after the Initial Term, you may not, directly or indirectly, commit any act of infringement, contest or aid others in contesting the validity of our and our affiliates' right to use the Licensed Marks, or take any other action in derogation of the

Licensed Marks. You must identify yourself as the Community's independent owner (in the manner we prescribe) in conjunction with any use of the Licensed Marks, including, without limitation, on invoices, contracts, timesheets, checks, receipts, employment and HR-related documents, and business stationery, and at conspicuous locations in the Community that we designate in writing. We may withdraw your ability to use the logo and/or Licensed Marks on any applicant or employee-facing communications, forms, or advertisements by providing 30 days' advance written notice. To the extent you use, with our permission, any Licensed Mark in employment-related materials, you must include a clear disclaimer that you (and only you) are the employer of the Community's employees and that we and our affiliates are not their employer or joint employer and do not engage in any employer-type activities for which only you are responsible, such as employee selection, promotion, termination, hours worked, rates of pay, other benefits, work assigned, discipline, adjustment of grievances and complaints, and working conditions. Notwithstanding your use of any Third-Party Materials or System Technology (if applicable), you are solely responsible and liable for understanding and following wage and hour laws as well as all other state, local, and federal laws applicable to the operations of the Community. You also must obtain an acknowledgment (in the form we specify or approve) from all Community employees that you (and not we or our affiliates) are their employer.

5.3 <u>Infringement.</u>

We and/or our affiliates have the sole right to handle disputes with third parties concerning our and our affiliates' ownership of and rights in, or your use of, the Licensed Marks or the BrightStar System. You must immediately notify us in writing if you receive notice or learn of any: (a) improper use of any Licensed Mark or element of the BrightStar System; (b) use by any third party of any mark, design, logo, or commercial symbol which, in your judgment, may be confusingly similar to any Licensed Mark; (c) use by any third party of any business practice which, in your judgment, unfairly simulates the BrightStar System in a manner likely to confuse or deceive the public; or (d) claim, challenge, suit, or demand asserted against you based on your use of the Licensed Marks or the BrightStar System.

We and our affiliates have the right to take any action we and they deem appropriate, including no action, and the sole right to control any legal proceeding or negotiation arising out of any infringement, challenge, or claim or otherwise relating to the Licensed Marks or the BrightStar System. You must not settle or compromise any claim, suit, or demand asserted against you and are bound by our and our affiliates' decisions in handling disputes regarding the Licensed Marks and the BrightStar System. You must cooperate fully with us and our affiliates and execute any documents and perform any actions that, in our judgment, are necessary, appropriate, or advisable to defend such claims, suits, or demands and to protect and maintain our and our affiliates' rights in the Licensed Marks and the BrightStar System. We agree to reimburse you for all damages, claims, and expenses you incur or for which you are liable in any proceeding challenging your right to use any Licensed Mark, provided your use of the Licensed Mark has been consistent with this Agreement, the Operations Manual, and System standards communicated to you and you have timely notified us and our affiliates of, and comply with our and our affiliates' directions in responding to, the proceeding. Because we and our affiliates will defend the third-party claim

relating to your right to use the Licensed Marks, we and they will not reimburse you for legal or other professional fees or costs you pay to independent legal counsel or others in connection with the matter.

5.4 <u>Operation Under Licensed Marks.</u>

You must use only the Licensed Marks we designate and must use them only in the manner we authorize. You agree to operate and advertise only under the names or marks we periodically designate for use by similarly-situated franchisees; to refrain from using the Licensed Marks to perform any activity or incur any obligation or indebtedness that might subject us to liability; to observe all laws regarding registration of trade names and assumed or fictitious names, include in any such application a statement that your use of the Licensed Marks is limited by this Agreement's terms, and give us a copy of any such application and other registration document(s); to observe such requirements regarding trademark and service mark registrations and copyright notices we periodically require, including, without limitation, affixing "SM," "TM," or "®" adjacent to all such Licensed Marks in any and all uses; and to utilize other appropriate notices of ownership, registration, and copyright we require.

5.5 <u>Modification/Replacement of Licensed Marks.</u>

We reserve the right to designate one or more new, modified, or replacement Licensed Marks for use by you and require your use of any such new, modified, or replacement Licensed Marks in addition to or in lieu of any previously-designated Licensed Marks. We and our affiliates need not reimburse your direct expenses for changing the Community's signage, your lost revenue due to any modified or discontinued Licensed Mark, or your expenses to promote a modified or substitute trademark or service mark. We and our affiliates' rights in this Section 5.5 apply to any and all of the Licensed Marks (and any portion of any Licensed Mark) this Agreement authorizes you to use. We and our affiliates may exercise these rights at any time and for any reason, business or otherwise, we and they think best. You acknowledge both our and our affiliates' right to take this action and your obligation to comply with our and their directions.

5.6 <u>Non-Exclusive License.</u>

The license to you of the Licensed Marks is nonexclusive, and we and our affiliates retain the right, among others, to (a) use the Licensed Marks ourselves in connection with selling Services, other services, and products, (b) grant other licenses for the Licensed Marks, and (c) develop and establish other systems using the Licensed Marks, similar marks, or any other marks, and grant licenses for them, without providing any further rights to you.

6. <u>STANDARDS OF OPERATION</u>

6.1 <u>Construction, Development, and Opening of First Community.</u>

You must complete certain steps in the first Community's construction and development process on or before <u>each</u> of the following specific deadlines:

(a) the Premises and Community must "Begin Construction" within one hundred eighty (180) days after we accept the site. If you or your affiliate purchased (or intends to purchase) the real property for the Premises and will own the Community's physical structure, the term "Begin Construction" means that (i) you (or your affiliate) have secured all financing required to construct, develop, and operate the Community in compliance with System standards, (ii) you (or your affiliate) have selected and formally hired (with a signed contract) a general contractor, (iii) you (or your affiliate) have obtained all relevant building permits, and (iv) the construction company has taken concrete steps consistent with beginning construction work in earnest, such as erecting a fence or clearing the site. [If you intend to lease (rather than purchase) the real property or building for the Premises and the Community's physical structure, the term "Begin Construction" means that (1) the developer has taken concrete steps consistent with beginning construction work in earnest, such as erecting a fence or clearing the site. [If you intend to lease (rather than purchase) the real property or building for the Premises and the Community's physical structure, the term "Begin Construction" means that (1) the developer has taken concrete steps consistent with beginning construction work in earnest, such as erecting a fence or clearing the site, (2) the developer has approved the lessee, and (3) you have secured all required financing, executed a twenty (20)-year lease for the Premises, and paid any relevant deposits.];

(b) the Premises and Community must have reached Complete Construction within five-hundred-and-fifty (550) days after the Effective Date. The term "Complete Construction" means that you have received a Certificate of Occupancy from local authorities; and

(c) the Community must Open for business within thirty (30) days after you receive the Certificate of Occupancy (the "Opening Date"). The term "Open" means that (i) the Community is fully licensed to operate as an assisted-living/memory-care BRIGHTSTAR CARE HOMES[®] Community in your market area, (ii) the Community is prepared to receive its first residents, and (iii) we approve the Community to open. However, you may not open and begin operating any Community without our prior written consent. Your failure to satisfy any of the deadlines identified in clauses (a) through (c) above will be a default of this Agreement.

6.2 <u>Construction, Development, and Opening of Second and Third Communities.</u>

(a) You must comply with our standards and specifications for the construction, development, and opening of your second and third Communities, including but not limited to the applicable requirements described in Section 6.1 above. In order to retain the exclusive right to establish and operate BRIGHTSTAR CARE HOMES[®] Communities whose premises are physically located within (i) the Protected Territory described in your (or your affiliate's) BRIGHTSTAR CARE® Agency Franchise Agreement(s), as described in Section 1.3(a)(i) above, or (ii) the Site Selection Area described in Exhibit H to this Agreement (if you or your affiliate is not an existing BrightStar Care Agency franchisee), as described in Section 1.3(a)(ii) above, you must Open for business (as defined in Section 6.1(c) above) your second and third Communities by the following deadlines:

(1) the second Community must Open for business on or before the two (2)-year anniversary of the date on which the first Community actually opens for business; and

(2) the third Community must Open for business on or before the eighteen (18)-month anniversary of the date on which the second Community actually opens for business.

(3) Notwithstanding the deadlines specified above in this Section 6.2(a) for you to retain your location exclusivity in the Protected Territory described in your (or your affiliate's) BRIGHTSTAR CARE® Agency Franchise Agreement(s) or, if you or your affiliate is not an existing BrightStar Care Agency franchisee, in the Site Selection Area described in Exhibit H, your right to construct, develop, and commence operating your second and third Communities expires five (5) years after the Effective Date. You have no right to commence operating an additional Community after the five (5)-year anniversary of the Effective Date.

6.3 <u>Pre-Sales.</u>

No later than ninety (90) days before the Opening Date, you must commence offering and selling residence and other care opportunities to be provided by the Community after it opens ("Pre-Sale Activities") once (a) we authorize you to engage in Pre-Sale Activities, (b) we accept the location and facilities from which you will engage in Pre-Sale Activities, (c) you or your Administrator has completed to our satisfaction the pre-opening training described in Section 6.5, and (d) you have secured all financing and permits necessary to develop, build, and fully equip the Community. You certify that, before engaging in any Pre-Sale Activities, you will obtain all necessary bonds and otherwise comply with all applicable laws relating to Pre-Sale Activities.

6.4 <u>Quality Standards for Fixtures, Equipment and Supplies.</u>

All fixtures, furnishings, furniture, equipment, signs, and other operating assets (collectively, "Operating Assets") and supplies that you acquire for the Community must meet our quality standards set forth in the Operations Manual or otherwise in writing, subject to compliance with applicable laws and regulations.

6.5 <u>Training.</u>

6.5.1 Throughout the Initial Term, an owner of yours must be designated your "Control Person," responsible personally for devoting his or her best efforts to the Community's construction, development, and operation and to whom we may give, and from whom we may receive, direction. You may have only one (1) Control Person at any given time. Your Control Person as of the Effective Date is identified in Exhibit K. The Control Person also may be the Community's Administrator, as long as he or she has a minimum of 5 years' experience working in an Assisted-Living and/or Memory-Care residence. If your Control Person transfers his or her ownership interest in you (with our approval) during the Initial Term, you must designate a new Control person (whom we must approve), and have that new Control Person attend and satisfactorily complete our required training (defined below), within the timeframe we specify.

6.5.2 Your Control Person must attend at your expense, within eight (8) weeks after the Effective Date, BRIGHTSTAR CARE HOMES[®] Getting Started Training (Phases 1 and 2) ("Getting Started Training"), which consists of up to three (3) days of training at our headquarters in Bannockburn, Illinois or another location we specify. We may supplement this training by e-learning through an Online Training System ("Online Training"). We also may shorten this training through the use of Online Training.

6.5.3 Before the Opening Date, the Control Person and your Administrator (if the Control Person is not also the Administrator) and salesperson each must attend and complete to our satisfaction the portions of BRIGHTSTAR CARE HOMES[®] Road to Opening Training designated for their positions. Road to Opening Training currently is scheduled for two (2) to four (4) days (depending on the trainee) at our headquarters in Bannockburn, Illinois or another location we specify. The duration of Road to Opening Training may be supplemented or reduced by Online Training. If we determine that the Control Person and/or Administrator cannot satisfactorily complete the portions of Road to Opening Training designated for him or her (and he or she, or a replacement, cannot satisfactorily complete a repeat training program), we may terminate this Agreement.

6.5.4 If any executive or managerial employee is not qualified or suitable to hold his or her position, you must pay our then-current fee to train replacements. Our training program may include a "train the trainer" module so that your senior-level personnel can learn how to train your other employees.

6.5.5 Your Administrator (who may also be the Control Person) may request additional or repeat training at the end of Getting Started Training and Road to Opening Training if he or she does not feel sufficiently trained to operate a BRIGHTSTAR CARE HOMES[®] Community. We and you will jointly determine the duration of any additional training, which is subject to our personnel's availability. However, if your Administrator (who may also be the Control Person) satisfactorily completes Getting Started Training and Road to Opening Training and does not expressly inform us that he or she does not feel sufficiently trained to operate a BRIGHTSTAR CARE HOMES[®] Community, then he or she will be deemed to have been trained sufficiently to operate BRIGHTSTAR CARE HOMES[®] Communities.

6.5.6 Replacement personnel of the following leadership positions— Administrators (even if also a replacement Control Person), Directors of Nursing, Office Manager, and other employees we designate—likewise must complete applicable training programs to our satisfaction after you hire them (typically within 60 days). You must pay our then-applicable fee for all such training and are responsible for all personnel and travel expenses relating to such training programs.

6.5.7 We may require your Administrator (who may be the Control Person), Director of Nursing, Office Manager, and other employees we designate to attend and satisfactorily complete supplemental training each year during the Initial Term at the times and locations we designate. We may charge reasonable registration or similar fees for these courses. You are responsible for all expenses incurred to attend such training. 6.5.8 If you request and we agree to provide, or we determine that you need, additional or special guidance, assistance, or training during the Initial Term, you agree to pay the per-diem charges (including wages) of our personnel as well as their travel, hotel, and living expenses.

6.5.9 Any training we provide is to provide you and your key representatives more information about brand standards and best practices. At all times, any individual you send to training is your employee, and you (not we) are solely responsible for all employment and personnel matters related to any employee you send to training. We do not reserve the right and expressly disavow any right or ability to control any employees you send to training. By sending an individual to training, you designate such employee as a key representative of your Agency to whom we can speak.

6.6 <u>Hiring and Supervision.</u>

6.6.1 You must hire a sufficient number of qualified, competent personnel in order to offer prompt, courteous, high quality, and efficient service to Community residents and their family members, and otherwise operate the Community in compliance with the BrightStar System and all applicable federal, state, and local requirements, in order to preserve, maintain, and enhance the System's reputation and goodwill.

- Required initial personnel for a Community include:
 - An Administrator (for up to 3 Communities). Some states require the Administrator to be licensed. It is your responsibility to determine your state's requirements.
 - A part-time salesperson. If you or your affiliate is a BrightStar Care Agency franchisee, these duties can be assumed by an existing salesperson in your (or your affiliate's) organization.
 - A part-time Director of Nursing (DON) (unless state license requires full-time). Recommend that she or he holds a RN license or higher with a minimum of 2 years of experience working in an assisted-living or a memory-care residence. If you or your affiliate is a BrightStar Care Agency franchisee, these duties may be assumed by an existing DON in your (or your affiliate's) organization so long as capacity within that role exists.
 - A staff of full-time and part-time caregivers on site is required to meet resident needs 24/7. A staffing ratio of 1:8 is our minimum standard requirement during traditional 1st and 2nd shift hours (i.e., 6:00 a.m.-10:00 p.m.). A staffing ratio of 1:14 during overnight hours is acceptable if appropriate based on resident needs. As resident needs increase, adding part-time or full-time staff on any shift may be

needed to provide a higher standard of resident care. It is your responsibility to determine your state's requirements and staff accordingly.

- Optional personnel to consider supporting business growth:
 - House Manager overseeing the day-to-day operations and sales efforts of the Community;
 - o Housekeeper (part-time)
 - o Life enrichment coordinator (part-time)
 - Cooking support (part-time)

Unless state regulations have higher standards for nursing oversight, you also must employ at least one part-time Registered Nurse to oversee the clinical components of each Community's operation, including resident care.

All employees must present a neat and clean appearance (including satisfying minimum dress standards) and render competent, empathetic service to the Community's residents and their family members. We may dictate the Community's minimum staffing requirements in order to ensure the Community operates in compliance with our then-current standards and specifications and thereby protect the reputation and goodwill of the Licensed Marks and businesses operating under the BrightStar System.

The Community must be under the Control Person's direct control and supervision. You will keep us informed at all times of the identity of any personnel required to sign the confidentiality and non-competition covenants under Section 12. Unless state regulations have higher standards for nursing oversight, you must employ, at a minimum, one part-time Registered Nurse to oversee the clinical components of the Community's operation, including resident care. You (and not us) are responsible for all employment decisions regarding the Community, including hiring, firing, training, promotion, wage and hour requirements, recordkeeping, supervision, and discipline. We will not, directly or indirectly, exercise or reserve control over employment and personnel matters and decisions involving your Community's employees. You must communicate clearly with your employees in your employment agreements, human resources manuals, written and electronic correspondence, paychecks, and other materials that you (and only you) are their employer and that we, as the franchisor of BRIGHTSTAR CARE HOMES® Communities, are not their employer or joint employer and do not engage in any employer-type activities (including those described above) for which only franchisees are responsible. Notwithstanding your use of any Third-Party Materials or System Technology (if we require you to use System Technology), you are solely responsible for understanding and following wage and hour laws as well as all other state, local and federal laws applicable to the operations of the Community. You also must obtain an acknowledgment (in the form we specify or approve) from all Community employees that we and our affiliates are not their employer or joint employer.

You must conduct criminal background checks on all prospective Community employees to determine a history of elder abuse, crimes involving elders, or similar crimes ascertainable on the public record. You must ensure that your employees pass applicable tests required by any governmental entity, submit to pre-employment and random drug tests (if allowed by state law), and participate actively in safety training seminars and programs. You are solely responsible for all hiring and other employment decisions and compliance with all applicable federal and state employment practices, including, but not limited to, wage and hour laws, overtime pay requirements, OSHA training, Hepatitis B vaccination, and Health Insurance Portability and Accountability Act (HIPAA). Notwithstanding your use of any Third-Party Materials or System Technology (if we require you to use System Technology), you are solely responsible and liable for understanding and following wage and hour laws as well as all other state, local and federal laws applicable to the operations of the Community.

6.7 <u>Maintenance of Premises.</u>

You agree at all times to maintain the Premises and the Community's Operating Assets in conformity with our then-current standards and to make such repairs and replacements as we require. You also must at all times meet and maintain all governmental standards and ratings applicable to the Community's operation or such higher minimum standards and ratings we set forth from time to time in our Operations Manual or otherwise in writing. You acknowledge and agree that we are not responsible for the safety and security of Community employees, clients, and visitors. Safety and security are your responsibility.

6.8 Operation of Community.

You agree to comply with all mandatory BrightStar System rules, regulations, policies, and standards in operating the Community, whether specified in this Agreement, the Operations Manual, or other materials provided by us to you from time to time. Without limiting the generality of the foregoing, you specifically agree as follows:

6.8.1 To purchase at your expense, and install and use, all Operating Assets we require, meet the specifications of the approved site layout and plan, and refrain from purchasing, or installing or using at the Premises or in connection with the Community's operation, any item not meeting our standards and specifications.

6.8.2 To maintain in sufficient supply, and use at all times, only operating products, materials, supplies, and expendables conforming to our then-current standards and specifications and refrain from using non-conforming items without our prior written consent.

6.8.3 To sell and offer for sale all Services and products we periodically require and only those Services and products we periodically approve, and do not subsequently disapprove, as meeting our quality standards and specifications. In addition to remodeling, repairs, replacement, and redecoration required by this Section 6 and elsewhere in this Agreement, in order to introduce new Services and products through all BRIGHTSTAR CARE HOMES[®]

Communities, we may require you to expend additional amounts on new, different, or modified Operating Assets necessary to offer such new Services or products.

6.8.4 To use only software and other Technology/Computer Requirements that we approve, which might include or be limited to System Technology. You may not use any other software or Technology/Computer Requirements without our prior written consent. We reserve the right to withdraw any such written consent at any time.

6.8.5 To use accounting systems (including chart of accounts), reporting forms, and other forms we periodically develop and file such forms with us as we require.

6.8.6 To record all billings and maintain all business information and records associated with the Community using the reporting systems and associated equipment we specify in the Operations Manual and maintain, without alteration, all information and categories we require to be programmed into the billing reporting system. You authorize us to access any and all information (other than information relating to the employment terms and conditions of the Community's employees, as you control exclusively your labor relations and employment practices) from such reporting systems and associated equipment, whether by inspection at the Premises or via retrieval by internet or other means. The reporting systems and associated equipment must be electronically accessible to us twenty-four (24) hours per day, every day of the year (including Sundays and holidays), and accessible in person during normal business hours. You agree not to impede our access to the reporting systems or associated equipment. While we have access to the information described in this Section and elsewhere in this Agreement, we do not own such information and you alone are responsible for protecting the privacy of the records of Community residents.

6.8.7 To fully comply with all policies and procedures regarding the collection, storage, use, processing, and transfer of personal data (i.e., any information identifying or capable of identifying an individual) that we may promulgate from time to time. You also agree to execute any agreements or other documents (including Business Associate Agreements), and to take any actions, we require you (or your affiliate) and all similarly-situated franchisees to execute or take from time to time to implement our data privacy or data security compliance program. We have the right (but no obligation), at our sole option at any time and from time to time during the Initial Term, to audit your (or your affiliate's) compliance with this Section 6.8.7 and to require you (at your expense) to enroll or maintain enrollment in a third-party audit and/or validation program we monitor.

6.8.8 To obtain all licensure required by state regulations to perform the BRIGHTSTAR CARE HOMES[®] Community business model to the fullest extent. You must immediately comply with all new or modified licensure requirements under applicable law. You also are responsible for complying with state and federal HIPAA standards.

6.8.9 To comply with our pricing standards and policies, which may (a) prohibit certain types of charges or billing practices we determine are misleading or otherwise detrimental to the System, and (b) require that you price consistently. We have the right, to the

extent allowed by applicable law, to control the minimum, maximum, and other prices for the Community's Services and products (including, if applicable, restrictions on your use of coupons and other price discounting practices). Our pricing standards and policies are not a representation or warranty by us that using such pricing standards and policies will produce, increase, or optimize your profits.

6.9 <u>Purchases.</u>

You agree that your (or your affiliate's) obligations in this Agreement and the Operations Manual are reasonable and necessary for the Community's operation and to maintain uniformity throughout the BrightStar System. You must adhere to the standards and specifications set forth in this Agreement and the Operations Manual, as periodically revised. You must use Operating Assets, supplies, inventory, marketing materials, and insurance and payroll services complying with our then-current standards and specifications, which we periodically may change. You agree to incur all associated costs.

6.9.1 Recognizing that preserving the BrightStar System depends upon product and service uniformity and maintaining our trade dress, you agree to purchase certain Operating Assets and inventory from us or approved or designated third-party suppliers we periodically specify in the Operations Manual or otherwise in writing. We, our affiliate and/or a third party may be one of several approved suppliers, or the only approved supplier, of any item. You agree to pay the then-current price in effect for the items you purchase from us or our affiliates. In some instances, the cost of items you purchase from us or our affiliates might be higher than the cost of similar supplies and products available on the market.

6.9.2 Except where we have designated a single source as noted in Section 6.9.1, if you wish to purchase any unapproved item and/or to acquire approved items from an unapproved supplier, you must provide us the proposed supplier's name, address, and telephone number, a description of the item you wish to purchase, and the item's purchase price, if known. You also must pay us our then-current supplier evaluation fee simply for beginning the supplier evaluation process. At our request, you must provide us, for testing purposes, a sample of the item you wish to purchase. We may base our approval of a proposed item or supplier on considerations relating not only directly to the item or supplier, but also indirectly to the uniformity, efficiency, and quality of operation we deem necessary or desirable for the System. We have no obligation to approve any particular supplier, or any particular number of suppliers, for a given item. We may revoke our approval of particular products or suppliers when we determine such products or suppliers no longer meet our standards or otherwise should not be offered through the System. You must cease purchasing products from such supplier upon receiving notice from us.

6.9.3 We periodically may establish business relationships with suppliers who produce Operating Assets, inventory, and other items according to our proprietary standards and specifications or private-label goods we have authorized or prescribed for sale by System franchisees ("System Suppliers"). You recognize that such items are essential to the Community's operation and the System generally. Your failure to pay System Suppliers could interfere with those suppliers' willingness to supply the System, resulting in other System franchisees' inability

to obtain product or ability to obtain product only on less favorable credit terms. Accordingly, you agree to pay System Suppliers as and when due.

6.9.4 We and our affiliates have the right to receive payments from suppliers on account of their actual or prospective dealings with us, our affiliates, you, and other franchisees and to use all amounts received without restriction for any purposes we and our affiliates deem appropriate.

6.10 <u>Hours of Business.</u>

You must at all times staff the Community with the number of employees necessary to satisfy either state licensing requirements or the staff profile described in the Operations Manual, whichever requires more staffing, to maximize both the quality of resident care and resident satisfaction. A manager on duty must be available at all times under the Control Person's direction.

6.11 <u>Printed Materials.</u>

You must use only business stationery, business cards, marketing materials, advertising materials, printed materials, and forms we approve in advance and meeting our standards.

6.12 <u>Identification of Community.</u>

In all advertising displays and materials at the Community, you must notify the public and your employees, in the form and manner we specify, that you operate the Community as an independent franchisee of ours.

6.13 <u>Third-Party Actions.</u>

You must notify us in writing within one (1) day after any written threat or the actual commencement of any action, suit or proceeding, or the issuance of any order, writ, injunction, award, or decree of any court, agency, or other governmental instrumentality, which might adversely affect the Community's operation or your (or your affiliate's) or the Community's financial condition.

6.14 <u>Inspection of Community Premises.</u>

To determine whether you and the Community are complying with this Agreement and all System standards, we and our designated agents and representatives (including quality assurance auditors) may at any time and without prior notice to you: (a) inspect the Community; (b) photograph the Community, digitally or otherwise, and observe and video its operation for consecutive or intermittent periods we deem necessary, including, but not limited to, through an Internet-accessible video monitoring system installed at the Community to which we have unrestricted access; (c) remove samples of any products and supplies; (d) interview the Community's personnel and clients; (e) inspect and copy any books, records, and documents regarding the Community's operation (other than those relating to the employment terms and conditions of the Community's employees, as you control exclusively your labor relations and employment practices); and (f) tour the Community for promotional purposes. You agree to cooperate fully with us and our agents and representatives in these activities. We may hire outside consultants and vendors to perform certain types of audits. If we exercise any of these rights, we will not interfere unreasonably with the Community's operation. If we (or our agents or representatives) inspect a Community, determine it is not operating in compliance with this Agreement and all System standards, and then must re-inspect the Community to determine whether you have corrected the operating deficiencies, we may require you to pay us up to Two Thousand Five Hundred Dollars (\$2,500) for the first and each subsequent re-inspection of the Community. You agree to give your clients the evaluation forms we periodically prescribe and to participate and/or request your clients to participate in any surveys performed by or for us. We agree to share the survey results with you.

6.15 <u>Possible Variation in Certain Standards.</u>

Because complete and detailed uniformity under many varying conditions might not be possible or practical, we specifically reserve the right and privilege to vary standards for any franchisee based upon the peculiarities of a particular site or circumstance, density of population, business potential, population of trade area, licensing requirements, existing business practices, or any other condition we deem important for the successful operation of such franchisee's business. You have no recourse against us on account of any variation from standards, specifications, and practices granted to any franchisee and are not entitled to require us to grant you a similar variation.

6.16 <u>Attendance at Annual Meetings.</u>

Your representative authorized by us must attend all annual franchisee and/or Community leadership meetings we conduct to address subjects relevant to the BrightStar System. We may use the annual meetings to offer continuing or advanced-level training instruction. If an annual meeting is held, we will determine the meeting's length and place and required attendees. We reserve the right to charge a registration fee (whether or not your authorized representative attends), and you must pay all travel and salary expenses. Your authorized representative need not attend more than three (3) business days of an annual meeting. We may combine annual meetings for BRIGHTSTAR CARE HOMES[®] Community franchisees with other brand conferences, which we may require your representative to attend. We may hold annual meetings via a virtual experience. Should we hold the annual franchise meeting via a virtual experience, you will be required to attend and pay the registration fee regardless of your participation in the virtual annual meeting.

6.17 Intellectual Property Belongs to Us.

If you or your employees or principals develop any new concept, process, or improvement for the operation or promotion of BRIGHTSTAR CARE HOMES[®] Communities, you must promptly notify us and send us all necessary related information, without compensation. Any such concept, process, or improvement will become our sole property, and we will be the sole owner of all related patents, patent applications, trademarks, copyrights, and other intellectual property rights. You and your principals hereby assign to us any and all such rights, including the right to modify such concept, process, or improvement, and otherwise waive and/or release all related rights of restraint and moral rights. You and your principals agree to assist us in obtaining and enforcing the intellectual property rights in such concept, process, or improvement in any and all countries and to execute all documents necessary for us to obtain and enforce such rights. You and your principals hereby irrevocably designate and appoint us as your agent and attorney-in-fact to execute and file any such documents and to do all other lawful acts to further the prosecution and issuance of patents or other intellectual property rights related to any concept, process, or improvement. If any part of this Section is invalid or otherwise unenforceable, you and your principals hereby grant to us a worldwide, perpetual, non-exclusive, fully-paid license to use and sublicense the use of the concept, process or improvement to the extent such use or sublicense would, absent this Agreement, directly or indirectly infringe your rights.

7. <u>OPERATIONS MANUAL</u>

7.1 <u>Compliance with Operations Manual.</u>

You must operate the Community in accordance with various written instructions, including technical bulletins and confidential manuals (referred to collectively as the "Operations Manual"), including amendments, as we publish from time to time, all of which you acknowledge belong solely to us and are available from us during the Initial Term via our Intranet. Our required standards or specifications are to protect the reputation and goodwill of the Licensed Marks and businesses operating under the BrightStar System, and to maintain standards of operations, rather than to exercise control over the Community's day-to-day operations, which is your exclusive responsibility. When any provision in this Agreement requires you to comply with any standard, specification, or requirement, such standard, specification, or requirement will, unless otherwise indicated, be as set forth in this Agreement or in the Operations Manual.

7.2 <u>Confidential Use.</u>

You must at all times use your best efforts to maintain the confidentiality of the Operations Manual and any other manuals (and their respective contents), materials, goods, and information created or used by us and designated for BRIGHTSTAR CARE HOMES[®] Communities and limit access to your employees on a need-to-know basis. You acknowledge that the unauthorized use or disclosure of our confidential information or trade secrets will cause irreparable injury to us, and damages are not an adequate remedy. You accordingly covenant that you will not at any time, without our prior written consent, disclose, use, permit the use of (except as may be required by applicable law or authorized by this Agreement), copy, duplicate, record, transfer, transmit, allow access to, or otherwise reproduce such information, in any form or by any means, in whole or in part, or otherwise make the same available to any unauthorized person or source. Any and all information, knowledge, and know-how not known about BRIGHTSTAR CARE HOMES[®] Communities and our products, services, standards, procedures, techniques, and such other information or material we designate as confidential will be deemed confidential for purposes of this Agreement.

7.3 <u>Revisions.</u>

We periodically may revise the Operations Manual's contents to implement new or different requirements for the Community's operation, and you expressly agree to comply with all changed requirements (none of which will change your fundamental status under this Agreement). We may provide Operations Manual updates electronically over our Intranet. Therefore, you agree periodically to check our Intranet, at least once per week, for such updates.

8. <u>ADVERTISING AND MARKETING</u>

8.1 <u>General Marketing Fund.</u>

8.1.1 Recognizing the value of marketing, advertising, promotional, CRM, public relations, and brand building and protection activities and programs (collectively, "Marketing") to the goodwill and public image of BRIGHTSTAR CARE HOMES[®] Communities, we have established a general marketing fund (the "Fund") to which you must contribute the General Marketing Fee specified in Section 4.4 above. BRIGHTSTAR CARE HOMES[®] Communities that we and our affiliates own and operate will contribute to the Fund on the same basis as franchisees (in terms of how such contribution is calculated). We have the right to collect for deposit into the Fund any allowances paid by suppliers who do business with BRIGHTSTAR CARE HOMES[®] CARE HOMES[®] Communities and with whom we agree to so deposit those allowances.

We will direct all Marketing that the Fund finances, with sole control over the creative concepts, graphics, materials, communications media, and endorsements used and their geographic, market, and media placement and allocation. The Fund may pay for creating, preparing, and producing video, audio, and written materials, graphics, and electronic and social media; developing, implementing, operating, and maintaining the Franchise System Website, Intranet and/or related strategies; administering national, regional, multi-regional, and local Marketing, including, without limitation, purchasing media advertising, conducting direct mail and other direct marketing campaigns, doing on-line Internet advertising and marketing (including search engine marketing / pay per click charges to search engines, banner advertising sources, and advertising host sites), conducting research and other Marketing tactics and using advertising, promotion, CRM, graphic design, marketing, and research agencies and other advisors to provide assistance; supporting public relations, market research, client satisfaction surveys, and other Marketing and research activities; paying dues for membership and participation in franchising and industry associations; paying third-party vendors to customize Marketing materials for local use by franchisees; and engaging in other brand enhancement activities. The Fund periodically may give you samples of Marketing formats and materials at no cost. We may sell you multiple copies of these materials at our direct production cost, plus any related shipping, handling, and storage charges.

8.1.2 We will account for the Fund separately from our other monies (although we need not keep Fund contributions in a separate bank account) and not use the Fund for any of our general operating expenses. However, we may use the Fund to pay the expenses we incur in activities reasonably related to directing the Fund and its programs, including, without

-27-

limitation, market research, public relations, creating, preparing, and producing Marketing materials, and collecting and accounting for Fund contributions; reasonable salaries and benefits of personnel who manage and administer the Fund; the Fund's other administrative costs, including taxes we must pay on Fund contributions we receive; travel expenses of personnel while they are on Fund business; meeting costs; overhead relating to Fund business; and franchisee conferences.

8.1.3 We periodically may assist franchisees to maintain high quality standards through client surveys, customer interviews, and other similar initiatives ("Surveys"). The Fund may pay for these programs, although we may charge these costs directly to you if your Survey results fall below established minimum standards in the System. Any such fees charged will be contributed to the Fund.

8.1.4 Although the Fund is not a trust, we will hold all Fund contributions for the contributors' benefit and use contributions only for the purposes described in this Section 8.1. We do not owe any fiduciary obligation to you for administering the Fund or any other reason. The Fund may spend in any fiscal year more or less than the total Fund contributions in that year, borrow from us, our affiliates, or others (paying reasonable interest) to cover deficits, or invest any surplus for future use. Deficits may be repaid from future contributions. We may accumulate monies in the Fund for the time periods we deem necessary or appropriate, with no obligation to spend any or all monies received in any fiscal year during that fiscal year. At the end of our fiscal year, any unused Fund contributions will roll over to the following fiscal year.

8.1.5 We will prepare an annual, unaudited statement of Fund collections and expenses and either give you a copy of the statement upon written request or post the statement on the Franchise System Website or Intranet. We may have the Fund audited annually, at the Fund's expense, by an independent certified public accountant we select. We may incorporate the Fund or operate it through a separate entity whenever we deem appropriate. The successor entity will have the rights and duties specified in this Section 8.1.

8.1.6 We intend the Fund to maximize recognition, and enhance system protection, of the Licensed Marks and increase occupancy of BRIGHTSTAR CARE HOMES[®] Communities. Although we will try to use the Fund to develop Marketing materials and execute Marketing activities and programs benefiting all BRIGHTSTAR CARE HOMES[®] Communities, we need not ensure that Fund expenditures in or affecting any geographic area are proportionate or equivalent to Fund contributions by BRIGHTSTAR CARE HOMES[®] Communities operating in that geographic area or that any BRIGHTSTAR CARE HOMES[®] Community benefits directly or in proportion to its Fund contributions from the development of Marketing materials or the execution of Marketing activities and programs. The Fund will not be used principally to develop materials and programs to solicit franchisees. However, media, materials, and programs, including the Franchise System Website, prepared using Fund contributions may describe the franchise program, reference the availability of franchises and related information, and process franchise leads.

8.1.7 We have the right, but no obligation, to use collection agents and institute legal proceedings at the Fund's expense to collect Fund contributions. We also may

forgive, waive, settle, and compromise all claims by or against the Fund. Except as expressly provided in this Section 8.1, we assume no direct or indirect liability or obligation to you for collecting amounts due to, maintaining, directing, or administering the Fund.

8.1.8 We may at any time defer or reduce a BRIGHTSTAR CARE HOMES[®] Community franchisee's contributions and, upon thirty (30) days' prior written notice to you, reduce the required Fund contribution or suspend Fund operations for one or more periods of any length and terminate (and, if terminated, reinstate) the Fund. If we terminate the Fund, we will at our option spend all remaining Fund monies on permitted Marketing or distribute all unspent monies to Fund contributors in proportion to their respective Fund contributions during the preceding fiscal year.

8.1.9 You agree that, in addition to the Fund activities described in this Section 8.1, the Fund may be used for joint or collective advertising campaigns with our direct or indirect parent corporations, subsidiaries, or affiliated companies designed to promote the BrightStar[®] trademark and other Licensed Marks (including, for example but without limitation, campaigns and other advertising activities promoting both BRIGHTSTAR CARE HOMES[®] Communities and BRIGHTSTAR SENIOR LIVING Communities).

8.2 <u>Local Marketing.</u>

Beginning with your required Pre-Sale Activities, you must spend at least Two Thousand Dollars (\$2,000) for each planned room in each Community for pre-opening Marketing activities. Beginning on the date on which a Community Opens for business and during each month afterward, you must spend at least One-Hundred Dollars (\$100) per month for local Marketing for each room in the Community if your occupancy rate is below 85%. If your occupancy rate is 85% or greater, you must spend at least Fifty Dollars (\$50) per month for local Marketing for each room in the Community. (We may increase these dollar amounts based on inflation.) Costs related to charges from referral companies (e.g., A Place for Mom, etc.) do not count towards your local Marketing spend requirement. Your local Marketing (including through digital, video, and social media platforms) must be creative from our libraries of approved marketing and advertising materials, be conducted in a dignified manner, and conform to our standards and requirements. You may not use any Marketing materials unless they are from our libraries of approved marketing and advertising and advertising materials. Any advertising or marketing materials not from our approved libraries must be submitted to us for approval and must align with our brand standards.

8.3 <u>Marketing Generally.</u>

You must place or display at the Community's Premises (interior and exterior) only the signs, emblems, lettering, logos, and displays and Marketing materials from our libraries of approved marketing and advertising materials. You may not use any Marketing materials we have not expressly approved or have disapproved. (Our failure to respond within ten (10) days will be deemed a disapproval.) All Marketing must prominently display, and comply with our standards for use of, the Licensed Marks. We may require you to stop using any Marketing materials within specific timeframes, at your sole cost and expense.

You agree to participate in system-wide and applicable regional promotions and advertising campaigns that we originate or approve. These may include promotions via the Internet, e-commerce, electronic media, or other technologically advanced media. You also agree to participate at your sole expense in, and comply with the rules and procedures of, all client loyalty, gift certificate, and similar programs we create.

8.4 <u>Website and Intranet Matters.</u>

8.4.1 You may not, without our prior written consent, develop, maintain, link to, or authorize any website mentioning or describing you or the Community, or displaying any of the Licensed Marks, other than the website hosted by us. "Website" means any part of the Internet (including social media) and any successor technology that enables the public to research and/or purchase services or goods by means of electronic commerce. We may establish a website providing information about the System and its Services and products (the "Franchise System Website"). We will be the webmaster, either directly or through a third party, with the right to control the Franchise System Website.

8.4.2 If we establish a Franchise System Website, we may provide you with a separate interior webpage (accessible only through the Franchise System Website) referencing the Communities and/or otherwise allow you to participate in the Franchise System Website. You will be trained on the CMS (content management system) to update your webpage. If you elect to hire an external vendor to make changes on your behalf, we must pre-approve the vendor. We must pre-approve your webpage before it is posted by our webmaster on the Franchise System Website. We have the continuing right to monitor and pre-approve your webpage's form, content, and quality during the Initial Term. We may reject your webpage if we consider your form or contents to be unacceptable. Your webpage always must comply with our System standards. You may modify your webpage only through and with the pre-approval of our webmaster. You must give us the information and materials we request for you to participate in the Franchise System Website. By giving us such information and materials, you represent they are accurate and not misleading and do not infringe any other party's rights. We will own all intellectual property and other rights in the Franchise System Website, your webpage, and all information they contain (including, without limitation, the log of "hits" by visitors and any personal or business data visitors supply).

We will control, and may use Fund contributions to develop, maintain, operate, update, and market, the Franchise System Website, including your webpage (if any). We will update information on your webpage, if any, or add information we approve as frequently as we deem appropriate. You must notify us whenever any information on your webpage changes or is not accurate. We have final approval rights over all information on the Franchise System Website, including your webpage, if any. We may implement and periodically modify System standards for the Franchise System Website.

We will maintain your webpage, if any, and otherwise allow you to participate in the Franchise System Website only while you are in substantial compliance with this Agreement and all System standards (including those for the Franchise System Website). If you are in material default of any obligation under this Agreement or System standards, we may, in addition to our other remedies, temporarily suspend your participation in the Franchise System Website until you fully cure the default. We will permanently terminate your access to and participation in the Franchise System Website upon this Agreement's expiration or termination.

8.4.3 We may, at our option, establish and maintain an Intranet and establish System standards for the Intranet's use. These System standards will address, among other things, (a) restrictions on using abusive, slanderous, or otherwise offensive language in electronic communications, (b) restrictions on communications among franchisees endorsing or encouraging breach of any franchisee's franchise agreement, (c) confidential treatment of materials we transmit via the Intranet, (d) password protocols and other security precautions, (e) grounds and procedures for our suspending or revoking a franchisee's access to the Intranet, and (f) a privacy policy governing our access to and use of electronic communications that franchisees post on the Intranet. We expect to adopt and adhere to a reasonable privacy policy. However, as the Intranet's administrator, we have the right to access and view any communication posted on the Intranet. You acknowledge that the Intranet and all communications posted to it are our property, free of any privacy or privilege claims that you or another person might assert.

Your obligation to maintain connection with the Intranet applies during the entire Initial Term (unless we dismantle the Intranet or suspend your access). If you fail to comply with any Intranet System standard, we may (in addition to any other rights we reserve in this Agreement) temporarily suspend your access to any chat room, bulletin board, list-serve, or similar feature the Intranet includes until you fully cure the breach.

8.4.4 You acknowledge that we are the lawful, rightful, and sole owner of the Internet domain name www.brightstarcarehomes.com and www.brightstarcarehome.com, and any other Internet domain names we register, and unconditionally disclaim any ownership interest in those or any similar Internet domain names. You agree not to register any Internet domain name in any class or category containing words used in or similar to any brand name owned by us or our affiliates or the names or likenesses of any executive or employee of us or our affiliates.

8.5 <u>Online Advertising.</u>

In addition to the Marketing obligations described above, at least ninety (90) days before your first Community opens, you must review listings and invest in Google Business Profile, SEM/PPC, Meta/Facebook and Yellowpages.com. The advertisement must conform to our standards and specifications, including the appropriate heading(s) we designate. We may use the Fund to cover online advertising as part of your local Marketing requirements. You must pay for advertising as part of your local Marketing requirement unless and until the Fund elects to pay for it. You must use our required vendor partner for any SEM/PPC advertising investments to ensure optimal campaign performance and brand continuity.

9. <u>MODIFICATIONS</u>

9.1 Modifications to the BRIGHTSTAR CARE HOMES[®] Community Program.

The business environment affecting our franchise program is continually changing. We may develop other business activities or modify existing business activities in response to changes in the business environment, including those resulting from technological advances, expansion into new markets, and other factors not presently anticipated. We reserve the right to change or modify the Licensed Marks, the BRIGHTSTAR CARE HOMES[®] Community concept, the Operations Manual, and any Technology/Computer Requirements that we require you to use. We may adopt and use new or modified trade names, trademarks, service marks, logos, equipment, furniture, furnishings, products, techniques, concepts, or Technology/Computer Requirements. We may add new and different Services and products and withdraw Services or products or change their names or image; redesign the trade dress, software programs, and equipment, furniture, or fixture standards; or discontinue them as we consider appropriate. You must accept and use the changes as if they were part of this Agreement. If changes relate to the Licensed Marks, you will have one hundred twenty (120) days from the notice date to implement such changes under this Section 9.1.

9.2 <u>Modifications not Requiring Significant Changes in Fixtures or Equipment.</u>

If any changes involving Services or products would not require installation of new fixtures or equipment, we may instruct you to begin offering the new Services or products on a date specified in a supplement to the Operations Manual or other notice. Likewise, if the withdrawal of a Service or product would not require removal of fixtures or equipment, we may direct you to stop offering the Service or product on a date specified in a supplement to the Operations Manual or other notice. You agree to comply with these instructions and directions by the required deadline. (You must comply with all Technology/Computer Requirements changes within the timeframe specified in Section 1.4.)

9.3 <u>Modifications Requiring Significant Changes in Fixtures or Equipment;</u> Upgrading the Community and CapEx Account.

You may not make any material changes to a Community's existing or planned construction, including any change in the number of rooms, without our prior written consent and complying with the conditions and procedures we periodically establish for those changes. Without limiting our rights and your obligations under Sections 9.1 and 9.2, we may require you at any time and from time to time during the Initial Term to upgrade or renovate the Community, including by altering the Community's appearance and/or replacing a material portion of improvements and/or furniture, fixtures, and equipment, to comply with then-current building décor, appearance, and trade dress standards and other aspects of the System that we have established and require for new, similarly-situated BRIGHTSTAR CARE HOMES[®] Communities. This upgrading or renovation may obligate you to invest additional capital in each of your Communities and/or incur higher operating costs. You agree to implement the upgrading and renovation within the time period we request, regardless of its cost or the point during the

Initial Term when we require you to do so, provided all such upgrades and renovations apply uniformly to all similarly-situated BRIGHTSTAR CARE HOMES[®] Communities.

To ensure you have funds available to make any necessary capital expenditures or leasehold improvements at the Community and comply with your obligations under this Section 9.3 (but without limiting those obligations), you must deposit into a separate account you control an amount equal to one percent (1%) of the Community's monthly Net Billings during the Initial Term. Upon our reasonable request, you will give us information concerning the funds in and expenditures from that account. You must use those funds only for making approved capital expenditures or leasehold improvements and complying with your upgrade and other obligations under this Section 9.3, although such obligations may require you to spend more than the amount then in that account.

9.4 <u>Test Marketing.</u>

If we permit you to participate in any new Service or product concept test, you must do so in compliance with our standards and requirements.

9.5 <u>Dual Branding.</u>

We may co-brand one or more concepts with BrightStar ("Dual Branding"). Dual Branding may involve changes to the Licensed Marks and Community. If Dual Branding occurs, the scope and type of Dual Branding may vary in different markets. If we elect to conduct Dual Branding in the Community's market, we will give you notice regarding the contemplated Dual Branding. Despite anything in this Section 9.5 to the contrary, however, we may elect not to make the Dual Branding opportunity available to you. You will have up to one hundred eighty (180) days from the notice date to implement any changes under this Section 9.5 at your own expense.

10. <u>CONDEMNATION AND DAMAGE</u>

10.1 <u>Condemnation.</u>

You must immediately notify us of any proposed taking of any portion of a Community by eminent domain, condemnation, or expropriation. If we agree that all or a substantial portion of the Community is to be taken, condemned, or expropriated, then upon your request we may (but have no obligation to) allow you to relocate the Community to a new location (within the same market) you select (subject to our acceptance) within four (4) months after the taking, condemnation, or expropriation. If you develop a new BRIGHTSTAR CARE HOMES[®] Community at a new location we accept (a "Relocated Community") and open that Relocated Community according to our specifications and this Agreement's other terms and conditions within twenty-four (24) months after closing the Community, the Relocated Community will then be deemed to be the Community, or one of the Communities, franchised under this Agreement.

If a taking, condemnation, or expropriation involving all or a substantial portion of a Community occurs, and following such taking, condemnation, or expropriation:

- (a) We elect not to allow you to develop a Relocated Community; or
- (b) You promptly notify us that you will not develop a Relocated Community;
- or

(c) You and we do not agree to a new location for a Relocated Community within the four (4) month period specified above;

either party has the right, effective immediately upon delivery of written notice to the other, (i) to terminate this Agreement, in the event that the Community that was taken, condemned, or expropriated is the only BRIGHTSTAR CARE HOMES[®] Community then in operation or under construction by you under this Agreement, or (ii) to remove from the scope of this Agreement the Community that was taken, condemned, or expropriated, in the event that such Community is not the only BRIGHTSTAR CARE HOMES[®] Community then in operation or under construction by you under this Agreement, or expropriated is not the only BRIGHTSTAR CARE HOMES[®] Community then in operation or under construction by you under this Agreement (in which case the remaining Communities will not be impacted).

If this Agreement is terminated pursuant to this Section 10.1 and you and your owners sign our then-current form of termination agreement and a general release, in a form satisfactory to us, of any and all claims against us and our owners, affiliates, officers, directors, employees, and agents, you need not pay termination damages under Section 15.2 at the time of termination. However, the termination agreement will provide that if you or any of your affiliates begins constructing a Competitive Business within or outside the Community's market area any time during the twenty-four (24) months following the effective date of termination, you and your owners must immediately pay us termination damages equal to Four Hundred Thousand Dollars (\$400,000). If you and your owners fail to sign the termination agreement and general release within a reasonable time after we deliver them to you, you must pay us termination damages under Section 15.2 at the time of termination, in addition to complying with your other post-termination obligations under this Agreement.

If we and you agree to a new location for a Relocated Community but you fail to develop and open the Relocated Community according to this Agreement's terms and conditions within twenty-four (24) months after the Community's closing (and the Community that was taken, condemned, or expropriated was the only BRIGHTSTAR CARE HOMES[®] Community then in operation or under construction by you under this Agreement), we may terminate this Agreement, effective immediately upon delivery of written notice to you. In that case, you must pay us termination damages under Section 15.2 in addition to complying with your other post-termination obligations under this Agreement.

10.2 Damage.

If a Community is damaged by fire or other casualty, you must notify us immediately. If the cost to repair the damage is less than or equal to the Damage Threshold (defined below), or if the cost to repair the damage exceeds the Damage Threshold but you notify us within thirty (30) days after the casualty that you intend to repair the damage and operate the Community as a BRIGHTSTAR CARE HOMES[®] Community, then you must repair the damage promptly

according to our System standards and this Agreement's other terms and conditions. The "Damage Threshold" means the greater of (a) fifty-one percent (51%) of the Community's replacement cost value immediately before the fire or other casualty, or (b) the amount of insurance proceeds made available to you in connection with the fire or casualty. If the damage or repair requires you to close all or any portion of the Community, you must commence reconstruction as soon as practicable (but in any event within four (4) months) after closing the Community and reopen for continuous business operations as a BRIGHTSTAR CARE HOMES[®] Community as soon as practicable (but in any event within twenty-four (24) months) after closing the Community, but not without complying with this Agreement's other terms and conditions.

If the cost to repair the damage from a fire or other casualty exceeds the Damage Threshold and you either fail to notify us within a reasonable time after the casualty that you intend to repair the damage and operate the Community as a BRIGHTSTAR CARE HOMES[®] Community, or notify us that you elect not to repair the damage and operate the Community as a BRIGHTSTAR CARE HOMES[®] Community, then we may terminate this Agreement and you must immediately pay us termination damages under Section 15.2. However, if you and your affiliates do not open a Competitive Business during the twenty-four (24) months after closing the Community, then the termination damages payable under Section 15.2 will not exceed the amount of any insurance proceeds you receive, and we will return the difference to you within twenty (20) days after the twenty-four (24) month period expires. You must send us the documentation we reasonably request to calculate the Damage Threshold and the insurance proceeds you receive in connection with any fire or other casualty.

10.3 Extension of Initial Term.

If the condemnation or casualty involves the first Community that you develop under this Agreement, the Initial Term will be extended for the period of time during which that Community is closed due to condemnation or casualty. (The Initial Term will not be extended if the condemnation or casualty involves the second or third Community developed under this Agreement.) You need not pay any Royalty/Continuing Fees or General Marketing Fees while a Community is closed due to condemnation or casualty unless you receive insurance proceeds compensating you for lost Net Billings during such period, as provided in the definition of Net Billings.

11. <u>STATEMENTS, RECORDS AND FEE PAYMENTS</u>

11.1 <u>Maintenance of Records.</u>

You must obtain the right to use, maintain, and update (including any required replacements) the accounting system we prescribe in the Operations Manual or otherwise. You must, in a manner satisfactory to us, maintain original, full, and complete records, accounts, books, data, licenses, contracts, and supplier invoices accurately reflecting all particulars relating to your business, and such statistical and other information or records we require, and keep all such information for not less than seven (7) years, even if this Agreement is no longer in effect. We have the right to access or receive copies of these materials (other than those relating to the

employment terms and conditions of the Community's employees, as you control exclusively your labor relations and employment practices).

11.2 Examination of Records.

We may at any time, and without prior notice to you, examine your and the Community's business, bookkeeping, and accounting records, sales and income tax records and returns, and other records (other than those relating to the employment terms and conditions of the Community's employees, as you control exclusively your labor relations and employment practices). You must cooperate fully with our representatives and independent accountants in any examination. We may require you to send records off-site for review. If an examination discloses an understatement of the Community's Net Billings, you must pay us within fifteen (15) days after receiving the examination report the Royalty/Continuing Fee, General Marketing Fees, Monthly Technology & Email Service Fee, and other monies due on the understated amount, our late fee, and interest on the understated amount from the date originally due until the date of payment. Furthermore, if an examination is necessary due to your failure to furnish reports, supporting records, or other information as required, or if our examination reveals a Royalty/Continuing Fee or other understatement exceeding two percent (2%) of the amount actually reported to us for the period examined, you must reimburse our costs for the examination, including, without limitation, the charges of attorneys and independent accountants and our employees' travel expenses, room and board, and compensation. These remedies are in addition to our other remedies and rights under this Agreement and applicable law.

11.3 <u>Reports and Financial Statements.</u>

Upon our request, you will prepare and send us signed reports and returns of Net Billings, bank statements, quarterly unaudited profit and loss statements, use and gross receipt taxes, complete copies of any state or federal income tax returns covering the Community's operation, and such other reports we reasonably request, all of which you must certify as true and correct. In addition, within forty-five (45) days after the close of your fiscal year, you must prepare and send us year-end financial statements, including statements of income, retained earnings, and changes in financial position and a balance sheet, which you must certify as true and correct.

12. <u>COVENANTS NOT TO COMPETE AND CONFIDENTIALITY</u> <u>OBLIGATIONS</u>

12.1 You Defined.

Unless otherwise specified, the term "you" as used in this Section 12 includes, collectively and individually, all of your owners, guarantors, officers, directors, members, managers, and partners.

12.2 <u>Confidential Information.</u>

For purposes of this Agreement, "Confidential Information" means confidential information, some of which constitutes trade secrets under applicable law, relating to constructing, developing, and operating BRIGHTSTAR CARE HOMES[®] Communities, including site selection criteria; specifications for Services; training and operations materials and manuals; methods, formats, specifications, standards, systems, procedures, sales and marketing techniques, knowledge, and experience used in developing and operating BRIGHTSTAR CARE HOMES[®] Communities; Marketing programs and materials for BRIGHTSTAR CARE HOMES[®] Communities; knowledge of specifications for and suppliers of Operating Assets and other items; knowledge of the operating results and financial performance of BRIGHTSTAR CARE HOMES[®] Communities other than your Communities; Technology/Computer Requirements; client communication and retention programs and data used or generated in connection with those programs; and graphic designs and related intellectual property.

12.3 <u>Non-Use and Non-Disclosure of Confidential Information.</u>

You agree that you will not acquire any interest in Confidential Information, other than the right to use it as we specify during the Initial Term to operate your Communities. Confidential Information is proprietary, includes our and our affiliates' trade secrets, and is disclosed to you only on the condition you agree, and you hereby do agree: (a) not to use Confidential Information in any other business or capacity; (b) to keep confidential each item deemed to be a part of Confidential Information, both during and after the Initial Term; (c) not to make unauthorized copies of any Confidential Information disclosed via electronic media or in written or other tangible form; (d) to adopt and implement reasonable procedures to prevent unauthorized use and disclosure of Confidential Information, including, without limitation, restricting its disclosure to Community personnel and using non-disclosure and non-competition agreements with the appropriate officers, directors, and other personnel having access to Confidential Information. We have the right to regulate or review the forms of non-disclosure and non-competition covenants you use - solely to ensure that you adequately protect Confidential Information and the competitiveness of BRIGHTSTAR CARE HOMES[®] Communities – and to be a third-party beneficiary of those provisions with independent enforcement rights; and (e) not to sell, trade, or otherwise profit in any way from Confidential Information except as authorized by this Agreement. Under no circumstances will we control or regulate the forms of employment agreements you use with your employees or otherwise be responsible for your labor relations or employment practices.

Confidential Information does not include information, knowledge, or know-how you can demonstrate lawfully came to your attention before we disclosed it to you; that, when we disclosed it to you, already had lawfully become generally known through publication or communication by others (without violating an obligation to us or our affiliates); that, after we disclose it to you, lawfully becomes generally known through publication or communication by others (without violating an obligation to us or our affiliates); or that you independently develop without access to or reliance on Confidential Information. However, if we include any matter in Confidential Information, anyone who claims it is not Confidential Information must prove one of the exclusions in this paragraph.

12.4 <u>Non-Compete Covenants.</u>

12.4.1 We have granted you the franchise in consideration of and reliance upon your agreement to deal exclusively with us in the assisted living and senior care industry in which BRIGHTSTAR CARE HOMES[®] Communities operate. You therefore agree that, during the Initial Term, neither you, any of your direct or indirect owners, nor any of such owners' spouses will:

12.4.1.1 have any direct or indirect controlling interest as an owner whether of record, beneficial, or otherwise—in a Competitive Business (defined below), wherever located or operating;

12.4.1.2 have any direct or indirect non-controlling interest as an owner—whether of record, beneficial, or otherwise—in a Competitive Business, wherever located or operating (except that less than a two percent (2%) equity ownership interest in a Competitive Business whose stock or other forms of ownership interest are publicly traded on a recognized United States stock exchange will not violate this clause);

12.4.1.3 perform services as a director, officer, manager, employee, consultant, representative, or agent for a Competitive Business, wherever located or operating;

12.4.1.4 divert or attempt to divert the Community's business or customers to a Competitive Business; or

12.4.1.5 engage in any other activity that might injure the goodwill of the Licensed Marks and System.

The term "Competitive Business" means (i) any business whose primary purpose is housing or caring for persons over the age of sixty (60), or (ii) any business that provides supplemental healthcare staff to institutional clients, such as hospitals, Medicare agencies, hospice agencies, assisted-living centers, nursing homes, and clinics, or (iii) any business that provides comprehensive care, including medical and non-medical services, to private-duty clients within their home, or (iv) any business granting franchises or licenses to others to operate the type of business specified in clauses (i), (ii), or (iii) (but not including any business operating under a franchise or license agreement signed with one of our affiliates).

12.4.2 We have granted you the franchise in consideration of and reliance upon your agreement to deal exclusively with us in the assisted living and senior care industry in which BRIGHTSTAR CARE HOMES[®] Communities operate. You therefore agree that, for a period of twenty-four (24) months after the expiration (without a successor franchise) or termination of this Agreement, regardless of the cause of termination, or within twenty-four (24) months after sale of the Communities or any interest in you, either directly or indirectly, neither you, any of your direct or indirect owners, nor any of such owners' spouses shall have a direct or indirect controlling or non-controlling ownership interest in, or perform services as a director, officer, manager, employee, consultant, representative, or agent for, any Competing Business: 12.4.2.1 Located at the premises of the former Community;

12.4.2.2 Located or operating within a thirty-five (35) mile radius of the former Community; or

12.4.2.3 Located or operating within a twenty-five-mile (25) radius of any other BRIGHTSTAR CARE HOMES[®] Community (whether owned by a franchisee, us, or our affiliates) in operation or under construction on the effective date of the expiration, termination, or transfer.

12.4.3 Neither you, your direct or indirect owners, nor any of such owners' spouses shall, for a period of twenty-four (24) months after the expiration (without a successor franchise) or termination of this Agreement, regardless of the cause of termination, or within twenty-four (24) months after sale of the Communities or any interest in you, solicit business from clients or referral sources of your former Community or Communities, or with whom the BrightStar System did business during the Initial Term, or contact any of our suppliers or vendors in connection with your ownership, management, operation, or maintenance of, or engagement or consulting with, any Competitive Business.

12.4.4 You agree to obtain similar covenants from the appropriate officers, directors, and other personnel. To give effect to your obligations in this Section 12.4, you acknowledge that neither you, any of your direct or indirect owners, nor any of their spouses will seek to violate this Section directly or through any other person with whom you or any other restricted party is acting in concert in connection with the prohibited activities. We may enforce this Section by acting against you, the other restricted parties, and all other persons with whom you are acting in concert in connection with the prohibited activities.

12.4.5 Each of the foregoing covenants will be independent of any other covenant or provision of this Agreement. If any court of competent jurisdiction or arbitrator determines that any aspect of the covenant is unreasonable, you and we agree that the court or arbitrator may determine an appropriate limitation to accomplish this Section's intent and purpose, and each party agrees to be bound by such determination. You and your owners and affiliates expressly acknowledge that you and they possess skills and abilities of a general nature and have other opportunities for exploiting these skills. Therefore, our enforcing the covenants made in Section 12.4 will not deprive you or your owners and affiliates of your and their personal goodwill or ability to earn a living.

13. TRANSFER AND ASSIGNMENT

13.1 <u>Transfer by Us.</u>

You acknowledge that we maintain a staff to manage and operate the System and staff members can change as employees come and go. You represent that you did not sign this Agreement in reliance on any particular owner, director, officer, or employee remaining with us in that capacity. We may change our ownership or form and/or assign this Agreement and any other agreement to a third party without restriction. After our assignment of this Agreement to a third party who expressly assumes this Agreement's obligations, we no longer will have any performance or other obligations under this Agreement. That assignment will constitute a release and novation with respect to this Agreement, and the new owner-assignee will be liable to you as if it had been an original party to this Agreement. Specifically, and without limiting the foregoing, you agree that we may sell our assets (including this Agreement), the Licensed Marks, or the System to a third party; offer our ownership interests privately or publicly; merge, acquire other business entities, or be acquired by another business entity; and/or undertake a refinancing, recapitalization, leveraged buyout, or other economic or financial restructuring.

13.2 <u>Transfer by You.</u>

The rights and duties this Agreement creates are personal to you (and your owners). We have granted you the franchise in reliance upon our perceptions of your (and your owners') collective character, skill, aptitude, attitude, business ability, and financial capacity. Accordingly, neither this Agreement (or any interest in this Agreement), the Communities or substantially all of their assets, an ownership interest in you (regardless of its size), a controlling ownership interest in an entity with an ownership interest in you, nor actual management control of the Communities' operation may be transferred without our prior written approval, which we will not unreasonably withhold or delay if the transfer conditions contained in this Section 13 are satisfied. A transfer of the Communities' ownership, possession, or management control, or substantially all of their assets, may be made only with a transfer of this Agreement. A transfer of the Communities' underlying real estate (which includes the Communities' buildings/physical structures), if you or your affiliate owns that real estate, may be made only with, and must be made only in connection with, a transfer of the Communities and this Agreement in compliance with this Agreement's terms. A transfer of the Communities' underlying real estate (which includes the Communities' buildings/physical structures), if an unaffiliated third party owns that real estate, may be made only if the proposed transferee first satisfies us that your possession and continuous operation of the Communities after the transfer in full compliance with this Agreement, and all of our rights under this Agreement, will not be affected in any way. Any transfer without our required approval is a breach of this Agreement and has no effect, meaning that you (and your owners) will continue to be obligated to us for all of your (and their) obligations under this Agreement. For the avoidance of all doubt, you may not transfer less than all of the Communities governed by this Agreement, and any proposed transfer of the Communities governed by this Agreement must be made to the same transferee.

In this Agreement, the term "transfer" includes a voluntary, involuntary, direct, or indirect assignment, sale, gift, or other disposition of any interest in: (a) this Agreement; (b) you; (c) the Communities or substantially all of their assets or their management control; (d) the Communities' underlying real estate (which includes the Communities' buildings/physical structures); or (e) your owners (if they are legal entities). An assignment, sale, gift, or other disposition includes the following events: (i) transfer of ownership of capital stock, a partnership or membership interest, or another form of ownership interest; (ii) merger or consolidation or issuance of additional securities or other forms of ownership interest; (iii) any sale of a security convertible to an ownership interest; (iv) transfer of an interest in you, this Agreement, the Communities or

substantially all of their assets, the Communities' underlying real estate (which includes the Communities' buildings/physical structures), or your owners in a divorce, insolvency, or entity dissolution proceeding or otherwise by operation of law; (v) if one of your owners, or an owner of one of your owners, dies, a transfer of an interest in you, this Agreement, the Communities or substantially all of their assets, the Communities' underlying real estate (which includes the Communities' buildings/physical structures), or your owner by will, declaration of or transfer in trust, or under the laws of intestate succession; or (vi) pledge of this Agreement (to someone other than us) or of an ownership interest in you or your owners as security, foreclosure upon the Communities or their underlying real estate (which includes the Communities' buildings/physical structures), or your owners as security, foreclosure upon the Structures), or your transfer, surrender, or loss of the Communities' possession, control, or management.

You may grant a security interest (including a purchase money security interest) in the Communities' assets (not including this Agreement) to a lender financing your acquisition, development and/or operation of the Communities without obtaining our prior written approval. You may not pledge this Agreement (to someone other than us), an ownership interest in you (regardless of your size), or a controlling ownership interest in an entity with an ownership interest in you unless, as a condition of that pledge, the pledgee agrees not to exercise its secured party rights without first satisfying the conditions we reasonably impose at that time to ensure each Community's continued operation in compliance with this Agreement and all System standards and no violation of any material provision of this Agreement.

If your affiliate owns a Community's underlying real estate (which includes the Communities' buildings/physical structures), you commit that such affiliate will sign, acknowledge, and deliver to us at the Effective Date or at such other times we specify, for any legitimate business purpose we specify, the Joinder Signature Page attached to this Agreement to confirm the affiliate's agreement to comply with both this Section 13 and Section 15.3 with respect to the Community's underlying real estate.

13.3 <u>Representations as to Ownership.</u>

You represent that, as of the Effective Date, your equity and voting control is owned as shown in Exhibit C.

13.4 <u>Conditions to Our Consent to Transfer.</u>

13.4.1 You understand and acknowledge the vital importance of your performance to the market position and overall image of the BRIGHTSTAR CARE HOMES[®] Community System. You also recognize the many subjective factors comprising the process by which we select a suitable franchisee. We will not unreasonably withhold our consent to a Transfer of any interest in this Franchise or any equity or voting interest in you, but such consent will remain a subjective determination that is subject to your and the transferee's compliance with and satisfaction of numerous conditions, including, but not limited to, the following:

13.4.1.1 (a) we determine that the transferee and its direct and indirect owners have sufficient business experience, aptitude, and financial resources to operate the Communities and otherwise are qualified under our then-current standards for the approval of new franchisees or owners of franchisees, and (b) the transferee and/or its owners are not a private equity firm or a Search Fund.

13.4.1.2 You have paid all Royalties/Continuing Fees, General Marketing Fees, Branded Email Fees (if applicable), Monthly Technology & Email Service Fees (if applicable), and other amounts owed to us, our affiliates, System Suppliers, and other third-party vendors; have submitted all required reports and statements; and have not violated any material provision of this Agreement or any other agreement with us during both the sixty (60) days before you requested our consent to the transfer and the period between your request and the transfer's proposed effective date.

13.4.1.3 Neither the transferee nor its owners or affiliates have an ownership interest (direct or indirect) in or perform services for a Competitive Business, wherever located or operating.

13.4.1.4 The transferee's Control Person must satisfactorily complete required training within the timeframe we specify.

13.4.1.5 The transferee has the right to occupy the Communities for the expected franchise term.

13.4.1.6 The transferee obtains and maintains all licenses and licensure necessary to operate the Communities to the fullest extent of the BRIGHTSTAR CARE HOMES[®] Community business model.

13.4.1.7 The transferee must (if the transfer is of this Agreement), or you must (if the transfer is of a controlling ownership interest in you), sign our then-current forms of franchise agreement and related documents ("related documents" include the Personal Guarantee and Agreement to be Bound), any and all of the provisions of which may differ materially from any and all of those contained in this Agreement, provided, however, the initial term of the new franchise agreement signed will be equal to this Agreement's unexpired Initial Term or, at our option, a full initial twenty (20) year term, and the Royalty/Continuing Fee, General Marketing Fees, and Monthly Technology & Email Service Fee (if applicable) payable under the new franchise agreement during its initial term will be the same as in this Agreement if that initial term is equal to this Agreement's unexpired Initial Term.

13.4.1.8 You or the transferee pays us a transfer fee equal to Fifteen Thousand Dollars (\$15,000) (plus any broker fees we incur in helping you find a transferee). No transfer fee is due if, upon a spouse's death, that spouse's ownership in you is transferred to the surviving spouse.

13.4.1.9 You (and, if applicable, your transferring owners) sign a general release, in a form satisfactory to us, of any and all claims against us and our owners, affiliates, officers, directors, employees, and agents.

13.4.1.10 We reasonably believe that the transferee's capital structure and the proposed purchase price and payment terms are not likely to affect adversely the transferee's operation of the Communities.

13.4.1.11 If you or your owners finance any part of the purchase price, you and/or your owners agree that (a) the transferee's obligations under promissory notes and other agreements cannot contain a security interest in this Agreement, the then-current form of franchise agreement signed by the transferee, or any franchise rights we grant to the transferee, and (b) the Promissory note or agreements must be subordinate to the transferee's obligation to pay Royalties/Continuing Fees, General Marketing Fees, Monthly Technology & Email Service Fees (if applicable), and Branded Email Fees (if applicable).

13.4.1.12 (a) you have corrected existing deficiencies in the Communities of which we have notified you on a punch-list or in other communications, and/or (b) the transferee agrees to upgrade, remodel, and refurbish the Communities according to our then-current requirements and specifications for BRIGHTSTAR CARE HOMES[®] Communities within the time period we specify following the transfer's effective date (we will advise the transferee, before the transfer's effective date, of the specific actions it must take and the time period within which it must do so).

13.4.1.13 You and your transferring owners (and your owners' spouses) will not, for twenty-four (24) months beginning on the transfer's effective date, engage in any activities proscribed in Sections 12.4.2 and 12.4.3.

13.4.1.14 You and your transferring owners will not directly or indirectly at any time or in any manner identify themselves in any business as a current or former BRIGHTSTAR CARE HOMES[®] Community or as one of our franchisees; use any Licensed Mark, any colorable imitation of a Licensed Mark, or other indicia of a BRIGHTSTAR CARE HOMES[®] Community for any purpose; or utilize for any purpose any trade name, trade or service mark, or other commercial symbol suggesting or indicating a connection or association with us.

You acknowledge that we have legitimate reasons to evaluate the qualifications of potential transferees and to analyze and critique the terms of their purchase contracts with you. Therefore, our contact with potential transferees to protect our business interests will not constitute improper or unlawful conduct. You expressly authorize us to investigate any potential transferee's qualifications, to analyze and critique the proposed purchase terms, to communicate candidly and truthfully with the transferee regarding your operation of the Communities, and to withhold consent to transactions as provided above. You waive any claim that the action we take in good faith to protect our business interests in connection with a proposed transfer constitutes breach of contract or tortious interference with contractual or business relationships. Similarly, we may

review all information regarding the Communities that you give the proposed transferee, correct any information we believe is inaccurate, and give the proposed transferee copies of any reports you have given us, or we have made, regarding the Communities.

Our consent to a transfer of this Agreement and the Communities, or any ownership interest in you or your owners, is not a representation of the fairness of any contract terms between you and the transferee, a guarantee of the Communities' or transferee's prospects of success, or a waiver of any claims we have against you (or your owners) or of our right to demand the transferee's full compliance with this Agreement.

13.5 <u>Transfer and Operation in the Event of Death or Disability.</u>

13.5.1 Upon the death of one of your Owners, the owner's executor, administrator, conservator, guardian, or other personal representatives must transfer the Owner's ownership interest in you to a third party (which may be the Owner's heirs, beneficiaries, or devisees). That transfer must be completed within a reasonable time, not to exceed nine (9) months from the date of death or disability, and is subject to all of the terms and conditions in this Section 13. Failure to transfer the Owner's ownership interest in you within this time period is a breach of this Agreement.

13.5.2 Upon the Owner's death or disability, a new Control Person (if the deceased Owner was the Control Person) must be appointed and satisfactorily complete our required training program within thirty (30) days. If the Communities are not being managed properly (in our judgment) any time after the Control Person's death or disability, we may, but need not, assume the Communities' management (or appoint a third party to assume its management). All funds from the Communities' operation while they are under our (or the third party's) management will be kept in a separate account to which all expenses will be charged. We may charge you (in addition to the Royalty/Continuing Fee, General Marketing Fees, Monthly Technology & Email Service Fees (if applicable), and other amounts due under this Agreement) the manager's then-current daily salary plus our direct expenses while the Communities are under our (or the third party's) management. We (or a third party) have a duty to use only reasonable efforts and, if we (or the third party) are not grossly negligent and do not engage in willful misconduct, will not be liable to you or your owners for any debts, losses, lost or reduced profits, or obligations the Communities incur, or to any of your creditors for any supplies, products, other assets, or services the Communities purchase, while we (or a third party) manage them. The term "disability" means a mental or physical disability, impairment, or condition that is reasonably expected to prevent or actually does prevent the Control Person from personally devoting his or her full time and best efforts to the Communities' operation.

13.6 <u>Our Right of First Refusal.</u>

If you, one of your owners, or the owner of a controlling ownership interest in an entity with an ownership interest in you at any time determines to sell or transfer (i) an interest in this Agreement and the Communities (including their buildings/physical structures if you or your affiliate owns them), or (ii) an ownership interest in you or a controlling ownership interest in the entity with an ownership interest in you (except to or among your current owners or between a current owner and his or her immediate family member, neither of which is subject to this Section) in a transaction that otherwise would be allowed under the provisions of Section 13 above, you (or your owners) must obtain from a responsible and fully disclosed buyer and then send us a true and complete copy of a bona fide, signed written offer (which may include a letter of intent) relating to an interest in you (or in the entity with an ownership interest in you) or in this Agreement and the Communities (including their buildings/physical structures if you or your affiliate owns them).

The offer must include details of the proposed sale or transfer's payment terms and the financing sources and terms of the proposed purchase price and provide for an earnest money deposit of at least five percent (5%) of the proposed purchase price. To be a valid, bona fide offer, the proposed purchase price must be a fixed-dollar amount, without any contingent payments of purchase price (such as earn-out payments), and the proposed transaction must relate exclusively to the rights granted by this Agreement and the Communities (including their buildings/physical structures if you or your affiliate owns them), an ownership interest in you, or a controlling ownership interest in the entity with an ownership interest in you. It may not relate to any other interests or assets. The right of first refusal process will not be triggered by a proposed transfer that would not be allowed under this Section 13 and therefore could not proceed. We may require you (or your owners) to send us copies of any materials or information you send to the proposed buyer or transferee regarding the possible transaction.

We may, by written notice delivered to you within thirty (30) days after we receive both an exact copy of the offer and all other information it requests, elect to purchase the interest offered for the price and on the terms and conditions contained in the offer, provided that: (a) we may substitute cash for any form of consideration proposed in the offer; (b) our credit will be deemed equal to the credit of any proposed buyer; (c) the closing of the purchase will not (unless we agree otherwise) be earlier than sixty (60) days after we notify you of our election to purchase or, if later, the closing date proposed in the offer; (d) you and your owners must sign the general release described in Section 13.1.4.9 above; and (e) we must receive, and you and your owners agree to make, all customary representations, warranties, and indemnities given by the seller of the assets of a business or of ownership interests in a legal entity, as applicable, including representations and warranties regarding ownership and condition of, and title to, assets and (if applicable) ownership interests; your and your owners' authorization to sell, as applicable, any ownership interests or assets without violating any law, contract, or requirement of notice or consent; liens and encumbrances on ownership interests and assets; validity of contracts and liabilities, contingent or otherwise, relating to the assets or ownership interests being purchased; and indemnities for all actions, events, and conditions that existed or occurred in connection with the Communities before the closing of our purchase. If the offer is to purchase all of your ownership interests, we may elect instead to purchase all of the Communities' assets (including their buildings/physical structures if you or your affiliate owns them), and not any of your ownership interests, on the condition that the amount we pay you for such assets equals the full value of the transaction as proposed in the offer (i.e., the value of all assets to be sold and of all liabilities to be assumed).

If we exercise our right-of-first-refusal and close the transaction, you and your transferring owners agree that, for two (2) years beginning on the closing date, you and they will be bound by the non-competition covenants contained in Sections 12.4.2 and 12.4.3 above.

If we do not exercise our right-of-first-refusal, you or your owners may complete the sale to the proposed buyer on the original offer's terms, but only if we approve the transfer as provided in this Section 13. This means that, even if we do not exercise its right of first refusal (whether or not it is properly triggered as provided above), if the proposed transfer otherwise would not be allowed under this Section 13, you (or your affiliate or owners) may not move forward with the transfer at all. If you or your owners do not complete the sale to the proposed buyer within sixty (60) days after we notify you that we do not intend to exercise our right-of-first-refusal, or if there is a material change in the sale's terms (which you agrees to tell us promptly), we will have an additional right-of-first-refusal during the thirty (30) days following either expiration of the sixty (60) day period or our receipt of notice of the material change(s) in the sale's terms, either on the terms originally offered or the modified terms, at our option.

We have the unrestricted right to assign this right-of-first-refusal to a third party (including an affiliate), which then will have the rights described in this Section 13.6. (All references in this Section 13.6 to "we", "us", and "our" include our assignee if we have exercised our right to assign this right-of-first-refusal to a third party.)

14. **DEFAULT AND TERMINATION**

14.1 <u>Termination With Notice and Without Opportunity to Cure.</u>

We may terminate this Agreement, effective upon delivery of written notice of termination to you and without providing you an opportunity to cure, for any of the following breaches or defaults:

14.1.1 Felonies. You or any of your owners is convicted of or pleads guilty or no contest to a felony.

14.1.2 Criminal Misconduct. You or any of your owners engages in any criminal or other misconduct (even if not felonious) that negatively impacts the Licensed Marks or a Community's operation.

14.1.3 Fraud or Misrepresentation during Operation. You or any of your owners commits any fraud or makes any misrepresentation in operating a Community.

14.1.4 Misrepresentation in Application. You or any of your owners makes any misrepresentation or omission in connection with your franchise application, including, but not limited to, any financial misrepresentation.

14.1.5 Unauthorized Surrender. You surrender control of a Community's operation without our prior written consent.

14.1.6 Unauthorized Transfers. You or any of your owners makes or attempts to make an unauthorized assignment of this Agreement, a Community, an ownership interest in you, or a controlling ownership interest in an entity with an ownership interest in you or interferes with our right of first refusal under Section 13.6.

14.1.7 Repeated Breaches. You or any of your owners (a) fails on three (3) or more separate occasions within any twelve (12) consecutive calendar-month period to comply with this Agreement, whether or not we notify you of the failures, and, if we do notify you of the failures, whether or not you correct the failures after our delivery of notice to you; or (b) fails on two (2) or more separate occasions within any six (6) consecutive calendar-month period to comply with the same obligation under this Agreement, whether or not we notify you of the failures, and, if we do notify you of the failures, whether or not you correct the failures after our delivery of notice to you.

14.1.8 Misuse of the Licensed Marks or Confidential Information. You or any of your owners materially violates any provision of this Agreement relating to the Licensed Marks or Confidential Information.

14.1.9 Violation of Law. You (a) violate any health, safety, or sanitation law, ordinance, or regulation, or operate a Community in an unsafe manner, and do not begin to cure the violation immediately after delivery of notice (from us or any government agency) or do not correct the violation within the timeframe mandated by us or, if applicable, the law or government agency, or (b) fail to maintain any licenses or permits required to operate a Community, as a result of which you would be legally obligated to cease operations, and you fail to secure those licenses and permits within the timeframe mandated by law, or (c) violate any other federal, state, or local law impacting a Community's operation and do not cure the violation within the timeframe mandated by law or a government agency.

14.1.10 Violation of In-term Restrictive Covenant. You or any of your owners violates the in-term restrictive covenant contained in Section 12.4.1.

14.1.11 Liens. You make an assignment for the benefit of creditors or admit in writing your insolvency or inability to pay your debts generally as they become due; you consent to the appointment of a receiver, trustee, or liquidator of all or the substantial part of your property; a Community (including its building/physical structure if owned by you or an affiliate) is attached, seized, subjected to a writ or distress warrant, or levied upon, unless the attachment, seizure, writ, warrant, or levy is vacated within thirty (30) days; or any order appointing a receiver, trustee, or liquidator of you, a Community (including its building/physical structure if owned by you or an affiliate) is not vacated within thirty (30) days following the order's entry.

14.1.12 Abandonment. You (a) abandon a Community, meaning you have deserted, walked away from, or closed the Community under circumstances leading us to conclude that you have no intent to resume the Community's operation, regardless of the number of days passing since the apparent abandonment, or (b) fail actively and continuously to operate a Community (a failure to operate the Community for two (2) or more consecutive days will be deemed a default under this clause (b), except where closure is due to fire, riot, flood, acts of terrorism, or natural disaster and you notify us within two (2) days after the particular occurrence to obtain our written approval to remain closed for an agreed-upon amount of time as is necessary under the circumstances before you must re-open).

14.1.13 Anti-Terrorist Activities. You or any of your owners fails to comply with the provisions of Section 27.

14.1.14 Personal Use of Community's Property. If any of your owners takes for its, his, or her personal use any assets or property of a Community, including employee taxes, FICA, insurance, or benefits.

14.1.15 Insufficient Funds. If there are insufficient funds in your bank account to cover (i) a check or EFT payment to us three (3) or more times within any twelve (12) month period, or (ii) payroll at any time, whether or not those violations are cured.

14.1.16 Unethical or Immoral Behavior. You or any of your owners engages in any dishonest, unethical, immoral, lewd, deviant, or similar conduct as a result of which you or the owner's association with a Community (or the owner's association with you) could, in our reasonable opinion, have a material adverse effect on the goodwill associated with the Licensed Marks.

14.1.17 Failure to satisfactorily complete Initial Training. If the Control Person (or a replacement) fails to satisfactorily complete the portions of Getting Started Training and Road to Opening Training designed for him or her.

14.1.18 Loss of Possession of Premises. You lose the right to possess the Premises of a Community, regardless of the reason.

14.2 <u>Termination Upon Notice and Five (5) Days' Opportunity to Cure.</u>

We have the right to terminate this Agreement if any of the following defaults remains uncured after expiration of a five (5) day cure period:

14.2.1 Insurance. You fail to maintain the insurance coverage we require or to repay us for insurance we acquire on your behalf, or you otherwise fail to adhere to the requirements of Section 17.

14.2.2 Failure to Personally Supervise Community Operations or Employ Adequate Personnel. The Control Person fails to supervise the Community's day-to-day operation personally, or you fail to employ a sufficient number of qualified, competent personnel.

14.2.3 Quality Control. You fail to maintain the strict quality controls required by this Agreement and/or the Operations Manual.

14.2.4 Endorsement of Checks. You fail to immediately endorse and deliver to us or another system franchisee any payments due to us or another system franchisee from any third party that is erroneously made to you.

14.2.5 Interruption of Service. You fail to maintain the prescribed hours of operation at the Community.

14.2.6 Closure of Bank Account. You terminate the authorization for our electronic transfer of funds from your account, or close the bank account from which we debit funds due from you, without prior notice to us and without first establishing a substitute bank account permitting such withdrawals.

14.3 <u>Termination Upon Notice and Fifteen (15) Days' Opportunity to Cure.</u>

We have the right to terminate this Agreement if any of the following defaults remains uncured after expiration of a fifteen (15) day cure period:

14.3.1 Nonpayment. If you fail to pay when due any sums owed to us, any of our affiliates, any System Suppliers, or any of our other vendors.

14.3.2 Under-Reporting of Net Billings. If any audit reveals that you have understated required Royalty/Continuing Fees or General Marketing Fees by more than two percent (2%).

14.3.3 Unauthorized Products or Services. You offer any unauthorized and unapproved services or products at, from, or through the Community.

14.3.4 Unapproved Purchases. You order or purchase unapproved Marketing materials, non-health insurance coverage, supplies, or Operating Assets or buy such items from an unapproved supplier.

14.4 <u>Termination With Notice and Thirty (30) Days' Opportunity to Cure.</u>

We have the right to terminate this Agreement if any of the following defaults remains uncured after expiration of a thirty (30) day cure period:

14.4.1 Failure to Comply with this Agreement. Except as noted above in this Section 14, any other failure by you to comply with this Agreement is subject to a thirty (30) day cure period.

14.4.2 Failure to Open. You fail to commence operating the Community by the Opening Date.

14.5 <u>Termination With Notice and One-Year Opportunity to Cure.</u>

We have the right to terminate this Agreement under the following additional circumstances. If you or one of you affiliates is a BrightStar Care Agency franchisee and the Net Billings of one or more of your or your affiliate's BrightStar Care Agencies decline on a year-over-year basis by twenty percent (20%) or more during your development and operation of your Communities, you or your affiliate must submit a plan to us (which we may share with the franchisor of BrightStar Care Agencies) within fifteen (15) days following notice from us of such a decline with a description of the proposed cure for that decline, which then must be implemented within the following ninety (90) days. The plan must include, at a minimum, additional infrastructure/roles or other revenue-producing activities proposed to support the operations in order to recover or recoup the year-over-year decline. Your or your affiliate's failure to recover or recoup the decline in the performance of your or its BrightStar Care Agency within twelve (12) months from the date on which we first notify you of the year-over-year decline in Net Billings of at least twenty percent (20%) will be good cause for our termination of this Agreement.

14.6 <u>Termination of Right to Operate a Community. Whenever we retain the right</u> to terminate this Agreement on account of a default by you occurring with respect to a specific Community (and not with respect to all of the Communities that you then are operating under this Agreement), we have the option (but not the obligation), instead of terminating this Agreement, to terminate your right to operate that particular Community, effective upon delivery of written notice of termination to you. In such circumstances, we may exercise the rights, and you must comply with the obligations, appearing in Sections 14.9 and 15 below but may do so with respect to only the affected Community.

14.7 <u>Nonwaiver.</u>

Our delay in exercising or failing to exercise any right or remedy under this Agreement, or our acceptance of any late or partial payment due, will not constitute a waiver of our rights against you.

14.8 <u>Termination by You.</u>

If you and your owners are fully complying with this Agreement and we materially fail to comply with this Agreement and do not correct the failure within thirty (30) days after you deliver to us written notice of the material failure or, if we cannot correct the failure within thirty (30) days, do not give you within thirty (30) days after your notice reasonable evidence of our effort to correct the failure within a reasonable time (which may extend beyond that thirty (30) days), you may terminate this Agreement effective an additional thirty (30) days after you deliver to us written notice of termination. (The time period during which we may cure any alleged material default after your delivery of notice is called the "Cure Period.") However, if we send you written notice during the Cure Period indicating that either (a) we do not agree that we have materially failed to comply with this Agreement or (b) we have fully corrected the failure, then you may not terminate this Agreement. If you disagree with our position and still wish to terminate this Agreement, seeking a

declaration of your right to terminate this Agreement. This Agreement will remain in full force and effect during the legal proceeding (unless we terminate it as provided in this Section 14). If there is a determination that we are in or did not fully correct a material default, we will have an additional thirty (30) days following the determination to correct the failure. If we fail to do so, you may terminate this Agreement immediately upon delivery of written notice. Your termination of this Agreement other than according to this Section 14.8 will be deemed a termination without cause and a breach of this Agreement.

14.9 <u>Step-In Rights.</u>

In addition to and without waiving any other rights or remedies we have under this Agreement (including in Section 13.5) or at law or in equity, you authorize us to operate one or more of your Communities (or designate a third party to do so) if:

(a) We consider such operation necessary and practical to prevent an interruption in a Community's operation;

(b) this Agreement expires or is terminated, and we are deciding whether to exercise our option to purchase the Communities under Section 15.3 below;

(c) this Agreement expires or is terminated, and we want to control the deidentification process under Section 15.1 below; or

(d) in our sole judgment, we deem you incapable of operating the Community because (i) your Control Person is absent, (ii) you have failed to pay when due all taxes and assessments against the Community, (iii) you have failed to pay when due any and all liens or encumbrances of any kind placed upon or against the Community, (iv) you have lost or had suspended your license or permit to operate the Community, or (v) we determine there are operational problems that could imperil the Community or its residents.

We are not obligated to exercise these step-in rights. However, if we choose to do so, we and you agree to cooperate and to review periodically the status of the Community's operation. Your cooperation includes using all lawful means (including management or other agreements) to help us (or our designee) obtain all licensure required to exercise our or its step-in rights. You must reimburse all of our (or our designee's) reasonable costs and overhead incurred in connection with operating the Community, including, without limitation, personnel costs for supervising and staffing the Community and their travel and lodging accommodations, and pay a fee of up to Two Thousand Dollars (\$2,000) per day. If we operate a Community under this Section, you agree to indemnify and hold harmless us (and our designees and employees) from and against any fines, claims, suits, or proceedings arising out of our operation of the Community.

15. <u>POST-TERM OBLIGATIONS</u>

15.1 <u>Obligations upon Termination or Expiration.</u>

15.1.1 Payment. You agree to pay us within fifteen (15) days after this Agreement expires or is terminated, or on any later date we determine the amounts due, the Royalties/Continuing Fees, General Marketing Fees, late fees and interest, and other amounts owed to us (and our affiliates) that then are unpaid, including all termination damages under Section 15.2.

15.1.2 Licensed Marks. When this Agreement expires or is terminated:

15.1.2.1 You may not directly or indirectly at any time or in any manner identify yourself in any business as a current or former BRIGHTSTAR CARE HOMES[®] Community or as one of our current or former franchisees; use any Licensed Mark, any colorable imitation of a Licensed Mark, or other indicia, trade dress, or distinguishing features of a BRIGHTSTAR CARE HOMES[®] Community for any purpose; or use for any purpose any trade name, trade or service mark, or other commercial symbol indicating or suggesting a connection or association with us.

15.1.2.2 You agree, within fifteen (15) days, to cancel all fictitious or assumed name or equivalent registrations covering your use of any Licensed Mark.

15.1.2.3 If we do not have or do not exercise an option to purchase the Communities' assets (including their buildings/physical structures if you or your affiliate owns them) under Section 15.3 below, you agree at your own cost and without any payment from us for such items to deliver to us, to make available to us for pick-up, or to destroy (at our option), in any case within twenty (20) days, all signs, Marketing materials, forms, and other materials we request containing any Licensed Mark or otherwise identifying or relating to a BRIGHTSTAR CARE HOMES[®] Community. If you fail to do so voluntarily when we require, we and our representatives may enter the Community at our convenience and remove these items without liability to you, any landlord, or another third party for trespass or any other claim. You must reimburse our costs of doing so.

15.1.2.4 If we do not have or do not exercise an option to purchase the Communities' assets (including their buildings/physical structures if you or your affiliate owns them) under Section 15.3 below, you agree within the timeframe we specify (which may be within thirty (30) days after the effective date of expiration or termination), and at your own expense, to take the closing, de-identification, and de-branding steps, and make the alterations to the Communities' trade dress and interior design, layout, and configuration, we specify in the Operations Manual (or otherwise communicate to you) to distinguish the Communities clearly from their former appearance and from other BRIGHTSTAR CARE HOMES[®] Communities in order to prevent public confusion and protect the Licensed Marks and the System. The de-identification process is intended to alert the public immediately that the Communities are not affiliated with the System. If you fail to do so voluntarily when we require, we and our representatives may enter the Communities at our and their convenience and take this action without liability to you, any landlord, or another third party for trespass or any other claim. We need not compensate you or any other party for any alterations. We also may exercise this right under Section 14.9.(c) above. You must reimburse our costs of de-identifying the Communities. If you fail to comply strictly with all de-identification obligations in this Section 15.1.2.4, you agree to pay us a royalty fee of Five Thousand Dollars (\$5,000) per day until de-identification is completed to our satisfaction.

15.1.2.5 You agree, within fifteen (15) days, to notify the telephone company and all telephone directory publishers (both web-based and print) of the termination or expiration of your right to use any telephone, facsimile, or other numbers and telephone directory listings associated with any Licensed Mark; to authorize, and not to interfere with, the transfer of these numbers and directory listings to us (or other party identified by us) or at our direction (including pursuant to a Conditional Assignment of Telephone Number(s), in the form of Exhibit G, previously signed by you); and/or to instruct the telephone company to forward all calls made to your numbers to numbers we specify. If you fail to do so, we may take whatever action and sign whatever documents we deem appropriate on your behalf to effect these events.

15.1.3 Confidential Information. When this Agreement expires or is terminated, you must immediately cease using any of our Confidential Information (including computer software or similar technology and digital passwords and identifications we have licensed to you or that otherwise are proprietary to us or the System) in any business or otherwise and return to us all copies of the Operations Manual and any other confidential materials to which we gave you access. You may not sell, trade, or otherwise profit in any way from any Confidential Information after the expiration or termination of this Agreement. You will have access to the Technology/Computer Requirements only for fifteen (15) days after the Initial Term for purposes of data migration.

15.1.4 List of Employees, Clients, and Vendors. When this Agreement expires or is terminated, you must send us within fifteen (15) days a complete list of your employees, clients, referral sources, and vendors and their respective addresses and any outstanding obligations you have to any third parties. We have the right to contact those individuals and entities to inform them that the Communities no longer are part of the BRIGHTSTAR CARE HOMES[®] Community System or that there will be a transition in ownership and operation. Our exercise of these rights will not constitute an interference with your contractual or business relationships. You acknowledge that the individuals and entities doing business with the Communities when they were part of the BRIGHTSTAR CARE HOMES[®] Community System constitute our customers.

15.1.5 Vacatur of Premises. When this Agreement expires or is terminated, you must immediately vacate each Premises if we exercise our rights under Section 15.3 below (or any of our step-in rights).

15.2 <u>Termination Damages.</u>

You acknowledge that we will suffer substantial damages as a result of this Agreement's termination, or the termination of your right to operate a specific Community, before the Initial Term expires. Some of those damages include lost Royalty/Continuing Fees, lost General Marketing Fees, lost market penetration and goodwill, loss of System representation in the Communities' (or Community's) market area, customer confusion, lost opportunity costs, and expenses we will incur in developing or finding another franchisee to develop BRIGHTSTAR CARE HOMES[®] Communities in the Communities' market area (collectively, "Brand Damages"). We and you acknowledge that Brand Damages are difficult to estimate accurately over a period of years, and proof of Brand Damages would be burdensome and costly, although those damages are real and meaningful to us. Therefore, upon termination of this Agreement for any reason before the Initial Term expires (except for an authorized termination by you), or upon the termination of your right to operate a specific Community, you agree to pay us, within fifteen (15) days after the date of termination, termination damages in a lump sum as calculated below.

15.2.1 For any Communities that have operated as BRIGHTSTAR CARE HOMES[®] Communities for at least two (2) years before the effective date of termination, the termination damages payable with respect to each affected Community will equal the product of (a) either thirty-six (36) or the number of months then remaining in the Initial Term as of the effective date of termination (had this Agreement or your right to operate the particular Community not been terminated), whichever is shorter, multiplied by (b) the sum of the average monthly Royalty/Continuing Fees, plus the average monthly General Marketing Fees, plus the average Monthly Technology & Email Service Fee (if applicable) that were due and payable from you to us with respect to each affected Community during the twelve (12) month period before the month of termination.

15.2.2 For each Community that has not yet opened for business with our authorization (unless you choose not to find sites for the second and third Communities that this Agreement authorizes you to develop and operate) or has not operated as a BRIGHTSTAR CARE HOMES[®] Community for at least two (2) years as of the effective date of termination, the termination damages payable with respect to each affected Community will be the fixed sum of One Hundred Thousand Dollars (\$100,000).

15.2.3 If this Agreement is terminated by us because you violate the noncompetition restrictions in Section 12.4.1 above, or if you terminate this Agreement without cause in breach of this Agreement, the termination damages payable with respect to each affected Community will equal one hundred fifty percent (150%) of the amount of termination damages otherwise payable under Sections 15.2.1 or 15.2.2 above (depending on the effective date of termination).

15.2.4 You agree that termination damages calculated under this Section 15.2 represent the best estimate of our Brand Damages arising from any termination of this Agreement, or your right to operate a particular Community, before the Initial Term expires. Your payment of termination damages to us will not be considered a penalty but, rather, a reasonable estimate of

fair and just compensation to us for the Brand Damages we will incur because this Agreement, or your operation of the particular Community, did not continue for the Initial Term's full length.

15.2.5 You acknowledge that payment of termination damages is full compensation to us only for the Brand Damages resulting from early termination of this Agreement, or your right to operate a particular Community, and is in addition to, and not in place of, your obligation to pay other amounts due to us under this Agreement as of the date of termination and to comply strictly with your de-identification and other post-termination obligations.

If any valid law or regulation governing this Agreement limits your obligation to pay, and our right to receive, the termination damages for which you are obligated under this Section 15.2, you will be liable to us for any and all Brand Damages we incur, now or in the future, as a result of your breach of this Agreement.

15.3 <u>Our Right to Purchase Community's Assets.</u>

Under the circumstances listed below, we have the right to acquire the Communities' assets (including their buildings/physical structures if you or your affiliate owns them), or to receive an assignment, lease, or sublease of the occupancy rights to the Premises, upon this Agreement's termination or expiration. This Section 15.3 survives this Agreement's termination or expiration.

15.3.1 Exercise of Option. Upon (a) our termination of this Agreement according to its terms and conditions, (b) your termination of this Agreement without cause in breach of this Agreement, or (c) expiration of this Agreement (if we offer, but you elect not to acquire, a successor franchise, or if we do not offer you a successor franchise because you failed to satisfy the conditions for a successor franchise set forth in Section 2.2 above), we have the option, exercisable by giving you written notice (a "Purchase Notice") before or within thirty (30) days after the effective date of termination or expiration,

(i) to purchase the Communities' assets (including their buildings/physical structures if you or your affiliate owns them); and/or

(ii) if we choose not to purchase the buildings/physical structures of the Communities (if you or your affiliate owns them), to purchase the remaining assets of the Communities and exercise the rights under Section 15.3.2 below.

We have the unrestricted right to assign this purchase option to another party.

15.3.2 Right to Occupy Site. If we choose not to purchase the buildings/physical structures of the Communities (if you or your affiliate owns them), or if an unaffiliated entity owns the buildings/physical structures of the Communities, then you, your affiliate, or the unaffiliated real estate owner must (at our option and as applicable) within sixty (60) days after we deliver a Purchase Notice: (a) assign the leasehold interest in the Communities' buildings/physical structures to us; (b) sublease the leasehold interest in the Communities'

buildings/physical structures to us for the remainder of the lease term on the same terms (including renewal options) as the lease; or (c) lease the Communities' buildings/physical structures to us for an initial ten (10) year term, with two five (5) year renewal terms (at our option), on commercially reasonable terms, including, without limitation, a fair market rental rate to be agreed upon within the first fifteen (15) days after we deliver a Purchase Notice. If we and you, your affiliate, or the unaffiliated real estate owner, as applicable, cannot agree upon the fair market rental rate within this fifteen (15) day period, the fair market rental rate will be determined pursuant to the appraisal process set forth in Section 15.3.4 below.

15.3.3 Purchase Price. The purchase price for the Communities' assets that we choose to buy and, if applicable, the Communities' buildings/physical structures will be their fair market value, although fair market value will not include any value for: (a) the franchise or any rights granted by this Agreement; (b) goodwill attributable to the Licensed Marks, the BrightStar brand image, and our other intellectual property; or (c) participation in the network of BRIGHTSTAR CARE HOMES[®] Communities. In all cases, we may exclude from the assets purchased any Operating Assets or other items not reasonably necessary (in function or quality) to the Communities' operation or that we have not approved as meeting System standards; the purchase price will reflect these exclusions. We will identify the exclusions in the Purchase Notice to the extent we then have the information necessary to assess the Operating Assets or other items. We and you, or your affiliate, as applicable, must work together in good faith to agree upon the assets' fair market value within fifteen (15) days after we deliver a Purchase Notice. If we and you cannot agree on fair market value within this fifteen (15) day period, fair market value will be determined by the appraisal process described in Section 15.3.4 below.

Appraisal. For purposes of this Section 15.3, whenever a value must be 15.3.4 established and there is no agreement on that value within fifteen (15) days (or as otherwise expressly provided in this Agreement), the value will be established by the following procedure. Value will be determined by one (1) independent accredited appraiser upon whom we and you agree who, in conducting the appraisal, will be bound by the criteria specified above. We and you agree to select the appraiser within fifteen (15) days after expiration of the fifteen (15) day period in Section 15.3.3. If we and you cannot agree on a mutually-acceptable appraiser within the fifteen (15) days, we will send you a list of three (3) independent appraisers, and you must within seven (7) days select one of them to be the designated appraiser to determine the purchase price. Otherwise, we have the right to select the appraiser. We and you will share equally the appraiser's fees and expenses. Within thirty (30) days after delivery of notice invoking the appraisal mechanism, we and you each must send the appraiser our and your respective calculations of the purchase price, with such detail and supporting documents as the appraiser requests and according to the criteria specified above. Within fifteen (15) days after receiving both calculations, the appraiser must decide whether our proposed purchase price or your proposed purchase price most accurately reflects the assets' fair market value. The appraiser has no authority to compromise between the two (2) proposed purchase prices; it is authorized only to choose one or the other. The appraiser's choice will be the purchase price and is final.

15.3.5 Closing. We will pay the purchase price at the closing, which will take place not later than sixty (60) days after the purchase price is determined, unless licensure or other

state requirements dictate a longer time period. However, we may decide after the purchase price is determined not to complete the purchase and will have no liability to you for choosing not to do so. We may set off against the purchase price, and reduce the purchase price by, any and all amounts you owe us (or our affiliates). At the closing, you agree to deliver instruments transferring to us: (a) good and merchantable title to the assets purchased, free and clear of all liens and encumbrances (other than liens and security interests acceptable to us), with all sales and transfer taxes paid by you; (b) all of the Communities' licenses and permits that may be assigned; and (c) possessory rights to the Communities' Premises.

If you cannot deliver clear title to all purchased assets, or if there are other unresolved issues, the sale will be closed through an escrow. You and your owners further agree to sign general releases, in a form satisfactory to us, of any and all claims against us and our affiliates and their respective owners, officers, directors, employees, agents, representatives, successors, and assigns. If we exercise our rights under this Section 15.3, then for two (2) years beginning on the closing date, you and your owners and affiliates will be bound by the non-competition covenants contained in Sections 12.4.2 and 12.4.3.

You may not under any circumstances sell the Communities' buildings/physical structures or any of their other assets until we have exercised or elected not to exercise our right to purchase those assets, as provided in this Section.

16. **DISPUTE RESOLUTION AND CORRESPONDING PROCEDURES**

16.1 <u>Meeting Between Senior Executives.</u>

We and you will attempt to resolve any controversy, claim, or dispute arising out of or relating to this Agreement or our and your relationship and involving you and us or any of our and your respective affiliates, owners, officers, directors, employees, agents or representatives (a "Dispute") promptly by a meeting between senior executives of the respective organizations who have authority to settle the Dispute ("Senior Executives"). Either party may give the other written notice ("Dispute Notice") of any Dispute that has not been resolved in the ordinary course of business. Within fifteen (15) days after delivery of the Dispute Notice, the receiving party must give to the other a written response ("Response"). The Dispute Notice and the Response must include: (a) a statement describing the position of the party giving the Dispute Notice and the Response and a summary of arguments supporting such position; and (b) the name of the Senior Executive and any other persons who will accompany the Senior Executive at the meeting at which the parties will attempt to resolve the Dispute. Within thirty (30) days after delivery of the Dispute Notice, the Senior Executives will meet at a mutually acceptable time and place, and then as often as they reasonably consider necessary, to attempt to resolve the Dispute. All reasonable requests for information made by one party to the other will be honored. If the Dispute has not been resolved within sixty (60) days after delivery of the Dispute Notice, or if the Senior Executives do not meet within thirty (30) days after delivery of the Dispute Notice, either party may initiate mediation of the Dispute. All meetings, discussions, and other communications under this Section will be treated as compromise and settlement discussions and communications.

16.2 <u>Mediation.</u>

All claims or disputes between you and us or any of your and our respective affiliates, owners, officers, directors, employees, agents, or representatives arising under, out of, in connection with, or in relation to this Agreement, our and your relationship, the Communities, or any of our and your respective rights and obligations arising out of this Agreement must be submitted first to mediation prior to initiating an arbitration proceeding (except as noted in Section 16.4 below). Such mediation will take place in Bannockburn, Illinois (or at our then-current headquarters) under the auspices of the American Arbitration Association ("AAA") in accordance with the AAA's Commercial Mediation Rules then in effect. You may not commence any formal legal action against us or our affiliates with respect to any such claim or dispute unless mediation proceedings have been terminated either: (a) as the result of a written declaration of the mediator(s) that further mediation efforts are not worthwhile; or (b) as a result of a written declaration by us. Each party will bear its own costs of mediation and share equally the filing fee imposed by AAA and the mediator's fees. We reserve the right to specifically enforce our right to mediation. Before mediation and before commencing any legal action against us or our affiliates with respect to any such claim or dispute, you must submit a notice to us specifying in detail the precise nature and grounds of such claim or dispute.

16.3 <u>Arbitration.</u>

If not resolved by mediation and except as qualified below, all controversies, disputes, or claims between us (and our affiliates and our and their respective owners, officers, directors, agents, and employees, as applicable) and you (and your affiliates and your and their respective owners, officers, and directors, as applicable) arising out of or related to:

a) this Agreement or any other agreement between you (or your owner) and us (or our affiliate) relating to the Communities or any provision of any such agreements;

b) our relationship with you;

c) the validity of this Agreement or any other agreement between you (or your owner) and us (or our affiliate) relating to the Communities, or any provision of any such agreements, and the validity and scope of the arbitration obligation under this Section; or

d) any standard, specification, or operating procedure relating to the Communities;

must be submitted for arbitration to the American Arbitration Association. Except as otherwise provided in this Agreement, such arbitration proceedings will be heard by one (1) arbitrator in accordance with the then-existing Commercial Arbitration Rules of the American Arbitration Association. All proceedings, including the hearing, will be conducted in Chicago, Illinois or, at our option, at a suitable location that is within ten (10) miles of where we have our principal business address when the arbitration demand is filed. The arbitrator will have no authority to select a different hearing locale other than as described in the prior sentence. All matters within

the scope of the Federal Arbitration Act (9 U.S.C. Sections 1 et seq.) will be governed by it and not by any state arbitration law.

The arbitrator has the right to award any relief he or she deems proper in the circumstances, including money damages (with interest on unpaid amounts from the date due), specific performance, injunctive relief, and attorneys' fees and costs (in accordance with Section 20.2 below), provided that: (1) the arbitrator has no authority to declare any Licensed Mark generic or otherwise invalid; and (2) as provided in Section 16.9, you waive to the fullest extent that applicable law permits any right to or claim for any punitive, exemplary, treble, and other forms of damages. The arbitrator's award and decision will be conclusive and bind all parties covered by this Section, and judgment upon the award may be entered in a court specified or permitted in Section 16.7 below.

We and you will be bound by any limitation under this Agreement or applicable law, whichever expires first, on the timeframe in which claims must be brought. We and you further agree that, in connection with any arbitration proceeding, each must submit or file any claim constituting a compulsory counterclaim (as defined by the then-current Rule 13 of the Federal Rules of Civil Procedure) within the same proceeding as the claim to which it relates. Any such claim not submitted or filed in the proceeding will be barred. The arbitrator may not consider any settlement discussions or offers either you or we made. We reserve the right, but have no obligation, to advance your share of the costs of any arbitration proceeding in order for the arbitration proceeding to take place and by doing so do not waive or relinquish our right to seek recovery of those costs in accordance with Section 20.2 below.

We and you agree that arbitration will be conducted on an individual basis and not in a class, consolidated, or representative action, that only we (and our affiliates and our and their respective owners, officers, directors, agents, and employees, as applicable) and you (and your affiliates and your and their respective owners, officers, and directors, as applicable) may be the parties to any arbitration proceeding described in this Section, and that no such arbitration proceeding may be consolidated or joined with another arbitration proceeding involving us and/or any other person. Despite the foregoing or anything to the contrary in this Section or Section 24.1, if any court or arbitrator determines that all or any part of the preceding sentence is unenforceable with respect to a dispute that otherwise would be subject to arbitration under this Section, then you and we agree that this arbitration clause will not apply to that dispute, and such dispute will be resolved in a judicial proceeding in accordance with Section 16 (excluding this Section 16.3).

This Section's provisions will continue in full force and effect after and notwithstanding expiration or termination of this Agreement.

To the extent permitted by applicable law, no issue of fact or law may be given preclusive or collateral estoppel effect in any arbitration, except to the extent such issue may have been determined in another proceeding between the parties. 16.4 <u>Exceptions to Arbitration.</u>

Notwithstanding Section 16.3, the parties agree that the following claims will not be subject to mediation or arbitration:

(a) each has the right to seek temporary restraining orders and temporary or preliminary injunctive relief from a court of competent jurisdiction, provided, however, each must contemporaneously submit its dispute for arbitration on the merits as provided in Section 16.3; and

(b) any action in ejectment or for possession of any interest in real or personal property.

16.5 <u>Third Party Beneficiaries.</u>

Our affiliates and their respective owners, officers, directors, agents, and employees are express third-party beneficiaries of this Agreement, each having authority to specifically enforce the right to mediate/arbitrate your claims asserted against such person(s).

16.6 <u>No Right to Offset.</u>

You may not withhold any payment to us or our affiliates on the grounds of our or their alleged nonperformance of our or their obligations, or set off any amount you owe us or our affiliates against any amount we or our affiliates might owe you, under this Agreement or any related agreements. However, we may set off any amounts you or your owners owe us or our affiliates against any amounts that we or our affiliates owe you or your owners, whether in connection with this Agreement or otherwise.

16.7 <u>Venue.</u>

We and you expressly agree, subject to our and your arbitration obligations under Section 16.3, to the exclusive jurisdiction and venue of any court of general jurisdiction in the city and state where our headquarters is located at the time the action is filed ("Home State") and the jurisdiction and venue of the United States District Court presiding over our Home State. You acknowledge that this Agreement has been entered into in the State of Illinois, and you will receive valuable and continuing services emanating from the Home State where our headquarters is located. In recognition of such services and their origin, you hereby irrevocably consent to the personal jurisdiction of the state and federal courts of our Home State as set forth above.

16.8 <u>Limitation on Actions.</u>

Except for the parties' indemnification obligations under Section 20.1 below, and except for claims arising from your non-payment or underpayment of amounts you owe us and our affiliates, we and you agree that no cause of action arising out of or under this Agreement or as a result of the parties' relationship may be maintained by either party against the other unless an arbitration or judicial proceeding, as mandated or permitted by this Agreement, is commenced

within two (2) years from the date on which the violation, act, or conduct giving rise to the claim occurs, regardless of when the party asserting the claim knew or should have known of the facts giving rise to the claim. Any action not brought within this period will be barred as a claim, counterclaim, defense, or set-off.

16.9 Waiver of Punitive, Exemplary, and Certain Other Damages.

You hereby waive to the fullest extent the law permits any right to or claim for any punitive, exemplary, incidental, indirect, special, or consequential damages (including, without limitation, lost profits) against us or our affiliates arising out of any cause whatsoever (whether such cause is based in contract, negligence, strict liability, other tort, or otherwise) and agree that, if there is a dispute, your recovery is limited to actual damages and, if appropriate, equitable relief.

16.10 Jury Trial Waiver.

Subject to the arbitration obligations in Section 16.3, we and you (and your owners) irrevocably waive trial by jury in any action, proceeding, or counterclaim, whether at law or in equity, brought by either us or you (or your owners). We and you (and your owners) acknowledge that we and you (and they) make this waiver knowingly, voluntarily, without duress, and only after considering this waiver's ramifications.

17. <u>INSURANCE</u>

17.1 Lines of Insurance.

You must, at your own expense and no later than upon commencement of each Community's build-out process, procure and then maintain in full force and effect throughout the Initial Term the minimum types and amounts of insurance enumerated in the Operations Manual. All insurance policies must name, as additional insureds, each of us, our affiliates, their respective officers, directors, and employees, and any other parties we designate. Required insurance includes the following:

17.1.1 Professional Liability on an occurrence basis with limits not less than \$1,000,000 per occurrence / \$3,000,000 aggregate per policy year. If you purchase this coverage on a claims-made policy form, you must purchase a minimum 3-year extended reporting endorsement (aka "tail") upon sale or closure of your business.

17.1.2 General Liability on an occurrence basis with limits not less than \$1,000,000 per occurrence / \$3,000,000 aggregate per policy year. The following minimum sublimits must be met: \$1,000,000 Personal & Advertising Injury, \$3,000,000 Products/Completed Ops Aggregate, \$100,000 Damage to Rented Premises, and \$5,000 Medical Expense. If you purchase this coverage on a claims-made policy form, you must purchase a minimum 3-year extended reporting endorsement (aka "tail") upon sale or closure of your business. 17.1.3 Abuse/Molestation Liability with limits not less than \$1,000,000 per occurrence and aggregate per policy year.

17.1.4 Hired and Non-Owned Automobile coverage not less than \$1,000,000 combined single limit each accident.

17.1.5 Special Form property insurance in an amount appropriate to cover the full replacement value of each Community and its contents, including flood and any other insurance that is appropriate for the Community's location if the primary policy has exclusions for such events. Business Income and Extra Expense must be included at a minimum 1.5 times annual revenue assuming full occupancy and covering an 18-month period following the loss.

17.1.6 Workers' Compensation and Employer's Liability insurance with minimum limits not less than \$500,000 per accident; \$500,000 policy limit by disease; and \$500,000 per employee for bodily injury by disease or higher limit as required by law in your state.

17.1.7 A \$50,000 Employee Dishonesty and \$50,000 Theft of Client Property.

17.1.8 Employment Practices Liability (EPL) with \$1,000,000 minimum limit covering indemnification and defense costs for employee allegations of harassment, discrimination, and wrongful termination practices. Coverage must include a "3rd Party Endorsement" to respond to client allegations of similar wrongful acts, to include at minimum \$100,000 for Wage & Hour defense costs.

17.1.9 Umbrella Liability with a \$2,000,000 minimum limit to extend over professional liability, general liability, abuse/molestation liability and employer's liability. The umbrella may contain a sub-limit of \$1,000,000 for abuse/molestation.

17.1.10 Cyber Liability with \$500,000 minimum limit to include both administrative costs of the privacy event and 3rd party losses.

17.1.11 Any other insurance not listed above but required by applicable law, rule, regulation, ordinance, or licensing requirements and any updates as specified in the Operations Manual.

17.1.12 Policy and Coverage level deductibles may not exceed \$25,000 for any coverages required unless we grant a written waiver.

We reserve the right to change the types and amounts of insurance required under this Agreement. You agree, at your own expense, to conform your insurance coverage to our requirements. You must obtain and maintain insurance coverage from the agency and carriers we specify. All insurance companies must carry an A.M. Best's rating of "A-/Excellent" or better or be approved by us in writing prior to placement of coverage.

17.2 <u>Insurance Certificates.</u>

You must make timely delivery to us of certificates of all required insurance. Each must contain a statement by the insurer that the policy will not be cancelled or materially altered without at least ten (10) days' prior written notice to us. In order to monitor claims activity on a national level and effectively assess program exposures, you must collect Loss History Statements ("Loss Runs") from carriers and send them to us annually.

17.3 <u>No Relief from Indemnity Requirement.</u>

Obtaining and maintaining the insurance required by this Agreement do not relieve you of any liability to us under any indemnity requirement in this Agreement.

17.4 <u>Administrative Fee.</u>

If you fail to comply with your minimum insurance requirements, we have the right to obtain such insurance and keep it in force and effect, in which case you must pay us, on demand, all premium costs plus an eighteen percent (18%) administrative fee. We have the right to increase or otherwise modify the minimum insurance requirements upon thirty (30) days' prior written notice to you. You must comply with any such modification within the time specified in the notice.

18. <u>YOUR OWNERSHIP AND ORGANIZATION</u>

18.1 <u>Disclosure of Ownership Interests.</u>

You and each owner represent, warrant, and agree that Exhibit C is current, complete, and accurate. You agree that updates to Exhibit C will be furnished promptly to us so that Exhibit C is at all times current, complete, and accurate. Each owner must be an individual acting in his or her individual capacity unless we waive this requirement. The spouses of each of your owners must sign a Spousal / Life Partner Consent in the form of Exhibit F to this Agreement.

18.2 <u>Organizational Documents.</u>

You and each owner represent, warrant, and agree that: (a) you are duly organized and validly existing under the laws of the state of your organization; (b) you have the authority to execute and deliver this Agreement and all related agreements and to perform your obligations under all such agreements; (c) the articles of incorporation, partnership agreement, or other organizational documents recite that the issuance, transfer, or pledge of any direct or indirect legal or beneficial ownership interest in you is restricted by this Agreement's terms; and (d) all certificates representing direct or indirect legal or beneficial ownership interests now or hereafter issued bear a legend in conformity with applicable law reciting or referring to such restrictions.

18.3 <u>Personal Guarantee Covenants and Assumption of Obligations.</u>

Each owner (regardless of the owner's ownership interest in you) reflected on Exhibit C must sign and irrevocably be bound by the Personal Guarantee and Agreement to be Bound

included as Exhibit D. If you are a corporation initially owned in majority by the corporation's 401(k) plan, you must sign Exhibit D.

19. TAXES, PERMITS AND INDEBTEDNESS

19.1 <u>Taxes.</u>

You must promptly pay when due any and all federal, state, and local taxes, including, without limitation, unemployment and sales taxes, levied or assessed with respect to any services or products furnished, used, or licensed under this Agreement and all accounts or other indebtedness of every kind you incur in operating the Communities.

19.2 <u>Permits.</u>

You must comply with all federal, state, and local laws, rules, and regulations and timely obtain any and all permits, certificates, and licenses for the full and proper conduct of the Communities.

19.3 <u>Full and Sole Responsibility for Debts and Obligations.</u>

You hereby expressly covenant and agree to accept full and sole responsibility for any and all debts and obligations incurred in operating the Communities.

20. INDEMNIFICATION AND INDEPENDENT CONTRACTOR

20.1 <u>Indemnification.</u>

You agree to indemnify and hold harmless us, our affiliates, and our and their respective owners, directors, officers, employees, agents, successors, and assignees (the "Indemnified Parties") against, and to reimburse any one or more of the Indemnified Parties for, all Losses (defined below) incurred as a result of a claim threatened or asserted or an inquiry made formally or informally, or a legal action, investigation, or other proceeding threatened or brought, by a third party and directly or indirectly arising out of or relating to: (i) a Community's operation; (ii) the business you conduct under this Agreement; (iii) your breach of this Agreement or any other agreement with us or our affiliates; (iv) your non-compliance or alleged non-compliance with any federal, state, or other law, ordinance, rule, or regulation, including those concerning a Community's design, construction, or operation, or employment or personnel practices (expressly including your compliance with wage and hour and other labor and employment laws), and including any allegation that we or another Indemnified Party is a joint employer or otherwise responsible for your acts or omissions relating to employees; or (v) claims alleging either intentional or negligent conduct, acts, or omissions by you (or your contractors or any of their employees, agents, or representatives) or by us or our affiliates (or our and their contractors or any of their employees, agents, or representatives). You also agree to defend the Indemnified Parties (unless an Indemnified Party chooses to defend at your expense as provided in the following paragraph) against any and all such claims, inquiries, actions, investigations, and proceedings.

However, you have no obligation to indemnify or hold harmless an Indemnified Party for any Losses to the extent they are determined in a final, unappealable ruling issued by a court or arbitrator with competent jurisdiction to have been caused solely and directly by the Indemnified Party's negligence, willful misconduct, or willful wrongful omissions (so long as the claim to which those Losses relate is not asserted on the basis of theories of vicarious liability—including agency, apparent agency, or joint employment—or our failure to compel you to comply with this Agreement).

For purposes of this indemnification and hold harmless obligation, "Losses" includes all obligations, liabilities, damages (actual, consequential, or otherwise), and reasonable defense costs that any Indemnified Party incurs. Defense costs include, without limitation, accountants', arbitrators', attorneys', and expert witness fees, costs of investigation and proof of facts, court costs, travel and living expenses, and other expenses of litigation, arbitration, or alternative dispute resolution, regardless of whether litigation, arbitration, or alternative dispute resolution is commenced. Each Indemnified Party, with its own counsel and at your expense, may defend and otherwise respond to and address any claim threatened or asserted or any inquiry made, or any action, investigation, or proceeding threatened or brought (instead of having you defend it with your counsel, as provided in the preceding paragraph), and, in cooperation with you, agree to settlements or take any other remedial, corrective, or other actions, for all of which defense and response costs and other Losses you are solely responsible (except as provided in the last sentence of the preceding paragraph).

Your obligations in this Section 20.1 will continue in full force and effect subsequent to and notwithstanding this Agreement's expiration or termination. An Indemnified Party need not seek recovery from any insurer or other third party, or otherwise mitigate its Losses, in order to maintain and recover fully a claim against you under this Section 20.1. You agree that a failure to pursue a recovery or mitigate a Loss will not reduce or alter the amounts that an Indemnified Party may recover from you under this Section 20.1.

20.2 <u>Cost of Enforcement.</u>

If we incur costs and expenses to enforce our rights or your obligations under this Agreement because you have failed to pay when due amounts owed to us, to submit when due any reports, information, or supporting records, or otherwise to comply with this Agreement, you agree to reimburse us for all costs and expenses we incur, including, without limitation, reasonable accounting, attorneys', arbitrators', and related fees. Your obligation to reimburse us arises whether or not we begin a formal legal proceeding against you to enforce this Agreement. If we do begin such a formal legal proceeding against you, the reimbursement obligation applies to all costs and expenses we incur preparing for, commencing, and prosecuting the legal proceeding and until the proceeding has come to a complete end (including appeals and settlements).

20.3 <u>No Fiduciary Relationship; Independent Contractor Status.</u>

In all dealings with third parties, including, without limitation, franchisees, employees, suppliers, and clients, you must disclose in an appropriate manner acceptable to us that you are an

independent entity licensed by us. Nothing in this Agreement is intended to create a fiduciary relationship between us and you or to constitute you as a subsidiary, joint venturer, partner, agent, or employee of us for any purpose whatsoever. Further, we will not be construed to be jointly liable for any of your acts or omissions under any circumstances. We are not the employer or joint employer of you or the Community's employees. You are an independent contractor and in no way authorized to make any warranty or representation on our behalf. You are not authorized to create any obligation or enter into any contract binding on us.

You are solely responsible for managing and operating the Community and all employment and personnel matters and decisions involving the Community, including but not limited to the hiring, firing, supervision, direction, scheduling, and compensation of the Community's employees. Notwithstanding your use of any Third-Party Materials or System Technology (if applicable), you are solely responsible for understanding and following wage and hour laws as well as all other state, local and federal laws applicable to the operations of the Community. We will not exercise direct or indirect control over the working conditions of Community personnel, except to the extent such indirect control is related to our legitimate interest in protecting the quality of our services or brand. We do not share or codetermine the employment terms and conditions of the Community's employees and do not affect matters relating to the employment relationship between you and the Community's employees, such as employee selection, promotion, termination, hours worked, rates of pay, other benefits, work assigned, discipline, adjustment of grievances and complaints, and working conditions. To that end, you must notify Community personnel that you are their employer and that we, as the franchisor of BRIGHTSTAR CARE HOMES[®] Communities, are not their employer or joint employer and do not engage in any employer-type activities for which only franchisees are responsible, such as employee selection, promotion, termination, hours worked, rates of pay, other benefits, work assigned, discipline, adjustment of grievances and complaints, and working conditions. You also must obtain an acknowledgment (in the form we specify or approve) from all Community employees that we are not their employer.

21. WRITTEN APPROVALS, WAIVERS, AND AMENDMENT

21.1 <u>Approval Process.</u>

Whenever this Agreement requires our prior approval, you must make a timely written request. Unless a different time period is specified in this Agreement, we will respond with our approval or disapproval within thirty (30) calendar days. In addition, our approval will not be unreasonably withheld.

21.2 <u>No Waiver.</u>

We and you may in writing unilaterally waive or reduce any contractual obligation of or restriction upon the other, effective upon delivery of written notice to the other or another effective date stated in the waiver notice. Any waiver granted is without prejudice to any other rights we or you have, is subject to continuing review, and may be revoked at any time and for any reason effective upon delivery of ten (10) days' prior written notice.

We and you will not waive or impair any right, power, or option this Agreement reserves (including, without limitation, our right to demand strict compliance with every term, condition, and covenant or to declare any breach to be a default and to terminate this Agreement before the Initial Term expires) because of any custom or practice varying from this Agreement's terms; our or your failure, refusal, or neglect to exercise any right under this Agreement or to insist upon the other's compliance with this Agreement, including, without limitation, any System standard; our waiver of or failure to exercise any right, power, or option, whether of the same, similar, or different nature, with other BrightStar franchisees; the existence of franchise agreements or license agreement; or our acceptance of any payments due from you after any breach of this Agreement. No special or restrictive legend or endorsement on any check or similar item given to us will be a waiver, compromise, settlement, or accord and satisfaction. We may remove any legend or endorsement, which then will have no effect.

Neither we nor you will be liable for loss or damage or be in breach of this Agreement if our or your failure to perform obligations results from: (a) compliance with the orders, requests, regulations, or recommendations of any federal, state, or municipal government; (b) acts of God; (c) fires, strikes, embargoes, war, acts of terrorism or similar events, or riot; or (d) any other similar event or cause. Any delay resulting from these causes will extend performance accordingly or excuse performance, in whole or in part, as may be reasonable, except that these causes will not excuse payment of amounts owed at the time of the occurrence or payment of Royalties/Continuing Fees, General Marketing Fees, and other amounts due afterward.

21.3 <u>Amendments.</u>

This Agreement is binding upon us and you and our and your respective executors, administrators, heirs, beneficiaries, permitted assigns, and successors in interest. Subject to our right to modify the Operations Manual and System standards, this Agreement may not be modified except by a written agreement signed by the parties' duly-authorized officers specifying their intent to modify this Agreement.

22. <u>NOTICES</u>

All written notices, reports, and payments permitted or required to be delivered by this Agreement or the Operations Manual will be deemed delivered: (a) at the time delivered by hand; (b) one (1) business day after transmission by facsimile or electronic mail if the sender has confirmation of successful transmission; (c) one (1) business day after being placed in the hands of a nationally recognized commercial courier service for next business day delivery; or (d) three (3) business days after placement in the United States Mail by Registered or Certified Mail, Return Receipt Requested, postage prepaid. Any notice to us must be sent to the address specified on the first page of this Agreement, although we may change this address for notice by giving you ten (10) days' prior notice by any of the means specified above. Any notice we send you may be sent to the Control Person at the postal address specified on Exhibit A. You may change the person and/or address for notice only by giving us ten (10) days' prior notice by any of the means specified above. Any required report we do not actually receive during regular business hours on or before

the date due (or postmarked by postal authorities at least two (2) days before then for reports unrelated to payments due under this Agreement) will be deemed delinquent. Payments must be actually received by us on or before their specific due dates.

23. <u>GOVERNING LAW</u>

ALL MATTERS RELATING TO ARBITRATION WILL BE GOVERNED BY THE UNITED STATES FEDERAL ARBITRATION ACT (9 U.S.C. §§ 1 ET SEQ.). EXCEPT TO THE EXTENT GOVERNED BY THE FEDERAL ARBITRATION ACT, THE UNITED STATES TRADEMARK ACT OF 1946 (LANHAM ACT, 15 U.S.C. SECTIONS 1051 ET SEQ.) OR OTHER FEDERAL LAW, THIS AGREEMENT, THE FRANCHISE, AND ALL CLAIMS ARISING FROM THE RELATIONSHIP BETWEEN US AND YOU WILL BE GOVERNED BY THE LAWS OF THE STATE OF ILLINOIS, WITHOUT REGARD TO ITS CONFLICT OF LAWS RULES, EXCEPT THAT ANY ILLINOIS LAW REGULATING THE SALE OF FRANCHISES OR GOVERNING THE RELATIONSHIP OF A FRANCHISOR AND ITS FRANCHISEE WILL NOT APPLY UNLESS ITS JURISDICTIONAL REQUIREMENTS ARE MET INDEPENDENTLY WITHOUT REFERENCE TO THIS SECTION.

24. <u>SEVERABILITY AND CONSTRUCTION</u>

24.1 <u>Severability.</u>

Except as expressly provided to the contrary in this Agreement, each section, paragraph, term, and provision of this Agreement is severable, and if, for any reason, any part is held to be invalid or contrary to or in conflict with any applicable present or future law or regulation in a final, unappealable ruling issued by any court, agency, or tribunal with competent jurisdiction, that ruling will not impair the operation of, or otherwise affect, any other portions of this Agreement, which will continue to have full force and effect and bind the parties. If any applicable and binding law or rule of any jurisdiction requires more notice than this Agreement requires of this Agreement's termination or of our refusal to grant a successor franchise, or some other action this Agreement does not require, or if, under any applicable and binding law or rule of any jurisdiction, any provision of this Agreement or any System standard is invalid, unenforceable, or unlawful, the notice and/or other action required by the law or rule will be substituted for the comparable provisions of this Agreement, and we may modify the invalid or unenforceable provision or System standard to the extent required to be valid and enforceable or delete the unlawful provision entirely. You agree to be bound by any promise or covenant imposing the maximum duty the law permits that is subsumed within any provision of this Agreement, as though it were separately articulated in and made a part of this Agreement.

24.2 <u>Execution in Counterparts.</u>

This Agreement may be executed in any number of counterparts, each of which when so executed and delivered will be deemed an original, but such counterparts together will constitute one and the same instrument.

24.3 <u>Headings and Captions.</u>

The headings and captions contained in this Agreement are for purposes of convenience and reference only and are not to be construed as part of this Agreement. All terms and words used will be construed to include the number and gender as the context of this Agreement may require. Each section of this Agreement will be construed independently of any other section or provision of this Agreement. The words "include" and "including" are meant to be illustrative and not exhaustive and are deemed to be read in all cases as "including, without limitation" and/or "including but not limited to."

24.4 Interpretation of Rights and Obligations.

The following provisions apply to and govern the interpretation of this Agreement, the parties' rights under this Agreement, and the relationship between the parties.

24.4.1 Our Rights. Whenever this Agreement provides that we have a certain right, that right is absolute, and the parties intend that our exercise of that right will not be subject to any limitation or review. We have the right to operate, administer, develop, and change the System in any manner not specifically precluded by this Agreement, although this right does not modify any express limitations set forth in this Agreement.

24.4.2 Our Reasonable Business Judgment. Whenever we reserve discretion in a particular area or where we agree to exercise our rights reasonably or in good faith, we will satisfy our obligations whenever we exercise Reasonable Business Judgment in making a decision or exercising a right. Our decisions or actions will be deemed to be the result of "Reasonable Business Judgment," even if other reasonable or even arguably preferable alternatives are available, if our decisions or actions are intended, in whole or significant part, to promote or benefit the System generally even if the decision or action also promotes our financial or other individual interests. Examples of items that will promote or benefit the System include, without limitation, enhancing the value of the Licensed Marks, improving client service and satisfaction, improving product quality, improving uniformity, enhancing or encouraging modernization, and improving the System's competitive position.

25. <u>ENTIRE AGREEMENT</u>

The Recitals and Exhibits are a part of this Agreement which, together with System standards contained in the Operations Manual (which we may periodically modify, as provided in this Agreement) or otherwise communicated to you, constitutes our and your entire agreement, and there are no other oral or written understandings or agreements between the parties, and no oral or written representations by us, relating to the subject matter of this Agreement, the franchise relationship, or a Community (any understandings or agreements reached by the parties, or any representations made by us, before this Agreement are superseded by this Agreement). You may not rely on any alleged oral or written understandings, agreements, or representations not contained in this Agreement. However, nothing in this Agreement or any related agreement is intended to disclaim our representations in our Franchise Disclosure Document.

Any policies we periodically adopt and implement to guide us in our decision-making are subject to change, are not a part of this Agreement, and are not binding on us. Except as expressly provided in this Agreement, nothing in this Agreement is intended or deemed to confer any rights or remedies upon any person or legal entity not a party to this Agreement.

26. <u>SPOUSAL/LIFE PARTNER CONSENT</u>

Each spouse of each of your owners personally and unconditionally guarantees, without notice, demand, or presentment, the payment of all of your monetary obligations under this Agreement as if each spouse were an original party to this Agreement in his or her individual capacity. All such spouses further agree to be bound by the restrictions upon your activities upon transfer, termination, or expiration of this Agreement as if each were an original party to this Agreement in his or her individual capacity. All such spouses must execute a spousal / life partner consent in the form attached as Exhibit F to this Agreement. In the event of divorce and remarriage, or subsequent marriage, your owners covenant and agree to provide us with a properly executed spousal / life partner consent in the form we prescribe.

27. <u>ANTI-TERRORIST ACTIVITIES</u>

You certify that you and your owners, directors, officers, and employees are not listed in Executive Order Annex is available at Annex 13224. (The the to http://www.treasury.gov/offices/enforcement/ofac/sanctions/terrorism. html.) You agree not to hire or have any dealings with a person listed in the Annex. You certify that you have no knowledge or information that, if generally known, would result in you and your owners, directors, officers, and employees being listed in the Annex to Executive Order 13224. You agree to comply with, and/or assist us to the fullest extent possible to comply with, the Anti-Terrorism Laws (as defined below). In connection with such compliance, you certify, represent, and warrant that your property and interests are not subject to being "blocked" under any Anti-Terrorism Law and that you and your owners, directors, officers, and employees are not otherwise in violation of any Anti-Terrorism Law. You are solely responsible for ascertaining what actions must be taken to comply with all such Anti-Terrorism Laws and specifically acknowledge and agree that your indemnification responsibilities as provided in Section 20.1 of this Agreement pertain to your obligations under this Section 27. Any misrepresentation by you under this Section or any violation of the Anti-Terrorism Laws by you or your owners, directors, officers, or employees will constitute grounds for immediate termination of this Agreement and any other agreement you have entered into with us or one of our affiliates in accordance with Section 14 of this Agreement. "Anti-Terrorism Laws" means Executive Order 13224 issued by the President of the United States, the Terrorism Sanctions Regulations (Title 31, Part 595 of the U.S. Code of Federal Regulations), the Foreign Terrorist Organizations Sanctions Regulations (Title 31, Part 597 of the U.S. Code of Federal Regulations), the Cuban Assets Control Regulations (Title 31, Part 515 of the U.S. Code of Federal Regulations), the USA PATRIOT Act, and all other present and future federal, state and local laws, ordinances, regulations, policies, lists, and any other requirements of any governmental authority (including, without limitation, the United States Department of Treasury Office of Foreign Assets Control) addressing or in any way relating to terrorist acts and acts of war.

28. <u>ACKNOWLEDGMENTS</u>

You acknowledge the truthfulness and accuracy of the representations signed by you and attached as Exhibit E.

29. NO WAIVER OR DISCLAIMER OF RELIANCE IN CERTAIN STATES

The following provision applies only to franchisees and franchises that are subject to the state franchise registration/disclosure laws in California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, or Wisconsin:

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on our behalf. This provision supersedes any other term of any document executed in connection with the franchise.

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement on the dates next to their signatures, to be effective on the date of our signature.

FRANCHISOR (US):

BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC

By:			
Name:			
Title:			
Date:			

FRANCHISEE (YOU):

By:		
Name:		
Title:		
Date:		

JOINDER SIGNATURE PAGE

TO THE FRANCHISE AGREEMENT BETWEEN

BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC

AND _____

Pursuant to Section 13.2 of the Franchise Agreement, this Joinder Signature Page will be executed by any person related to or any entity affiliated with Franchisee that owns the underlying real estate of a Community (as defined in the Agreement), including its building/physical structure.

_____, by signing below, joins in the Franchise Agreement solely to confirm his, her, or its agreement to be bound by all terms in Sections 13 and 15.3 of the Franchise Agreement.

[Name of Person or Entity]

By:_____ [Signature]

Title:_____
Date: _____

EXHIBIT A

PREMISES

1. **PREMISES**. The only Premises from which the Communities may operate pursuant to Sections 1.1 and 1.2 of the Agreement are:

Community #1:

Community #2: _____

Community #3: _____

FRANCHISOR (US):

BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC

By:		
Name:		
Title:		
Date:		

FRANCHISEE (YOU):

By:	
Name:	
Title:	
Date:	

EXHIBIT B-1

AUTHORIZATION TO INITIATE DEBIT ENTRIES FOR FEES AND PAYMENTS DUE FROM FRANCHISEE

, the undersigned franchisee, hereby authorizes BrightStar Senior Living Franchising, LLC, an Illinois limited liability company, to initiate debit entries to the checking account indicated and at the depository identified below (the "Depository") to debit to such account the amounts of the entries reflecting Royalty/Continuing fees, General Marketing Fees, Monthly Technology & Email Service Fees, and other amounts that become due from the undersigned to BrightStar Senior Living Franchising, LLC and/or its affiliates:

Depository Name:	
Depository Branch:	
Depository Address:	
	CityStateZip
Routing Number:	
Account Name:	
Account Number:	

For good and valuable consideration, the receipt and sufficiency of which the undersigned hereby acknowledges, the undersigned agrees that:

This authorization is to remain in full force and effect until BrightStar Senior Living Franchising, LLC has received advance written notification from the undersigned of its termination in such manner as to afford BrightStar Senior Living Franchising, LLC and Depository reasonable time and opportunity to act upon such notification.

IN WITNESS WHEREOF, this authorization has been executed on ______.

FRANCHISEE (YOU):

By:			
Name	:		
Title:			

Phone No.:_____

Exhibit B-1-1

EXHIBIT B-2

AUTHORIZATION TO INITIATE DEBIT ENTRIES FOR FEES AND PAYMENTS DUE FROM FRANCHISEE

, the undersigned franchisee, hereby authorizes BrightStar Technology Group, LLC, a Delaware limited liability company, to initiate debit entries to the checking account indicated and at the depository identified below (the "Depository") to debit to such account the amounts of the entries reflecting technology fees and other amounts that become due from the undersigned to BrightStar Technology Group, LLC and/or its affiliates:

Depository Name:	
Depository Branch:	
Depository Address:	
	CityStateZip
Routing Number:	
Account Name:	
Account Number:	

For good and valuable consideration, the receipt and sufficiency of which the undersigned hereby acknowledges, the undersigned agrees that:

This authorization is to remain in full force and effect until BrightStar Technology Group, LLC has received advance written notification from the undersigned in such manner as to afford BrightStar Technology Group, LLC and Depository reasonable time and opportunity to act upon such notification.

IN WITNESS WHEREOF, this authorization has been executed on ______.

FRANCHISEE (YOU):

By:		
Name		
Title:		

Phone	No.:				
-------	------	--	--	--	--

EXHIBIT C

FRANCHISEE OWNERSHIP AND MANAGEMENT INFORMATION

1. Form of Entity of Franchisee.

a. <u>Corporation</u>. You were incorporated on _____, under the laws of the State of ______. You have not conducted business under any name other than your corporate name. The following is a list of all of your directors and officers as of ______.

Name of Each Director/Officer	Position(s) Held
-------------------------------	-------------------------

b. <u>Limited Liability Company Franchisee</u>. You are a limited liability company formed on _____, under the laws of the State of _____. You have not conducted business under any name other than your limited liability company name. The following is a list of all of your members as of _____.

Name of Each Member (Identify the Managing Member)

2. <u>Owners</u>.

You and each of your owners represent and warrant that the following is a complete and accurate list of all your owners, including the full name and mailing address of each owner, and fully describes the nature and extent of each owner's interest in you. You and each owner, as to his or her ownership interest, represent and warrant that each owner is the sole and exclusive legal and beneficial owner of his or her ownership interest in you, free and clear of all liens, restrictions, and encumbrances of any kind or nature, other than those required or permitted by this Agreement.

Owner's Name and Address	Description of Interest, Including Percentage of Ownership Interest

Exhibit C-1

Submitted by you on		·
		_, a
limited liability company	_ corporation/	
nimited habinity company		
By:		_
Print Name:		_
Title:		-
Owners:		
(Signature)		(Signature)
Print Name:		_ Print Name:
(Signature)		(Signature)
Print Name:		

Accepted by BrightStar Senior Living Franchising, LLC, and made a part of the BRIGHTSTAR CARE HOMES[®] Community Franchise Agreement as of _____, 20__.

BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC (US)

By:			
Name:			
Title:			

EXHIBIT D

PERSONAL GUARANTEE AND AGREEMENT TO BE BOUND PERSONALLY BY THE TERMS AND CONDITIONS OF THE FRANCHISE AGREEMENT

In consideration of, and as an inducement to, the execution of that certain Franchise Agreement (the "Agreement") on this date by BrightStar Senior Living Franchising, LLC, an Illinois limited liability company ("we", "us", or "our"), each of the undersigned personally and unconditionally (a) guarantees to us and our successors and assigns, for the term of the Agreement (including, without limitation, any extensions of its term) and afterward as provided in the Agreement, that ("you" "vour") or will punctually pay and perform each and every undertaking, agreement, and covenant set forth in the Agreement (including, without limitation, any amendments or modifications of the Agreement) and (b) agrees to be personally bound by, and personally liable for the breach of, each and every provision in the Agreement (including, without limitation, any amendments or modifications of the Agreement), including (i) monetary obligations, (ii) obligations to take or refrain from taking specific actions and to engage or refrain from engaging in specific activities, including, but not limited to, the non-competition, confidentiality, and transfer requirements, and (iii) the enforcement and other provisions in the Agreement, including the arbitration provision.

Each of the undersigned consents and agrees that: (1) his or her direct and immediate liability under this Guaranty will be joint and several, both with you and among other guarantors; (2) he or she will render any payment or performance required under the Agreement upon demand if you fail or refuse punctually to do so; (3) this liability will not be contingent or conditioned upon our pursuit of any remedies against you or another person; (4) this liability will not be diminished, relieved, or otherwise affected by any extension of time, credit, or other indulgence we may from time to time grant to you or to another person, including, without limitation, the acceptance of any partial payment or performance or the compromise or release of any claims (including, without limitation, any release of other guarantors), none of which will in any way modify or amend this Guaranty, which will continue and be irrevocable during the term of the Agreement (including, without limitation, any extensions of its term) and afterward for so long as any performance is or might be owed under the Agreement by you or any of your owners and for so long as we have any cause of action against you or any of your owners; (5) this Guaranty will continue in full force and effect for (and as to) any extension or modification of the Agreement and despite the transfer of any interest in the Agreement or you, and each of the undersigned waives notice of any and all renewals, extensions, modifications, amendments, or transfers; and (6) any of your indebtedness to the undersigned, for whatever reason, whether currently existing or hereafter arising, will at all times be inferior and subordinate to any indebtedness owed by you to us or our affiliates.

Each of the undersigned waives: (i) all rights to payments and claims for reimbursement or subrogation which the undersigned may have against you arising as a result of the undersigned's execution of and performance under this Guaranty; (ii) acceptance and notice of acceptance by us of his or her undertakings under this Guaranty, notice of demand for payment of any indebtedness or non-performance of any obligations hereby guaranteed, protest and notice of default to any party with respect to the indebtedness or nonperformance of any obligations hereby guaranteed, and any other notices to which he or she may be entitled; and (iii) all rights to assert or plead any statute of

Exhibit D-1

limitations or other limitations period as to or relating to this Guaranty. The undersigned expressly acknowledges that the obligations under this Guaranty survive expiration or termination of the Agreement.

If we seek to enforce this Guaranty in an arbitration, judicial, or other proceeding and prevail in that proceeding, we are entitled to recover our reasonable costs and expenses (including, but not limited to, attorneys' fees, arbitrators' fees, expert witness fees, costs of investigation and proof of facts, court costs, other arbitration or litigation expenses, and travel and living expenses) incurred in connection with such proceeding. If we are required to engage legal counsel in connection with the undersigned's failure to comply with this Guaranty, the undersigned must reimburse us for any of the above-listed costs and expenses we incur, even if we do not commence a judicial or arbitration proceeding.

Subject to the arbitration obligations (as set forth in the Agreement) and the provisions below, each of the undersigned agrees that all actions arising under this Guaranty or the Agreement, or otherwise as a result of the relationship between us and the undersigned, must be brought exclusively in the state or federal court of general jurisdiction in the state, and in (or closest to) the city, where we have our principal business address when the action is commenced, and each of the undersigned irrevocably submits to the jurisdiction of those courts and waives any objection he or she might have to either the jurisdiction of or venue in those courts. Nonetheless, each of the undersigned agrees that we may enforce this Guaranty and any arbitration orders and awards in the courts of the state or states in which he or she is domiciled. WE AND THE UNDERSIGNED IRREVOCABLY WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING, OR COUNTERCLAIM, WHETHER AT LAW OR IN EQUITY, BROUGHT BY ANY OF THEM. EACH ACKNOWLEDGES THAT THEY MAKE THIS WAIVER KNOWINGLY, VOLUNTARILY, WITHOUT DURESS, AND ONLY AFTER CONSIDERATION OF THIS WAIVER'S RAMIFICATIONS.

PERSONAL GUARANTORS:

(signature)		(signature)		
Print Name, Individually		Print Name, Individually		ıally	
	Address			Address	
City	State	Zip Code	City	State	Zip Code
	Telephone	2		Telephone	2

EXHIBIT E

FRANCHISEE REPRESENTATIONS RELATING TO BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC BRIGHTSTAR CARE HOMES[®] COMMUNITY FRANCHISE AGREEMENT

(These Franchisee Representations will not be used if the franchise is to be operated in, or you are a resident of, California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, or Wisconsin)

BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC <u>FRANCHISEE REPRESENTATIONS</u>

THIS DOCUMENT SHALL NOT BE SIGNED BY YOU, AND WILL NOT APPLY, IF THE OFFER OR SALE OF THE FRANCHISE IS SUBJECT TO THE STATE FRANCHISE REGISTRATION/DISCLOSURE LAWS IN THE STATES OF CALIFORNIA, HAWAII, ILLINOIS, INDIANA, MARYLAND, MICHIGAN, MINNESOTA, NEW YORK, NORTH DAKOTA, RHODE ISLAND, SOUTH DAKOTA, VIRGINIA, WASHINGTON, OR WISCONSIN.

Important Instructions: BrightStar Senior Living Franchising, LLC ("we," "us," or "our") and you are preparing to enter into a Franchise Agreement and other agreements for your construction, development, and operation of one or more BRIGHTSTAR CARE HOMES[®] Communities. This document's purpose is to determine whether any statements or promises were made to you that we have not authorized, that do not appear in or are inconsistent with our franchise documents, and that may be untrue, inaccurate, or misleading. We also want to be sure that you understand certain terms of the agreements you will sign and their ramifications. If you sign this document, you are confirming that what it says is true. In addition, if you sign it, we will take actions in reliance on the truth of what it says. Please review each of the following statements carefully. *Initial the spaces after the statements to confirm your understanding and the accuracy of the statements*.

Franchisee: ______(the "Franchisee")

BRIGHTSTAR CARE HOMES[®] Community Location (if available at signing):

(the "Community")

Each of the undersigned represents that all of the following statements are true:

1. Each of the undersigned has independently investigated us; the BrightStar System (as that term is used in our Franchise Agreement); the risks, burdens, and nature of the business you will conduct under the Franchise Agreement; the Community and its proposed site; and the Community's proposed market area.

*Insert initials into the following blank to confirm this statement: _____

2. Each of the undersigned understands that the business you will conduct under the Franchise Agreement involves risk and that success or failure will be substantially influenced by your ability and efforts and the viability of the Community's site.

*Insert initials into the following blank to confirm this statement: _____

3. Each of the undersigned understands that we previously might have entered into franchise agreements with provisions different from the provisions of the Community Franchise Agreement and we have the right to enter into franchise agreements in the future with provisions different from the provisions of the Community Franchise Agreement.

*Insert initials into the following blank to confirm this statement: _____

4. If we unilaterally made material changes in your final, ready-to-be-signed copies of the Franchise Agreement and related documents (other than as a result of our negotiations with you), you have had those documents in hand for at least seven (7) calendar days before signing them and have had ample opportunity to consult with your attorneys, accountants, and other advisors concerning those documents.

*Insert initials into the following blank to confirm this statement: _____

5. You have received a franchise disclosure document ("FDD") as required by law at least fourteen (14) calendar days before signing the Franchise Agreement, and at least fourteen (14) calendar days before paying any consideration to us or an affiliate in connection with this franchise, and have had ample opportunity to consult with your attorneys, accountants, and other advisors concerning the FDD.

6. Except as provided in our FDD, we have made no representation, warranty, promise, guaranty, prediction, projection, or other statement, and given no information, as to the future, past, likely, or possible income, sales volume, or profitability, expected or otherwise, of the Community or any other business, except: (None, unless something is filled-in here).

*Insert initials into the following blank to confirm this statement: _____

7. Each of the undersigned understands that:

7.1 Except as provided in the FDD, we do not authorize our officers, directors, or employees to furnish any oral or written representation, warranty, promise, guaranty, prediction, projection, or other statement or information concerning actual or potential income, sales volume, or profitability, either generally or of any BRIGHTSTAR CARE HOMES[®] Community.

*Insert initials into the following blank to confirm this statement: _____

7.2 Actual results vary from time period to time period, and we cannot estimate, project, or predict the results of any particular BRIGHTSTAR CARE HOMES[®] Community.

*Insert initials into the following blank to confirm this statement: _____

7.3 We have specifically instructed our officers, directors, and employees that, except as provided in our FDD, they are not permitted to make any representation, warranty, promise, guaranty, prediction, projection, or other statement or give other information as to income, sales volume, or profitability, either generally or with respect to any particular BRIGHTSTAR CARE HOMES[®] Community.

*Insert initials into the following blank to confirm this statement: _____

7.4 If any unauthorized representation, warranty, promise, guaranty, prediction, projection, or other statement or information is made or given, the undersigned should not (and will not) rely on it.

*Insert initials into the following blank to confirm this statement: _____

8. Before signing the Franchise Agreement and any related documents, the undersigned has had ample opportunity: (a) to discuss the Franchise Agreement, any related document, and the business you will conduct with its/his/her own attorneys, accountants, and real estate and other advisors; (b) to contact our existing franchisees; and (c) to investigate all statements and information made or given by us and our officers, directors, employees, and agents relating to the BrightStar System, BRIGHTSTAR CARE HOMES[®] Communities, and any other subject.

*Insert initials into the following blank to confirm this statement:

9. Each of the undersigned understands that the Franchise Agreement licenses certain rights for up to three, and only three, BRIGHTSTAR CARE HOMES[®] Communities, located only at the sites specified or to be specified in the Franchise Agreement in accordance with its terms, and that no "exclusive," "expansion," "protected," "non-encroachable," or other territorial rights, rights of first refusal, or rights of any other kind are granted or have been promised concerning the contiguous or any other market area of the Communities or any other existing or potential BRIGHTSTAR CARE HOMES[®] Community or geographical area.

*Insert initials into the following blank to confirm this statement: _____

10. Each of the undersigned understands that the Franchise Agreement (including any riders and exhibits) constitutes the entire agreement between the parties and supersedes all prior and contemporaneous oral or written agreements, statements, representations (except for those in the FDD), or understandings of us, the undersigned, and you.

*Insert initials into the following blank to confirm this statement: _____

11. Each of the undersigned understands that nothing stated or promised by us that is not specifically set forth in the Franchise Agreement or the FDD can be relied upon by the undersigned or you.

*Insert initials into the following blank to confirm this statement: ______

12. The only state(s) in which each of the undersigned is a resident is (are):

.

*Insert initials into the following blank to confirm this statement: ______

13. Each of the undersigned understands the importance of the Community's location. The undersigned and you have had or will have ample opportunity and the means to investigate, review, and analyze independently the Community's site, the market area and all other facts relevant to the selection of a site for a BRIGHTSTAR CARE HOMES[®] Community, and the lease or purchase documents for such location.

*Insert initials into the following blank to confirm this statement: _____

14. Each of the undersigned understands that neither our acceptance or selection of any site, nor our negotiation or acceptance of any lease or purchase contract, implies or constitutes any warranty, representation, guarantee, prediction, or projection that the site will be profitable or successful or that the lease or purchase contract is on favorable terms, its often being the case that such documents contain very tough terms.

*Insert initials into the following blank to confirm this statement: _____

15. Each of the undersigned understands that site selection is a difficult and risky proposition. We have not given any representation, warranty, promise, guaranty, prediction, projection, or other statement or information relied upon by the undersigned or you regarding a site's prospects for success or other attributes, even if we reviewed the lease or purchase contract. You will have any lease or purchase contract reviewed by your own attorney and other advisors.

*Insert initials into the following blank to confirm this statement: _____

16. Each of the undersigned understands that the estimated initial investment ranges disclosed in Item 7 of our FDD are for BRIGHTSTAR CARE HOMES[®] Communities of a certain size, at certain types of sites, and having certain characteristics that we consider to be fairly standard for BRIGHTSTAR CARE HOMES[®] Communities. Your actual investment to develop your Communities could be incrementally or materially higher than the investment ranges disclosed in Item 7 if you choose to develop a larger Community or a Community that otherwise is atypical when compared with standard BRIGHTSTAR CARE HOMES[®] Communities.

*Insert initials into the following blank to confirm this statement: _____

17. The covenants and restrictions concerning competition contained in the Franchise Agreement are fair and reasonable and will not impose an undue hardship on the undersigned or you. Each of the undersigned has other considerable skills, abilities, opportunities, and experience

in other matters and of a general nature enabling it/him/her to derive satisfactory income from other endeavors.

*Insert initials into the following blank to confirm this statement: _____

18. There is no fiduciary or confidential relationship between us and the undersigned or between us and you. Each of the undersigned expects us to deal, and will act as if we are dealing, with it/him/her at arm's length and in our best interests.

*Insert initials into the following blank to confirm this statement: _____

19. We have advised the undersigned and you to consult with your and their own advisors on the legal, financial, and other aspects of the Franchise Agreement, this document, the Community, any lease, sublease, or purchase contract for the site, and the business contemplated. Each of the undersigned has either consulted with such advisors or deliberately declined to do so.

*Insert initials into the following blank to confirm this statement: _____

20. Neither we nor our employees have provided the undersigned or you with services or advice that are legal, accounting, or other professional services or advice.

*Insert initials into the following blank to confirm this statement: _____

21. We may communicate directly with your trade suppliers at any time during your operation of the Community and obtain from them any sales and purchasing information relating to their dealings with you.

*Insert initials into the following blank to confirm this statement: _____

22. We may sell or transfer our assets, our trademarks, or the BrightStar System outright to a third party; may go public; may engage in a private placement of some or all of our securities; may merge, acquire other corporations, or be acquired by another corporation; and/or may undertake a refinancing, a recapitalization, a leveraged buy-out, or other economic or financial restructuring.

*Insert initials into the following blank to confirm this statement: _____

23. Each of the undersigned understands that we have (or will have) agreements in place with our vendors that will prohibit them from doing business with the undersigned if our relationship with the undersigned ends for whatever reason. These restrictions are in place to protect the BrightStar System.

*Insert initials into the following blank to confirm this statement: _____

24. Each of the undersigned understands that the Franchise Agreement contains termination damages provisions that apply in certain circumstances and acknowledges that those provisions are fair and reasonable.

*Insert initials into the following blank to confirm this statement:

The statements made in this document supplement and are cumulative to 25. statements, warranties, and representations made in other documents, such as the Franchise Agreement. The statements made in this document or the Franchise Agreement are made separately and independently. They are not intended to be, and will not be, construed as modifying or limiting each other.

*Insert initials into the following blank to confirm this statement: _____

FRANCHISEE (YOU):

Print Name of Legal Entity

By: _____ Signature

Print Name: _	
Title:	
Date:	

Owners of your legal entity must sign below in their individual capacities.

Signature Print Name: _____ Date:

	Signature	
Print Name:		
Date:		

EXHIBIT F

SPOUSAL / LIFE PARTNER CONSENT

NOTE: THE SPOUSE / LIFE PARTNER OF EACH OWNER OF FRANCHISEE MUST SIGN THIS SPOUSAL / LIFE PARTNER CONSENT.

The individual(s) listed below represents to BrightStar Senior Living Franchising, LLC ("Company") that each is the spouse (including common law spouse) or life partner of the individual(s) who is an owner of the franchisee entity ("you" or "your") that has signed a Franchise Agreement with the Company dated ______.

In consideration of the Company's grant to you of the rights under the Franchise Agreement, each of the individual spouses (including common law spouse) or life partners listed below agrees, in consideration of benefits received and to be received by each of them, jointly and severally, and for themselves and their heirs, legal representatives, and assigns, that they and each of them:

1. must be firmly bound by all of the terms, provisions, and conditions in the Franchise Agreement;

2. unconditionally guarantee the full and timely performance by you of all of your obligations under the Franchise Agreement, including, without limitation, any of your indebtedness arising under or by virtue of the Franchise Agreement; and

3. agree to be bound by the non-competition covenants in the Franchise Agreement.

EXHIBIT G

CONDITIONAL ASSIGNMENT OF FRANCHISEE'S TELEPHONE NUMBERS

2. The conditional agreement will become effective automatically upon termination or expiration of Assignor's Franchise Agreement. Upon the occurrence of that condition, Assignor must do all things the telephone company requires to assure the effectiveness of this assignment of telephone numbers as if the Assignee had been originally issued such telephones, telephone numbers, and telephone listings.

3. Assignor agrees to pay the telephone company on or before the effective date of assignment all amounts owed for use of the telephone number(s), including, without limitation, Yellowpages.com advertising. Assignor further agrees to reimburse Assignee for any sums Assignee must pay the telephone company to effectuate this agreement and to cooperate with the telephone company and Assignee fully in effectuating this assignment.

ASSIGNOR:

BY:	Date:
NAME:	
TITLE:	

ASSIGNEE:

BrightStar Senior Living Franchising, LLC

By:	
Name	
Title:	

EXHIBIT H

SITE SELECTION ADDENDUM

BrightStar Senior Living Franchising, LLC ("we", "us", or "our") and _____

("you") have as of ______ (the "Effective Date"), entered into the Franchise Agreement (the "Franchise Agreement") granting to you the right to construct, develop, and operate up to three BRIGHTSTAR CARE HOMES[®] Communities providing various services for seniors and other individuals in need, including residential assisted living, memory care, and ancillary services (the "Communities"), and desire to supplement its terms, as set forth below. The parties therefore agree as follows:

1. Within one year (365 days) after the Effective Date (the "First Site Acquisition Deadline"), you must locate and secure, at your own expense, a site for the first Community to be constructed, developed, and operated under the Franchise Agreement, which site we will review and evaluate as provided below. The site for the first Community (as well as the sites for your second and third Communities permitted under the Franchise Agreement, if you choose to develop them) must be physically located within the following site selection area: <u>If you or your affiliate operates a BrightStar Care Agency, the Site Selection Area will match the franchisee's "Protected Territory" under its BSC Franchise Agreement. If you or your affiliate does not operate a BrightStar Care Agency, the Site Selection Area will be defined by a collection of contiguous zip codes with a total combined population of approximately 200,000 to 300,000 people and then specifically described here.]</u>

2. Your failure to obtain a site for your first Community by the First Site Acquisition Deadline will constitute a default under the Franchise Agreement and this Site Selection Addendum. Time is of the essence. The provisions below also will govern your selection of a site for the second and third Communities permitted under the Franchise Agreement if you choose to develop them.

3. Prior to your acquisition by lease or purchase of a site for the Community, you must submit to us, in the form we specify, such information or materials we may reasonably require regarding the proposed site. We will have twenty-one (21) calendar days after receiving such information and materials from you to accept or reject the site as a location for the Community. No proposed site will be deemed accepted unless we have expressly accepted it in writing.

4. We will grant you access to the Operations Manual, which outlines such site selection guidelines we deem advisable as part of our evaluation of your request for site acceptance. We will not, however, provide on-site evaluation for any proposed site before we receive the information and materials required by Paragraph 3. We may, but are not obligated to, make one trip to your market at our own cost to visit sites you are considering for your Communities so that we can accept or not accept the sites. If we make more than one on-site evaluation, you must reimburse our reasonable expenses, including the costs of travel, lodging, and meals.

Exhibit H-1

5. If you will occupy the Community's premises under a lease, you must send us the lease for our written acceptance before you sign it. We will have twenty-one (21) calendar days after receiving the lease to accept or reject it. Our acceptance will be conditioned upon your execution of a Collateral Assignment of Lease in the form we prescribe (a sample of which is attached to this Addendum, but which may change based on the site's circumstances) and the lease's inclusion of the following terms and conditions:

- That the initial term of the lease will be for a minimum of twenty years.
- That use of the premises be restricted solely to the operation of the Community;
- That you be prohibited from subleasing or assigning all or any part of your occupancy rights or extending the term of or renewing the lease without our prior written consent;
- That the lessor provides us copies of any and all notices of default given to you under the lease;
- That we (or our designee) have the right to enter the premises to make modifications necessary to protect the Licensed Marks or the System or to cure any default under the Agreement or under the lease;
- That we (or our designee) have the option, upon default, expiration, or termination of the Franchise Agreement or upon an early termination of the lease, and upon notice to the lessor, to assume all of your rights under the lease terms, including the right to assign or sublease.

6. You must give us a copy of the executed lease and Collateral Assignment of Lease within fourteen (14) days after signing them.

7. After we have accepted a site for the Community in writing and you have acquired the site by lease or purchase, the site will constitute the Community's Premises referred to in Section 1.1 of the Franchise Agreement and will be identified on Exhibit A to the Franchise Agreement.

8. You acknowledge and agree that our acceptance of a site does not constitute an assurance, representation or warranty of any kind, express or implied, as to the site's suitability for a BRIGHTSTAR CARE HOMES[®] Community or any other purpose. Our acceptance of the site indicates only that we believe the site complies with acceptable minimum criteria that we have established solely for our purposes as of the time of the evaluation. Both parties to this Agreement acknowledge that application of criteria that have been effective with respect to other sites and premises may not predict the potential for all sites and that, after our acceptance of a site, demographic and/or economic factors included in or excluded from our criteria could change, thereby altering the site's potential. Such factors are unpredictable and beyond our control. We will not be responsible for the failure of a site we accept to meet your expectations as to revenue or operational criteria. You further acknowledge and agree that your acceptance of a franchise for

Exhibit H-2

the BRIGHTSTAR CARE HOMES[®] Communities granted by the Franchise Agreement is based on what is and will be your own independent investigation of the suitability of the sites for your Communities.

9. This Site Selection Addendum constitutes an integral part of the Agreement between the parties and will control the subject matter it addresses. Except as modified or supplemented by this Site Selection Addendum, the Franchise Agreement's terms are hereby ratified and confirmed.

IN WITNESS WHEREOF, the parties hereto have duly executed this Addendum as of the Effective Date.

FRANCHISOR (US): BRIGHTSTAR FRANCHISING, LLC

By:		
Name:		
Title:		

FRANCHISEE (YOU):

By:		
Name:		
Title:		
By:		

Name:		
Title:		

This instrument was prepared by

and after recording return to:

[]
[]
[]
ſ	1

COLLATERAL ASSIGNMENT OF LEASE

("Assignor"), and **BRIGHTSTAR SENIOR LIVING FRANCHISING**, **LLC**, an Illinois limited liability company ("BSLF"); and

In consideration of the mutual promises and covenants herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged by each party, the parties agree as follows:

1. The Assignor hereby assigns, transfers, and sets over unto BSLF all of Assignor's right, title, and interest as tenant in, to, and under that certain Lease Agreement dated <______> and Rider to Lease Agreement dated concurrently herewith between Landlord and Assignor, as amended (with BSLF's consent) (collectively, the "Lease"), for certain real property located at <______> and more particularly described on the attached **Exhibit A** (the "Premises"). This Assignment is for collateral purposes only and, except as specified herein, BSLF shall have no liability or obligation of any kind whatsoever arising from or in connection with this Assignment unless BSLF shall exercise its rights by providing the BSLF Notice (as defined herein).

2. Assignor represents and warrants to BSLF that it has full power and authority to so assign the Lease and its interest therein, and Assignor has not previously assigned or transferred, and is not obligated to assign or transfer, the Lease or any of its interest in the Lease or the Premises.

3. In the event (i) of any default by Assignor under the Lease which remains uncured after expiration of any applicable cure period; or (ii) that certain Franchise Agreement dated <______> between BSLF and Assignor,

including all amendments and renewals thereof (the "Franchise Agreement"), for a BRIGHTSTAR

CARE HOMES[®] Community to be located at Premises should, for any reason, be terminated or expire, BSLF shall have the right for a period of thirty (30) days after the event in (i) or (ii) herein to exercise its rights pursuant to this Assignment by providing written notice to Landlord and Assignor ("BSLF Notice"). If BSLF exercises its rights by providing the BSLF Notice: (a) BSLF shall thereupon become tenant under the Lease with all rights and obligations of tenant commencing upon, first accruing, and effective from and after the date of the BSLF Notice; (b) BSLF shall have the right to take possession of the Premises, expel Assignor therefrom without being guilty of trespass, forcible entry or detainer, or other tort; (c) Assignor shall have no further right, title, or interest in the Lease or the Premises; and (d) Assignor shall peaceably and promptly vacate the Premises and (subject to BSLF's right to acquire any such property pursuant to the Franchise Agreement) remove its personal property therefrom and any property not removed or otherwise disposed of by the Assignor shall be deemed abandoned. Nothing in this Assignment shall restrict, limit, terminate, waive, or otherwise affect Landlord's rights against Assignor.

4. In no event shall BSLF be or become liable for any liability or obligation of Assignor accruing or applicable to the period prior to the date of the BSLF Notice. All rents and other obligations under the Lease shall be prorated as of the date of the BSLF Notice. BSLF shall have no liability or obligation to the Landlord under the Lease unless and until it exercises its right by providing the BSLF Notice.

5. Assignor agrees it will not allow or permit any surrender, termination, amendment, or modification of the Lease which adversely affects the rights of BSLF under the Lease or this Assignment without the prior written consent of BSLF, which consent shall not be unreasonably withheld, conditioned, or delayed. Through the term of the Franchise Agreement and any renewals thereof, Assignor agrees that it shall elect and exercise all options to extend the term of or renew the Lease not less than thirty (30) days prior to the last day on which such option must be exercised, unless BSLF otherwise agrees in writing. Upon BSLF's failure to otherwise agree in writing, and upon Assignor's failure to so extend or renew the Lease as stated herein, Assignor hereby appoints BSLF as its true and lawful attorney-in fact to exercise such extension or renewal options in the name, place, and stead of Assignor for the sole purpose of effecting such extension or renewal.

6. Anything herein to the contrary notwithstanding, and without waiving any rights BSLF may have under the Franchise Agreement, in the event Assignor acquires fee simple title to the Premises during the term or any extension of the Lease, Assignor shall enter into an option agreement with BSLF (utilizing BSLF's then-current form) granting BSLF the right to lease the Premises on the same terms as the Lease in the event Assignor ceases to own the Premises or the Franchise Agreement should for any reason be terminated or expire.

7. Landlord consents to this Assignment and agrees to recognize BSLF as tenant under the Lease upon exercising the rights in favor of BSLF as set forth in this Assignment and in the Lease.

8. All notices (including the BSLF Notice) shall be deemed sufficient and properly given in writing (except as otherwise expressly provided herein) if delivered by one of the following methods: (i) by personal delivery or by a reputable delivery service at the street address specified below, or (ii) by first-class, registered or certified mail, postage prepaid, to the post office

box specified below or to the street address if no post office box is given. The hand delivery address and mailing address for receipt of notice or other documents by such parties are as follows:

If to Landlord:	<	>	
	<	>	
	<	>	
	Attention: <		>
If to Assignor:	<	>	
U	<	>	
	<		
	Attention: <		>
If to BSLF:	BrightStar Senior L 2275 Half Day Roa	d, Suite 210	sing LLC
	Bannockburn, Illing	ois 60015	

Attention: CEO

Any of the above-mentioned parties may, by like notice, designate any further or different addresses to which subsequent notices shall be sent. Any notice hereunder signed on behalf of the notifying party by its identified attorney-at-law shall be valid and effective to the same extent as if signed by such party. Any notice or other communications shall be deemed delivered when actually delivered to the address of the party to whom directed or, if sent by mail, three (3) days after such notice or document is deposited in the United States mail, as provided above.

9. Time is of the essence.

10. This Assignment shall inure to the benefit of and be binding upon their respective heirs, successors, representatives, and permitted assigns.

11. This Assignment shall be governed by, and construed in accordance with, the laws of the state in which the Premises are located.

[Signature Page Follows]

IN WITNESS WHEREOF, the parties hereto have duly executed, sealed and delivered this Assignment on the day and year first above written.

		LANDLORD:		
		<>		
	By:			
	Name:			
	Title:			
State of	f			
County	of			
2				
	The fo	regoing instrument was acknowledged be	fore me this <	>
by) of
<		>, on behalf of the company.		
		(SEAL)	Notary Public	
		My commission ex	xpires:	

ASSIGNOR:

<____>

By: _____

Name: _____

Title: _____

State of _____

County of	f
-----------	---

	The	foregoing	instrument	was	acknowledged	before	me	this
<		>	by		,			of
<		>, on b	ehalf of the con	mpany.				

(SEAL)

Notary Public

My commission expires:

RRICHTSTAR SENIO	R LIVING FR		
DRIGHTSTAR SERIO		ANCHISING, LL	С
Ву:			
Name:			
Title:			
State of			
County of	_		
The foregoing instr <> by			
SENIOR LIVING FRANCHISING, L			

(SEAL)

Notary Public

My commission expires:

EXHIBIT A

TO COLLATERAL ASSIGNMENT OF LEASE LEGAL DESCRIPTION OF THE PREMISES

Exhibit H-1-7

EXHIBIT I

FRANCHISOR AND FRANCHISEE ACKNOWLEDGMENTS

We and you acknowledge the following:

You will be a corporation owned initially in majority by that corporation's 401(k) plan and in minority by _______ in (their/his/her) individual legal capacity. The 401(k) plan will not be a guarantor of franchise royalty payments or any other payment obligations under the Franchise Agreement. This does not relieve ______

______ in (their/his/her) individual legal capacity from (their/his/her) obligation to personally guarantee performance under the Franchise Agreement.

Additionally, ______ will be treated as the Control Person for purposes of operational obligations, training requirements, or other provisions in the Franchise Agreement requiring performance of the "Control Person," since instead the majority shareholder will at least initially be the corporation's 401(k) plan. The mix of 401(k) and individual ownership outlined in the above paragraph will be permitted under the Franchise Agreement.

Finally, transfer fee provisions and restrictions contained in the Franchise Agreement will not apply to redemptions of stock by the corporation that is initially held by the 401(k) plan. The purpose of this paragraph is to permit _______ to cause the corporation to redeem stock from the 401(k) so that (they/he/she) will eventually own and control 100% of the stock in (their/his/her) individual legal capacity.

This Acknowledgment supersedes any contradictory provisions in the Franchise Agreement or any modifications to that Agreement and is incorporated into the Franchise Agreement in its entirety. It does not change the enforceability of the Franchise Agreement in any manner against ______ acting in (their/his/her) individual legal capacity in using (their/his/her) own assets to guarantee performance under the Franchise Agreement.

BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC (US)

By:	
Name:	
Title:	
Date:	

FRANCHISEE (YOU):

By:	
Name:	
Title:	
Date:	

EXHIBIT J

BUSINESS ASSOCIATE AGREEMENT

Exhibit J-1

BUSINESS ASSOCIATE AND CONFIDENTIALITY AGREEMENT

("Covered Entity") and BrightStar Senior Living Franchising, LLC ("BrightStar" or "Business Associate") hereby enter into this Agreement, effective as of Effective Date of the Franchise Agreement (the "Effective Date").

Recitals

A. BrightStar arranges for certain products and services (the "Services") as a franchisor to Covered Entity as a franchisee, as set forth in the BrightStar Senior Living Franchising BRIGHTSTAR CARE HOMES[®] Community Franchise Agreement between Covered Entity and BrightStar ("Franchise Agreement").

B. The Parties' performance under the Franchise Agreement may or will require Covered Entity to disclose and/or provide to Business Associate private and/or protected health and/or medical information as defined under, and governed by, applicable state law and Individually Identifiable Health Information and/or Electronic Protected Health Information as defined in the Health Insurance Portability and Accountability Act ("HIPAA"), the Health Information Technology for Economic and Clinical Health Act ("HITECH"), and/or regulations promulgated under such laws (state law, HIPAA, and HITECH are hereafter referred to collectively as "Privacy Laws") and may or will require BrightStar to receive, access, review, maintain, retain, modify, record, store, forward, produce, hold, use, create, disclose, and/or destroy such information (the "PHI").

C. BrightStar's performance of the Services may give rise to certain legal obligations under Privacy Laws and BrightStar may be considered a "business associate" and franchisee may be a "covered entity" as those terms are defined in 45 C.F.R. § 160.103. This Agreement will not apply to relationships between the parties where BrightStar is not considered a "business associate" as defined in 45 C.F.R. 160.103.

Accordingly, the parties hereto ("Parties") agree to the terms and conditions set forth below:

Terms of Agreement

1. <u>Performance and Compliance with Law</u>. The Parties will work together in good faith to determine applicability of Privacy Laws, to comply with applicable Privacy Laws, and to amend this Agreement as necessary for Covered Entity and Business Associate to comply with applicable Privacy Laws, as modified and/or supplemented from time to time.

2. <u>Interpretation</u>. Any ambiguity herein must be resolved in favor of a meaning that permits both Covered Entity and Business Associate to comply with applicable Privacy Laws, consistent with the Franchise Agreement. Terms that are not otherwise defined herein shall have the meanings given to them by HIPAA, the HITECH Act, and their implementing regulations.

3. <u>Privileges and Protections</u>. This Agreement does not constitute or evidence a waiver of, nor does it amend, the attorney-client privilege, the attorney work-product doctrine, and/or any other applicable privileges or protections.

4. <u>Business Associate's Obligations</u>.

4.1 <u>Handling of the PHI and Safeguards</u>. Business Associate agrees to prevent access, use and/or disclosure of PHI other than as permitted or required by this Agreement and/or applicable Privacy Laws, and will implement and use, at all times, appropriate administrative, physical and technical safeguards to (i) prevent access, use or disclosure of PHI other than as permitted by this Agreement and/or Privacy Laws; and (ii) reasonably and appropriately protect the confidentiality, integrity, security, and availability of PHI.

4.2 <u>Minimum Necessary Use and Disclosure</u>. Business Associate will determine the amount of PHI necessary for performance of the Services and will make reasonable efforts to limit the receipt, use, and disclosure of PHI to the minimum necessary as required by the Privacy Laws.

4.3 <u>Data Aggregation and De-identification</u>. Business Associate may use PHI to aggregate information to provide reports regarding health care operations to Covered Entity. The parties agree that Business Associate may de-identify PHI as defined by and in compliance with Privacy Laws and may use such de-identified information as permitted by applicable law.

4.4 <u>Management and Administration</u>. Business Associate may use and disclose PHI for management and administrative purposes. In doing so, Business Associate will comply with all applicable Privacy Laws and with Covered Entity's obligations under subpart E of 45 CFR Part 164.

4.5 <u>Disclosures to Subcontractors and/or Third Parties</u>. Business Associate shall ensure that all representatives, subcontractors, persons and/or entities (other than entities that are merely conduits) to whom Business Associate discloses or provides the PHI execute a written Business Associate Agreement, as required under the Privacy Laws, in which such third persons and/or entities expressly agree to the same restrictions and conditions that apply to Business Associate hereunder, as applicable. If a Business Associate Agreement is not required by the Privacy Laws, Business Associate shall obtain reasonable assurances from all persons and entities who have access to or are recipients of the PHI that: (i) the PHI will be held confidential and used or further disclosed only as required by law or for the purposes for which it was disclosed to the third party; and (ii) the third party will promptly notify Business Associate of any Compromise of PHI, and Business Associate will, in turn, notify Covered Entity.

4.6 <u>Access to, or Amendment of, PHI</u>. If Business Associate maintains any PHI in a Designated Record Set on behalf of Covered Entity, Business Associate agrees:

(a) to provide access to the PHI in a Designated Record Set to authorized individuals as required by Privacy Laws and in the time, manner, and

format designated by such individuals to the extent required by Privacy Laws; and

(b) to make any amendment(s) to PHI in a Designated Record Set as requested by Covered Entity and/or authorized individuals pursuant to 45 C.F.R. § 164.526.

4.7 <u>Restrictions on PHI</u>. Business Associate will comply with any patient restrictions on the Use and Disclosure of PHI requested by Covered Entity under Section 5.3 below.

4.8 <u>Accounting of PHI Disclosures</u>. Business Associate will document and report to Covered Entity all disclosures of PHI that are required for Covered Entity to provide an accounting under 45 C.F.R. § 164.528 and/or the Privacy Laws. If an individual contacts Business Associate directly for such an accounting, Business Associate will direct the individual to contact Covered Entity.

4.9 <u>Reporting of Violations and Security Incidents.</u> Business Associate will promptly report to Covered Entity any impermissible use or disclosure under Privacy Laws of which it becomes aware that Compromises the security or privacy of the PHI ("Breach"). Business Associate will include in the report of Breach the following information if known or can be reasonably obtained:

- (a) Contact information for individuals who may be impacted by the Breach;
- (b) The date of Breach and a brief description of the circumstances surrounding the Breach;
- (c) A description of the type of information involved; and
- (d) What Business Associate is doing to investigate the Breach and mitigate harm to individuals.

In addition, Business Associate will report attempted or successful unauthorized access, use, disclosure, modification, or destruction of information or interference with system operations in an information system that does not Compromise the security or privacy of the PHI ("Security Incidents"). Business Associate will identify and respond internally to suspected or known Security Incidents, and will mitigate, to the extent practicable, their harmful effects, document their outcomes, and provide such documentation to Covered Entity upon request. Notice is hereby deemed provided, and no further notice will be given, with respect to routine unsuccessful attempts at unauthorized access to the PHI such as pings and other broadcast attacks on firewalls, denial of service attacks, failed login attempts, and port scans.

The parties will meet and confer in good faith before notifying affected individuals, government agencies, and/or commencing any legal action regarding any suspected or actual Breach or Security Incident and/or breach of this Agreement and shall comply with applicable

Exhibit J-4

Privacy Laws regarding the need for and nature of any notification of individuals or reporting to government agencies.

4.10 <u>Mitigation and Notification</u>. Mitigation efforts by Business Associate shall not require Business Associate to pay the costs of credit monitoring or other similar credit protection services unless required by law. Business Associate will not be responsible for notifying individuals of a Breach and will not be responsible for any notification costs.

4.11 <u>Audits and Inspections</u>. Business Associate will make its internal practices, books, and such records as are not protected by applicable legal privilege or work product protection relating to the use, disclosure, and/or compromise of PHI available to Covered Entity to determine compliance with applicable Privacy Laws and this Agreement, and to the Secretary of the United States, Department of Health and Human Services and/or other authorized lawful authority as required by law or authorized by Covered Entity in writing.

4.12 <u>Prohibition on Sale of PHI and use of PHI for Marketing.</u> Business Associate will not directly or indirectly receive remuneration in exchange for any PHI and will not use or disclose PHI for Fundraising and/or Marketing purposes, except with prior written consent of Covered Entity and in accordance with applicable Privacy Laws.

5. <u>Covered Entity's Obligations</u>.

5.1 <u>HIPAA Compliance</u>. Covered Entity agrees that it will comply with Privacy Laws, including, but not limited to, using appropriate safeguards to protect the privacy, security, and integrity of electronic PHI and training workforce members on the appropriate use of the property management and other electronic information systems BrightStar's Franchise Agreement requires Covered Entity to use in its operations.

5.2 <u>Notice of Privacy Practices</u>. Covered Entity is responsible for using the services of Business Associate in accordance with its Notice of Privacy Practices (NPP). Covered Entity will provide a copy of its NPP to Business Associate upon request.

5.3 <u>Restrictions and Revocations</u>. Covered Entity will promptly notify Business Associate in writing of any patient-requested restrictions, changes to, or revocation of consent and/or authorization to use and/or disclose PHI that may affect Business Associate's ability to perform its obligations under this Agreement or the Franchise Agreement.

5.4 <u>Compliant Requests</u>. Covered Entity will not request or cause Business Associate to make a Use or Disclosure of PHI in a manner that does not comply with Privacy Laws.

5.5 <u>Authorizations</u>. Covered Entity will obtain all consents and authorizations necessary and/or required by law for Covered Entity and Business Associate to fulfill their obligations under applicable Privacy Laws and this Agreement.

5.6 <u>Accounting of PHI Disclosures</u>. Covered Entity will include in individual accountings requested under the Privacy Laws, including without limitation, 45 C.F.R. § 164.528, any disclosures by Business Associate, to the extent such disclosures are made.

5.7 <u>Meet and Confer</u>. Upon any suspected or actual Breach, unauthorized disclosure of the PHI or breach of this Agreement, Covered Entity will meet and confer in good faith with Business Associate before notifying affected individuals, government agencies, and/or commencing any legal action.

6. <u>Term and Termination</u>.

6.1 <u>Term</u>. The term of this Agreement will commence upon receipt by Business Associate of any PHI, or the date set forth below, whichever is earlier, and will terminate upon discharge of Business Associate's obligations under the Franchise Agreement and this Agreement, including the obligations set forth in Section 6.3 below, and/or performance of the Services.

6.2 <u>Effects of Termination</u>. Covered Entity acknowledges that due to the relationship with Business Associate, it will not be feasible for Business Associate to return or destroy PHI after the termination of the Franchise Agreement. Business Associate will continue to extend the protections of this Agreement to PHI and limit further Uses and Disclosures of such PHI to those purposes that make the return or destruction infeasible or contrary to the Privacy Laws, for so long as Business Associate maintains such PHI.

6.3 <u>Breach</u>. If either party hereto breaches its obligations under this Agreement, the non-breaching party will provide the other with notice and a thirty (30) day period to cure the breach. If the breaching party fails to cure the breach or cure is not possible within thirty (30) days, the non-breaching party may terminate this Agreement immediately upon written notice and without further legal action or declaration.

7. <u>Miscellaneous</u>.

7.1 <u>Entire Agreement</u>. This Agreement and the Franchise Agreement, the consistent terms of which are incorporated herein, constitute the entire agreement between the Parties and supersede all prior negotiations, discussions, representations, or proposals, whether oral or written, unless expressly incorporated herein, related to the subject matter of this Agreement. Unless otherwise expressly provided herein, this Agreement may not be modified unless in writing signed by the duly authorized representatives of the parties.

7.2 <u>Severability</u>. If any provision of this Agreement or part thereof is found to be invalid, the remaining provisions will remain in full force and effect.

7.3 <u>Waiver</u>. Any failure of a party to insist upon strict compliance with any term, undertaking, or condition of this Agreement will not be deemed to be a waiver of such term, undertaking, or condition. To be effective, a waiver must be in writing, signed and dated by the parties to this Agreement.

7.4 <u>No Third-Party Beneficiaries</u>. Except as otherwise provided in the Privacy Laws or this Agreement, there are no third-party beneficiaries to this Agreement. Business Associate's obligations are to Covered Entity only.

7.5 <u>Successors and Assigns</u>. This Agreement will inure to the benefit of, and be binding upon, the successors and assigns of the parties. However, this Agreement is not assignable by any party without the prior written consent of the other parties.

7.6 <u>Dispute Resolution</u>. If at any time during or after the term of this Agreement either party hereto believes that a dispute exists between them, then the parties agree that they shall follow the Dispute Resolution process outlined in the Franchise Agreement.

7.7 <u>Counterparts</u>. This Agreement may be executed in counterparts, by manual, electronic, or facsimile signature, each of which will be deemed an original and all of which together will constitute one and the same instrument.

IN WITNESS WHEREOF the undersigned duly authorized representatives of the parties hereby execute this Business Associate Agreement as of ______.

COVERED ENTITY:	BUSINESS ASSOCIATE:		
By:	By:		
Name:	Name:		
Its:	Its:		

Exhibit J-7

EXHIBIT K

CONTROL PERSON ADDENDUM

Exhibit K-1

EXHIBIT K CONTROL PERSON ADDENDUM

1. <u>Control Person</u>. One of your owners must be the Control Person. We require that your Communities at all times be under your designated Control Person's direct supervision. You represent and warrant to us that the following person, and only the following person, is the Control Person:

NAME <u>TITLE</u> <u>ADDRESS</u>

2. <u>Control Person's Responsibilities</u>. Your Control Person must meet our standards and requirements and must have successfully completed the BRIGHTSTAR CARE HOMES[®] Community initial training program. Your Control Person is the individual who has the authority to actively direct your business affairs regarding the Communities, is responsible for overseeing the general management of the Communities, and has authority to sign all contracts. You must seek our approval if you want the Control Person to be someone other than your owner during the franchise term. Prior to there being any change to your designated Control Person, you must give us advance written notice of this change, and we must confirm that your proposed replacement Control Person meets our then-current Control Person standards and requirements.

2. <u>Default of the Agreement</u>. You will be in default under the Franchise Agreement if any of the following occurs: (i) at any time during the Franchise Agreement's term, your Communities are not under the designated Control Person's direct supervision, (ii) you designate a replacement Control Person without first providing us with advance written notice of this change and we do not approve the change, or (iii) any replacement Control Person does not meet our then-current Control Person standards and requirements. Upon any of these occurrences, we have the right, after giving you advance written notice of the default and a 15-day opportunity to cure the default, to terminate the Franchise Agreement.

FRANCHISOR: BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC (US)

By:		
Name:		
Title:		

FRANCHISEE (YOU):

By:		
By: Name:		
Title:		
By:		
By: Name:		
Title		

EXHIBIT C

TO BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC BRIGHTSTAR CARE HOMES COMMUNITY FRANCHISE DISCLOSURE DOCUMENT

OPERATIONS MANUAL TABLE OF CONTENTS

OUR OPERATIONS MANUAL CONTAINS A TOTAL OF 180 PAGES AND COVERS THE FOLLOWING TOPICS:

TABLE OF CONTENTS

Volume 1 – Getting Started – Table of Contents

Chapte	er 1: Overview	6
1.	How to Use this Manual	6
2.	The Company	6
3.	You are the CEO of your own business	9
4.	Franchisor/Franchisee Relationship	9
5.	Initial Training Program	16
Chapte	er 2: Setting up Your Business	17
1.	Selecting Professional Advisors	17
2.	Setting up the Business	20
3.	Insurance	27
4.	Telephone, E-mail and Listings	31
5.	Start-up Equipment and Supplies	33
Chapte	er 3: Vendors	
1.	Definitions	
2.	Working with Vendors	36
3.	Vendor Variance	37
4.	Vendor Feedback	31
5.	Start-up Equipment and Supplies	33
Volum	ne 2 – Setting up Your Business – Table of Contents	
1.	Chapter 1: Leasing	
2.		
3.		
Chapte	er 4: Office Layout	12
Volum	ne 3 – Getting Started – Table of Contents	
СНАРТ	TER 1: EMPLOYEE SET UP	1
1.	Staff Requirements	1
2.	Setting A Pay Scale	3
3.	The Hiring Process	5
4.	Preparing For New Hires	17
5.	Paperwork	26
СНАРТ	FER 2: MANAGING STAFF	1
1.	Setting Standards	1
2.	Employee Behavior and Training	4
3.	Counseling and Documentation	6
4.	Reviews and Evaluations	7
5.	Retention	8

6.	Termination	11
CHAPT	ER 3: RECRUITING	1
	Lead Sources	
	Strategies	
3.	Conclusion	

Volume 4 – Franchise Administration – Table of Contents

1.	Client Care Standards	1
2.	Communication	4
3.	Continuing Payments	7
4.	Record Keeping and Accounting	10
5.	Conclusion	22

Volume 5 – Sales & Marketing – Table of Contents

1.	What We Do	1
	How You Make Money	
3.	Getting Referrals	.14
4.	Institutional Sales	.45
5.	Private Duty – Inside Sales	62
6.	Client Retention	87
7.	Contracts and Agreements	89
8.	Graphic Standards	.98
9.	BrightStart/ReStart Standards Guide	103
	Social Media Policy	
11.	Co-Op Policy	.112
12.	Conclusion	.113

Volume 6 – Business Operations – Table of Contents

1.	Hours of Operations	1
	Operations Standards	
	The Daily Routine	
	Accounting	
	Weekly Activities	
7.	Monthly Activities	1
8.	Reports	1
9.	Conclusion	1

Volume 7 – ABS User Guide – Table of Contents

Chapter 1: What is ABS?	21
Chapter 2: What Does ABS Manage?	23
Chapter 3: ABS Basics	26

Chapter 4: My ABS	56
Chapter 5: Location Management	67
Chapter 6: User Management	90
Chapter 7: Product Catalog	
Chapter 8: Employee Management	
Chapter 9: Customer Management	
Chapter 10: Daily Operations	
Chapter 11: ABS Mobile	
Appendix 1: Role Based Security	608
Appendix 2: PTO Accrual Rates	612
Appendix 3: Pay Rate Hierarchy	613
Appendix 4: Live In Operating Procedures	614
Appendix 5: Mobile Maintenance Screen Exceptions	616
Appendix 6: Notifications	620
Index	622
Volume 9 – Clinical – Table of Contents	
Chapter 1: Getting Started Clinically	7
Chapter 2: Initial Orientation Training	8
Chapter 3: Registered Nurse Oversight, and Responsibilities	
Chapter 4: Documentation Standards	14
Chapter 5: OSHA Standards	15
Chapter 6: Client Privacy/Confidentiality	24
Chapter 7: Joint Commission Accreditation	25
Chapter 8: State and Business Licensure	27
Volume 10 – Skilled Operations – Table of Contents	
Chapter 1: Overview and Definitions	4
Chapter 2: Organizational Structure	17
Chapter 3: Care and Quality Standards	22
Chapter 4: Business Processes Within the Client Lifecycle	

Chapter 5: Selling Skilled Services	.44
Appendix	52

Total Number of Pages.....1,049

EXHIBIT D

TO BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC BRIGHTSTAR CARE HOMES COMMUNITY FRANCHISE DISCLOSURE DOCUMENT

CONFIDENTIALITY, NON-DISCLOSURE AND NON-COMPETITION AGREEMENT

CONFIDENTIALITY, NON-DISCLOSURE AND NON-COMPETITION AGREEMENT

WHEREAS, the undersigned is an officer, director, member, manager, key employee, partner or owner of an interest in the equity or voting interests of _______, the Franchisee under, and signatory to, that certain Franchise Agreement dated ______, (the "Franchise Agreement") entered into with BrightStar Senior Living Franchising, LLC ("Company") granting Franchisee the right to own and operate a BRIGHTSTAR CARE HOMES Community on the terms and conditions stated in the Franchise Agreement.

WHEREAS, the undersigned acknowledges that, in order to induce Company to enter into the Franchise Agreement, Franchisee must cause certain persons owning an interest in Franchisee, or who are associated with Franchisee in an executive or similar capacity, to sign this Confidentiality, Non-Disclosure and Non-Competition Agreement ("Agreement") for the benefit of Company.

NOW, THEREFORE, the undersigned, having read this Agreement and understanding its terms, hereby agrees as follows:

1. <u>Nondisclosure of Confidential Information</u>.

a. The undersigned agrees not to disclose, duplicate, sell, reveal, divulge, publish, furnish, or communicate, either directly or indirectly, any Confidential Information (as defined below) to any other person, firm or entity, unless authorized in writing by Company.

b. The undersigned agrees not to use any Confidential Information for his or her own personal gain or to further the purposes of others, whether or not the Confidential Information has been conceived, originated, discovered or developed, in whole or in part, by the undersigned or represents the undersigned's work product. To the extent the undersigned has assisted in the preparation of any information which Company considers to be Confidential Information or has prepared or created such information by himself or herself, the undersigned hereby assigns any rights that he or she may have in such information as creator to Company, including all ideas made or conceived by the undersigned.

c. The undersigned understands and agrees that the use, publication or duplication of the Confidential Information for any purpose not authorized by this Agreement constitutes an unfair method of competition by the undersigned.

d. The provisions of this paragraph 1 shall survive the expiration or termination of all contracts between Company and Franchisee.

e. The provisions concerning non-disclosure of Confidential Information shall not apply if disclosure of Confidential Information is legally compelled in a judicial or administrative proceeding, provided the undersigned shall have used its best efforts, and shall have afforded Company the opportunity, to obtain an appropriate protective order or other assurance satisfactory to Company of confidential treatment for the information required to be disclosed. f. The term "Confidential Information" shall have the same meaning assigned to it in the Franchise Agreement. A copy of the relevant Sections of the Franchise Agreement are attached to this Agreement as Schedule 1.

2. <u>Return of Proprietary Materials</u>.

Upon expiration or termination of the Franchise Agreement, the undersigned shall surrender to Franchisee, or, if directed by Company, directly to Company, all materials in the possession of the undersigned relating to or concerning any Confidential Information. The undersigned understands and agrees that such materials shall be and remain the sole property of Company.

3. <u>Agreements Regarding Competition</u>.

a. For purposes of this Paragraph 3, the term "Competing Business" shall mean (a) any business whose primary purpose is housing or caring for persons over the age of sixty (60), or (b) any business granting franchises or licenses to others to operate the type of business specified in clause (a).

b. For as long as the Franchise Agreement shall remain in effect, the undersigned agrees that he or she shall neither directly nor indirectly, whether individually, or through, on behalf of, as a member of, or in conjunction with any person, persons, partnership, corporation, or other legal entity, operate, advise on the development or operation of, be employed in a managerial or sales capacity by, or have any interest as an owner of any business, other than as a Franchisee of the Company, that would be considered to be a Competing Business located anywhere in the world, provided, however, the restrictions stated in this paragraph shall not apply to the undersigned after twenty-four (24) months from the date that the undersigned ceases to be an officer, director, member, manager, key employee, partner or owner of an interest in the equity or voting interests in Franchisee.

c. For a period of twenty-four (24) months after expiration or termination of the Franchise Agreement, or an event of transfer as defined in the Franchise Agreement, whichever occurs first, the undersigned agrees that it, he or she shall neither directly nor indirectly, whether individually, or through, on behalf of, as a member of, or in conjunction with any person, persons, partnership, corporation, or other legal entity, operate, advise on the development or operation of, be employed in a managerial or sales capacity by, or have any interest as an owner of any business, other than as a Franchisee of the Company, that would be considered to be a Competing Business within 25 mile radius of any franchise or corporate-owned BRIGHTSTAR CARE HOMES Communities.

d. This Agreement does not prohibit the undersigned from owning 2% or less of the voting stock of a company that is in a Competing Business and whose shares are publicly traded on a national exchange.

e. The parties understand and agree that the undersigned may engage in any activities not expressly prohibited by this Agreement. However, in connection with permitted activities, the undersigned shall not (i) use the Confidential Information or any of the Licensed

Marks (as that term is defined in the Franchise Agreement); (ii) engage in any conduct or activity which suggests or implies that Company endorses, or authorizes, the undersigned's activities; or (iii) induce any person to engage in conduct prohibited by this Agreement.

4. <u>Interference</u>.

The undersigned agrees not to, directly or indirectly, for itself or on behalf of any other person, divert, or attempt to divert, any business or customer of any BRIGHTSTAR CARE HOMES Community to any competitor by direct or indirect inducement or perform any act which directly or indirectly could, or may, injure or prejudice the goodwill and reputation of the Licensed Marks or the BrightStar System.

5. <u>Irreparable Harm to Company</u>.

a. The undersigned understands and agrees that Company will suffer irreparable injury not capable of precise measurement in monetary damages if it discloses or misuses any Confidential Information or if the undersigned breaches the covenants contained in this Agreement. Accordingly, in the event of a breach of this Agreement by the undersigned, the undersigned consents to entry of interim relief, including, without limitation, the entry of a temporary restraining order, preliminary injunction, permanent injunction, writ of attachment, appointment of a receiver, and any other equitable relief which the court deems necessary in order to prevent irreparable injury, all without the requirement that bond be posted.

b. The undersigned agrees that the award of equitable remedies to Company in the event of such breach is reasonable and necessary for the protection of the business and goodwill of Company.

6. <u>Survival</u>.

The agreements made by the undersigned shall survive the expiration or termination of all contracts between Company and Franchisee.

7. <u>Validity; Conformity With Applicable Law</u>.

a. Wherever possible, each provision of this Agreement shall be interpreted in a manner as to be valid under applicable law, but if any provision of this Agreement shall be invalid or prohibited thereunder, the provision shall be ineffective only to the extent of the prohibition or invalidity without invalidating the remainder of this Agreement.

b. If any provision of Paragraph 3 is void or unenforceable under Illinois law, but would be enforceable as written or as modified under the laws of any state having jurisdiction over the undersigned (the "Local Laws"), the parties agree that the Local Laws shall govern any dispute concerning or involving the construction, interpretation, validity or enforcement of the provisions of this Agreement regarding competition, but only with respect to the subjects covered in Paragraph 3.

8. <u>Miscellaneous</u>

a. Any waiver granted to the undersigned by Company excusing or reducing any obligation or restriction imposed under this Agreement shall be evidenced by a writing signed by Company in order to be effective and shall be effective only to the extent specifically allowed in such writing. No waiver granted by Company shall constitute a continuing waiver. Any waiver granted by Company shall be without prejudice to any other rights Company may have. The rights and remedies granted to Company are cumulative. No delay on the part of Company in exercising any right or remedy shall preclude Company from fully exercising such right or remedy or any other right or remedy.

b. This Agreement sets forth the entire agreement made by the undersigned pertaining to the subject matter hereof, fully superseding any and all prior agreements or understandings that may exist between the undersigned and the Company pertaining to such subject matter. No amendment, change, modification or variance to or from the terms and conditions contained in this Agreement shall be binding on the undersigned unless it is contained in a writing and duly signed by the undersigned and Company.

c. This Agreement shall be binding on the undersigned's heirs, executors, successors and assigns as though originally signed by such persons.

d. The parties agree that all capitalized terms in this Agreement shall have the same meaning assigned to them in any Franchise Agreement between Company and Franchisee and incorporate such definitions into this Agreement.

IN WITNESS WHEREOF, the undersigned has entered into this Agreement as of the date shown above the undersigned's signature.

Signature of Individual

DATED: _____

Print Name of Individual

SCHEDULE 1 CONFIDENTIAL INFORMATION

The following Section of the Franchise Agreement shall define the term "Confidential Information" for purposes of this Agreement: Section 12

EXHIBIT E

TO BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC BRIGHTSTAR CARE HOMES COMMUNITY FRANCHISE DISCLOSURE DOCUMENT

STATE SPECIFIC ADDENDA

NO WAIVER OR DISCLAIMER OF RELIANCE IN CERTAIN STATES

The following provision applies only to franchisees and franchises that are subject to the state franchise registration/disclosure laws in California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, or Wisconsin:

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on our behalf. This provision supersedes any other term of any document executed in connection with the franchise.

ADDITIONAL DISCLOSURE DOCUMENT DISCLOSURES REQUIRED BY THE STATE OF CALIFORNIA

THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE DISCLOSURE DOCUMENT 14 DAYS PRIOR TO THE EXECUTION OF ANY AGREEMENT.

SECTION 31125 OF THE CALIFORNIA CORPORATIONS CODE REQUIRES US TO GIVE TO YOU A DISCLOSURE DOCUMENT, IN A FORM CONTAINING THE INFORMATION THAT THE COMMISSIONER MAY BY RULE OR ORDER REQUIRE, BEFORE A SOLICITATION OF A PROPOSED MATERIAL MODIFICATION OF AN EXISTING FRANCHISE.

OUR WEBSITE HAS NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION & INNOVATION. ANY COMPLAINTS CONCERNING THE CONTENT OF THIS WEBSITE MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION & INNOVATION AT www.dfpi.ca.gov.

The following risk factor is added to the page of the Franchise Disclosure Document entitled "Special Risks to Consider about *This* Franchise":

You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control.

Item 1, Additional Disclosures. The following statement is added to Item 1:

Attached as Appendix 1 to this "Additional Disclosure Document Disclosures Required by the State of California" is a list of other California laws that may apply to your franchise.

Item 3, Additional Disclosure. The following statement is added to Item 3:

Neither BrightStar Senior Living Franchising, LLC nor any person listed in Item 2 is subject to any currently-effective order of any national securities association or national securities exchange, as defined in the Securities Exchange Act of 1934, 15 U.S.C.A. 78a *et seq.*, suspending or expelling such persons from membership in such association or exchange.

Item 5, Additional Disclosure. The following paragraph is added to the end of Item 5 of the Disclosure Document:

Despite the payment provisions above, we will defer the payment of all initial fees you owe to us or our affiliates until such time as we have fulfilled all initial obligations owed to you under the Franchise Agreement or other agreements and you have commenced doing business.

Item 6, Additional Disclosure. The following statement is added to Item 6:

The highest interest rate in California is 10%.

Item 12, Additional Disclosures. The following paragraph is added to Item 12:

The Franchise Agreement gives you the right to construct, develop, and operate 3 BRIGHTSTAR CARE HOMES[®] Communities in a specific market area. If you develop all 3 BRIGHTSTAR CARE HOMES[®] Communities by the required deadlines, that market area will be exclusive to you. However, if you fail to do so, you will lose the territorial exclusivity in the market area. If that happens, we will have the right to establish and operate BRIGHTSTAR CARE HOMES[®] Communities whose premises are physically located within either your "protected territory" or your Site Selection Area, depending on which one you received under your Franchise Agreement.

Item 17, Additional Disclosures. The following statements are added to Item 17:

California Business and Professions Code Sections 20000 through 20043 provide rights to the franchisee concerning termination, transfer, or non-renewal of a franchise. If the Franchise Agreement contains a provision that is inconsistent with the law, the law will control.

The Franchise Agreement contains a covenant not to compete which extends beyond the termination of the franchise. These provisions may not be enforceable under California law.

The Franchise Agreement requires binding arbitration. The arbitration will occur in the city and state in which Franchisor's headquarters are located at the time the action is filed with the costs being borne by each party, provided, however, that the single arbitrator's fee will be shared equally by the parties.

Prospective franchisees are encouraged to consult private legal counsel to determine the applicability of California and federal laws (such as Business and Professions Code Section 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provisions of a Franchise Agreement restricting venue to a forum outside the State of California.

The Franchise Agreement provides for termination upon bankruptcy. This provision may not be enforceable under federal bankruptcy law (11 U.S.C.A § 101, <u>et</u>. <u>seq.</u>).

The Franchise Agreement contains a liquidated damages clause. Under California Civil Code Section 1671, certain liquidated damages clauses are unenforceable.

The Franchise Agreement requires application of the laws of the State of Illinois. This provision may not be enforceable under California law.

Any condition, stipulation, or provision in the Franchise Agreement which would result in your waiver of compliance with any provision of the California Franchise Relations Act is void to the extent that such provision violates such law.

You must sign a general release if you renew or transfer your franchise. California Corporations Code Section 31512 voids a waiver of your rights under the Franchise Investment Law (California Corporations Code Section 31000 – 31516). Business and

Professions Code Section 20010 voids a waiver of your rights under the Franchise Relations Act (Business and Professions Code Sections 2000 – 20043).

No disclaimer, questionnaire, clause, or statement signed by a franchisee or prospective franchisee in connection with the commencement of the franchise relationship shall alone be construed or interpreted as a waiver of any claim of fraud in the inducement, whether common law (unless allowed by existing common law) or statutory, or as alone disclaiming reliance on or the right to rely upon any statement made or information provided by any franchisor, broker or other person acting at the direction of the franchisor that was a material inducement to a franchisee's investment. This provision supersedes any other or inconsistent term of any document executed in connection with the franchise.

APPENDIX 1 TO DISCLOSURES

LAWS THAT APPLY IN THE STATE OF CALIFORNIA

Below is a list of other California laws that may apply to your franchise. We do not provide assistance in determining which specific federal, state, or local laws apply to the operation of your Community. It is critical before you purchase a franchise for the Community that you determine whether you will qualify under applicable laws to own and operate the franchise. You should also determine the initial and annual cost of compliance with those laws. We therefore strongly urge you to retain competent local counsel before you purchase a franchise for the Community. You should consult with that counsel regarding the possible application of the laws and regulations described below to the operation of your Community. You should also ask for advice on other laws and regulations that may apply to the operation of your Community.

- a. Financial Management (See Cal. Code Regs. tit. 22 §§ 47417). These provisions include requirements relating to operating budget, billing and collection, accounts receivable and payable, payroll processing, and bad debt.
 - Charge Verification. (See Cal. Code Regs. Tit. 22 §§ 47417)
 - Billing and Collections. (See Cal. Code Regs. Tit. 22 §§ 47417)
 - Accounts Receivable Review. (See Cal. Code Regs. Tit. 22 §§ 47417)
 - Bad Debt Policy. (See Cal. Code Regs. Tit. 22 §§ 47417)
 - Cash Receipts. (See Cal. Code Regs. Tit. 22 §§ 47417)
 - Purchasing Authorization and Accounts Payable. (See Cal. Code Regs. Tit. 22 §§ 47417)
 - Fixed Assets and Depreciation. (See Cal. Code Regs. Tit. 22 §§ 47417)
 - Payroll Processing. (See Cal. Code Regs. Tit. 22 §§ 47417)
 - Allocation of Time Worked. (See Cal. Code Regs. Tit. 22 §§ 47417)
 - Annual Operating Budget. (See Cal. Code Regs. Tit. 22 §§ 47417)
 - Financial Management and Control. (See Cal. Code Regs. Tit. 22 §§ 47417)
 - Fiscal Solvency. (See Cal. Code Regs. Tit. 22 §§ 47417)
 - Financial Reports. (See Cal. Code Regs. Tit. 22 §§ 47417)
 - Fee Determination. (See Cal. Code Regs. Tit. 22 §§ 47417)
- b. Referral Prohibitions (Speier Bill) (Cal. Bus. & Prof. Code § 650.01)

Note: This is similar to the federal Stark law and prohibits referrals to certain health care providers if the practitioner or immediate family member has a financial interest.

- c. California Anti-Kickback/Anti-Fee Splitting Statute (Cal. Bus. & Prof. Code § 650)
- d. Medical Referral Services (Cal. Health & Safety Code § 445)

Note: This is a criminal statute, related to Anti-Kickback Statute at Cal. Bus. & Prof. Code § 650, which prohibits anyone from referring or recommending a person, "for profit," to a physician, hospital, health-related Community, or dispensary, for "any type of medical care or treatment of any ailment or physical condition."

ADDENDUM TO THE FRANCHISE AGREEMENT FOR USE IN CALIFORNIA

This Addendum to the Franchise Agreement ("Franchise Agreement") dated ________between BrightStar Senior Living Franchising, LLC ("we," "us," or "our") and _______ ("you" or "your") is entered into simultaneously with the execution of the Franchise Agreement.

1. <u>Background</u>. The provisions of this Addendum form an integral part of, and are incorporated into, the Franchise Agreement. This Addendum is being signed because: (a) the offer or sale of the franchise for the BRIGHTSTAR CARE HOMES[®] Community you will operate under the Franchise Agreement was made in the State of California and you are a California resident, or (b) the BRIGHTSTAR CARE HOMES[®] Community will operate in California.

2. <u>Initial Fees</u>. Section 4.1 of the Franchise Agreement is amended by adding the following language to the end:

Despite the payment provisions above, we will defer the payment of all initial fees that you owe to us or our affiliates until such time as we have fulfilled all initial obligations owed to you under the Franchise Agreement and you have commenced doing business.

IN WITNESS WHEREOF, the parties have executed and delivered this Addendum as of the date set forth under our signature below.

BRIGHTSTAR SENIOR LIVING

By: ______[Signature]

FRANCHISING, LLC an Illinois limited liability company

FRANCHISEE:

Print Name of Legal Entity

By:____

[Signature]

Print Name:	
Title:	
DATED:	

Print Name:	-
Title:	
Date:	

ADDITIONAL DISCLOSURE DOCUMENT DISCLOSURES REQUIRED BY THE STATE OF ILLINOIS

1. The following is added to the end of Item 5:

Despite the payment provisions above, we will defer the payment of all initial fees that you owe us or our affiliates until such time as we have fulfilled all initial obligations owed to you under the Franchise Agreement or other agreements and you have commenced doing business. This financial assurance requirement was imposed by the Office of the Illinois Attorney General due to our financial condition.

2. The following statement is added to Item 17:

The conditions under which your franchise can be terminated and your rights upon nonrenewal may be affected by Illinois law, 815 ILCS 705/19 and 705/20.

Section 41 of the Illinois Franchise Disclosure Act states that "any condition, stipulation, or provision purporting to bind any person acquiring any franchise to waive compliance with any provision of this Act is void." To the extent that any provision in the Agreement is inconsistent with Illinois law, Illinois law will control.

Franchisees' rights upon termination and non-renewal are set forth in Sections 19 and 20 of the Illinois Franchise Disclosure Act.

In conformance with Section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation, or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void. However, this waiver shall not apply to the extent prohibited by Section 705/41 of the Illinois Franchise Disclosure Act of 1987 or Illinois Regulations at Section 260.609.

ADDENDUM TO THE FRANCHISE AGREEMENT FOR USE IN ILLINOIS

This Addendum to Franchise Agreement ("Franchise Agreement") dated ______ between BrightStar Senior Living Franchising, LLC ("we," "us," or "our") and _______ ("you" or "your") is entered into simultaneously with the execution of the Franchise Agreement.

1. **Background**. The provisions of this Addendum form an integral part of, and are incorporated into, the Franchise Agreement. This Addendum is being signed because: (a) you are a resident of the State of Illinois; or (b) the franchise offer and sale activity occurred in Illinois and the BRIGHTSTAR CARE HOMES[®] Community will be located in Illinois.

2. <u>Initial Fees</u>. Section 4.1 of the Franchise Agreement is amended by adding the following language to the end:

Despite the payment provisions above, we will defer the payment of all initial fees that you owe to us or our affiliates until such time as we have fulfilled all initial obligations owed to you under the Franchise Agreement and you have commenced doing business. This financial assurance requirement was imposed by the Office of the Illinois Attorney General due to our financial condition.

3. <u>Illinois Franchise Disclosure Act</u>. The following language is added as a new Section 16.11 of the Franchise Agreement:

16.11. ILLINOIS FRANCHISE DISCLOSURE ACT.

In conformance with Section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation, or provision purporting to bind any person acquiring any franchise to waive compliance with any provision of the Act or any other law of Illinois is void. However, that Section shall not prevent any person from entering into a settlement agreement or executing a general release regarding a potential or actual lawsuit filed under any provision of the Act, nor shall it prevent the arbitration of any claim pursuant to the provisions of Title 9 of the United States Code.

IN WITNESS WHEREOF, the parties have executed and delivered this Addendum as of the date set forth under our signature below.

[Signature Page Follows]

BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC an Illinois limited

liability company

By: ______[Signature]

Print Name:	
Title:	
DATED:	

FRANCHISEE:

Print Name of Legal Entity

By:_____

[Signature]

Print Name:	
Title:	
Date:	

ADDITIONAL DISCLOSURE DOCUMENT DISCLOSURES REQUIRED BY THE STATE OF MARYLAND

1. The following is added to the end of Items 5 and 7:

Based upon our financial condition, the Maryland Securities Commissioner has required a financial assurance. Therefore, all initial fees and payments owed by you shall be deferred until we complete our pre-opening obligations under the Franchise Agreement and the BRIGHTSTAR CARE HOMES[®] Community has opened for business.

2. The "Summary" sections of Items 17(c) and (m) in the Franchise Disclosure Document, captioned "Requirements for you to renew or extend" and "Conditions for our approval of transfer," are amended by adding the following:

The general release required as a condition of renewal, sale, and/or assignment/transfer will not apply to any liability under the Maryland Franchise Registration and Disclosure Law.

3. The "Summary" section of Item 17(h) in the Franchise Disclosure Document, captioned "Cause' defined – non-curable defaults," is amended by adding the following:

The Agreement provides for termination upon your insolvency. This provision might not be enforceable under federal bankruptcy law (11 U.S.C. Sections 101 *et seq.*), but we will enforce it to the extent enforceable.

4. The "Summary" section of Item 17(v) in the Franchise Disclosure Document, captioned "Choice of forum," is amended to read as follows:

Subject to arbitration requirements, litigation generally must be in courts located closest to where we have our principal business address when the action is commenced (it currently is in Illinois), although you may, subject to your arbitration obligations, commence a lawsuit against us in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law.

5. The "Summary" section of Item 17(w) in the Franchise Disclosure Document, captioned "Choice of law," is amended to read as follows:

Except for federal law and claims arising under the Maryland Franchise Registration and Disclosure Law, Illinois law applies.

6. The following is added to the end of Item 17 of the Franchise Disclosure Document:

Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the

effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on our behalf. This provision supersedes any other term of any document executed in connection with the franchise.

ADDENDUM TO THE FRANCHISE AGREEMENT FOR USE IN MARYLAND

This Addendum to Franchise Agreement ("Franchise Agreement") dated ______ between BrightStar Senior Living Franchising, LLC ("we," "us," or "our") and _______ ("you" or "your") is entered into simultaneously with the execution of the Franchise Agreement.

1. **Background**. The provisions of this Addendum form an integral part of, and are incorporated into, the Franchise Agreement. This Addendum is being signed because: (a) you are a resident of the State of Maryland; or (b) the BRIGHTSTAR CARE HOMES[®] Community will be located in Maryland.

2. <u>**Releases**</u>. Sections 2.2.3, 10.1, 13.4.1.9, 13.6, and 15.3.5 of the Franchise Agreement are amended by adding the following:

The general release required as a condition of renewal, sale, and/or assignment/transfer will not apply to any liability under the Maryland Franchise Registration and Disclosure Law.

3. <u>Initial Fees</u>. Section 4.1 of the Franchise Agreement is amended by adding the following language to the end:

Based upon our financial condition, the Maryland Securities Commissioner has required a financial assurance. Therefore, all initial fees and payments that you owe shall be deferred until we complete our pre-opening obligations under the Franchise Agreement and the BRIGHTSTAR CARE HOMES[®] Community has opened for business.

4. **Termination**. Section 14.1 of the Franchise Agreement is amended by adding the following:

Termination upon bankruptcy might not be enforceable under federal bankruptcy law (11 U.S.C. Sections 101 et seq.), but we intend to enforce this provision to the extent enforceable.

5. <u>Consent to Jurisdiction</u>. Section 16.7 of the Franchise Agreement is amended by adding the following language:

However, subject to your arbitration obligations, nothing in this Section affects your right under the Maryland Franchise Registration and Disclosure Law to bring a lawsuit in Maryland for claims arising under that law.

6. <u>Limitation of Claims</u>. Section 16.8 of the Franchise Agreement is amended by adding the following language:

Any limitation of claims will not act to reduce the three (3) year statute of limitations afforded you for bringing a claim under the Maryland Franchise Registration and Disclosure Law.

7. <u>Non-Waiver</u>. The following language is added to the end of Section 16.9 of the Franchise Agreement:

All representations requiring prospective franchisees to assent to a release, estoppel or waiver of liability are not intended to nor shall they act as a release, estoppel, or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law.

8. <u>Governing Law</u>. Section 23 of the Franchise Agreement is amended by adding the following language:

Despite anything to the contrary stated above, Maryland law will apply to claims arising under the Maryland Franchise Registration and Disclosure Law.

IN WITNESS WHEREOF, the parties have executed and delivered this Addendum as of the date set forth under our signature below.

BRIGHTSTAR SENIOR LIVING

FRANCHISING, LLC an Illinois limited liability company

FRANCHISEE:

Print Name of Legal Entity

By:	By:
[Signature]	[Signature]
-	Print Name:
Print Name:	Title:
Title:	Date:
DATED:	

ADDITIONAL DISCLOSURE DOCUMENT DISCLOSURES REQUIRED BY THE STATE OF MINNESOTA

1. The following paragraph is added to the end of Item 5 of the Franchise Disclosure Document:

Despite the payment provisions above, we will defer the payment of all initial fees that you owe to us or our affiliates until such time as we have fulfilled all initial obligations owed to you under the Franchise Agreement or other agreements and you have commenced doing business.

2. The following statement is added to Item 13 of the Franchise Disclosure Document:

We will protect your right to use the Marks or indemnify you from any loss, costs, or expenses arising out of any claim, suit, or demand regarding the use of the name to the extent required by Minn. Stat. Sec. 80C.12, Subd.1(g).

3. The following paragraphs are added to the end of Item 17 of the Franchise Disclosure Document:

With respect to franchises governed by Minnesota law, we will comply with Minnesota Statute § 80C.14, subdivisions 3, 4, and 5 which require, except in certain specified cases, that you be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice for non-renewal of the Franchise Agreement.

Minnesota Statutes, Section 80C.21 and Minnesota Rule 2860.4400J prohibit us from requiring litigation to be conducted outside Minnesota, requiring waiver of a jury trial, or requiring you to consent to liquidated damages, termination penalties, or judgment notes. In addition, nothing in the Franchise Disclosure Document or agreement(s) can abrogate or reduce any of your rights as provided for in Minnesota Statutes, Chapter 80C, or your rights to any procedure, forum, or remedies provided for by the laws of the jurisdiction.

Minnesota Rule 2860.4400D prohibits us from requiring you to assent to a release, assignment, novation, or waiver that would relieve any person from liability imposed by Minnesota Statute \$

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on our behalf. This provision supersedes any other term of any document executed in connection with the franchise.

ADDENDUM TO THE FRANCHISE AGREEMENT FOR USE IN MINNESOTA

This Addendum to the Franchise Agreement ("Franchise Agreement") dated _______between BrightStar Senior Living Franchising, LLC ("we," "us," or "our") and ______ ("you" or "your") is entered into simultaneously with the execution of the Franchise Agreement.

1. **Background**. The provisions of this Addendum form an integral part of, and are incorporated into, the Franchise Agreement. This addendum is being signed because: (a) the franchise offer or sale activity occurred in Minnesota; or (b) the BRIGHTSTAR CARE HOMES[®] Community will be located in Minnesota.

2. **Renewal and Termination**. The following sentence is added to the end of Sections 2.1 and 14:

With respect to franchises governed by Minnesota law, we will comply with Minnesota Statute § 80C.14, subdivision 3, 4, and 5 which require, except in certain cases, that you be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice for non-renewal of the Franchise Agreement.

3. **<u>Releases</u>**. The following sentence is added to the end of Sections 2.2 and 13.4:

Notwithstanding the foregoing, you will not be required to assent to a release, assignment, novation, or waiver that would relieve any person from liability imposed by Minnesota Statute \$ 80C.01 – 80C.22.

4. **Initial Fees**. Section 4.1 is amended by adding the following language to the end:

Despite the payment provisions above, we will defer the payment of all initial fees that you owe to us or our affiliates until such time as we have fulfilled all initial obligations owed to you under the Franchise Agreement and you have commenced doing business.

5. <u>Governing Law, Jurisdiction, and Venue</u>. The following sentences are added to the end of Sections 16.3 and 16.7:

Minnesota Statute § 80C.21 and Minnesota Rule 2860.4400J prohibit us from requiring arbitration or litigation to be conducted outside Minnesota. In addition, nothing in the Disclosure Document or the Franchise Agreement can abrogate or reduce any of your rights as provided for in Minnesota Statutes, Chapter 80C, or your rights to any procedure, forum, or remedies provided for by the laws of the jurisdiction.

6. **Limitations of Actions.** The following is added to the end of Section 16.8 of the Franchise Agreement:

Minnesota law provides that no action may be commenced under Minn. Stat. Sec. 80C.17 more than 3 years after the cause of action accrues.

IN WITNESS WHEREOF, the parties have executed and delivered this Addendum as of the date set forth under our signature below.

BRIGHTSTAR SENIOR LIVING

FRANCHISEE:

FRANCHISING, LLC an Illinois limited liability company

Print Name of Legal Entity

By: ______[Signature]

Print Name:	
Title:	
DATED:	

By:	
-	[Signature]
Print Name: _	
Title:	
Date:	

BRIGHTSTAR CARE HOMES FDD (FINAL 2024) ACTIVE\1606838690.2

ADDITIONAL DISCLOSURE DOCUMENT DISCLOSURES REQUIRED BY THE STATE OF NEW YORK

1. The following information is added to the cover page of the Franchise Disclosure Document:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN EXHIBIT A OR YOUR PUBLIC LIBRARY FOR SOURCES OF INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN THAT ANYTHING IN THE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND NEW YORK STATE DEPARTMENT OF LAW, BUREAU OF INVESTOR PROTECTION AND SECURITIES, 28 LIBERTY STREET, 21ST FLOOR, NEW YORK, NEW YORK 10005.

THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, THE FRANCHISOR CANNOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A PROSPECTIVE FRANCHISEE TO ACCEPT TERMS WHICH ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

2. The following is added to the end of Item 3:

Except as provided above, with regard to the franchisor, its predecessor, a person identified in Item 2, or an affiliate offering franchises under the franchisor's principal trademark:

- A. No such party has an administrative, criminal or civil action pending against that person alleging: a felony, a violation of a franchise, antitrust or securities law, fraud, embezzlement, fraudulent conversion, misappropriation of property, unfair or deceptive practices, or comparable civil or misdemeanor allegations.
- B. No such party has pending actions, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature or financial condition of the franchise system or its business operations.
- C. No such party has been convicted of a felony or pleaded nolo contendere to a felony charge or, within the 10-year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antifraud, or securities law; fraud;

embezzlement; fraudulent conversion or misappropriation of property; or unfair or deceptive practices or comparable allegations.

- D. No such party is subject to a currently effective injunctive or restrictive order or decree relating to the franchise, or under a Federal, State, or Canadian franchise, securities, antitrust, trade regulation or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent.
- 3. The following is added to the end of Item 4:

Neither the franchisor, its affiliate, its predecessor, officers, or general partner during the 10-year period immediately before the date of the offering circular:

- (a) filed as debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code;
- (b) obtained a discharge of its debts under the U.S. Bankruptcy Code; or
- (c) was a principal officer of a company or a general partner in a partnership that either filed as a debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code or that obtained a discharge of its debts under the U.S. Bankruptcy Code during or within 1 year after that officer or general partner of the franchisor held this position in the company or partnership.
- 4. The following is added to the end of Item 5:

The initial franchise fee constitutes part of our general operating funds and will be used as such in our discretion.

5. The following is added to the end of the "Summary" sections of Item 17(c), titled "Requirements for franchisee to renew or extend," and Item 17(m), titled "Conditions for franchisor approval of transfer":

However, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force; it being the intent of this proviso that the

non-waiver provisions of General Business Law Sections 687.4 and 687.5 be satisfied.

6. The following replaces the "Summary" section of Item 17(d), titled "Termination by franchisee":

You may terminate the agreement on any grounds available by law.

7. The following is added to the end of the "Summary" section of Item 17(j), titled "Assignment of contract by franchisor":

However, no assignment will be made except to an assignee who in good faith and judgment of the franchisor, is willing and financially able to assume the franchisor's obligations under the Franchise Agreement.

8. The following is added to the end of the "Summary" sections of Item 17(v), titled "Choice of forum," and Item 17(w), titled "Choice of Law":

The foregoing choice of law should not be considered a waiver of any right conferred upon the franchisor or upon the franchisee by Article 33 of the General Business Law of the State of New York.

ADDENDUM TO THE FRANCHISE AGREEMENT FOR USE IN NEW YORK

This Addendum to the Franchise Agreement ("Franchise Agreement") dated _______between BrightStar Senior Living Franchising, LLC ("we," "us," or "our") and ______ ("you" or "your") is entered into simultaneously with the execution of the Franchise Agreement.

1. **Background**. The provisions of this Addendum form an integral part of, and are incorporated into, the Franchise Agreement. This Addendum is being signed because: (a) the offer or sale of the franchise was made in the State of New York; or (b) you are a resident of New York and the BRIGHTSTAR CARE HOMES[®] Community will be located or operated in New York.

2. <u>**Releases**</u>. The following sentence is added to the end of Sections 2.2.3, 10.1, 13.4.1.9, 13.6, and 15.3.5:

, provided, however, that all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force; it being the intent of the proviso that the non-waiver provisions of GBL 687 and 687.5 be satisfied.

3. **<u>Transfer</u>**. The following sentence is added to the end of Section 13.1:

We will not assign our rights under the Franchise Agreement except to an assignee who in our good faith and judgment is willing and able to assume our obligations under the Franchise Agreement.

4. <u>Termination by You</u>. Section 14.7 of the Franchise Agreement is amended by adding the following as the last sentence:

You also may terminate this Agreement on any grounds available by law under the provisions of Article 33 of the General Business Law of the State of New York.

5. **<u>Governing Law</u>**. The following sentence is added to the end of Section 23:

HOWEVER, THE GOVERNING CHOICE OF LAW SHALL NOT BE CONSIDERED A WAIVER OF ANY RIGHT CONFERRED UPON YOU BY THE PROVISIONS OF ARTICLE 33 OF THE GENERAL BUSINESS LAW OF THE STATE OF NEW YORK.

6. **<u>Binding Effect</u>**. Section 25 of the Franchise Agreement is amended by adding the following language:

Modifications to the Operations Manual will not unreasonably affect your obligations, including economic requirements, under this Agreement.

IN WITNESS WHEREOF, the parties have executed and delivered this Addendum as of the date set forth under our signature below.

ADDITIONAL DISCLOSURE DOCUMENT DISCLOSURES REQUIRED BY THE STATE OF NORTH DAKOTA

1. The following language is added as the last paragraph of Items 5 and 7:

Despite the payment provisions above, we will defer your payment of the initial franchise fee due to us under the Franchise Agreement until we have fulfilled all of our initial obligations to you under this Agreement and you have commenced operating the BRIGHTSTAR CARE HOMES[®] Community. You must pay us the initial franchise fee on the day you begin operating your BRIGHTSTAR CARE HOMES[®] Community.

2. The "Summary" sections of Items 17(c) and 17(m) are amended by adding the following:

However, any release required as a condition of renewal and/or assignment/transfer will not apply to the extent prohibited by the North Dakota Franchise Investment Law.

3. The "Summary" section of Item 17(i) is amended by adding the following:

The Commissioner has determined termination or liquidated damages to be unfair, unjust, and inequitable within the intent of Section 51-19-09 of the North Dakota Franchise Investment Law. However, we and you agree to enforce these provisions to the extent the law allows.

4. The "Summary" section of Item 17(r) is amended by adding the following:

Covenants not to compete such as those mentioned above generally are considered unenforceable in North Dakota. However, we will seek to enforce them to the extent enforceable.

5. The "Summary" section of Item 17(v) is amended by adding the following:

To the extent required by the North Dakota Franchise Investment Law, but subject to your arbitration obligations, you may bring an action in North Dakota.

6. The "Summary" section of Item 17(w) is amended by adding the following:

Except for federal law, North Dakota law applies.

7. The following paragraph is added to the end of Item 17:

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on our behalf. This provision supersedes any other term of any document executed in connection with the franchise.

ADDENDUM TO THE FRANCHISE AGREEMENT FOR USE IN NORTH DAKOTA

This Addendum to the Franchise Agreement ("Franchise Agreement") dated ________ between BrightStar Senior Living Franchising, LLC ("we," "us," or "our") and _______ ("you" or "your") is entered into simultaneously with the execution of the Franchise Agreement.

1. **Background.** The provisions of this Addendum form an integral part of, and are incorporated into, the Franchise Agreement. This addendum is being signed because (a) you are a resident of the State of North Dakota and the BRIGHTSTAR CARE HOMES[®] Community you will operate under the Franchise Agreement will be located or operated in North Dakota, and/or (b) any of the franchise offer or sales activity occurred in North Dakota.

2. **<u>Releases</u>**. The following language is added to the end of Sections 2.2.3, 10.1, 13.4.1.9, 13.6, and 15.3.5 of the Franchise Agreement:

Any release executed will not apply to the extent otherwise prohibited by applicable law with respect to claims arising under the North Dakota Franchise Investment Law.

3. **Initial Fees**. Section 4.1 of the Franchise Agreement is amended by adding the following language to the end:

Despite the payment provisions above, we will defer your payment of the initial franchise fee due to us under the Franchise Agreement until we have fulfilled all of our initial obligations to you under this Agreement and you have commenced operating the BRIGHTSTAR CARE HOMES[®] Community. You must pay us the initial franchise fee on the day you begin operating your BRIGHTSTAR CARE HOMES[®] Community.

4. <u>Covenant Not to Compete</u>. Section 12.4 of the Franchise Agreement is amended by adding the following:

Covenants not to compete such as those mentioned above are generally considered unenforceable in the State of North Dakota. However, you acknowledge and agree that we intend to seek enforcement of these provisions to the extent allowed under the law.

5. <u>Arbitration</u>. The third sentence of Section 16.3 of the Franchise Agreement is amended to read as follows:

All proceedings, including the hearing, will be conducted at a suitable location that is within ten (10) miles of where we have our principal business address when the arbitration demand is filed, provided, however, that to the extent required by the North Dakota Franchise Investment Law (unless such a requirement is preempted by the Federal Arbitration Act), arbitration proceedings will be held at a site to which we and you agree. 6. <u>Consent to Jurisdiction</u>. The following language is added to the end of Section 16.7 of the Franchise Agreement:

However, to the extent required by applicable law, but subject to your arbitration obligations, you may bring an action in North Dakota.

7. **Limitation of Claims.** The following sentence is added to the end of Section 16.8 of the Franchise Agreement:

The statutes of limitations under North Dakota law apply with respect to claims arising under the North Dakota Franchise Investment Law.

8. <u>Waiver of Punitive Damages and Jury Trial</u>. If and then only to the extent required by the North Dakota Franchise Investment Law, Sections 16.9 and 16.10 of the Franchise Agreement are deleted in their entirety.

9. <u>Governing Law</u>. The following language is added to the end of Section 23 of the Franchise Agreement:

Notwithstanding the foregoing, to the extent required by the North Dakota Franchise Investment Law, North Dakota law will apply to this Agreement.

IN WITNESS WHEREOF, the parties have executed and delivered this Addendum as of the date set forth under our signature below.

BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC an Illinois limited liability company

FRANCHISEE:

Print Name of Legal Entity

By:		
•	[Signature]	

Print Name:	U			
Title:				
DATED:				

By:	
•	[Signature]

Print Name:	:
Title:	
Date:	

ADDITIONAL DISCLOSURE DOCUMENT DISCLOSURES REQUIRED BY THE STATE OF RHODE ISLAND

1. The following language is added to the end of the "Summary" section of Item 17(v), entitled <u>Choice of forum</u>:

Subject to arbitration requirements, litigation generally must be where we have our principal business address at the time the action is commenced (it currently is in Chicago, Illinois), except that, to the extent required by the Rhode Island Franchise Investment Act, you may bring an action in Rhode Island.

2. The following language is added to the end of the "Summary" section of Item 17(w), entitled <u>Choice of law</u>:

Except for Federal Arbitration Act and other federal law, and except as otherwise required by the Rhode Island Franchise Investment Act, Illinois law applies.

ADDENDUM TO THE FRANCHISE AGREEMENT FOR USE IN RHODE ISLAND

This Addendum to the Franchise Agreement ("Franchise Agreement") dated _____ between BrightStar Senior Living Franchising, LLC ("we," "us," or "our") and ______ ("you" or "your") is entered into simultaneously with the execution of the Franchise Agreement.

1. **Background**. The provisions of this Addendum form an integral part of, and are incorporated into, the Franchise Agreement. This addendum is being signed because (a) you are a resident of Rhode Island and the BRIGHTSTAR CARE HOMES® Community that you will operate under the Franchise Agreement will be located in Rhode Island; and/or (b) any of the franchise offer or sales activity occurred in Rhode Island.

2. **Consent to Jurisdiction/Governing Law**. The following language is added to the end of Sections 16.7 and 23 of the Franchise Agreement:

Section 19-28.1-14 of the Rhode Island Franchise Investment Act provides that "A provision in a franchise agreement restricting jurisdiction or venue to a forum outside this state or requiring the application of the laws of another state is void with respect to a claim otherwise enforceable under this Act." To the extent required by applicable law, Rhode Island law will apply to claims arising under the Rhode Island Franchise Investment Act.

IN WITNESS WHEREOF, the parties have executed and delivered this Addendum as of the date set forth under our signature below.

BRIGHTSTAR SENIOR LIVING

FRANCHISING, LLC an Illinois limited liability company

FRANCHISEE:

Print Name of Legal Entity

By: ______[Signature] Print Name: Title: _____ DATED:

By:_____[Signature]

Print Name:	
Title:	
Date:	

ADDITIONAL DISCLOSURE DOCUMENT DISCLOSURES REQUIRED BY THE STATE OF SOUTH DAKOTA

1. The following language is added as the last paragraph of Items 5 and 7:

Despite the payment provisions above, we will defer your payment of the initial franchise fee due to us under the Franchise Agreement until we have fulfilled all of our initial obligations to you under this Agreement and you have commenced operating the BRIGHTSTAR CARE HOMES[®] Community. You must pay us the initial franchise fee on the day you begin operating your BRIGHTSTAR CARE HOMES[®] Community.

ADDENDUM TO THE FRANCHISE AGREEMENT FOR USE IN SOUTH DAKOTA

This Addendum to the Franchise Agreement ("Franchise Agreement") dated _____ between BrightStar Senior Living Franchising, LLC ("we," "us," or "our") and ______ ("you" or "your") is entered into simultaneously with the execution of the Franchise Agreement.

1. **Background.** The provisions of this Addendum form an integral part of, and are incorporated into, the Franchise Agreement. This Addendum is being signed because you will operate the BRIGHTSTAR CARE HOMES® Community in South Dakota.

2. Initial Fees. Section 4.1 of the Franchise Agreement is amended by adding the following language to the end:

Despite the payment provisions above, we will defer your payment of the initial franchise fee due to us under the Franchise Agreement until we have fulfilled all of our initial obligations to you under this Agreement and you have commenced operating the BRIGHTSTAR CARE HOMES[®] Community. You must pay us the initial franchise fee on the day you begin operating your BRIGHTSTAR CARE HOMES[®] Community.

IN WITNESS WHEREOF, the parties have executed and delivered this Addendum as of the date set forth under our signature below.

BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC an Illinois limited liability company

FRANCHISEE:

Print Name of Legal Entity

By:	
-	

[Signature] Print Name: Title: DATED:

	[Signature]
Print Name:	
Title:	
Date:	

By:

ADDITIONAL DISCLOSURE DOCUMENT DISCLOSURES REQUIRED BY THE STATE OF VIRGINIA

1. The following risk factor is added to the page of the Franchise Disclosure Document titled "Special Risks to Consider about *This* Franchise":

Estimated Initial Investment. You will be required to make an estimated initial investment ranging from \$1,225,916 to \$1,966,720 (if you intend to buy the Community's land and building) and \$201,308 to \$329,850 (if you intend to lease the Community's land and building). This amount exceeds the franchisor's stockholders' negative equity as of January 1, 2023, which is (\$2,804,765).

2. <u>Item 5, Additional Disclosure</u>. The following paragraph is added to the end of Item 5 of the Disclosure Document:

The Virginia State Corporation Commission's Division of Securities and Retail Franchising requires us to defer payment of the initial franchise fee and other initial payments owed by franchisees to us until we have completed our pre-opening obligations under the Franchise Agreement.

3. <u>Item 17, Additional Disclosure</u>. In recognition of the restrictions contained in Section 13.1-564 of the Virginia Retail Franchising Act, the Franchise Disclosure Document for BrightStar Senior Living Franchising, LLC for use in the Commonwealth of Virginia shall be amended as follows:

"According to Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any ground for default or termination stated in the Franchise Agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable."

4. The following is added to the end of Item 17 of the Franchise Disclosure Document:

No statement, questionnaire, or acknowledgement signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on our behalf. This provision supersedes any other term of any document executed in connection with the franchise.

ADDENDUM TO THE FRANCHISE AGREEMENT FOR USE IN VIRGINIA

1. <u>Background</u>. The provisions of this Addendum form an integral part of, and are incorporated into, the Franchise Agreement. This Addendum is being signed because the BRIGHTSTAR CARE HOMES[®] Community that you will operate under the Franchise Agreement will be located in Virginia.

2. <u>Initial Fees</u>. Section 4.1 of the Franchise Agreement is amended by adding the following language to the end:

The Virginia State Corporation Commission's Division of Securities and Retail Franchising requires us to defer payment of the initial franchise fee and other initial payments owed by you to us until we have completed our pre-opening obligations under the Franchise Agreement.

IN WITNESS WHEREOF, the parties have executed and delivered this Addendum as of the date set forth under our signature below.

BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC an Illinois limited liability company

FRANCHISEE:

Print Name of Legal Entity

By:_____

[Signature]

Print Name: ______ Title: ______ DATED: _____

By:		
-	[Signature]	
Print Name:		

Title:	
Date:	

WASHINGTON ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT, FRANCHISE AGREEMENT, FRANCHISE REPRESENTATIONS, AND RELATED AGREEMENTS

This Addendum to the Franchise Agreement ("Franchise Agreement") dated _______between BrightStar Senior Living Franchising, LLC ("we," "us," or "our") and _______ ("you" or "your") is entered into simultaneously with the execution of the Franchise Agreement.

1. **Background**. The provisions of this Addendum form an integral part of, and are incorporated into, the Franchise Agreement. This Addendum is being signed because: (a) you are domiciled in Washington; and/or (b) the BRIGHTSTAR CARE HOMES[®] Community you will operate under the Franchise Agreement will be located or operated in Washington; and/or (c) any of the franchise offer or sales activity occurred in Washington.

2. <u>Initial Fees</u>. Section 4.1 of the Franchise Agreement is amended by adding the following language to the end:

Despite the payment provisions above, we will defer your payment of the initial franchise fee due to us under the Franchise Agreement until we have fulfilled all of our initial obligations to you under this Agreement and you have commenced operating the BRIGHTSTAR CARE HOMES[®] Community. You must pay us the initial franchise fee on the day you begin operating your BRIGHTSTAR CARE HOMES[®] Community.

3. **Entire Agreement.** The first paragraph of Section 25 is amended to read as follows:

The Recitals and Exhibits are a part of this Agreement which, together with System standards contained in the Operations Manual (which we may periodically modify, as provided in this Agreement) or otherwise communicated to you, constitutes our and your entire agreement, and there are no other oral or written understandings or agreements between the parties, and no oral or written representations by us, relating to the subject matter of this Agreement, the franchise relationship, or a Community. However, nothing in this Agreement or any related agreement is intended to disclaim our representations in our Franchise Disclosure Document.

4. <u>Washington Law</u>. The following paragraphs are added to the end of the Franchise Agreement:

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.

RCW 19.100.180 may supersede the Franchise Agreement in your relationship with us, including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the Franchise Agreement in your relationship with us, including the areas of termination and renewal of your franchise.

In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the Franchise Agreement, you may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

A release or waiver of rights executed by a franchisee may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.

Transfer fees are collectable to the extent that they reflect our reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the Franchise Agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.

RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the Franchise Agreement or elsewhere are void and unenforceable in Washington.

[Signature page follows]

IN WITNESS WHEREOF, the parties have executed and delivered this Addendum as of the date set forth under our signature below.

BRIGHTSTAR SENIOR LIVING
FRANCHISING, LLC an Illinois limited
liability company

FRANCHISEE:

By: ______[Signature]
Print Name: ______
Title: _____
DATED: _____

By:	
	[Signature]
Print Name:	
Title:	
Date:	

EXHIBIT F

TO BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC BRIGHTSTAR CARE HOMES COMMUNITY FRANCHISE DISCLOSURE DOCUMENT

FINANCIAL STATEMENTS

Financial Statements Years Ended December 31, 2023 and January 1, 2023

The report accompanying these financial statements was issued by BDO USA, P.C., a Virginia professional corporation, and the U.S. member of BDO International Limited, a UK company limited by guarantee.



Financial Statements Years Ended December 31, 2023 and January 1, 2023

Contents

Independent Auditor's Report	3-4
Financial Statements	
Balance Sheets as of December 31, 2023 and January 1, 2023	5
Statements of Operations for the Years Ended December 31, 2023 and January 1, 2023	6
Statements of Member's Deficit for the Years Ended December 31, 2023 and January 1, 2023	7
Statements of Cash Flows for the Years Ended December 31, 2023 and January 1, 2023	8
Notes to Financial Statements	9-14



Tel: 312-856-9100 Fax: 312-856-1379 www.bdo.com

Independent Auditor's Report

BrightStar Senior Living Franchising, LLC Bannockburn, Illinois

Opinion

We have audited the financial statements of BrightStar Senior Living Franchising, LLC (the Company), which comprise the balance sheets as of December 31, 2023 and January 1, 2023, and the related statements of operations, member's deficit, and cash flows for the years then ended, and the related notes to the financial statements.

In our opinion, the accompanying financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2023 and January 1 2023, and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audits in accordance with auditing standards generally accepted in the United States of America (GAAS). Our responsibilities under those standards are further described in the Auditor's Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities, in accordance with the relevant ethical requirements relating to our audits. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Emphasis of Matter

As described in Note 2 to the financial statements, the Parent Company, BrightStar Group Holdings, Inc., has agreed to support the operating, investing, and financing activities of the Company through at least one year and a day beyond the report date. Additionally, the Parent and affiliated companies (see Note 3) will not call their notes through at least one year and a day beyond the report date. Our opinion is not modified with respect to these matters.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern within one year after the date that the financial statements are available to be issued.

BDO USA, P.C., a Virginia professional corporation, is the U.S. member of BDO International Limited, a UK company limited by guarantee, and forms part of the international BDO network of independent member firms.



Auditor's Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the financial statements.

In performing an audit in accordance with GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the financial statements, whether due to fraud or error, and design and perform audit procedures responsive to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control-related matters that we identified during the audit.

BDO USA, P.C.

April 29, 2024

Balance Sheets

	D	ecember 31, 2023	January 1, 2023
Assets			
Current Assets Cash and cash equivalents Accounts receivable, net Prepaid expenses	\$	184,103 85,323 3,798	\$ 1,890 101,779 6,790
Total Current Assets		273,224	110,459
Long-Term Assets Equipment, net		1,057	2,806
Total Assets	\$	274,281	\$ 113,265
Liabilities and Member's Deficit			
Current Liabilities Accrued salaries and payroll tax Other current liabilities	\$	59,713 20,642	\$ 79,925 36,637
Total Current Liabilities		80,355	116,562
Long-Term Liabilities Deferred revenue (non-refundable) Notes payable, affiliated companies, net		264,934 2,882,929	244,933 2,556,535
Total Long-Term Liabilities		3,147,863	2,801,468
Total Liabilities		3,228,218	2,918,030
Member's Deficit		(2,953,937)	(2,804,765)
Total Liabilities and Member's Deficit	\$	274,281	\$ 113,265

Statements of Operations

Year ended	December 31, 2023	January 1, 2023	
Net Revenues			
Royalty fees	\$ 469,667 \$	357,302	
General marketing fund fees	72,476	71,460	
System fees	27,693	25,500	
Other	400	3,384	
Total Net Revenues	570,236	457,646	
Expenses			
Payroll and related expenses	769,614	525,490	
Broker fees	-	32,000	
Professional fees	232,536	204,927	
Marketing	15,786	2,567	
Travel and meals	60,888	38,815	
IT services	6,767	20,205	
Occupancy and office expense	7,309	7,854	
Depreciation	1,750	1,854	
Miscellaneous expenses	3,787	5,118	
Total Expenses	1,098,437	838,830	
Net Loss	\$ (528,201) \$	(381,184)	

Statements of Member's Deficit

	Member's Deficit	
Balance, January 2, 2022	\$ (2,423,581)	
Net loss	(381,184)	
Balance, January 1, 2023	(2,804,765)	
Contributions	379,029	
Net loss	(528,201)	
Balance, December 31, 2023	\$ (2,953,937)	

Statements of Cash Flows

Year ended		ecember 31, 2023	January 1, 2023	
Cash Flows from Operating Activities Net loss Adjustments to reconcile net loss to net cash	\$	(528,201) \$	(381,184)	
used in operating activities: Depreciation expense Decrease (increase) in:		1,750	1,854	
Accounts receivable Prepaid expenses Increase (decrease) in:		16,456 2,992	(65,317) 2,747	
Other current liabilities Deferred revenue (non-refundable) Accrued salaries and payroll tax		(15,995) 20,001 (20,212)	28,007 75,000 30,777	
Net Cash Used in Operating Activities		(523,209)	(308,116)	
Cash Flows from Financing Activities Contributions Notes payable, net, affiliated companies		379,029 326,393	- (131,818)	
Net Cash Provided by (Used in) Financing Activities		705,422	(131,818)	
Net Increase (Decrease) in Cash and Cash Equivalents		182,213	(439,934)	
Cash and Cash Equivalents, beginning of year		1,890	441,824	
Cash and Cash Equivalents, end of year	\$	184,103 \$	1,890	

1. Description of Business and Operations

BrightStar Senior Living Franchising, LLC (the Company) is a member of a consolidated group whose parent is BrightStar Group Holdings, Inc. (the Parent), which consists of 14 additional members-BrightStar Technology Group, LLC; 24-7 BrightStar Operations, LLC; 24-7 BrightStar Healthcare, LLC (BrightStar Nevada); BrightStar Franchising, LLC; BrightStar Senior Living, LLC; BrightStar CRDM, LLC; BrightStar Senior Living Development of Mason, LLC and Brightstar Senior Living Operations of Mason, LLC; BrightStar Owned LLC; BrightStar Scottsdale Operations, LLC; BrightStar FL/AL Operations, LLC; BrightStar Spartanburg/Greenville Operations, LLC; BrightStar TN Operations, LLC: and BrightStar Wisconsin Operations, LLC. The Company is organized as a limited liability company under the laws of the state of Illinois and it is a franchiser of businesses that has developed a system for constructing, operating, identifying, and promoting assisted living and memory care facilities under the BRIGHTSTAR SENIOR LIVING® trademark and related commercial symbols. These facilities provide various services for seniors and other individuals in need, including residential assisted living, memory care, and ancillary assisted living services and are marketed as BrightStar Senior Living Communities. The Company also operates a smaller residential care home model for assisted living for seniors and other individuals in need that is marketed as BrightStar Care Homes Communities.

The Company's fiscal year ends on the nearest Sunday to December 31. There were 52 weeks included in each of the years ended December 31, 2023 and January 1, 2023.

2. Summary of Significant Accounting Policies

Basis of Presentation

Liquidity and Management's Plans

The accompanying financial statements have been prepared assuming the Company will continue as a going concern, which contemplates the realization of assets and settlement of obligations in the normal course of business. The Company incurred operating losses from operations of \$528,201, had member's deficit of \$2,953,937, and had negative cash flows from operating activities of \$523,209 for the year ended December 31, 2023. Additionally, during 2023, the Company required additional advances of \$326,393 from its Parent to fund operations and address cash flow needs. These conditions create uncertainty as to the Company's ability to meet its obligations as they come due with at least one year and a day post issuance of these financial statements.

Management has evaluated its plans whereby the Parent has committed to provide financial support to the Company by providing written assurances that it will provide, as required, additional advances of up to the amount of \$1,200,000 to support the operating, investing and financing activities of the Company through at least one year and a day beyond the report date.

In addition, the Company has historically received advances primarily through BrightStar Franchising, LLC from the Parent. These notes are due on demand. The Parent and affiliated companies (see Note 3) will not call these notes for payment through at least one year and a day beyond the report date of these financial statements.

The Company's ability to continue as a going concern is dependent on the continued financial support of its Parent and the extension of the due date of the notes and the ability of the Company to execute its plan.

Basis of Accounting

The Company maintains its books and records on the accrual basis of accounting in conformity with accounting principles generally accepted in the United States of America (U.S. GAAP). Under the accrual method of accounting, revenues are recorded when earned and expenses are recorded when incurred.

Cash and Cash Equivalents

For purposes of the statements of cash flows, cash equivalents include the general checking account held by the Company. From time-to-time, cash balances may exceed federally insured limits of \$250,000. The Company believes the associated credit risk to be minimal.

Accounts Receivable

In June 2016, the Financial Accounting Standards Board (FASB) issued Accounting Standards Update (ASU) 2016-13, *Financial Instruments - Credit Losses: Measurement of Credit Losses on Financial Instruments*, which applies primarily to the Company's accounts receivable impairment loss allowances. The guidance provides a revised model whereby the current expected credit losses are used to compute impairment of financial instruments. The new model requires evaluation of historical experience and various current and expected factors, which may affect the estimated amount of losses and requires determination of whether the affected financial instruments should be grouped in units of account. The guidance was effective for the Company on January 2, 2023. Adoption of the ASU did not have material impact to the consolidated balance sheets, consolidated statements of income, cash flows, and related disclosures.

The Company grants credit to its customers in the normal course of business. Accounts receivable are recorded at the invoiced amount and do not bear interest. The Company may create payment plans for customers with significant past-due balances. The Company maintains an allowance for credit losses to cover potential credit losses relating to its accounts receivable. The allowance is based on the Company's historical collection experience, as well as an analysis of specific past-due accounts and forward-looking information. All accounts or portions thereof deemed to be uncollectible by management are written off in the period in which that determination is made. Balances outstanding after management has used reasonable collection efforts are written off through a charge to the valuation allowance and a credit to accounts receivable. There was no allowance for credit losses recorded as of December 31, 2023 and January 1, 2023.

Revenue Recognition

Under Accounting Standards Codification (ASC) 606, *Revenue from Contracts with Customers*, the Company accounts for a contract when it has approval and commitment from both parties, the rights of the parties are identified, payment terms are identified, the contract has commercial substance, and collectability of consideration is probable. The Company recognizes revenue only when it satisfies a performance obligation by transferring a promised good or service to the customer. The good or service is considered to be transferred when the customer obtains control, meaning when the customer has the ability to direct the use and obtain substantially all of the remaining benefits of the good or service. At each contract inception, the Company determines whether control of a good or service transfers to a customer over time or at a point in time.

Franchise Revenue

The terms of the franchise agreements for BrightStar Senior Living Communities are typically for 20 years. As of December 31, 2023 and January 1, 2023 there were three and four franchises, respectively. No new franchises are currently in consideration.

The terms of the franchise agreements for the BrightStar Care Homes Communities model are typically for 20 years. As of December 31, 2023 and January 1, 2023 there were five and four franchises operating, respectively.

As part of each agreement, the Company identified one performance obligation that requires the Company to provide a combination of the following:

- Intellectual Property (IP) Licenses grant a non-exclusive right to establish and operate an assisted living and memory care facility under the trademarks and systems established as part of the BrightStar license during the term of the agreement.
- Operations Manual and Brand-Specific Training Services These services provide training programs, operating manuals, development of standards and pricing policies, specific to the BrightStar brand.
- *Marketing Services* These services are ongoing local and national advertising programs that are implemented throughout the course of the contract term.
- *Continuing Consulting Services* These services include the review of plans and equipment and merchandise selection.

The Company determined that the services noted above represent a set of integrated or highly interrelated tasks/services and are, therefore, accounted for as a single-performance obligation of providing the franchise license.

In addition, the Company has also determined the following performance obligation is included in the contracts but is distinct from the above performance obligation:

• Non-Brand-Specific Training - This training relates to educating the franchisees as to the appropriate business model for operating a franchise. The services could be provided by another third party to the franchisee and do not relate specifically to the license of the BrightStar intellectual property.

The contracts the Company enters contain several types of payments, as follows:

BrightStar Senior Living Communities

- Initial Fees Franchise agreements require a nonrecurring and nonrefundable initial fee of \$50,000. Initial franchise fees are due and payable when a contract is signed. The initial franchise fee is not in exchange for any particular products, services, or assistance but instead solely in consideration of the Company's granting the franchise to the franchisee.
- *Royalty Fees* The Company receives monthly royalty payments based on a percentage of each franchisee's net billings. The franchisee is required to meet a minimum revenue level for the payment of these royalty fees in any given month.

- General Marketing Fund Fees These fees are based on the greater of \$2,000 or a portion of franchisee net billings to contribute to the local franchise and to the national brand advertising activities directed by the franchisor. These fees are typically due on a monthly basis over the course of the contract term.
- *Monthly Technology Fees* The Company receives monthly fees for technology software and support based on the greater of \$250/month or 0.83% of franchisee net billings.
- *Renewal Fees* Franchise agreements include renewal options and require the franchisee to pay a renewal fee.
- *Transfer Fees* Franchise agreements include transfer options and require the franchisee to pay a transfer fee.

BrightStar Care Homes Communities

- Initial Fees Franchise agreements require a nonrecurring and nonrefundable initial fee of \$50,000 for an affiliate that is not an existing franchisee. For existing franchisees, an initial fee of \$30,000 will be charged, with an additional \$10,000 for each additional community under the agreement (up to two additional locations in one agreement). Initial franchise fees are due and payable when a contract is signed. The initial franchise fee is not in exchange for any particular products, services, or assistance but instead solely in consideration of the Company's granting the franchise to the franchisee.
- *Royalty Fees* The Company receives monthly royalty payments based on a 5% of each franchisee's net billings during the preceding calendar month. The franchisee is required to meet a minimum revenue level for the payment of these royalty fees in any given month.
- General Marketing Fund Fees These fees are based on the greater of 2.50% of the prior month's net billings or \$500 per month, per community (up to three per franchise agreement) to contribute to the local franchise and to the national brand advertising activities directed by the franchisor. These fees are typically due on a monthly basis over the course of the contract term.
- *Monthly Technology Fees* The Company receives monthly fees for technology software and support based on the greater of \$100/month or 0.83% of franchisee prior month net billings for each community operating under the franchise agreement.
- *Renewal Fees* Franchise agreements include renewal options and require the franchisee to pay a renewal fee.
- *Transfer Fees* Franchise agreements include transfer options and require the franchisee to pay a transfer fee of \$15,000.

The Company recognizes revenue for each performance obligation identified within the customer contracts when, or as, the performance obligation is satisfied by transferring the promised goods or services. All revenue is recognized over time. The Company recognizes revenue over time for the combined franchise license performance obligation and for the non-brand-specific training, as these performance obligations involve a continuous transfer of control to the customer.

The standard specifically identifies franchise rights as an example of a symbolic license. This type of license is satisfied over time since the customer simultaneously receives and consumes the benefit as the entity performs its obligation to provide access and, therefore, meets the criterion

of recognizing revenue over time. Royalty fees represent the majority of consideration the Company receives under franchise agreements and are recognized over time at the greater of the actual royalty earned or the contract monthly minimum each month. Revenue related to upfront fees allocated to this single-performance obligation is recognized over time using a straight-line (time-lapse) measure of progress as the control of various services is provided to the customer ratably over the term of the contract for the initial upfront fee. The renewal option provides continued access for the franchise rights (symbolic license) for an extended period of time and, therefore, would also be recognized over time (over the course of the renewal term) as it meets the above-mentioned criterion.

For the non-brand-specific training, the Company also recognizes revenue over time since the customer simultaneously receives and consumes the benefit as the entity performs the training. Revenue for this performance obligation is recognized over time using an input measure of progress based upon hours of training performed.

General Marketing Fund

The General Marketing Fund (GMF) funds received from the franchisees are recognized as revenue over time when incurred and marketing expenses are recognized when incurred within the Company's statements of operations. GMF income and GMF expense for the year ended December 31, 2023 were \$72,476 and \$53,634, respectively, and are included within the statements of operations. GMF income and GMF expense for the year ended January 1, 2023 were \$71,460 and \$23,866, respectively, and are included within the statements of operations.

Contract Assets and Liabilities

Contract assets include accrued revenue for items not invoiced as of period end due to billing occurring subsequent to the performance obligation being fulfilled. Contract assets were \$85,323 and \$101,779 as of December 31, 2023 and January 1, 2023, respectively, included within accounts receivable. Contract liabilities include initial franchise fees received upfront for the right to use the BrightStar Senior Living brand. These fees are recognized ratably over the initial term of the individual contacts, which typically lasts 20 years. The total balance of the contract liabilities, reflected as deferred revenue (non-refundable), was \$264,934 and \$244,933 as of December 31, 2023 and January 1, 2023, respectively.

Income Taxes

The Company's Parent has elected to be taxed as an S-Corporation under provisions of the Internal Revenue Code. Under those provisions, the Company's Parent does not pay federal income taxes. Instead, the Parent's stockholder is liable for individual federal income taxes on the Parent's consolidated taxable income.

Use of Estimates

The preparation of financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements, and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

3. Notes Payable - Affiliated Companies, Net

The Company has historically received advances from and provided advances to the Parent and its wholly owned affiliates. These notes are subject to voluntary prepayment at any time, in whole or in part, without penalty. No formal repayment plan exists for these uncollateralized notes. A right of offset exists with these notes. The notes are due on demand. These affiliates will not call these notes for payment through at least one year and a day beyond the report date (see Note 2), resulting in a classification of long-term in the balance sheets for the years ended December 31, 2023 and January 1, 2023.

The fiscal year-end balance on these notes are as follows:

	D	ecember 31, 2023	January 1, 2023
BrightStar Franchising, LLC	\$	(3,809,963) \$	(3,229,877)
BrightStar Senior Living Development of Fort Wayne, LLC		-	(85,644)
BrightStar Senior Living Operations of Fort Wayne, LLC		-	461,375
BrightStar Group Holdings, Inc.		(11,099)	(11,099)
BrightStar Senior Living Operations of Mason, LLC		1,054,625	416,537
BrightStar Senior Living Development of Mason, LLC		(116,492)	(107,827)
Total	\$	(2,882,929) \$	(2,556,535)

4. Contingencies

From time-to-time, the Company is involved in legal proceedings, claims, or investigations that are incidental to the conduct of the Company's business. Although the ultimate outcome of any legal matter cannot be predicted with certainty, based on present information, including management's assessment of the merits of any particular claim, the Company does not expect that these legal proceedings or claims will have any material adverse impact on its future financial position or results of operations.

The Company is exposed to various risks of loss related to torts; theft of, damage to, and destruction of assets; errors and omissions; and natural disasters for which the Company carries commercial insurance. There have been no significant reductions in coverage from the prior year, and settlements, if any, have not exceeded coverage.

5. Subsequent Events

The Company has evaluated subsequent events through April 29, 2024, the date the financial statements became available for issuance.

Financial Statements For the Years Ended January 1, 2023 and January 2, 2022





Financial Statements For the Years Ended January 1, 2023 and January 2, 2022

Contents

Independent Auditor's Report	3-4
Financial Statements	
Balance Sheets as of January 1, 2023 and January 2, 2022	5
Statements of Operations for the Years Ended January 1, 2023 and January 2, 2022	6
Statements of Member's Deficit for the Years Ended January 1, 2023 and January 2, 2022	7
Statements of Cash Flows for the Years Ended January 1, 2023 and January 2, 2022	8
Notes to Financial Statements	9-15



Tel: 312-856-9100 Fax: 312-856-1379 www.bdo.com

Independent Auditor's Report

BrightStar Senior Living Franchising, LLC Gurnee, Illinois

Opinion

We have audited the financial statements of BrightStar Senior Living Franchising, LLC (the Company), which comprise the balance sheets as of January 1, 2023 and January 2, 2022, the related statements of operations, member's deficit, and cash flows for the years then ended, and the related notes to the financial statements.

In our opinion, the accompanying financial statements present fairly, in all material respects, the financial position of the Company as of January 1, 2023 and January 2, 2022, and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audits in accordance with auditing standards generally accepted in the United States of America (GAAS). Our responsibilities under those standards are further described in the *Auditor's Responsibilities for the Audit of the Financial Statements* section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities, in accordance with the relevant ethical requirements relating to our audits. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Emphasis of Matter

As described in Note 2 to the financial statements, the Parent Company, BrightStar Group Holdings, Inc., has agreed to support the operating, investing, and financing activities of the Company through at least one year and a day beyond the report date. Additionally, the Parent and affiliated companies (see Note 3) will not call their notes through at least one year and a day beyond the report date. Our opinion is not modified with respect to these matters.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern within one year after the date that the financial statements are issued or available to be issued.

BDO is the brand name for the BDO network and for each of the BDO Member Firms.

BDO USA, LLP, a Delaware limited liability partnership, is the U.S. member of BDO International Limited, a UK company limited by guarantee, and forms part of the international BDO network of independent member firms.



Auditor's Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and, therefore, is not a guarantee that an audit conducted in accordance with GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the financial statements.

In performing an audit in accordance with GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the financial statements, whether due to fraud or error, and design and perform audit procedures responsive to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control-related matters that we identified during the audit.

BDO USA, LLP

March 16, 2023

Balance Sheets

	January 1, 2023			January 2, 2022	
Assets					
Current Assets Cash and cash equivalents Accounts receivable, net Prepaid expenses	\$	1,890 101,779 6,790	\$	441,824 36,462 9,537	
Total Current Assets		110,459		487,823	
Long-Term Assets Equipment, net		2,806		4,660	
Total Assets	\$	113,265	\$	492,483	
Liabilities and Member's Deficit					
Current Liabilities Accrued salaries and payroll tax Other current liabilities	\$	79,925 36,637	\$	49,148 8,630	
Total Current Liabilities		116,562		57,778	
Long-Term Liabilities Deferred revenue (non-refundable) Notes payable, affiliated companies, net		244,933 2,556,535		169,933 2,688,353	
Total Long-Term Liabilities		2,801,468		2,858,286	
Total Liabilities		2,918,030		2,916,064	
Member's Deficit		(2,804,765)		(2,423,581)	
Total Liabilities and Member's Deficit	\$	113,265	\$	492,483	

Statements of Operations

Year ended	January 1, 2023	January 2, 2022	
Net Revenues			
Royalty fees	\$ 357,302	\$	280,095
General marketing fund fees	71,460		56,019
System fees	25,500		25,390
Initial franchisee fees	-		3,200
Other	3,384		2,307
Total Net Revenues	457,646		367,011
Expenses			
Payroll and related expenses	525,490		419,478
Broker fees	32,000		-
Professional fees	204,927		196,295
Marketing	2,567		21,823
Travel and meals	38,815		25,115
IT services	20,205		28,033
Occupancy and office expense	7,854		7,795
Insurance	-		953
Depreciation	1,854		2,240
Miscellaneous expenses	5,118		8,731
Total Expenses	838,830		710,463
Net Loss	\$ (381,184)	\$	(343,452)

Statements of Member's Deficit

	Member's Deficit
Balance, January 3, 2021	\$ (2,080,129)
Net loss	(343,452)
Balance, January 2, 2022	(2,423,581)
Net loss	(381,184)
Balance, January 1, 2023	\$ (2,804,765)

Statements of Cash Flows

Year ended	January 1, 2023 January 2			ary 2, 2022
Cash Flows from Operating Activities Net loss Adjustments to reconcile net loss to net cash used in operating activities:	\$	(381,184)	\$	(343,452)
Depreciation expense		1,854		2,240
Decrease (increase) in: Accounts receivable Prepaid expenses Increase (decrease) in:		(65,317) 2,747		(11,217) 11,883
Other current liabilities Deferred revenue (non-refundable) Accrued salaries and payroll tax		28,007 75,000 30,777		2,820 86,800 18,508
Net Cash Used in Operating Activities		(308,116)		(232,418)
Cash Flows from Financing Activities Notes payable, net, affiliated companies		(131,818)		475,987
Net Cash Provided by (Used in) Financing Activities		(131,818)		475,987
Net (Decrease) Increase in Cash and Cash Equivalents		(439,934)		243,569
Cash and Cash Equivalents, beginning of year		441,824		198,255
Cash and Cash Equivalents, end of year	\$	1,890	\$	441,824

1. Description of Business and Operations

BrightStar Senior Living Franchising, LLC (the Company) is a member of a consolidated group whose parent is BrightStar Group Holdings, Inc. (the Parent), which consists of 16 additional members-BrightStar Technology Group, LLC: 24-7 BrightStar Operations, LLC: 24-7 BrightStar Healthcare, LLC (BrightStar Nevada); BrightStar Franchising, LLC; BrightStar Senior Living, LLC; BrightStar CRDM, LLC; BrightStar Senior Living Development of Fort Wayne, LLC; BrightStar Senior Living Operations of Fort Wayne, LLC; BrightStar Senior Living Development of Mason, LLC; and Brightstar Senior Living Operations of Mason, LLC; BrightStar Owned LLC; BrightStar Scottsdale Operations, LLC: BrightStar FL/AL Operations, LLC: BrightStar Spartanburg/Greenville Operations, LLC; BrightStar TN Operations, LLC; and BrightStar Wisconsin Operations, LLC. The Company is organized as a limited liability company under the laws of the state of Illinois and it is a franchiser of businesses that has developed a system for constructing, operating, identifying, and promoting assisted living and memory care facilities under the BRIGHTSTAR SENIOR LIVING® trademark and related commercial symbols. These facilities provide various services for seniors and other individuals in need, including residential assisted living, memory care, and ancillary assisted living services and are marketed as BrightStar Senior Living Communities. The Company also operates a smaller residential care home model for assisted living for seniors and other individuals in need that is marketed as BrightStar Care Homes Communities.

The Company's fiscal year ends on the nearest Sunday to December 31. There were 52 weeks included in each of the years ended January 1, 2023 and January 2, 2022.

2. Summary of Significant Accounting Policies

Basis of Presentation

Liquidity and Management's Plans

The accompanying financial statements have been prepared assuming the Company will continue as a going concern, which contemplates the realization of assets and settlement of obligations in the normal course of business. The Company incurred operating losses from operations of \$381,184, had member's deficit of \$2,804,765 and had negative cash flows from operating activities of \$308,116 for the year ended January 1, 2023. Additionally, during 2022, the Company required additional advances of \$124,136 from its Parent to fund operations and address cash flow needs. These conditions create uncertainty as to the Company's ability to meet its obligations as they come due with at least one and a day post issuance of these financial statements.

Management has evaluated its plans whereby the Parent has committed to provide financial support to the Company by providing written assurances that it will provide, as required, additional advances of up to the amount of \$1,000,000 to support the operating, investing and financing activities of the Company through at least one year and a day beyond the report date.

In addition, the Company has historically received advances primarily through BrightStar Franchising, LLC from the Parent. These notes are non-interest bearing and due on demand. The Parent and affiliated companies (see Note 3) will not call these notes for payment through at least one year and a day beyond the report date of these financial statements.

The Company's ability to continue as a going concern is dependent on the continued financial support of its Parent and the extension of the due date of the notes and the ability of the Company to execute its plan.

Basis of Accounting

The Company maintains its books and records on the accrual basis of accounting in conformity with accounting principles generally accepted in the United States of America (U.S. GAAP). Under the accrual method of accounting, revenues are recorded when earned and expenses are recorded when incurred.

Cash and Cash Equivalents

For purposes of the statements of cash flows, cash equivalents include the general checking account held by the Company. From time-to-time, cash balances may exceed federally insured limits of \$250,000. The Company believes the associated credit risk to be minimal.

Accounts Receivable

The Company grants credit to its customers in the normal course of business. Accounts receivable are recorded at the invoiced amount and do not bear interest. The Company may create payment plans for customers with significant past-due balances. The Company maintains an allowance for doubtful accounts to cover potential credit losses relating to its accounts receivable. The allowance is based on the Company's historical collection experience, as well as an analysis of specific past-due accounts. All accounts or portions thereof deemed to be uncollectible by management are written off in the period in which that determination is made. Balances outstanding after management has used reasonable collection efforts are written off through a charge to the valuation allowance and a credit to accounts receivable. There was no allowance for doubtful accounts recorded as of January 1, 2023 and January 2, 2022.

Revenue Recognition

Under Accounting Standards Codification (ASC) 606, *Revenue from Contracts with Customers*, the Company accounts for a contract when it has approval and commitment from both parties, the rights of the parties are identified, payment terms are identified, the contract has commercial substance, and collectability of consideration is probable. The Company recognizes revenue only when it satisfies a performance obligation by transferring a promised good or service to the customer. The good or service is considered to be transferred when the customer obtains control, meaning when the customer has the ability to direct the use and obtain substantially all of the remaining benefits of the good or service. At each contract inception, the Company determines whether control of a good or service transfers to a customer over time or at a point in time.

Franchise Revenue

The terms of the franchise agreements for BrightStar Senior Living Communities are typically for 20 years. As of January 1, 2023 and January 2, 2022 there were four franchises. No new franchises are currently in consideration.

The terms of the franchise agreements for the BrightStar Care Homes Communities model are typically for 20 years. As of January 1, 2023 and January 2, 2022 there were four franchises operating.

As part of each agreement, the Company identified one performance obligation that requires the Company to provide a combination of the following:

- Intellectual Property (IP) Licenses grant a non-exclusive right to establish and operate an assisted living and memory care facility under the trademarks and systems established as part of the BrightStar license during the term of the agreement.
- Operations Manual and Brand-Specific Training Services These services provide training programs, operating manuals, development of standards and pricing policies, specific to the BrightStar brand.
- *Marketing Services* These services are ongoing local and national advertising programs that are implemented throughout the course of the contract term.
- *Continuing Consulting Services* These services include the review of plans and equipment and merchandise selection.

The Company determined that the services noted above represent a set of integrated or highly interrelated tasks/services and are, therefore, accounted for as a single-performance obligation of providing the franchise license.

In addition, the Company has also determined the following performance obligation is included in the contracts but is distinct from the above performance obligation:

• Non-Brand-Specific Training - This training relates to educating the franchisees as to the appropriate business model for operating a franchise. The services could be provided by another third party to the franchisee and do not relate specifically to the license of the BrightStar intellectual property.

The contracts the Company enters contain several types of payments, as follows:

BrightStar Senior Living Communities

- Initial Fees Franchise agreements require a nonrecurring and nonrefundable initial fee of \$50,000. Initial franchise fees are due and payable when a contract is signed. The initial franchise fee is not in exchange for any particular products, services, or assistance but instead solely in consideration of the Company's granting the franchise to the franchisee.
- *Royalty Fees* The Company receives monthly royalty payments based on a percentage of each franchisee's net billings. The franchisee is required to meet a minimum revenue level for the payment of these royalty fees in any given month.
- General Marketing Fund Fees These fees are based on the greater of \$2,000 or a portion of franchisee net billings to contribute to the local franchise and to the national brand advertising activities directed by the franchisor. These fees are typically due on a monthly basis over the course of the contract term.
- *Monthly Technology Fees* The Company receives monthly fees for technology software and support based on the greater of \$250/month or 0.83% of franchisee net billings.

- *Renewal Fees* Franchise agreements include renewal options and require the franchisee to pay a renewal fee.
- *Transfer Fees* Franchise agreements include transfer options and require the franchisee to pay a transfer fee.

BrightStar Care Homes Communities

- Initial Fees Franchise agreements require a nonrecurring and nonrefundable initial fee of \$50,000 for an affiliate that is not an existing franchisee. For existing franchisees, an initial fee of \$30,000 will be charged, with an additional \$10,000 for each additional community under the agreement (up to two additional locations in one agreement). Initial franchise fees are due and payable when a contract is signed. The initial franchise fee is not in exchange for any particular products, services, or assistance but instead solely in consideration of the Company's granting the franchise to the franchisee.
- *Royalty Fees* The Company receives monthly royalty payments based on a 5% of each franchisee's net billings during the preceding calendar month. The franchisee is required to meet a minimum revenue level for the payment of these royalty fees in any given month.
- General Marketing Fund Fees These fees are based on the greater of 2.50% of the prior month's net billings or \$500 per month, per community (up to three per franchise agreement) to contribute to the local franchise and to the national brand advertising activities directed by the franchisor. These fees are typically due on a monthly basis over the course of the contract term.
- *Monthly Technology Fees* The Company receives monthly fees for technology software and support based on the greater of \$100/month or 0.83% of franchisee prior month net billings for each community operating under the franchise agreement.
- *Renewal Fees* Franchise agreements include renewal options and require the franchisee to pay a renewal fee.
- *Transfer Fees* Franchise agreements include transfer options and require the franchisee to pay a transfer fee of \$15,000.

The Company recognizes revenue for each performance obligation identified within the customer contracts when, or as, the performance obligation is satisfied by transferring the promised goods or services. All revenue is recognized over time. The Company recognizes revenue over time for the combined franchise license performance obligation and for the non-brand-specific training, as these performance obligations involve a continuous transfer of control to the customer.

The standard specifically identifies franchise rights as an example of a symbolic license. This type of license is satisfied over time since the customer simultaneously receives and consumes the benefit as the entity performs its obligation to provide access and, therefore, meets the criterion of recognizing revenue over time. Royalty fees represent the majority of consideration the Company receives under franchise agreements and are recognized over time at the greater of the actual royalty earned or the contract monthly minimum each month. Revenue related to upfront fees allocated to this single-performance obligation is recognized over time using a straight-line (time-lapse) measure of progress as the control of various services is provided to the customer ratably over the term of the contract for the initial upfront fee. The renewal option provides continued access for the franchise rights (symbolic license) for an extended period of time and, therefore, would also be recognized over time (over the course of the renewal term) as it meets the above-mentioned criterion.

For the non-brand-specific training, the Company also recognizes revenue over time since the customer simultaneously receives and consumes the benefit as the entity performs the training. Revenue for this performance obligation is recognized over time using an input measure of progress based upon hours of training performed.

General Marketing Fund (GMF)

The GMF funds received from the franchisees are recognized as revenue over time when incurred and marketing expenses are recognized when incurred within the Company's statements of operations. GMF income and GMF expense for the year ended January 1, 2023 were \$71,460 and \$23,866, respectively, and are included within the statements of operations. GMF income and GMF expense for the year ended January 2, 2022 were \$56,019 and \$3,898, respectively, and are included within the statements of operations.

Contract Assets and Liabilities

Contract assets include accrued revenue for items not invoiced as of period end due to billing occurring subsequent to the performance obligation being fulfilled. Contract assets were \$101,779 and \$36,462 as of January 1, 2023 and January 2, 2022, respectively, included within accounts receivable. Contract liabilities include initial franchise fees received upfront for the right to use the BrightStar Senior Living brand. These fees are recognized ratably over the initial term of the individual contacts, which typically lasts 20 years. The total balance of the contract liabilities, reflected as deferred revenue (non-refundable), was \$244,933 and \$169,933 as of January 1, 2023 and January 2, 2022, respectively.

Income Taxes

The Company's Parent has elected to be taxed as an S-Corporation under provisions of the Internal Revenue Code. Under those provisions, the Company's Parent does not pay federal income taxes. Instead, the Parent's stockholder is liable for individual federal income taxes on the Parent's consolidated taxable income.

Use of Estimates

The preparation of financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements, and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Recent Accounting Pronouncements

In June 2016, Financial Accounting Standards Board (FASB) issued Accounting Standards Update (ASU) 2016-13, *Financial Instruments - Credit Losses: Measurement of Credit Losses on Financial Instruments*, which applies primarily to the Company's accounts receivable impairment loss allowances. The guidance provides a revised model whereby the current expected credit losses are used to compute impairment of financial instruments. The new model requires evaluation of historical experience and various current and expected factors, which may affect the estimated amount of losses and requires determination of whether the affected financial instruments should be grouped in units of account. This standard is effective for private entities for annual periods

Notes to Financial Statements

beginning after December 15, 2022. The Company is currently assessing the impact of this standard on the Company's financial position, results of operations, and cash flows.

In January 2021, the FASB issued ASU 2021-02, *Franchisors - Revenue from Contracts with Customers* (*Subtopic 952-606*): *Practical Expedient*, related to revenue recognition of pre-opening services for franchisors that are not public business entities. The amendments introduce a new practical expedient that simplifies the application of the guidance about identifying performance obligations. The practical expedient permits franchisors that are not public business entities to account for pre-opening services provided to a franchisee as distinct from the franchise license if the services are consistent with those included in a predefined list within the guidance. In addition, a policy election can be made to recognize pre-opening services as a single-performance obligation. The Company is currently evaluating the impact that adoption of ASU 2021-02 will have on its financial statements and related disclosures.

3. Notes Payable - Affiliated Companies, Net

The Company has historically received advances from and provided advances to the Parent and its wholly owned affiliates. These notes are non-interest bearing and due on demand. These affiliates will not call these notes for payment through at least one year and a day beyond the report date (see Note 2), resulting in a classification of long-term in the balance sheets for the years ended January 1, 2023 and January 2, 2022.

The following is a summary of transactions with affiliated entities that can be settled on a net basis:

	Jan	uary 1, 2023	Jar	nuary 2, 2022
BrightStar Franchising, LLC	\$	(3,229,877)	\$	(3,105,741)
BrightStar Senior Living Development of Fort Wayne, LLC		(85,644)		(34,287)
BrightStar Senior Living Operations of Fort Wayne, LLC		461,375		408,214
BrightStar Group Holdings, Inc.		(11,099)		(11,099)
BrightStar Senior Living Operations of Mason, LLC		416,537		70,403
BrightStar Senior Living Development of Mason, LLC		(107,827)		(15,843)
Total	\$	(2,556,535)	\$	(2,688,353)

4. Contingencies

From time-to-time, the Company is involved in legal proceedings, claims, or investigations that are incidental to the conduct of the Company's business. Although the ultimate outcome of any legal matter cannot be predicted with certainty, based on present information, including management's assessment of the merits of any particular claim, the Company does not expect that these legal proceedings or claims will have any material adverse impact on its future financial position or results of operations.

The Company is exposed to various risks of loss related to torts; theft of, damage to, and destruction of assets; errors and omissions; and natural disasters for which the Company carries commercial insurance. There have been no significant reductions in coverage from the prior year, and settlements, if any, have not exceeded coverage.

5. Subsequent Events

The Company has evaluated subsequent events through March 16, 2023, the date the financial statements became available for issuance.

EXHIBIT G

TO BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC BRIGHTSTAR CARE HOMES COMMUNITY FRANCHISE DISCLOSURE DOCUMENT

LIST OF FRANCHISEES

OPEN COMMUNITIES

Principals	Franchisee	Address	City	State	Zip	Phone
					Code	
Caroline	Bright Blooms,	3940 S. Summerset	Boise	ID	83709	(208) 338-7879
Moore	LLC	Way				
Caroline	Bright Blooms,	2903 W. Taft Street	Boise	ID	83703	(208) 338-7879
Moore	LLC					
Caroline	Bright Blooms,	2222 S. Stephen	Boise	ID	83706	(208) 338-7879
Moore	LLC	Avenue				
Carolina	Bright Blooms,	5578 N. Eagle Road	Boise	ID	83713	(208) 639-9992
Moore	LLC					
Caroline	Bright Blooms,	4073 W. Prickly Pear	Eagle	ID	83616	(208) 338-7879
Moore	LLC	Drive				

UNDER CONSTRUCTION

Principals	Franchisee	Address	City	State	Zip	Phone
					Code	
Caroline	Bright Blooms,	5578 N. Eagle Rd.	Boise	ID	83713	(208) 338-7879
Moore	LLC	-				
Caroline	Bright Blooms,	3336 N. Meridian Rd	Meridian	ID	83646	(208) 338-7879
Moore	LLC					

SIGNED

Principals	Franchisee	City	State	Phone
Michael Perez & Sharon	Casa Del Corazon, LLC	Santa Barbara	CA	805-335-1200
Holland-Perez				
Leonard Daye	Leonard Daye	El Paso	TX	470-985-8508
Alexander & Tammara	Alexander & Tammara	Salt Lake City	UT	801-555-3999
Brown	Brown			
Scott & Lori Ranney	NorthStar Senior Care Inc.	Savage	MN	612-428-0224
Karl Larsen and Nathan	Karl Larsen and Nathan	Bellingham /	WA	360-920-6140 &
Jackson	Jackson	Mount Vernon		916-605-8448
Wu (Ethan) Youyi	Wu (Ethan) Youyi	South Jefferson	LA	646-875-0288
		Parish		

CONSENSUAL MUTUAL TERMINATIONS

Franchisee	Last Known Address	Phone	<u>Email</u>
*Anant Khokhar & Masha	713 Live Oak St., New	(386) 233-9009	Andy.khokhar@brightstarcare.com
Sandhu	Smyrna Beach, FL 32168		
*Christopher & Alison	2734 Chancellor Drive	(859) 692-9500	Chris.mccreary@brightstarcare.com
McCreary	Suite 207, Crestview Hills,		
	KY 41017		
**Timothy Estepp	1857 Bopp Road, St, Louis,	314-898-8687	Tim@onewest.com
	MO 63131		

*Chose to expand their BrightStar Care Agency instead or moving forward with BrightStar Care Homes opportunity.

**Did not proceed due to economic factors in their territory.

EXHIBIT H

TO BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC BRIGHTSTAR CARE HOMES COMMUNITY FRANCHISE DISCLOSURE DOCUMENT

RELEASE OF CLAIMS

RELEASE OF CLAIMS THIS FORM IS SUBJECT TO CHANGE OVER TIME

For and in consideration of the agreements and covenants described below, BrightStar Senior Living Franchising, LLC ("Franchisor") and ______ ("Franchisee") enter into this Release of Claims ("Agreement").

RECITALS

A. Franchisor and Franchisee entered into a BrightStar Senior Living Franchising, LLC Care Homes Franchise Agreement dated ______.

B. [NOTE: Describe the circumstances relating to the release.]

C. Subject to and as addressed with greater specificity in the terms and conditions identified below, Franchisor and Franchisee now desire to settle any and all disputes that may exist between them relating to the Franchise Agreement.

AGREEMENTS

1. **Consideration.** [NOTE: Describe the consideration paid.]

2-3. [NOTE: Detail other terms and conditions of the release.]

5. **Release of Claims by Franchisee.** In consideration of the other terms and conditions of this Agreement, the receipt and sufficiency of which are hereby acknowledged, Franchisee, for himself and for each of his heirs, executors, administrators, insurers, attorneys, agents, representatives, successors, and assigns, does hereby release and forever discharge Franchisor and each of its respective affiliated corporations, subsidiaries, divisions, insurers, indemnitors, attorneys, successors, and assigns, together with all of their past and present directors, officers, employees, attorneys, agents, assigns and representatives in their capacities as such, of and from any and all actions, suits, proceedings, claims (including, but not limited to, claims for attorneys' fees), complaints, charges, judgments, and executions, whether liquidated or unliquidated, known or unknown, asserted or unasserted, absolute or contingent, accrued or not accrued, related to the Franchise Agreement.

[The following is additional language for California-based franchisees – remove for all other states]

Each party granting a release acknowledges a familiarity with Section 1542 of the Civil Code of the State of California, which provides as follows:

"A general release does not extend to claims that the creditor or releasing party does not know or suspect to exist in his or her favor at the time of executing the release and that, if known by him or her, would have materially affected his or her settlement with the debtor or released party."

Each party granting a release and its authorized signatories hereto recognize that he, she, or it may have some claim, demand, or cause of action against Franchisor and the other Franchisor Released Parties of which he, she, or it is unaware and unsuspecting, and which he, she, or it is giving up by signing this Addendum. Each party granting a release and its authorized signatories hereby waive and relinquish every right or benefit which he, she, or it has under Section 1542 of the Civil Code of the State of California, and any similar statute under any other state or federal law, to the fullest extent that such right or benefit may lawfully be waived.

6. **Reservation of Claims Against Non-Settling Parties.** Franchisor and Franchisee expressly reserve their right and claims against any non-settling persons, firms, corporations, or other entities for whatever portion or percentage their damages are found to be attributable to the wrongful conduct of said non-settling parties.

7. **Entire Agreement.** This Agreement constitutes the entire agreement between the parties relative to the subject matter contained in this Agreement, and all prior understandings, representations and agreements made by and between the parties relative to the contents contained in this Agreement are merged into this Agreement.

8. **Voluntary Nature of Agreement.** The parties acknowledge and agree that they have entered into this Agreement voluntarily and without any coercion. The parties further represent that they have had the opportunity to consult with an attorney of their own choice, that they have read the terms of this Agreement, and that they fully understand and voluntarily accept the terms.

9. **Governing Law and Jurisdiction.** This Agreement will be construed and enforced in accordance with the laws of the state of ______.

10. **Attorneys' Fees.** All rights and remedies under this Agreement shall be cumulative and none shall exclude any other right or remedy allowed by law. In the event of a breach of this Agreement that requires one of the parties to enforce the terms and conditions of this Agreement, the non-prevailing party shall pay the prevailing party's attorneys' fees and costs incurred by reason of the breach.

This release does not apply with respect to claims arising under the Washington Franchise Investment Protection Act, RCW 19.100, and the rules adopted thereunder.

Dated:	BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC
	By
	Its
Dated:	FRANCHISEE:
	By
	Its

EXHIBIT I

TO BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC BRIGHTSTAR CARE HOMES COMMUNITY FRANCHISE DISCLOSURE DOCUMENT

MICROSOFT TEAMS/ONE-DRIVE/WEB VERSION OFFICE PACKAGE OPT-IN ADDENDUM

MICROSOFT TEAMS/ONE-DRIVE/WEB VERSION OFFICE PACKAGE OPT-IN ADDENDUM

,_____,

This addendum (the "Addendum") to the BrightStar Senior Living Franchising, LLC BRIGHTSTAR CARE HOMES Community Franchise Agreement is made and entered into as of ______ (the "Execution Date") by and between BrightStar Senior Living Franchising, LLC, an Illinois limited liability company with its principal place of business at 2275 Half Day Road, Suite 210, Bannockburn, Illinois 60015 ("we," "us," or "our"), and ______, with an address at _______

___("you" or "your").

BACKGROUND

A. On ______, 20___, the parties entered into a BrightStar Senior Living Franchising, LLC BRIGHTSTAR CARE HOMES Community Franchise Agreement (collectively, the "Franchise Agreement") pursuant to which you were granted the right to open and operate a BRIGHTSTAR CARE HOMES Community at the premises defined in Exhibit A to the Franchise Agreement (the "Community").

B. You have requested us to provide you with Microsoft Teams, One-Drive, and The Web version of Microsoft's Office package, which also includes a storage limit of 1TB per account for Teams/One Drive (collectively, the "Teams Package").

C. All capitalized terms not defined in this Addendum have the meanings ascribed to them in the Franchise Agreement. To the extent the terms of this Addendum are inconsistent with any terms of the Franchise Agreement, the terms of this Addendum will supersede and govern.

D. The parties wish to amend the terms of the Franchise Agreement pursuant to the terms and conditions of this Addendum.

AGREEMENT

NOW THEREFORE, in consideration of the agreements, covenants and promises contained in this Addendum and the Franchise Agreement, and for other consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

1. <u>Incorporation</u>. The "Background" provisions above are incorporated into and made a part of this Addendum by this reference.

2. By execution of this Addendum, you acknowledge and agree that beginning the month you are provided the Teams Package, you will be charged \$7.80 per user per month based on active Teams Package users during the prior month. The \$7.80 per user per month is not prorated (e.g., whether you are issued a Teams Package user account for you or an employee on the 1^{st} of the month or on the 30^{th} of the month you will be charged \$7.80). The monthly fee is

subject to change. We will provide a 30 day notice prior to any fee changes. You may elect, upon 30 days' notice to us, to terminate your and your Agency's staff's use of the Teams Package.

4. <u>Entire Agreement</u>. The Franchise Agreement and this Addendum constitute the entire, full, and complete agreement between the parties concerning the Agency. In the event of a conflict between the terms of the Franchise Agreement and the terms of this Addendum, the terms of this Addendum shall control. Except as amended hereby, all the other terms and conditions of the Franchise Agreement are ratified and confirmed.

IN WITNESS WHEREOF, the parties hereto, intending to be legally bound hereby, have duly executed and delivered this Addendum the date and year first written above.

BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC (FRANCHISOR)

FRANCHISEE:

By:	
Printed Name:	
Title: Director, Franchise Administration	tion

By:_____ Printed Name:_____ Title:_____

By:	
Printed Name:	
Title:	

NEW YORK REPRESENTATIONS PAGE

FRANCHISOR REPRESENTS THAT THIS FRANCHISE DISCLOSURE DOCUMENT DOES NOT KNOWINGLY OMIT ANY MATERIAL FACT OR CONTAIN ANY UNTRUE STATEMENT OF A MATERIAL FACT.

State Effective Dates

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered or exempt from registration, as of the Effective Date stated below:

State	Effective Date
California	Pending
Illinois	April 29, 2024
Indiana	April 29, 2024
Maryland	Pending
Michigan	April 29, 2024
Minnesota	Pending
New York	Pending
North Dakota	Pending
Rhode Island	Pending
South Dakota	April 29, 2024
Virginia	Pending
Washington	Pending
Wisconsin	April 29, 2024

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

EXHIBIT J

TO BRIGHTSTAR SENIOR LIVING FRANCHISING, LLC BRIGHTSTAR CARE HOMES COMMUNITY FRANCHISE DISCLOSURE DOCUMENT

RECEIPTS

RECEIPT

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully. If BrightStar Senior Living Franchising, LLC offers you a franchise, it must provide this disclosure document to you 14 calendar-days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale.

[Michigan law requires that we give you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.]

If BrightStar Senior Living Franchising, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the appropriate state agency identified on Exhibit A.

The franchisor is BrightStar Senior Living Franchising, LLC, located at 2275 Half Day Road, Suite 210, Bannockburn, Illinois 60015. Its telephone number is (888) 684-8250.

The issuance date of this Franchise Disclosure Document is April 29, 2024.

The name, principal business address, and telephone number of each franchise seller offering the franchise: Pete First, Chris Connolly, Amy Burnett, Dean Ulizio, Shannon Stemen, Shelly Sun, and David Pallaschke, 2275 Half Day Road, Suite 210, Bannockburn, Illinois 60015 847-693-2035.

We authorize the respective state agents identified on Exhibit A to receive service of process for us in the particular states. I received a Franchise Disclosure Document from BrightStar Senior Living Franchising, LLC dated as of April 29, 2024, that included the following Exhibits:

A – List of State Administrators and Agents for Service of Process; B – BRIGHTSTAR CARE HOMES Community Franchise Agreement (including exhibits); C – Operations Manual Table of Contents; D – Confidentiality, Non-Disclosure and Non-Competition Agreement; E – State Specific Addenda; F – Financial Statements; G – List of Franchisees; H – Release of Claims; I – Microsoft Teams/One-Drive/Web Version Office Package Opt-in Addendum; J – Receipts.

Date:	Signed:		
	Print Name:		
	City:	State	
	Phone ()		
Date:	Signed:		
	Print Name:		
	City:		
	Phone ()		

Prospective Franchisee's Copy

RECEIPT

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully. If BrightStar Senior Living Franchising, LLC offers you a franchise, it must provide this disclosure document to you 14 calendar-days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale.

[Michigan law requires that we give you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.]

If BrightStar Senior Living Franchising, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and the appropriate state agency identified on Exhibit A.

The franchisor is BrightStar Senior Living Franchising, LLC, located at 2275 Half Day Road, Suite 210, Bannockburn, Illinois 60015. Its telephone number is (888) 684-8250.

The issuance date of this Franchise Disclosure Document is April 29, 2024.

The name, principal business address, and telephone number of each franchise seller offering the franchise: Pete First, Chris Connolly, Amy Burnett, Dean Ulizio, Shannon Stemen, Shelly Sun, and David Pallaschke, 2275 Half Day Road, Suite 210, Bannockburn, Illinois 60015 847-693-2035.

We authorize the respective state agents identified on Exhibit A to receive service of process for us in the particular states. I received a Franchise Disclosure Document from BrightStar Senior Living Franchising, LLC dated as of April 29, 2024, that included the following Exhibits:

A – List of State Administrators and Agents for Service of Process; B – BRIGHTSTAR CARE HOMES Community Franchise Agreement (including exhibits); C – Operations Manual Table of Contents; D – Confidentiality, Non-Disclosure and Non-Competition Agreement; E – State Specific Addenda; F – Financial Statements; G – List of Franchisees; H – Release of Claims; I – Microsoft Teams/One-Drive/Web Version Office Package Opt-in Addendum; and J – Receipts.

Date:	Signed:	
	Print Name:	
	City:	
	Phone ()	
Date:	Signed:	
	City:	
	Phone ()	